SIEMENS

SIMATIC S5

COM 525

Programming Package for the Communications Processors CP 524 and CP 525 (S5-DOS)

Manual

Volume 2/2

Order No. 6ES5998-1DB21 Release 08

We have checked the contents of this manual for agreement with the hardware and software described. Since deviations cannel be precluded entirely, we cannot guarantee full agreement. However, the data in this manual are reviewed regularly and any necessary corrections included in subsequent editions. Suggestions for improvement are welcomed

Technical data

bjea to change

Siemens Aktiengesellschaft

The reproduction, transmission or use cl this document or its contents is not permitted without express written authority. Offenders will be liable for damages. All rights, including rights created by attent grant or ragl\$ItalonC4+ utility model or design. are reserved.

Copyright © Siemens AG 1991 All rights reserved

6ES5 998-1DB21 Elektronikwerk-Karlsruhe Printed in the Federal Republic of Germany

Contents		
Volume 1		
Volume 2		
Computer Link with RK 512 Example of Application	C79000-B8576-C540 -05	1
Event Output and Listing with the PT88/PT89 Printer Example of Application	C79000-B8576-C542-05	2
Notes on the Operating System PCP/M-86 and S5-DOS User's Guide	ns C79000-B8576-C543-03	3
Programming Package COM 525 User's Guide	C79000-B8576-C544-07	4
COM 525-Messages Reference Manual	C79000-B8576-C545-03	5
PROM 525 User's Guide	C79000-B8576-C546-04	6
Using the Handling Blocks Reference Manual	C79000-B8576-C547-06	7
		8
		9
		10

-

6ES5 998-1DB21 Release 08

`

SIEMENS

SIMATIC S5 Computer Link with RK 512

Example of Application

هه_{مه ک}رسر رسی و به بند بیا به ایر دید

,

С79000-В8576-С540-05

and the second state and the

~



CONTENTS

- - - -

1	Introduction	1
2	Aims	2
3	Hardware	3
4	Programing Software COM 525	7
4.1 4.2 4.3 4.3.1 4.3.1.1 4.3.1.2 4.3.1.3 4.4 4.4.1 4.4.1.1 4.4.1.2	<pre>Installing COM 525 on the PC 685 Calling COM 525 CP 525 User Program for PC 1 (135U) Interpreter and Procedure Copying from COMLIB into the User Program Assigning Parameters to the Procedure Transferring to the CP 525 User Memory CP 525 User Program for PC 2 (150U) Interpreter and Procedure changing the Procedure Parameters Transfer to the CP 525 User Memory</pre>	7 8 9 12 12 14 16 18 19 19 20
5	SEND Job from Data Block to Data Block	22
5.1 5.2 5.2.1 5.2.2 5.3 5.4 5.4.1 5.4.2 5.5	<pre>CP 525 User Program for PC 1 (135U) STEP 5 User Program for PC 1 (R Processor) Program Start-up Cyclic Program CP 525 User Program for PC 2 (150U) STEP 5 User Program for PC 2 (150U) Program Start-up Cyclic Program Test</pre>	24 26 27 30 30 30 32 33

i

Page

б	FETCH Job from Data Block to Data Block	34
6.1	CP 525 User Program for PC 1 (135U)	36
6.2	STEP 5 User Program for PC 1 (R Processor)	38
6,3	CP 525 User Program for PC 2 (150U)	40
6.4	STEP 5 User Program for PC 2 (150U)	40
6.5	Test	41
7	SEND Job, Flags to Data Block	42
7.1	CP 525 User Program for Pc 1 (135U)	44
7.2	STEP 5 User Program for PC 1 (R Processor)	44
7.3	CP 525 User Program for Pc 2 (150U)	45
7.4	STEP 5 User Program for Pc 2 (150U)	47
7.5	Test	48
8	Coordination Flags	49
8.1	CP 525 User Program for PC 1 (135U)	51
8.2	STEP 5 User Program for PC 1 (R Processor)	52
8.3	CP 525 User Program for PC 2 (150U)	53
8.4	STEP 5 User Program for Pc 2 (150U)	55
8.5	Test	55
9	PSEUDO WRITE Job, Input to Data Block	56
9.1	CP 525 User Program for PC 1 (135U)	59
9.2	STEP 5 User Program for PC 1 (R Processor)	59
9.3	CP 525 User Program for PC 2 (150U)	61
9.4	STEP 5 User Program for PC 2 (150U)	62
9.5	Test	62
10	Reading the Error Message Area in the SYSTAT	63
10.1	STEP 5 User Program for PC 1 (R Processor)	63
10.2	Test	65

ii

11	Date/Time PC Jobs	66
11.1 11.2 11.3	Read Date/Time Setting the Date/Time Synchronizing Several CP Hardware Clocks	66 68 69
12	Info	73
13	Program Documentation	75

iii

-3

. بوري و د د م م م م م م م م م م م م

.

and the state of the state

1 Introduction

This is an **example** of **an** application to illustrate the use of the communications **processor CP 525** for a point- **to-point computer** link. When you work **through** and implement this **example** you will have a finished **CP 525** user program in which all the most important functions of the **computer** link have been implemented.

First, the required hardware settings are explained. Then, not only the generation of a CP 525 user program with the COM 525 programming software, but also the corresponding STEP 5 user program for the CPU of the programmable controller are explained.

For this example you require the following hardware:

two CP 525 modules with RAM memory submodules
- one CP 525 <--> CP 525 cable connector (TTY or V.24)

- -onePG685programmer
- one cable connector PC <--> CP 525
- one S5-135U programmable controller
- one Rprocessorwith RAM memory **submodule** one cable connector **PG** <--> R processor
- one S5-150U programmable controller
- one cable connector \mathbf{PG} <--> AS 511

and if possible

one digital input module (24 V, at least 8 channels)
one digital output module (24 V, at least 8 channels)
one SIMATIC S5 simulator

the following software is also required:

- the S5 DOS programming package COM 525
- the STEP 5 basic package for the PG 685 programmer (supplied with the programmer)
- handling blocks for the R processor
- handling blocks for the **S5-150U**

2 Aims

First you set up the hardware. Then you install the software necessary for writing your user programs.

The first step is to **send** data from a data block in one PC to a data block in **the** other PC. Then data is fetched from the data block of the other PC and stored **in** a data block. The next step is to send a flag word to a data word on the partner cyclically. The reception of this data sent cyclically is then disabled by a coordination flag.

To demonstrate the PSEUDO READ/WRITE function, a PSEUDO WRITE job is programmed to send ϵn input byte to a data word on the partner.

You implement the reading of the error message area in the **SYSTAT.**

Both the reading and setting of the CP 525 hardware clock is demonstrated and the synchronization of several CP hardware clocks explained.

At the end of the example there is an explanation of the information and documentation functions available with the CPM 525 **prog**ramming software.

The individual tasks are restricted to what is absolutely necessary and in some respects (though not **all**) build on one another.

The DIRECT jobs (i.e., jobs which trigger job processing) are entered in the CP 525 (queue, which holds 10 jobs, and are processed in the order in which they were entered. Jobs already in the queue are not entered a second time. Since in this example of an **application**, there are less than 10 **DIRECT** jobs programmed, a DIRECT job will be processed **whenever** it is **triggered**. Interlocking of jobs is therefore not necessary in this example.

3 Hardware

In the course of this example, a point-to-point link between two SIMATIC S5 programmable controllers is set up, The two programmable controllers used are an S5-135U with R processor and an S5-150U.

In principle, the **procedure** for **S5-115U**, **135U** and **150U** is identical. Nothing whatsoever changes in the **CP** 525 user program. The handling blocks have different block numbers with the different PCs, however, they are identical in terms of programming. Only the handling of **interprocessor** communication (IPC) flags is different in the various PC'S (see Section **8**).

Plug in the Reprocessor into one of the slots (11, 19, 27 or 35) in the S5-135U. Carry out an overall reset of the CPU and switch the mode selector to STOP.

Plug in the central processor and memory modules in the correct slots in the **S5-150U** frame. **Carry** out an overall reset of the **S5-150U**.

On one **CP 525** set the module address 2; i.e., **insert** jumper 7-10 on jumper block 16. On this CP 525 enable the **IPC** flag bytes 0 to 31. To do this, insert jumper 8-9 at jumper block 25 on this module. Plug this **CP** into the **S5-135U**. The slots available for the **CP** are 11, 19, 27, 35, 43, 51, 59 and 67.

On the other **CP** set the module address O, **and** disable all the **IPC** flags (i.e., do not insert jumpers, either at jumper block 16 or block 25). Plug the **CP** 525 into the **S5-150U. The** slots available are 107, 115, 123 and **131.**

Connect the upper **device** interfaces (IF 1) of **both** CPS using the corresponding cable **connector**. Whether your **transmission** is carried out with **TTY** or **V.24** signals depends only on the **type** of cable **connector** used. Both signals are always available at the port of the **CP** 525. If the transmission distance exceeds 20 meters, you must use TTY signals.

Set the module address) on the digital input **mohule** and the digital output module. **This makes input byte** IB) and output byte QB O available. Plug in the modules in any free slots on the **S5-135U**. Connect the simulator to the 1/0 modules.

Assignment of input byte IB 0:

Bit 0 : not used Bit 1 : trigger SEND 1 PC job (see Section 5) Bit 2 : trigger FETCH 2 PC job (see Section 6) Bit 3 : trigger PSEUDO-WRITE PC job (see Section 9) Bit 4 : reset error message area in the SYSTAT (see Section 10) Bit 5 : trigger date/time synchronization (see Section 11.3) Bit 6 : trigger setting of date/time (see Section 11.2) Bit 7 : reset coordination flags (see Section 8)

Assignment ofoutputbyte QBO:

If you have no I/O modules orno simulator available, then insteadof input byte IB O, use flag byte FY O. These flags can be set and reset with the STEP 5 online function CONTROL VARIABLE.

Distribution module 756

In the remainder of thi; section the S5-135U, wh ch includes the digital 1/0 modules, is called PC 1 and the S5-150U is called PC 2.



SYSTEM CONFIGURATION:

w.,

Connect the **PG** 685 **usir**; the corresponding cable **connector** to device interface (II? 2) of the **CP** 525; or the **programming** interface of the R **processor** of the **135U**; or the **programming device** interface module AS 511 in the **150U**, depending on which **module** you wish **to** program. When programming the **CP** 525, you must switch its mode selector to **PGR**.

Instead of a PG 685, you can also use a PG 635, PG 675 or PG 695.

GENERATING PROGRAMS:



....

4 Programming Software COM 525

This chapter explains $h \cdot w$ to set up the software for programming the computer link jobs **ased** on the example.

4.1 Installing COM 525 on the PG 685

A **PG 685** programmer in which the STEP **5 basic** package has already been installed must be available.

Go to drive B: user mumber O by entering

0:

Copy the contents of the three COM 525 floppy disks with

PIP B:=A:*.*[RV]

Option V= verify cryying R = copy SYS files

onto the hard disk user number 0.

Assign the system and read only attributes to the files with

SET S5?EC?5X.CMD[SYS RO] SET S5PEP05X.CMD[SYS RO] SET COMLIB*.525[SYS RO]

You can work with COM 525 under any user number.

Exit user area 0 which should be reserved for system files, with
n: (n = required user number).

4.2 Calling COM 525

By entering

S5

you call the package **selection** mask. Place the **cursor** in the line 'COM 525 programming package for the CP 525/524' and using the function key F1(PACKAGE) select the COM 525 programming software for the CP 525.

The COM 525 basic mask then appears:

OPYRIGH ASIC	T (C) BY SII MASK	EMENS		SIMATIC s5 / COM52
	CCCCCCC CC CC CC CC CC CCCCCCCC	0000000 Do 0 C Do 0 C OO 70 OO 70	MM MM 55555555 22222 MMM MMM 55 22 1 MM MMM 55 22 1 MM MM 55 22 1 MM MM 55 22 1 MM MM 55 22 MM MM 55 22	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
	Program Version/Iss	ing packag sue: A [.] '4	e for the communications proc and the communications proc Serial no.: 7994-007	cessor CP 525 cessor CP 524 4654321
F 1 Select	F 2 System	F 3	F4 F5 F6	F7 F8

4.3 CP 525 User Program for PC 1 (135U)

Press F1 (SELECT PROGRAM) to call the mask 'PROGRAM SELECTION'.

BASIC MASK -> PROGRAM SELECTION	SIMATIC S5 / com525
DRIVE: B PROGRAM NAME: CL512PC1 COMPONENT: CL Plant designation: application exa Generated by: Fred Generated on: 07.10.87	mple
PGdate - time: D M Y H M 07.10.87 - 11:11	
F 1 F 2 F 3 F 4 F 5 F SELECTION	6 F 7 F 8 HELP EXIT

As drive, select the hard disk "B". Enter a program name (in this case "CL512PC1") and the required component "CL" for computer link. The plant designation is called 'application example" in this case. Under "generated by" enter your own name. The date and time of the PG 685 hardware clock can be set by overwriting the input fields.

. . .

.

·~~~

In the <code>`SELECTION'</code> mask, which you call with Fl, you can select the basic COM 525 functions.

BASIC MASK -> PROG. S E L E C T I O N	SELECTION ->	SIMATIC S5	/ COM525
	DRIVE: B PROGRAM NAME: CL512PC1 COMPONENT: CL		
	Plant designation: application example Generated by: Fred Generated on: 07.10.87		
1			
F 1 F 2 PROGRAM USER DATA TRANSFER	F 3F 4F 5F 6DELETEINFOCONDENSECONTINUE	F 7	F 8 EXIT

By pressing F6 (CONTINUE) you call the second set of function keys (second menu).

F 1	F 2	F 3	F 4	F 5	F 6	F 7	F 8
	PROC.PARA	PRI.PARA.	LISTING		CONTINUE		EXIT
	•						

By pressing F6 you return to the basic set of **function** keys.

1836 - C

الأه فتنابعونا وعاوفته والمحادي والا

-

When programming with COM 525 you generate the user program first on floppy or hard disk (storage medium FD), and then transfer it to the CP 525 user memory and then test it.

The S5 DOS file on the FD is automatically named program name.525, in this case CL512PC1.525.

4.3.1 Interpreter and Procedure

Without an interpreter and procedure no data exchange is possible between the CP 525 and a partner.

4.3.1.1 Copying from COMLIB into the User Program

The interpreter and procedure must first be copied into the user program. They are located in the library **COMLIBn** (n = version number) which is supplied on a separate floppy disk with the COM 525 software.

Ress F2 (TRANSFER) and F5 (FD->FD) in the 'Selection mask to call the function for transferring from FD to FD. Press F7 (HELP) in the 'TRANSFER' mask and enter drive "B" and program name "COMLIBN" (n = version number) as source. The destination is automatically the program CL512PC1 selected in the 'PROCRAM SELECTION' mask.

->PROG.SELECTION -> SELE TRANSFER 	CTION ->	SIMATIC S5 / COM525	
	Source:	DeSt.:	
STORAGE MEDIUM:	FD	FD	
DRIVE:	B	B	
INTERFACE NUMBER: PROGRAM NAME: COMPONENT:	COMLIB01	CL512PC1 CL	
Plant designation:	standard library	Application example	
Generated by:	GW Karlsruhe	Fred	
Generated on:	03.09.87	07.10.87	
F1 F2 F	3 F 4 F 5	F6 F7 F8	
INTE	R-	JOB	
PRET	ER PROCEDURE	BLOCK HELP EXIT	

B8576540-03

ويريد والمتحاف والمحاف

,

....

~

. .08

Press F3 to **call** the 'INTERPRETER' transfer mask. Press F7 (**HELP**) in this mask to select the interpreter "**RK512**" (component "**CL**") to be transferred **F1** (TRANSFER) starts the transfer. The successful transfer is displayed with '**MESS.002**: Completed'.

-

->PROG. SELECTION -> INTERPRETER	SIMATIC S5	/ (331525			
	Source:		Dest.:		
STORAGE MEDIUM: DRIVE:	FD B		FD B		
PROGRAM NAME: COMPONENT:	COMLIB01		CL512PC1 CL		
Source:	COMPONENT NA CL RK5	ME VERSION 12 01			
F 1 F 2 TRANSFER	F3 F4	F 5	F6	F 7 HELP	F 8 EXIT

With ${\bf F8}$ (EXIT) you return to the 'TRANSFER' mask.

With F4 you call the 'PROCEDURE' transfer mask. Select procedure "P3964R" (procedure 3964 with block check) using F7 (HELP). The procedureistransferred the same way that the interpreter is.

Press F8 (EXIT) twice to return to the 'SELECTION' mask from the 'PROCEDURE' mask.

 $\phi_{i} \leftarrow \phi_{i} \phi_$

4.3.1.2 Assigning Parameters to the Procedure

With F6 (CONTINUE) **call the** second set of function keys in the 'SELECTION' mask,

e 1

F2 branches to the 'ASSIGN **PROC. PARA.'.** The procedure **P3964R** belonging to your program **CL512PC1** is displayed with its version number. Select the data rate "9600" and priority "**HIGHER**".

In PC 2 (150U) the priority must be programmed as LOWER. If there is an initialization conflict, i.e., both CP 525s attempt to transmit simultaneously, the CP 525 in PC 2 (150U) desists.

> PROG. SELEC SSI GN	TION •> SELECTIO PROC. PARA	N ->	s	IMATIC S5 / COM525
		DRIVE: B	PROGRAM: CL512PC1	COMPONENT: CL
Procedure:	COMPONENT: CL	NAME: P396	4R VERSION: 01	
Baud rate:	9600		Char. length	8
Number of st	top bits: 1		Pri ori ty:	HI GHER
Pari ty:	EVEN			
F1 F	F 2 F 3	F 4	F5 F6 SAVE	F7 F8 HELP EXIT

الاهي والارداد والمتحافظ والافاق

,

Save the procedure parameters on the hard disk with F6. 'MESS.003: Saved!' indicates that the information has been saved. With F8 (EXIT) and F6 (CONTINUE) return to the 'SELECTION' mask (first set of function keys).

No parameters can be assigned to the **RK512** interpreter.

ins ledies de 🔍 🔾

۶.

4.3.1.3 Transferring to the CP 525 User Memory

The interpreter and procedure must still be transferred to the user memory of the CP 525 in PC 1 (135U). The CP 525 must be connected to the PG 685 and its mode selector must be set to "PGR" (programming).

Using F2 (TRANSFER) and F3 (FD->CP) in the 'SELECTION' mask call the function for transferring from FD to CP 525. In the 'TRANSFER' mask enter the CP 525 interface "1" as the destination. Make the transfer with F3 (INTERPRETER) and F4 (PROCEDURE).

->PROG. SELECTION -> SELI T R A N S F E R	SIMATIC S5	/ COM525		
	Source:	Dest.:		
STORAGE MEDIUM: DRIVE:	FD B	CP525		
INTERFACE NUMBER : PROGRAM NAME: COMPONENT :	CL512PC1 CL	1		
Plant designation: Generated by: Generated on:	application example Fred 07.10.87			
ERR.724: Interpreter/pr F 1 F 2 F 2 COLD TOTAL INTE RESTART PROGRAM PRET	ocedure not present 3 F 4 F 5 R- ER PROCEDURE	F 6 Job Block	F7 HELP	F 8 EXIT

The message **ERR.724** appears only if you have not yet loaded data in **the CP**.

'MESS.002: Completed!' shows that the transfer is completed. When the interpreter and procedure are transferred to the CP 525 memory you must cold restart the CP 525 with F1 (COLD RESTART). If you switch the CP 525 mode selector to RUN, the red LED of interface 1 (IF 1) must go out,

Press F8 (EXIT) twice to return to the 'PROGRAM SELECTION' selection. Press F8 (EXIT) again so that the prompt 'ACK.001: terminate program?' appears on the PG 685 screen. Confirm that you wish to terminate the program with F1 (YES) and exit the COM 525 programming package by again pressing F8 (EXIT),

Press F8 (BACK) and confirm the prompt 'EXIT S5 COMMAND INTERPRETER?' with the enter key to return to the PCP/M Operating system,

All the preparations are now made on the ${\bf CP}$ in PC 1 for programing the computer ${\bf link}$ functions,

17

- 20

.

- 4

وسياسه فعفوه وتوار

B8576540-03

 γ_{4}

યુક્યુ છે. સંદર્ભાતન

4.4 CP 525 User Program for PC2 (150U)

Copy the CP 525 user program for PC 1 into a file for the CP 525 user program for PC 2 with

PIP CL512PC2.525=CL512PC1.525

The programs differ from each only in the priority of the procedures. By typing in

S5

call the package selection **mask**. Using the cursor select the programming package COM 525 and press **Fl** (PACKACE).

From the <code>`BASIC MASK'</code>, <code>F1</code> (SELECT PROCRAM) leads to the <code>'PROGRAM</code> SELECTION' mask.

BASIC MASK -> P R O G R A M S E	LEC TION		SIMATIC S5	/ COM525
	DRIVE: B PROGRAM NAME: CL512 COMPONENT: C L	PPC2		
	Plant designation: Generated by: Generated on:	application example Fred 07.10.87		
	PGdate - tire:	DMYHM 07.10.87-1 3:13		
F 1 F 2 SELECTION	F 3 F 4	F 5 F 6	F 7 HELP	F 8 Exit

Press F7 (HELP) to select the program name "CL512PC2" in this mask.

For the component, plant designation and "generated by" no new entries are required. They are the same as those for program <code>C1512PC1_</code>

4.4.1 Interpreter and Procedure

The interpreters in PC 1 and PC 2 are identical. The procedures in both **communications** processors must have the same data rate but different priority.

4.4.1.1 Changing the Procedure Parameters

Press **F1** to call the 'SELECTION' mask and then F6 (CONTINUE) to call the second set of function keys.

F2 branches into the mask 'ASSIGN **PROC. PARA.';** the procedure belonging to program **CL512PC2** is displayed along with the version number. Leave the baud rate of 9600 as it is but change the priority using F7 (HELP) to "LOWER",

In PC 1 (135U) the priority was programed as HIGHER. If there is an initialization conflict, i.e. , both CP 525s attempt to transmit simultaneously, CP 525 in PC 2 (150U) desist.

		DRI VE:	В	PROGRAM:	CL512PC2	COMPONENT:	CI
Procedure: 🕻	OMPONENT: CL	NAME:	P3964F	R VER	SI ON: 01		
Baud rate:	%00			Cha	r. length:	8	
Number of stop	bits: 1			Prio	ority: L	OWER	
Parity:	EVEN						
F 1 F 2	F 3	F 4	F	5 S.	F 6 F AVE HE	F 8 F 8	Т

٠

Save the procedure parameters on the hard disk with F6. 'MESS.003: Saved!' indicates that the information is saved. With F8 (EXIT) and F6 (CONTINUE) return to the '.SELECTION^t mask (first set of function keys).

4.4.1.2 Transfer to the CP 525 User Memory

The interpreter and procedure must still be transferred to the user memory of the **CP** 525 **in** PC 2 **(150U)**. The **CP** 525 must be **connected** to the **PG** 685 **and** its mode selector must be set to **PGR** (programming).

e

Press F2 (TRANSFER) and F3 (FD->CP) in the 'SELECTION' mask to call the function for transferring from FD to CP 525. In the 'TRANSFER' mask enter the CP 525 interface "1" as the destination. Make the transfer with F3 (INTERPRETER) and F4 (PROCEDURE).

-> PROG. SELECTION -> SEL TRANSFER	SIMATIC S5 / COM525				
	Source:	Dest.:			
STORAGE MEDIUM: DRIVE:	FD B	CP525			
INTERFACE NUMBER: PROGRAM NAME: COMPONENT:	CL512PC2 CL	1			
Plant designation: Generated by: Generated on:	application example Fred 07.10.87				
ERR.724: Interpreter/procedure not present F 1 F 2 F 3 F 4 F 5 F 6 F 7 F 8 COLD TOTAL INTER- RESTART PROGRAM PRETER PROCEDURE BLOCK HELP EXIT					

The message $\ensuremath{\textit{ERR.724}}$ appears only if you have not yet loaded data in the $\ensuremath{\textit{CP}}.$

'MESS.002: Completed!' shows that the transfer has been completed. When the interpreter and procedure have been transferred to the CP 525 memory you must cold restart the CP 525 with F1 (COLD RESTART). If you switch the CP 525 mode selector to RUN the red LED of interface 1 (IF 1) must go out.

You can now program the computer link functions on the ${\bf CP}$ 525 in Pc 2.

SEND Job from Data Block to Data Block 5

Begin with a SEND job from PC 1 (135U) to PC 2 (1SOU).

Four data words of data block DB 10 (from data word DW 0 onwards) in the R processor are to be transferred to data block **DB** 11 (from data word DW 4) in the S5-150U. You can see that the source and destination specified can be completely different.

iγe i

For this the following program structure is required:

PROGRAMMABLECONTROLLER 1

PROGRAMMABLE CONTROLLER 2



*) DT = DATA TRANSFER

....

The interpreter and **proc**edure are already loaded in the user memories of the two **CP** 525s (see Sections 4.3.1.3 and 4.4.1.2).

The SEND job needs to be programmed only on the CP 525 of the active PC. Active means that this PC initiates the data exchange and that you need to program a SEND job only for the CP 525 in PC 1 (135U).

The SEND job is initiated when the CPU calls the handling block SEND DIRECT with the job number 'n'. Select job number 1. The same job number 'n' is also given to the corresponding job on the CP 525. This job contains information about the destination of the data in PC 2 (150U).

CP 525 fetches the data only when called by the handling block SEND ALL. This call transfers the data from the CPU/DB 10 to the CP 525 in PC 2.

The CP 525 in PC 2 (150U) receives the data and calls the RECEIVE ALL to store it in the destination indicated by PC 1.

The SEND AIL (also **known** as SEND 0) has the job number A-NR 0. It is called unconditionally in every program cycle and checks whether the **CP** 525 (dual-port **RAM**) requires data. If it does, the SEND ALL fetches it from the specified source and transfers it to the **CP** 525, If it does not, the function block is exited and the cyclic program continues.

The **RECEIVE ALL** (also known as RECEIVE O) has the job number A-NR O. It is called unconditionally in every program cycle and checks whether the CP 525 (dual-port RAM) needs to transfer data to the CPU. If it does, the RECEIVE ALL stores the data in the destination address specified by the CP 525. If it does not, the function block is exited and the cyclic program continues.



-v 19 a L 1 1 - - -

5.1 CP 525 User Program for PC 1 (135U)

Switch the mode selector on the CP 525 to PGR.

Press F1 (PROGRAM USER DATA) in the 'SELECTION' mask to call **up** the 'JOB BLOCK'. Enter the job number with which this job is to be triggered by the CPU, Use the default 'job no. :'"1".

Press F5 (PROGRAM JOB) to call the 'PROGRAM JOB' mask.

Use the defaults 'job:' "SEND" and 'job type' "DATA BLOCK". The 'dest. - word address' is "4" D (decimal) and "4" H (hexadecimal); the data block number is "11". No entry is necessary for the CPU number since you are not concerned with multiprocessor operation. Do not specify any coordination flags.

Press F6 (ENTER JOB) to write the job to the RAM memory of the PG 685. With F8 (EXIT) you return to the 'JOB BLOCK' mask. With F6 (SAVE JOB BLOCK) the SEND job is written to the hard disk.

Press F8 (EXIT) to return to the 'SELECTION' mask.

'24

بلانون والجار فيرواد فافتها مريحا والمرو

۰

.

-> SELECTION -> JOB PROGRAM JOB	SIMATIC S5 / COM525				
	DRI VE:	В	PROGRAM:	CL512PC1	COMPONENT: CL
	0 L	В			
Job 110.:	001				
Job:	SEND				
Job type	DATA BLOCK				
CPU no. :					
DB no. :	011				
Dest word addre	ess: 00004 D	0004	Н		
If required with	n coordination i	fl ag:			
FI F2 ON PAGE PRINTER BACKWARDS	F 3 F 4 PAGE FORWARDS		F 5 Delete Job	F 6 ENTER JOB	F 7 F 8 HELP EXIT

Transfer the job with F2 (TRANSFER) and F3 (FD -> CP) to the user memory of the CP 525. This is similar to transferring the interpreter and procedure (see Section 4.3.1.3).

Press F6 (JOB **BLOCK)** in the **'TRANSFER'** mask (see Section 4,3.1.3) to transfer all the jobs to the **CP** 525 user memory.

Perform a CP 525 cold restart by pressing F1 in the 'TRANSFER' mask.

Switch the **CP** 525 to **"RUN"**. The red LED for device interface **1** (IF 1) must **now go** off. If it does not, check that the interpreter and procedure are actually present.

5.2 STEP 5 User Program for PC 1 (Reprocessor)

Call the S5 package LAD, CSF, **STL. Program** in **STL** (statement list) and call your STEP 5 program file "B:CLPCIRST. S5D".Transfer the handling blocks for the R processor into this program file. Connect the R processor to the PG 685.

The STEP 5 program must perform the following:

- synchronize the CPU and CP 525 during the PC start-up
- trigger the SEND job
- transfer the data from the CPU to the CP 525.

5.2.1 Program Start-up

All the start-up organization blocks OB 20, OB 21 and OB 22 call the SYNCHRON handling block unconditionally.

The interface number SSNR for device interface 1 (IF 1) is the same as that set for the module address "2". Select field length "2" which allows data to be sent in strings of up to 32 bytes at once between the CPU and CP 525. This number is sufficient for this example since no more than 10 words (20 bytes) will be transferred at any time, Any other field length could also be used.

If you wish to transfer **more** data than the field Length set allows, the CP **525** requests the **remaining** data with further SEND ALL calls.

With the selection of the field length you can decide the **amount** of data to be transferred with each SEND **ALL** (and also RECEIVE AIL). With a smaller field length the data exchange may be distributed **over** several cycles, but puts very little load **on** the cycle. With larger field lengths more data at **a time can be** transferred. The cycle time **may** well be extended.

Use flag byte "FY10" as the parameter assignment error byte PAFE.

The start-up organization blocks OB 20, OB 21 and OB 22 appear as follows:

	:JU	FB125	SYNCHRONIZATION CP 525 <> CPU
W	:SYN	ICHRON	
SSNR	:	KY0,2	INTERFACE NUMBER 2
BLGR	:	KY0,2	FIELD LENGTH 2 (MAX, 32 BYTES)
P m	:	FY10	PARAMETER ASSIGNMENT ERROR BYTE
	:		
	BE		

A parameter assignment error is to be indicated at digital output Q 0.0. In the start-up organization blocks there is no process image; therefore no digital 1/0s are processed and the indication can be made only in the cyclic program (OB 1, see Section 5.2.2). Normally an error evaluation program is called up at this point. If an error occurs, check whether the interface number and module address are the same. If they are correct a hardware fault is suggested. In this case a computer link cannot be established.

5.2.2 cyclic Program

The beginning of the **cyclic** program checks whether a parameter assignment error occurred during the start-up (see Section 5.2.1).

The SEND job is triggered by a SEND DIRECT **call**, i.e., the handling block SEND is called with a job number other than O; in this case, job number 1. The job number in the SEND DIRECT must be the same as the job **in** the **CP** 525. The job is to be carried out on the positive **going** edge at digital input I 0.1.

The interface number SSNR "2" and the job number A-NR "1" must be programme d. Select condition codeword ANZW "FW 11" and parameter assignment error byte PAFE "FY 15". The condition codeword requires two words. The job status and any error messages are indicated in flag word FW 11. The number of pieces of data to be transferred per SEND ALL call is indicated in flag word FW 13.

In the handling block there must be details about the source of the data in the CPU for the SEND job. The data source should be a data block (source type QTYP "DB") with the number DBNR "10". The source start QANF is data word "O" and source length QLAE is "4" data words.

The SEND job is triggered with the result of logic. operation **(RLO)** 1. If **RLO** equals 0 when the call is made, only the condition codeword will be updated.

The termination of the $j_{0}b$ with an error and the (ccurrence of a parameter assignment error is indicated at digital output Q 0.1. In this situation, an error evaluation program tailored to the particular application is usually called. If an error occurs, evaluate the error numbers in the parameter assignment error byte PAFE, condition codeword ANZW and error message area in the SYSTAT (see aLso Section 10).

The actual data transfer **from** data block DB 10 to the **CP** 525 is carried out by the SEND **ALL** call. The interface number **SSNR** must also be "2". The job number **A-NR** is "0". Select condition codeword **ANZW "FW16"** and parameter assignment error byte **PAFE "FY18"**. In the **ALL** function the condition codeword is only **one** word **long**. The *lower* order byte (FY 17) indicates the **number** of the job for which the SEND AIL is currently transferring **data** from the CPU to the CP; in this case, job number 1.

The SEND ALL can be called up with any RLO.

On the following page yo 1 can see the program in the cyclic **manimum tion block OB** 1.

	:A	F 10,0	IF SYNCHRON-PARA. ASS. ERROR
	:=	Q 0.0	OCCURRED> SET OUTPUT
	:AN	I 0.1	EDGE EVALUATION:
	:R	F 2.1	THE RESULT OF LOGIC OPERATION
	:A	10.1	(RLO) IS SET FOR ONE CYCLE, IF
	:AN	F 2.1	THE SIGNAL CHANGES FROM O TO 1 AT
	:S	F 2.1	INPUT I 0.1.
NAME	:JU :SEN	FB120 D	TRIGGER SEND JOB 1
SSNR A-NR ANZW QTYP DBNR QANF QLAE PAFE	: : : : :	KY0,2 KY0,1 FW11 KSDB KY0,10 KF+0 KF+4 FY15	INTERFACE NUMBER 2 JOB NUMBER 1 CONDITION CODEWORD SOURCE TYPE DATA BLOCK SOURCE DATA BLOCK DB 10 SOURCE START DATA WORD O SOURCE LENGTH 4 DATA WORDS PARAMETER ASSIGNMENT ERROR BYTE
	:0	F 12.3	JOB 1 TERMINATED WITH ERROR
	:0	F 15.0	OR PARAMETER ASSIGNMENT ERROR
	:=	Q 0.1	INDICATE AT OUTPUT Q 0.1
NAME	: JU : SEN	FB126 D-A	SEND ALL CALL
SSNR	BE	KY0,2	INTERFACE NUMBER 2
A-NR		KY0,0	JOB NUMBER 0
ANZW		FW16	CONDITION CODEWORD
PAFE		FY18	PARAMETER ASSIGNMENT ERROR BYTE

All that is still missing is the data source, data block DB 10. To be sure that it is long enough for the other examples, it should be at least 31 data words long. Enter values other than 0 in the first four data words DW 0 to DW 3.

Transfer the R processor handling blocks FB 120 to FB 127, OB 20, OB 21, OB 22, OB 1 and DB 10 to the user memory of your R processor. Carry out a cold restart on the R processor. The green LED (RUN) must light up.
5.3 CP 525 User Program for PC 2 (150U)

Nothing is changed in the CP 525 user program in Section 4.4, since no job is required in the passive PC 2.

Switch the mode selector on the **CP** 525 to "RUN". The red **LED** on the CP 525 should now go out. If it does not, the interpreter and/or procedure is missing or there is a **CP** 525 hardware fault.

5.4 STEP 5 User Program for PC 2 (150U)

Connect the interface module AS 511 to the PC 685.

Call the **S5** package IAK), CSF, **STL**. Program in **STL** (statement list) and call your STEP **5** program file **"B:CLPC25ST.S5D".Trans**-fer the handling blocks for the R processor into this program file.

The STEP 5 program must perform the following:

- synchronize the CPU and ${\bf CP}$ 525 during the PC start-up - transfer the data received from the CP 525 to the CPU

5.4.1 Program Start-up

All the start-up **organization** blocks OB 20, **OB** 21 and **OB** 22 call the handling block **SYNCHRON** unconditionally.

The interface number SSNR for device interface 1 (IF 1) is the same as that set for the module address "01'. Select the smallest field length "1" which allows data to be sent in strings of up to 16 bytes at once between the CPU and CP 525. This number is sufficient for this example since no more than four words (8 bytes) will be transferredat any time. Any other field length could also be used.

If you want to transfer more than 16 bytes, the CI 525 simply stores the remaining data in the CPU with further RECEIVE ALL calls.

< •

With the selection of the field length you can decide the **amount** of data to be transferred with each RECEIVE ALL (and also SEND ALL). With a smaller field length the data ex-change may be distributed over several cycles, but puts very **li**t**le** load on the cycle. With larger field lengths more data can be transferred at once but the cycle time may well be extended.

Use flag byte "FY10" as the parameter assignment error byte ${\rm P\,m}$.

The start-up **organization** blocks **OB** 20, **OB** 21 and OB 22 appear as follows:

	:JU	FB185	SYNCHRONIZATION CP	525 <> CPU
NAME	:SYN	CHRON		
SSNR	:	KYO,O	INTERFACE NUMBER O	
BLGR	:	KY0,1	FIELD LENGTH 1 (MAX.	16 BYTES)
Ρm	:	FY10	PARAMETERASSICNMENT	ERROR BYTE

BE

A parameter assignment **error** is to be indicated by flag F 10.0. If an error occurs, check that the interface **mumber** and module address are the same. If they are correct, a hardware fault is suggested. In this case a computer link cannot be established.

31

حرسي بالاستهامية الديبا

.

5.4.2 Cyclic Program

The data transfer from the CP 525 to the data block DB 11 is carried out by a RECEIVE ALL call. The interface number SSNR must be "O". The job number A NR is "O". Select condition codeword ANZW "FW 11". In the ALL function the condition code is only one word long. The coordination flag byte number of the job that has just sent data (see Section \$) is indicated in the lower order byte (FY 12). In this example no coordination flags have been programed so FFH appears here. The parameter assignment error byte PAFE is flag byte "FY 13".

The data destination is specified by the **CP** 525. Therefore, there is no need to enter a destination in the handling block (specify dest. type **ZTYP** as "NN"). The following three parameters - data block number **DBNR**, destination start **ZANF** and destination length **ZLAE** - are therefore automatically not evaluated.

The RECEIVE **ALL** can be **called** up with any result cf logic **operation** (**RLO**).

The program in the cyclic **organization block OB 1** appears as follows:

:JU FB181 NAME :RECEIVE SSNR : KYO,O A-NR : KYO,O ANZW : FW11 ZTYP : KSNN DBNR : KYO,O ZANF : KY+O ZLAE : KY+O "Pm : FY13 RECEIVE ALL CALL

INTERFACE NUMBER O JOB NUMBER O CONDITION CODEWORD NO DESTINATION SPECIFIED NOT EVALUATED NOT EVALUATED NOT EVALUATED PARAMETER ASSIGNMENT ERROR BYTE

BE

Only the data destination data block DB 11 is now missing. This block must be at least 8 data words long.

32

المحري يرام رام راف ما ما ما ما ما ما م

Transfer the **S5-150U** handling blocks **FB** 180, FB **1&1** and FB 185, **OB** 20, **OB** 21, OB 22, **OB** 1 and **DB** 11 to the user memory of your S5 **150U**. Carry out a cold restart. The green LED (RUN) **must light** up.

5.5 **Test**

Using the PG 685 online function CONTROL VARIABLE observe the data words DW 4 to DW 7 of data block DB 11 in PC 2 (lSOU), Then switch the digital input I 0.1 on PC 1 from 0 to 1. The values from PC 1 DB 10 now appear in PC 2 DB 11.

If there is no digital input module with a simulator available, use the flag F 0.1 instead of input I 0.1. Set (ox reset) the flag with the PG 685 online function CONTROL VARIABLE. Observe the condition codeword FW 11 and the parameter assignment byte FY 15 using CONTROL VARIABLE.

s - <u>-</u> s

6 FETCH Job from Data Block to Data .3lock

The data sent to PC 2 in Section 5 will be fetched back from PC 2 (150U) to PC 1 (135U) using a FETCH job.

This means that four data words from DB 11 starting from data word DW 4 in PC 2 (150U) are fetched and transferred to data block DB 10 from data word DW 5 in PC 1 (135U/R processor). The source and destination specified are different.

The following program structure is required:

PROGRAMMABLE CONTROLLER 1 PROGRAMMABLE CONTROLLER 2 STEP 5 CP 525 STEP5 CP 525 user user user user program program program program Reprocessor CPU CPU CP 525 CP 525 CPU _____ - .. --. -. -. -.-.-. FETCH PROCE-PROCE т DURE DIRECT DURE R DT Job no. 2 3964R 3964R Destt = DB10 4 DW from DW 5 L G G DT DT ER RHECEINE INTER-**INTER** -1 SEND ALL PRETER PRETER DT ì ALL DT (JOB NO. 0) **RK512 RK512** DT DT **JOB NUMBER 2** FETCH Data Data DB 10 Source = DW4 **DB** 11 DW 5... 8 in DB 11 DW 4...7,

*) DT = DATA TRANSFER

The interpreter and procedure are already loaded in the **user** memory **of** both **CP** 525s (see Section 4).

The FETCH **job** needs to be programmed only on the **CP** 525 in the active PC. Active means that this PC triggers the exchange of data and that you do not need to program a FETCH job for the **CP** 525 on PC 1.

The FETCH job is triggered when the CPU calls the handling block **FETCH** DIRECT with the job number 'n'. Select job number 2.

The corresponding job on the CP 525 has the same job number 'n'. This job includes specifications about the data source in PC 2; the data destination in PC 1 is specified in the FETCH DIRECT.

The CP 525 in PC 1 (135U) sends a request telegram to the CP 525 in PC 2 (150U) with details about the location from which the data is to be fetched. The CP 525 in PC 2 (150U) retches the requested data using a SEND ALL from data block DB 11 of the CPU, and sends the data to PC 1. The CP 525 receives the data in PC 1 and writes it into data block DB 10 with a RECEIVE ALL.

The SEND ALL (also known as SEND O) has the job number A-NR O. It is called unconditionally in every program cycle and checks whether the CP 525 (dual-port RAM) requires data. If the CP does, the SEND ALL fetches it from the specified source and transfers it to the CP 525. If the CP does not, the function block is exited and the cyclic program continues.

The **RECEIVE ALL** (also known as **RECEIVE 0**) has the job number A-NR O. It is called unconditionally in every program cycle and checks whether the CP 525 (dual port RAM) needs to transfer data to the CPU. If it does, the **RECEIVE ALL** stores the data in the destination address specified by the CP 525. If it does not, the function block is exited and the cyclic program continues.

6.1 CP 525 User Program for PC 1 (13.5U)

Switch the mode **selector on** the **CP** 525 to "PGR".

Call the program "CL512PC1" in the COM 525 mask 'PROGRAM SELECTION' with F7 (HELP) .

Press **F1** (PROGRAM USER DATA) in the 'SELECTION' mask to call the 'JOB **BLOCK'** mask. Enter the job number with which the job will be triggered by the CPU. **Select** 'Job no. :' "2".

Press F5 (PROGRAM JOB) to call the 'PROCRAM JOB' mask.

Using F7 (HELP) enter the 'job: '"FETCH" and 'job type' "DATA BLOCK". The 'source - word address' in PC 2 is "4" D (decimal) and also "4" H (hexadecimal). The data block number is "11". No CPU number need be specified since PC 2 (150U) is capable only of single processor operation. Do not enter any coordination flags.

The job is written to the RAM on the PC 685 with F6 (ENTER JOB). Press F8 (EXIT) to **return** to the 'JOB **BLOCK'** mask,

Press F6 (SAVE JOB **BLOCK)** to save the whole job block; i.e., the FETCH and the SEND job (see Section 5) are saved on the hard disk. Depending on whether you wish to overwrite **the** existing job block, answer the prompt **'ACK.011:** Job block already exists - overwrite?' with **F1** (YES) or F3 (NO).

Press F8 (EXIT) to return to the 'SELECTION' mask.

	DRI VE:	В	PROGRAM:	CL512PC1	COMPONENT: (CL
		7 O	В			
Job no.:	002					
Job:	FETCH					
Job type	OATA BLOCK					
CPU no.:						
DB no.:	011					
Source - word add	dress: 0000+ 0	0004	1 H			
If required with	coordination flag:	:				
F1 F2	F3 F4		F 5	F 6	F7 F8	3
RINTER BACKWARDS	FORWARDS		JOB	JOB	HELP EXI	Т

Transfer the job in the 'SELECTION' mask with F2 (TRANSFER) and F3 (F'D -> CP) to the user memory of the CP 525. This is similar to transferring the' interpreter and procedure (see Section 4.3.1.3).

Press F6 (JOB **BLOCK)** in the 'TRANSFER' **mask** (see Section 4.3.1.3) to transfer all the jobs to the **CP** 525 user memory.

Performa **CP 525** cold **restart by pressing Fl in** the `TRANSFER* mask.

Switch the **CP** 525 to **"RUN".** The red **LED** for device interface **1** (IF 1) must **now go** off, If it does not, check that the interpreter and procedure are actually present.

6.2 STEP 5 User Program for PC 1 (R Processor)

The **FETCH** job is **trigger** d by a FETCH DIRECT call with a job number other than 0, in this case, job number 2. The job number in the FETCH DIRECT block and the number of the **job** in the **CP** 525 must be the same. The job is to be carried out on the positive going edge at digital input I 0.2.

Interface number SSNR "2" and job number A-NR "2" must be programmed. Select the flag word "FW 20" as the condition codeword ANZW and "FY 24" as the parameter assignment error byte PAFE. The condition codeword indicates the job status and any error messages.

The FETCH **DIRECT** requires details of the destination of the data in PC 1. The data destination should be the data block (dest. type QTYP "DB") with the number DBNR "10". The start of the destination ZANF is data word "5" and the **destination** length ZLAE is "4" data words.

The FETCH job is triggered with the result of logic operation (RLO) 1. If RLO equals 0 when the call is made, only the condition codeword is updated

The tennination of the job with an error and the ∞ currence of a parameter assignment error is indicated at digital output Q 0.2. In this situation, an **error** evaluation program tailored to the particular application is usually called. If an error occurs, evaluate the error numbers in the parameter assignment error byte **PAFE**, condition codeword **ANZW** and error message area in the **SYSTAT** (see also Section 10).

The data sent from PC 2 (150U) is written to the Reprocessor by the CP 525 of PC 1 (135U) using RECEIVE ALL. The interface number SSNR must also be "2". The job number A-NR is "O". Select condition codewordANZW "FW25° and parameter assignment error byte PAFE "FY 27". In the AIL function the condition codeword is only one word long. The lower order byte (FY 26) indicates the number of the job for which the RECEIVE ALL is currently transferring data from the CP to the CPU; in this case job number 2.

The **RECEIVE** AIL can be celled up with any RLO.

From the STEP 5 presetting mask, call the **program** file "B:CLPCIRST.S5D" and add the following statements to the organization block OB 1:

:AN I 0.2 :R F 2.2 :A I 0.2 :AN F2.2 :S F 2.2	EDGE EVALUATION: THE RESULT OF LOGIC OPERATION (RLO) IS SET FOR ONE CYCLE IF THE SIGNALAT INPUT I 0.2 CHANGES FROM O TO 1.
:JU FB122 NAME :FETCH	TRIGGER FETCH JOB 2
SSNR : KY0,2 A-NR : KY0,2 ANZW : FW20 ZTYP : KSDB DBNR : KY0,10 ZANF : KF+5 ZLAE : KF+4 PAFE : FY24	INTERFACE NUMBER 2 JOB NUMBER 2 CONDITION CODEWORD DEST. TYPE DATA BLOCK DEST. DATA BLOCK DB 10 DEST. START DATA WORD 5 DEST. LENGTH 4 DATA WORDS PARAMETER ASSIGNMENT ERROR BYTE
:0 F 21.3 :0 F 24.0 := QO.2	JOB 2 TERMINATED WITH ERROR OR PAMMETER ASSIGNMENT ERROR INDICATED AT OUTPUT Q 0.2 RECEIVE ALL CALL
W :KEC-A SSNR : KY0,2 A-NR : KY0,0 ANZW : FW25 PAFE : FY27	INTERFACE NUMBER 2 JOB NUMBER O CONDITION CODEWORD PARAMETER ASSIGNMENT ERROR BYTE

Data block DB 10 was programed with sufficient length in the **previous** example.

Transfer the extended $OB\ 1$ to the user memory of your R processor. Carry out a cold restart on the R processor. The green LED (RUN) should light up.

39

•

6.3 CP 525 User Program for PC 2 (15)U)

Nothing is changed in the CP **525** user program in Section **4.4** since no job is required in the passive PC **2.**

Switch the mode selector on the CP 525 to "RUN". The red LED on the device interface 1 of the CP 525 should now go out. If it does not, the interpreter and/or procedure is missing or there is a CP 525 hardware fault.

6.4 STEP 5 User Program for PC 2 (150U)

The transfer of the data requested by PC 1 (135U) from the CPU (data **block DB** 11) to the **CP** 525 in PC 2 (150U) is performed by a SEND ALL call. The interface number SSNR must be '0". The job number A-NR is "O". Select "FW14" as the condition codeword **ANZW**. In ALL functions the condition codeword is one word long. The coordination flag byte number of the job for which the data is fetched is indicated in the lower order byte (J?Y 15). In this example there are no coordination flags in the FETCH job for the CP 525 in PC 1 (135U) so FFH appears here.

CP 525 specifies the data source for the SEND ALL. No source need be specified in the **handling** block (specify source type **QTYP** "NN"). Thenext three parameters - data **block number DBNR**, source start **QANF and source length QLAE - are then automatically not** evaluated.

The parameter assignment error byte PAFE is flag byte "FY 16".

Connect the interface module AS 511 with the **PG** 6 35 and call up the program file "B: :CLPC:SST.S5D" in the STEP 5 p :esettings mask.

40

....

یک سو رو د و د را دیوه الیک عمر و یه جانب یه

Organisation block OB 1 must be extended as follows:

:JU	FB180	SEND ALL CALL
NAME :SE	ND	
SSNR :	KY0,0	INTERFACE NUMBER O
A-NR :	KY0,0	JOB NUMBER O
ANZW :	FW14	CONDITION CODEWORD
QTYP :	KSNN	NO SOURCE SPECIFIEL
DBNR :	KY0,0	NOT EVALUATED
QANF :	КҮ+0	NOT EVALUATED
QLAE :	КҮ+0	NOT EVALUATED
PAFE :	FY16	PARAMETER ASSIGNMENT ERROR BYTE

You already programmed the data block **DB11** in the previous **example**.

Transfer the extended OB 1 to the user memory of your **S5-150U**. **Carry out** a cold restart The green LEO (RUN) must light up.

6.5 **Test**

Observe the data words DW 0 to DW 3 and DW 5 to DW 9 of data block DB 10 in PC 1 (135U) using the PC 685 online *function* CONTROL VARIABLE.

Switch the digital input I 0.1 and then input I 0.2 on PC 1 from O to 1. Four data words from data word DW O to DW 3 are then sent to PC 2 (150U). Data words DW 5 to DW 9 are transferred back (see also Section 5). The values of data words DW O to DW 3 must now be located in data words DW 5 to DW 9.

If you do not have any digital 1/0 **modules** with a simulator available, then use **flags** F 0.1 and F 0.2 instead of the inputs I 0.1 and I 0.2. Set (or reset) the flags with the **PG** 685 online function CONTROL **VARIABLE.** Observe the condition codewords FW 11 and FW 20 and the **parameter** assignment error bytes FY 15 and **FY** 24 with **CONTROL VARIABLE.**

7 SEND Job, Flags to Data Block

A SEND job is to be **programmed** in the opposite direction, from PC 2 to **PC** 1.

In every second program cycle, flag word 8 (i.e. flag bytes 8 and 9) are to be transferred from PC 2 (150U) to data block DB 10 - data word DW 4 - in PC 1 (135U).

The following program structure is required:



*) DT = DATA TRANSFER?

The interpreter and $proc \ dure$ are already loaded in the user memories of the two CP 5?5s (see Section 4) .

The SEND job needs to be programed only on the CP 525 of the active PC. Active means that this PC initiates the data exchange and that you need to **program** a SEND job only for the CP 525 in PC 2.

The SEND job is initiated when the CPU calls the handling block SEND DIRECT with the job number 'n'. Select job number '1'. The same job number 'n' is also given to the corresponding job on the CP 525. This job contains information about the destination of the data in PC 1 (135U).

The CP 525 fetches the data only when called by tl.e handling block SEND ALL which transfers the data from the CPU/FW8 to the CP 525 in PC 2.

The CP 525 in PC 1 (135U) receives the data and calls the **RECEIVE** ALL to store it in the destination indicated by PC 2.

The **SEND ALL** (also **known** as SEND 0) has the job number A-NR 0. It is called unconditionally? in every program cycle **and** checks whether the **CP** 525 (dual-port RAM) requires data. If the **CP** does, the SEND **ALL** fetches it from the specified source and transfers it **to** the **CP** 525. If the **CP** does not, the function block is exited and the cyclic program continues.

The **RECEIVE ALL** (also known as **RECEIVE 0**) has the job number A-NR O. It is called unconditionally in every program cycle and checks whether the **CP** 525 (dud-port RAM) wishes to transfer data to the CPU. If **it** does, the RECEIVE ALL stores the data in the destination address specified by the **CP** 525, If it does not, the **func**tion block is exited **and** the cyclic **program** continues.

The content of flag word FW 8 is to be incremented by l in each program cycle. In this way it is easy to recognize in PC 1 that the SEND job has been carried out successfully.

7.1 CP 525 User Program for PC 1 (13:50)

Nothing is changed in the CP 525 user program since no job is required in the passive JC 1.

Switch the mode selector on the CP 525 to "RUN". The red LED on the CP 525 should now go out, If this is not the case, the interpreter and/or procedure is missing or there is a CP 525 hardware fault.

7.2 STEP 5 User Program for PC 1 (Reprocessor)

The transfer of the data received by the CP 525 to data block DB 10 is performed by the RECEIVE ALL function which is called up cyclically. Since this, and the data block DB 10, were programmed in the previous examples, nothing needbechanged in the STEP 5 programon the R processor.

Switch the R processor to "RUN". The green LED must light up.

7.3 CP 525 User Program for PC 2 (15)U)

Switch the mode selector on the CP 525 to "PCR" and call the program "CL512PC2" in the COM 525 mask 'PROGRAM SELECTION'.

Press F1 (PROCRAM USER DATA) in the 'SELECTION' mask to call up the 'JOB BLOCK'. Enter the job number with which this job is to be triggered by the CPU. Select 'Job no.:'"1".

Press F5 (PROGRAM JOB) to call the 'PROGRAM JOB' rask.

Using F7 (HELP) enter 'job:' "SEND" and 'job type' 'DATA BLOCK". The 'dest. - word address' is "4" D (decimal) and the data block number is "10". No entry is necessary for the CPU number **as PC 1** (135U) is in single **processor** operation. Do not specify any coordination flags.

The job is written to the RAM memory of the PG 685 by pressing F6 (ENTER JOB). With F8 (EXIT) you can return to the 'JOB BLOCK' mask. With F6 (SAVE JOB BLOCK) the SEND job is written to the hard disk.

F8 (EXIT) returns you to the 'SELECTION' mask,

15 5.56

-> SELECTION -> JOE PROGRAM JO	BLOCK ·> B		S	IMATIC S5 / COM525
	DRI VE:	B PROGRAM	CL512PC2	COMPONENT: CL
	J	0 B		
Job no.:	001			
Job:	SEND			
Job type	DATA BLOCK			
CPU no.:				
OB no.:	010			
Dest word addr	ress: 00004 D	0004 H		
lf required wi	th coordination	flag: .		
F 1 F 2 ON PAGE PRINTER BACKWARDS	F 3 F 4 PAGE FORWARDS	F 5 DELETE JC%	F 6 ENTER JOB	F7 F8 HELP EXIT

Transfer the job with F2 (TRANSFER) and F3 (FD \rightarrow CP) to the user memory of the CP 525. This is similar to transferring the interpreter and procedure (see Section 4.4.1.2).

Ress F6 (JOB **BLOCK)** in the '**TRANSFER**' mask (see Section 4,4.1.2) to transfer all the jobs to the **CP** 525 user memory.

Perform a CP 525 cold restart by pressing F1 in the 'TRANSFER' mask.

Switch the **CP** 525 to **"RUN"**. The red LED for device interface **1** (IF 1) must **now go** off. If it does not, check that the interpreter and procedure are actually present.

7.4 STEP 5 User Program for PC 2 (15JU)

The SEND job is triggered by a SEND DIRECT call, with a job number other than O, in this case job number 1. The job number in the SEND DIRECT must be the same as that of the job in the **CP** 525. The job is to be carried out in each second program cycle,

The interface **number** SSNR "O" and job number A-NR "1" must be programmed. Select condition codeword ANZW "FW 20" and parameter assignment error byte PAFE "FY 25". The condition codeword requires two words. The job status and any error messages are indicated in flag word FW 20. The number of pieces of data transferred during the current SEND ALL call is indicated in flag word FW 22.

In the handling block there must be details about the source of the data in the CPU for the SEND job. The data source should be flag bytes (source **type** QTYP "FY"). The **parameter** data **block DBNR** is therefore not evaluated. The source start **QANF** is flag byte "8" and source length **QIAE** is "2" flag bytes,

The SEND job is triggered with the result of logic operation (RLO) 1. If RLO equals 0 when the call is made, only the condition codeword is updated. Inverting the flag bit F 3.0 means that the RLO is 0 in one cycle and 1 in the next.

The termination of the job with an error is indicated in the condition codeword in bit 3. The occurrence of a parameter assignment error can be recognized by the setting of bit 0 in the parameter assignment error byte. In this situation, an error evaluation pro-gram tailored to the particular application is usually called. If an error occurs, evaluate the error numbers in the parameter as-signment error byte PAFE, condition codeword ANZW and error roes-sage area in the SYSTAT (see also Section 10).

The actual data transfer of the flag to the CP 525 is carried out by the SEND AIL call. This was programed in the last example.

47

ی بر در ماند و مربع از درو

B8576540-03

Connect the PG 685 with the programmer interface module AS 511, and call up the program file "B:CLPC25ST.S5D" in the STEP 5 presetting mask. Extend the organization block OB 1 with the following statements.

	:L :L :+F :T	FW8 KF+1 FW8	INCREMENT CONTENT OF THE FLAG WORD BY 1 IN EACH PROG. CYCLE
	: AN :=	F 3.0 F 3.0	INVERT FLAG BIT IN EACH PROGRAM CYCLE
NAME	: JU : SEN	FB180 D	TRIGGER SEND JOB 1
SSNR	:	KY0,0	INTERFACE NUMBER O
A-NR	:	KY0,1	JOB NUMBER 1
ANZW	:	FW20	CONDITION CODEWORD
QTYP	:	KSMB	SOURCE TYPE FLAG
DBNR	:	KY0,0	NOT EVALUATED
QANF	:	KF+8	SOURCE START FLAG BYTE 8
QLAE :		KF+2	SOURCE LENGTH 2 FLAG BYTES
PAFE :		FY24	PARAMETER ASSIGNMENT ERROR BYTE

Transfer OB 1 to the user memory of your S5 -150U. Perform $a\,\,{\rm cold}\,\,$ restart on the PC. The green LED (RUN) must light up.

7.5 **Test**

Observe the data word DW 4 in data block DB 10 on PC 1 (1350) using the PG 685 online function CONTROL VARIABLE. The value in the data word must increase constantly.

48

فالإلى والمستان والماسيات

•

** *

8 Coordination Flags

Coordination flags with a SEND job

By using coordination flags you can protect the data destination of a SEND job (if this **is** a data block) from being overwritten. The coordination flag is specified in the SEND job. The byte and bit number of this flag are transferred to the partner along with the data.

The byte number of the coordination flag appears in the destination PC for one cycle in the condition codeword of the RECEIVE ALL. Based on this byte number, you can recognize in the destination CPU which job is responsible for the data that has arrived. If you set the coordination flag belonging to the job in the STEP **5** user program, you can prevent the job from being repeated. **As** seen as the data **from** the destination area is saved or otherwise processed, you can have the user program reset the coordination flag; the job is then free to be executed again.

Coordination flags with a FETCH job

By using coordination flags you can also protect the data source of a FETCH job (only possible for **DB** or DX) in the source CPU from being read. A coordination flag is specified in the FETCH job of the **CP** 525. The byte and bit number of **this** flag are transferred in the request telegram to the partner.

The byte number of the coordination flag appears in the PC in which the data source is located. It appears for **one** cycle in the **condition codewor**doftheSENDAIL Basedonthisbytenumberyou can recognise when and by which job data is requested. If you set the coordination **flag belonging** to this job in the STEP 5user program, **you** can prevent the source data area for this job being read. Assoonasthedatain **the source area is ready**, you can have the coordination flag reset; data is then once again transferred in response to the request telegram from the job. Note that in the condition codeword of the SEND ALL or **RECEIVE** ALL only the byte and not the bit number of the coordination flag appears. You must therefore make sure that the byte number allocated to a coordination flag is unique if you wish to evaluate the information in the condition codeword in the program. You can use any flag as a coordination flag, however, only byte numbers from 1 to 223 can be indicated in the condition codeword.

The coordination flags must be declared in the STEP 5 program as **IPC** output flags and enabled on the **CP** 525 with jumpers at **jumper** block 25. The **IPC** flags (and therefore also the coordination flags) are part of the normal flag area.

IPC flags with the R processor

The IPC flags must be entered in data block $DB \ 1$ as IPC outputs. As soon as DB 1 is programmed, the digital 1/0 bytes must also be entered. The IPC flags are then handled by the operating system as digital I/Os. Before the cyclic program is run, the IPC input flags are transferred from the dual-port RAM of the CP to the flag area of the CPU. IPC input flags are not used with the CP 525. At the end of the cyclic program IPC output flags are transferred from the flag area of the CPU. RAM.

IPC flags with the S5-1150

The data block DB 1 must also be programmed. Digital I/0s do not need to be entered with the S5-115U.

IPC flags with the S5-1500

The **IPC** flags must be transferred in each program cycle between the **CPU and** the dual-port **RAM of** the **CP** 525 by function **block FB** 186 or by system **commands.FB** 186 is supplied with the handling blocks. ADB 1 for exchanging **IPC** flags and/or digital **I/Os** does **not exist in the S5-150U**.

In every second program cycle, PC 2 (150U) triggers a SEND job (job number 1). It is transferred to data **block DB** 10 - data word DW 4 in PC 1 with the flag word FW 8. The job was programed in Section 7 and only needs to have the details of the coordination flag F 5.3 added. The content of the flag word is incremented by 1 in each program cycle.

× , a

The program in PC 1 (R processor) recognises from the condition codeword of the RECEIVE ALL that data with the coordination flag byte number5fromaSEND jobhasarrived. CoordinationflagF 5.3 is then set. The jobs with the same coordination flag numbers (same byte and bit number) are then rejected with the error number 32H in the reply telegram, 9H in the condition codeword and 30H in the SYSTAT. Therefore data word DW 4 is protected from being overwritten.

With digital **input** I 0.7 you can reset the coordination flag again so that the SEND job is again accepted.

8.1 CP 525 User Program for PC 1 (135U)

The CP 525 program does not need to be changed. In any case no job is required on PC 1. The IPC flag bytes 0 to 31 **have** already been enabled by means of jumpers (see Section 3).

Switch the **CP** 525 to "RUN". The red LED on the **CP** 525 **must** go **out.** If this is not the case, either the interpreter and/or procedure is missing or there is a hardware fault on the CP 525.

8.2 STEP 5 User Program for PC 1 (R Processor)

The transfer of the **data** received by the CP **525** to the **data** block **DB** 10 is performed by the RECEIVE ALL, which is **called up** cyclically. The RECEIVE **ALL** and data block **DB** 10 have already been programmed.

As soon as data arrives from the SEND job (with job number 1) of PC 2 the coordination flag byte number appears in the lower order **byte** of the RECEIVE ALL condition codeword. If the byte number is 5, the coordination flag F 5.3 is set and SEND jobs from PC 2 are rejected.

The coordination flag is reset by a positive edge at digital input I 0.7 allowing the job to be accepted again.

Following the next SEND job from PC 2 (with job number 1) the coordination flag is set again immediately.

Extend organization block OB 1 as follows:

:L FY26	COORDINATION FIAGBYTENUMBERFROM
:L KF+5	CONDITION CODEWORD OF THE RECEIVE
:!=F	ALL EQUALS 5? IF YES, THEN SET
:S F 5.3	COORDINATION FLAG
:	AS LONG AS THE COORDINATION FLAG IS SET, THE DATADESTINATIONIS PROTECTED FROM BEING OVERWRITTEN, THE SEND JOB WITH COORDINATION FLAG F 5.3 WILL EJECTED.
:AN 10.7	EDGE EVALUATION:
:R F 2.7	THE RESULT OF LOGIC OPERATION
:A I 0.7	(RLO) IS SET FOR ONE CYCLE, IF
:AN F 2.7	THE SIGNALAT I 0.7 CHANGES FROM
:S F 2.7	0 TO 1.
R F 5.3	RESET COORDINATION FLAG WITH 0->1 EDGEATINPUTI0.7

 $\sim 10^{-10}$

ショ・モンロー・マトト 愛 かち 御知 ・・・マ

TIMER BLOCK LENGTH

IPC output flag byte 5, input byte 0 and output byte 0 must be entered in data $block\ DB\ 1$ with the STEP 5 programming function INPUT (F1) MASK (F4) :

DB 1 ASSIGNMENT OF 1/0s

DIGITAL INPUTS	:,	Ο,	,	,	,	•	•	•
DIGITAL OUTPUTS	· ,	0,	,	,	,			
IPC FLAG INPUT'S	:,		;	, ,	,			
IPC FLAG OUTPUTS	:,	5:	,	,	,	•		

: .

Transfer the extended OB 1 and DB 1 to the R processor. Perform a cold restart on the R processor. The green **LED** must now light up.

8.3 CP525 User Program in PC2 (150U)

Switch the mode selector of the CP 525 to "PGR" and call up the program "CL512PC2" in the COM 525 mask ' <code>PROGRAM SELECTION'</code> .

Using **F1** (PROGRAM USER DATA) in the 'SELECTION' mask, you call up the 'JOB BLOCK' mask. Enter the job number "l" of the SEND job programmed in Section 7.3.

Press F5 (PROGRAM JOB) to call up the 'PROGRAM JOB ' mask.

Leave the destination parameters as they are and enter the coordination flag F ``5. ~3`` .

• بعرب و • • راه • RK 512 Example of Application

	DRI VE:	В	PROGRAM:	CL512PC2	COMPONENT:	CL
		J 0	В			
Job no.:	001					
Job:	SENO					
Job type	DATA BLOCK					
CPU no.:						
OB no. :	010					
Source - word a	ddress: 00004 D	0004	, H			
If required wit	h coordination flag:	005.	3			
F1 F2	F3 F4		F 5	F 6	F7 F	8
UN PAGE RENTER BACKWARD	PAGE OS FORWARDS		JOB	ENTER JOS F	HELP E	XIT

The job is entered in the **RAM** of the **PG** 685 with F6 (ENTER JOB). With F8 (EXIT) you rem to the 'JOB **BLOCK'** mask. With F6 (SAVE JOB **BLOCK)** the SEND job is written to the hard disk. Return to the 'SELECTIONI mask with F8 (EXIT).

Transfer the job in the 'SELECTION' mask with F2 (TRANSFER) and F3 (FD -> CP) to the user memory of the CP 525. This is similar to transferring the interpreter and procedure (see Section 4.4.1.2). Ress F6 (JOB BLOCK) in the 'TRANSFER' mask (see Section 4.4.1.2) to transfer all the jobs to the CP 525 user memory. Perform a CP 525 cold ~ by pressing F1 in the 'TRANSFER' mask.

Switch the **CP** 525 to **"RUN"**. The red **LED** for device interface 1 (IF 1) must now go off. If it does not, check that the interpreter and procedure are actually present.

8.4 STEP 5 User Program for PC 2 (150U)

Nothing needs changing on the previous STEP 5 program. The SEND job with job number 1 is triggered as before (see Section 7.4).

Switch the CPU to "RUN". The green LED must light up.

8.5 Test

Observe the data word DW 4 in data block DB 10 on PC 1 (135U) with the PG 685 online function CONTROL VARIABLE. The value does not change.

Switch input I 0.7 from 0 to 1. The **value** in data word **DW** 4 must change once and then remain constant since the coordination flag is set **immediately** after the job is accepted.

Also observe condition codeword FW 20 of the SEND job with job number 1 on the S5-150U. If the coordination flag is set in the R processor, this SEND job is rejected by PC 1 (135U) and KM = 00001001 00101000 appears briefly in the condition codeword. This means that the job is completed with error 9H, DB or DX inhibited by coordination flag. The error number 30H appears in the error message area of the SYSTAT (see Section 10) .

9 PSEUDO WRITE Job, Input to Data Block

You can have the source and destination of a job specified by the STEP 5 program and change them while the program is running. For this purpose there are special jobs on the CP 525 with the reserved job numbers 190 to 199. To dist 'inguish them from normal SEND and FETCH jobs, these functions are known as PSEUDO WRITE and PSEUDO READ. These functions have nothing to do with the READ/WRITE functions of the handling blocks. In the READ/WRITE functions, neither CPU numbers nor coordination flags can be specified.

No job needs to be programmed CM the CP 525.

The source and destination must be specified in a data block or extended data block (the parameter data block) as follows

n	:	KS= QTYP	source type (not XX, RW or NN)
n	+1:	KY= DBNR	data block number for QTYP DB or DX
n	+ 2 :	KF= QANF	source start address
n	+3:	KF= QLAE	source length SEND (any length for
			FETCH)
n	+ 4 :	KS= ZTYP	dest. type (only DB, DX, RS, AS)
n	+ 5 :	KY= DBNR	data block number for ZTYP DB
n	+ 6 :	rf= Zanf	dest. start address
n	+7:	KF= ZLAE	dest. length FETCH (any length for
			SEND)
n	+ 8 :	KY= BYTE, BIT	coordination flag byte and bit number
n	+9:	KF= CPU No	CPU number in partner

The data type DB or **DX**, the data block number and the start **word** address n of the **set** of parameters must be specified when the **SEND** DIRECT or FETCH DIRECT is called. The **CP** 525 automatically fetches the 10 data words with the transfer **parameters**.

The **PSEUDO WRITE function** is triggered by a SEND DIRECT call with a job number between 190 and 199. The special job **first** fetches the source and destination parameters **from** the parameter data block specified with the SEND DIRECT by means of a SEND ALL. With a second SEND ALL call, the **CP** 525 fetches the data to be transferred from the source specified **in** the parameter data block

and transfers it to the **partner**. At the partner the data is stored under the specified destination address in the parameter data block by a RECEIVE ALL.

The **PSEUDO** READ function is triggered by a FETCH **DIRECT** call with a job number between 190 and 199. The special job first fetches the source and destination parameters from the parameter data block specified with the FETCH DIRECT by means of a SEND AIL. The **CP** 525 then sends a request telegram with the source address specified in the parameter data block to the **partner**. This uses a SEND ALL to fetch the data from the data source and sends it to the requesting CP 525. The **CP** stores the data using a RECEIVE **ALL** under the address specified in the parameter data block.

For example, input byte **IB** O in **in** PC 1 **(135U)** is sent to data word **DW** O in data block DB 11 of PC 2 **(150U)** using a PSEUDO WRITE job.

The interpreter and procedure are already loaded in the user memory of both CP 525s (see Section 4).

The special jobs 190 to 199 for the PSEUDO READ and PSEUDO WRITE jobs are permanently installed on the **CP** 525 so they do not need programming and transferring to the CP user memory. The job numbers 190 to 199 are **reserved** in the **computer** link for PSEUDO READ and PSEUDO **WRITE** functions.

The PSEUDO WRITE job is triggered by calling SEND DIRECT. Select the job number 190. The **CP** 525 first fetches the source and destination parameters from the parameter data block by means of a SEND AIL. Then the **input** byte IB O is fetched with a second SEND ALL and transferred to the **CP** 525 in PC 2 (150U).

The **CP** 525 in PC 2 (1500) receives the data and stores it in the destination specified by PC 1 (1350) by means of a RECEIVE ALL.

When only one byte is transferred to a data word, the CP 525 writes the data into the left **data** byte DL and overwrites the right **data** byte DR with O.

به برین است به ب

•



') DT = DATA TRANSFER

The **SEND ALL** (also known as SEND O) has the job number A-NR O. It is called unconditionally in **every** program **cycle** and checks whether the **CP** 525 (dual-port RAM) **requires** data. If it does, the SEND ALL fetches it from the specified address and transfers it to the **CP** 525. If the **CP** does not require data, the function block is exited and the cyclic program continues.

58

ж.,

The RECEIVE ALL (also known as RECEIVE 0) has the job number A-NR 0. It is called unconditionally in every program cycle and checks whether the CP 525 (dual-port RAM) wishes to transfer data to the CPU. If it does, the RECEIVE ALL stores the data in the destination address specified by the CP 525. If it does not, the function block is exited and the cyclic program continues.

9.1 CP 525 User Program for PC 1 (135U)

The job 190 is already available on the CP 525 and does not need tobeprogrammed.

Switch the CP 525 to "RUN". The red LED on device interface 1 (IF 1) must now go off. If this is not the case check whether the interpreter and procedure actually exist.

9.2 STEP 5 User Program for PC 1 (R Processor)

The PSEUDOWRITE job is to be triggered by **a signal change** from O to 1 at digital input I 0.3.

The PSEUDO WRITE job is triggered by a SEND DIRECT call; i.e., the handling block SEND with a job number in the range between 190 and 199 is called. Select job number 190.

The interface number SSNR must be programmed as "2" and the job number A-NR "190". Select "FW 30" as the condition codeword ANZW and "FY 34" as the parameter assignment error byte PAFE. The condition codeword requires two words. The flag word FW 30 indicates the job status and any error messages. The number of data to be transferred with each SEND ALL call is located in flag word FW 32.

and the second second

The start address of **th**: transfer parameters **mus**: be specified for the PSEUDO **WRITE** job. The parameters should **be** located in the data block (QTYP "DB") with the number **DBNR** "10" from data word **QANF** "21". The **CP** 525 automatically requests 10 data words as transfer parameters; the parameter QLAE is therefore not **eval**uated.

The SEND job is called unconditionally and triggered by result of logic operation (RLO) 1. If the RLO is 0 at the time of the call, only the condition codeword is updated.

The termination of the job with errors and the occurrence of **a** parameter assignment error are indicated at digital output Q 0.3. In this situation an error evaluation program tailored to the particular application is usually called. If an error occurs, evaluate the error numbers in the parameter assignment error byte **PAFE**, condition codeword **ANZW** and the error message area in the **SYSTAT** (see also Section 10).

Connect the **PG** 685 with the R processor and call up the program file "**B:CLPCIRST.S5D**" in the STEP 5 presetting mask. Extend the organization block OB 1 with the following statements.

	: AN	I 0.3	EDGE EVALUATION:
	:R	F 2.3	THE RESULT OF LOGIC OPERATION
	:A	10.3	(RLO) IS SET FOR ONE CYCLE, IF THE
	: AN	F 2.3	SIGNALAT INPUT I 0.3 CHANGES
	:S	F 2.3	FROM 0 TO 1.
	:		
	:JU	FB120	TRIGGER SEND JOB 190
NAME	: SEN	D	
SSNR	:	KY0,2	INTERFACE NUMBER 2
A-NR	:	KY0,190	JOB NUMBER 190
ANZW	:	FW30	CONDITION CODEWORD
QTYP	:	KSDB	SOURCE TYPE DATA BLOCK
DBNR	:	KY0,10	PAMMETERFROMDATABLOCKDB10
QANF	:	KF+21	START OF PARAMETER BLOCK DW 21
QLAE	:	KF+0	NOT EVALUATED
Ρm	:	FY34	PARAMETER ASSIGNMENT ERROR BYTE
	:		
	:0	F 31.3	JOB 190 TERMINATED WITH ERROR
	:0	F 34.0	OR PARAMETER ASSIGNMENT ERROR
	;=	Q0.3	INDICATE AT OUTPUT Q 0.3

्रेडर संदर्भ

The SEND AIL call required for transferring the transfer parameters and the input byte are already programmed.

Program the source and destination in data block DB 10:

21 :	KS= IB	QTYP source type input byte
22 :	KY= 0,0	DBNR DB number not necessary
23 :	KF= +0	QANF source start is input byte IB O
24 :	KF= +1	QLAE source length is 1 Syte
25 :	KS= DB	ZTYP dest. type data block
26 :	KY= 0,11	DBNR dest. data block is DB 11
27 :	KF= +0	ZANF dest. start is data word DW 0
28 :	KH= FFFF	ZLAE irrelevant with SEND
29 :	KY= 255,255	no coordination flag
30 :	KF= +0	CPU number

Transfer organization block OB 1 and data block DB 10 to the user memory of your ${\bf S5-135U}.$ Perform a cold restart on the PC. The green LED (RUN) must light up.

9.3 CP 525 User Program for PC 2 (150U)

Nothingneeds changing inthe existing CP525 program, since no job is required on PC $\mathbf{2}.$

Switch the **CP 525** to "RUN". The red **LED** on device interface 1 on **CP 525 must** now go off. If it does not, check whether the interpreter and procedure **actually** exist or if there is a **CP** 525 hardware error.

2000 C

9.4 STEP 5 User Program for PC 2 (150U)

The transfer of the data received by the CP 525 to data block DB 11 is carried out by the RECEIVE **ALL** which is called cyclically. RECEIVE ALL and data block DB 11 are already programmed; therefore nothing needs to be changed in the STEP 5 pr-.

Switch the CPU to "RUN". The green RUN LED must light up.

9.5 **Test**

Observe the data word DW 0 of data block DB 11 in PC 2 (150U) with the PG 685 online function CONTROL VANABLE.

Switch the digital input I 0.3 on PC 1 (135U) from O to 1. The bit pattern of input byte IB O appears in the higher order part of data word DW O (left-hand data DL O). The **dats** to the right DR O is always overwritten with **0**.

If you do not have any digital 1/0 modules with a simulator available in PC 1 (135U), then use flag byte FY O instead of the input byte IB O. Set (or reset) the flags with the PG 685 online function CONTROL VARIABLE. Observe the condition codeword FW 30 and the parameter assignment error bytes FY 34 with CONTROL VARIABLE.

B8576540-03

10 Reading the Error Message Area in the SYSTAT

The **SYSTAT** is a memory area on the **CP** 525 that contains status information separately for each interface.

Within the error message area of the **SYSTAT** the causes of errors are coded in detail. The error message area is 4 bytes or 2 words long and appears as follows



E - set, if error is entered in the SYSTAT
O - set, if there is an error overflow (more than 3 errors)
B - set, if there is a BREAK on the interface
R - irrelevant for the computer link

The PC job RECEIVE DIRECT 200 is reserved for reading the error message area. This job is carried out immediately; i.e. , it is not entered in the CP 525 queue and does not require a RECEIVE ALL. No job needs programming on the CP 525.

10.1 STEP 5 user Program for PC 1 (R Processor)

The error information is to be transferred to data block **DB** 10 into data words DW 9 and **DW** 10. The job is called unconditionally **in** every program cycle but executed **only** if an error occurs. It must be called by the result of logic operation **(RLO)** 1.

B8576540-03

RK 512 Example of Application

, the pays and an equilate an ender some some and and and an ender some some some some some some some and the s

When **RESET** 200 is called, the error message **area** of the interface addressed (in this example **SSNR** = 2 for device interface IF 1 of the **CP** 525) is reset. The data words **DW** 9 and **DW** 10 in data block **DB** 10 **in** which the error numbers are written by RECEIVE 200, must be reset separately.

The error message area is reset by a positive going edge at digital input I 0.4.

Extend the **organization block OB 1** by the following statements:

	:0 F 3.0 :ON F 3.0	GENERATE RLO = 1
	:JU FB121	READ ERROR MESSAGE AREA SYSTAT
NAME	:RECEIVE	
SSNR	: KY0,2	INTERFACE NUMBER 2
A-NR	: KY0,200	JOB NUMBER 200
ANZW	: FW35	CONDITION CODEWORD
ZTYP	: KSDB	THE DEST. IS A DATA block
DBNR	: KYO,10	DATA BLOCK NUMBER 10
ZANF	: KF+9	from data word 9
ZLAE	: KF+2	2 DATA WORDS
PAFE	: FY39	PARAMETER ASSIGNMENT ERROR BYTE
	:	
	:O F 36.3	JOB TERMINATED WITH ERROR
	:O F 39.0	OR PARAMETER ASSIGNMENT ERROR
	:= 00.4	INDICATEDAT OUTPUTO0.4
	~	~
	:AN 10.4	EDGE EVALUATION:
	:R F2.4	THE RESULT OF LOGIC OPERATION
	:A T 0 4	(RLO) IS SET FOR ONE CYCLE IF THE
	: AN F2 4	STGNAL AT INPIT I 0.4 CHANGES
	•S F2 4	FROM 0 TO 1
NAME	:JC FB100 :R-SYSTAT	

B8576540-03

Function block FB100

	:JU	FB124	RESET SYSTAT
NAME	:RES	SET	
SSNR	:	KY0,2	INTERFACE NUMBER 2
A-NR	:	KY0,200	JOB NUMBER 200
PAFE	:	FY40	PAWMETER ASSIGNMENT ERROR BYTE
	1		
	:C	DB10	IN DATA BLOCK DB 10
	:L	KF+0	DELETE DATA WORDS DW 9 AND DW 10
	:T	DW9	
	:T	DWIO	
	BE		

Transfer the organisation block OB 1 and function block FB 100 to the user memory of the R processor. Perform a cold restart. The green RUN **LED** must light up.

10.2 **Test**

Observe the data block **DB** 10 **DW** 9 and **DW** 10 with the PG 685 online function **CONTROL** VARIABLE and disconnect the cable connector between the two CP 525s.

In data word DW 9, KH = OEFF appears, and in DW 10, KH = FFFF appears, (BREAK). This error number is signalled repeatedly until the situation has been resolved. For this reason the error buffer in the SYSTAT overflows; bit 2 in the first byte is set. Transfer jobs are terminated with error FH in the ANZW.

Reconnect the two **CP** 525s. Reset the **SYSTAT errcr** message area by switching input I 0.4 from 0 to 1. Check that it has been reset using CONTROL VARIABLE. **Initiate** the **SEND** job with job number 1 on PC **1** at input I 0.1.

Switch the mode selector of the **CP** 525 in **the S5-150U** to STOP and trigger the SEND job with the job number 1 on PC 1 (135U). KH = 0838 appears in data word DW 9 of DB 10, i.e. the error message that the partner **CP** is switched to "PGR" or "STOP".
B8576540-03

RK 512 Example of Application

11 Date/Time PC Jobs

The CP 525 has an internal battery-backed hardware clock that continues to run even if there is a power failure, If necessary it can be used as a calendar and timer. The clock can be set, read and synchronized with the hardware clocks of other CPS in the same PC.

The PC job number 218 is reserved for reading and writing the date and time on the CP 525. This special job is executed immediately; i.e., it is not entered in the job buffer of the CP 525 anddoesnotrequire a SEND ALL or RECEIVE ALL. The job is initiated and the transfer carried out in one call. You do not need to program a job on the CP 525.

11.1 Read Date/Time

The current date and time values are to be transferred cyclically from the CP 525 hardware clock to data block DB 10 from data word DW 11. In DB 10 they are stored in the following format:

11	:	кн= 0100	master identifier
12	:	KH= mmss	1/10 s / 1/100 s / seconds
13	:	KH= mmhh	minutes / hours
14	:	KH= ddmn	day / month
15	:	КH= уу00	year /

When the time is to be transferred to a master, which is to remain a master and which can read the time, the **master identifier must** be set. When the power is switched **on** the master identifier on the **CP** 525 is reset. It is, however, not influenced by the **SYNCHRON function** block. The setting of the master identifier is described in Sections 11.2 and 11.3.

The transfer is to take place in every cycle. The termination of the job with an error is tobe indicated at output Q 0.5.

Extend OB ${\bf 1}$ by the statements below to trigger a corresponding PC job:

	:0	F 3.0	
	:ON	F 3.0	RLO = 1
	:		
	:JU	FB121	
NAME	:RECI	EIVE	
SSNR	:	KY0,2	INTERFACE NUMBER 2
A-NR	:	KY0,218	JOB NUMBER 218
m	:	FW41	CONDITION CODEWORD
ZTYP	:	KSDB	THE DEST. IS A DATA BLOCK
DBNR	:	KY0,10	DATA BU3CK NUMBER 10
ZANF	:	KF+11	FROM DATA WORD 11
ZLAE	:	KF+5	5 DATA WORDS
PAFE	:	FY45	PARAMETER ASSIGNMENT ERROR BYTE
	:		
	:0	F 42.3	JOB TERMINATED WITH ERROR
	:0	F 45.0	OR PARAMETER ASSIGNMENT ERROR
	:=	Q0.5	INDICATED AT OUTPUT Q 0.5

Transfer the extended organization block OB 1 to the R processor and perform a cold restart. The green "RUN" **LED must** light **up**.

Test

Observe data block DB 10 fromDW 11 to DW 15 with the PG 685 online function CONTROL VARIABLE. The date and time are not entered because the master identifier is not set after the CP 525 has started following power up.

The job is terminated without errors in the condition codeword; nothing is entered in the destination data block.

B8576540-03

RK 512 Example of Application

- - - - - - ----

11.2 Setting the Date/Time

Enter the value to be set (31st December, 1987, 23 hours, 58 minutes and 30 seconds) in data block DB 10 from data word DW 16 in BCD format. The CP 525 is to be the master. The master identifier (bit 8) in data word DW 16 must be set (see also Sections 11.1 and 11.3).

16	:	KH= 0100	master identifer set
17	:	KH= 0030	1/10 s, 1/100 s / seconds
18	:	KH= 5823	minutes/hours
19	:	KH= 3112	day/month
20	:	KH= 8700	year/

The set **function** is to be triggered by a signal change from 0 to 1 at input bit I 0.6. If the job is terminated with an error, the error is indicated at output Q 0.6.

Extend **OB 1 with** the statements below to trigger a corresponding PC job:

	:AN	10.6	EDGE EVALUATION:
	:R	F 2.6	THE RLO IS SET FOR ONE CYCLE IF
	:A	10.6	THE SIGNALAT INPUT I 0.6 CHANGES
	:AN	F 2.6	FROM O TO 1.
	:S	F 2.6	
	:		
	:JU	FB120	CALL PC JOB SET TIME
NAME	:SEN	D	
SSNF	? :	KY0,2	INTERFACE NUMBER 2
A-NR	:	KY0,218	JOB NUMBER 218
ANZW	: •	FW46	CONDITION CODEWORD
QTYP	:	KSDB	SOURCE IS ADATA BLOCK
DBNR	:	KY0,10	DATA BI.XX'KNUK8ER10
QANF	:	KF+16	FROM DATA WORD 16
QLAE	:	KF+5	5 DATA WORDS
Ρm	:	FY50	PARAMETER ASSIGNMENT ERROR BYTE
	:		
	:0	F 47.3	JOB TERMINATED WITH ERROR
	:0	F 50.0	OR PARAMETER ASSIGNMENT ERROR
	: ===	QO.6	INDICATEDAT OUTPUTQ0.6

B8576540-03

Transfer organization **block** OB 1 and data block JB 10 to the user memory of the R processor. Perform a cold restart. The green "RUN" LED must light up.

Test

Switch the digital input I 0.6 from 0 to 1. The master identifier, date and time of the **CP** 525 hardware clock is set to the time programmed in data block DB 10. The success of the operation can be recognised by the date and time values that are written cyclically to data words **DW** 11 to **DW** 15; tenths of seconds, seconds and minutes must change **continuously**. Shortly afterwards the **CP** 525 indicates a change of year. The **CP** 525 recognizes a change of year and also leap years; it counts 28 or 29 days in February correctly.

11.3 Synchronizing Several CP Hardware Clocks

To synchronize the hardware clocks of several **CP** modules, the CPU must read the date and time from a **CP** declared as time master and transfer these values **to** other **communi**cations **processors** which **are declared as time** slaves.

Reading the time is carried out with a RECEIVE DIRECT 218 (see Section 11.1) and setting the time with a SEND DIRECT 218 (see Section 11.2).

A CP is declared master by means of a SEND DIRECT 218 if bit8is set in the first data word transferred. If this bit is 0 the CP is a time slave.

If only one data word is transferredwithSENDDIRECT218 (QLAE = 1), only the master identifier is affected; the time remains unchanged. If five data words are transferred with SEND DIRECT 218 (QLAE = 5) the master identifier, date and time are transferred. The hardware clock is set correspondingly and continues to run with the new value.

For example, **suppose** there are three CP 525 modules plugged into an **S5-135U** with R processor. They have the module addresses 0, 2 and 4. The **CP** 525 with module **address** 2 is already declared master and has the correct time.

The date, time and master identifier are read by the master clock (interface 2) in every program cycle in the **comman**d string from Section 11.1 and written into data block **DB** 10 from data word **DW** 11. Before the date and time of the master clock can be transferred to the slave clocks (interfaces 0 and 4), the master identifier in data block 10 data word **DW 11** must be reset. Carry out the **synchronization on a** positive **going edge at input I 0.5**.

To have an **exact** synchronization, the time read **by the master clock must be** transferred to the slaves in the same program cycle; it is, however, **not** essential that the **tw** slaves are synchronized in the same program cycle.

Following the 0->1 edge at input I 0.5, the initiation flags F 3.2 for interface O (SSNR O) and F 3.3 for interface 4 (SSNR 4) are set.

The two set jobs (SEND DIRECT 218) are executed only if the corresponding initiation flags are set and the read job (RECEIVE DIRECT 218) has been executed. You can recognise that the time has been read successfully because the master identifer that is reset in each cycle in DB 10 DW 11 is set immediately following the RECEIVE DIRECT call. Once a slave has been synchronized (complete without error), the initiation flag is reset.

Output Q 0.5 indicates any errors that occur when the error clock is being read. Output Q 0.7 indicates any errors that occur when the slaves are being set. Normally in this situation en error evaluation program tailored to the particular application is called up. In the case of errors, evaluate the error numbers in the condition codewords ANZW, parameter assignment error bytes PAFE and the error message area of the SYSTAT for the different interfaces (see Section 10).

B8576540-03

The program in organization $_{block}\ OB\ 1$ for synchronizing the three CP 525s appears as follows

:0	F 3.0		*)
:ON	F 3.0	RLO = 1	*)
1			
:Ju	FB121		*)
:REC	EIVE		*)
:	KYO , 2	INTERFACE NUMBER 2	*)
:	KYO, 218	JOB NUMBER 218	*)
:	FW41	CONDITION CODEWORD	*)
:	KSDB	DEST. IS A DATA BLOCK	*)
:	KYO, 10	data block number 10	*)
:	KF+11	FROM DATA WORD 11	*)
:	KF+5	5 DATA WORDS	*)
:	FY45	PARAMETER ASSIGNMENT ERROR BYTE	*)
1			
:0	F 42.3	JOB TERMINATED WITH ERROR	*)
:0	F 45.0	OR PARAMETER ASSIGNMENT ERROR	*)
:-	Q 0.5	INDICATED AT OUTPUT Q 0.5	*)
:			
:C	DB10		
:A	D 11.8	IFMASTER IDENTIFIER 1, THEN	
:=	F 3.1	SET "READING SUCCESSFUL" BIT	
∶R	D 11.8	AND RESET MASTER IDENTIFIER	
:AN	I 0.5	EDGE EVALUATION:	
∶R	F 2.5	THE RLO IS SET FOR ONE CYCLE IF	
:A	I 0.5	THE SIGNAL AT I 0.5 CHANGES	
:AN	F 2.5	FROM O TO 1	
:S	F 2.5		
;			
:S	F 3.2	SET INITIATION FLAG SSNR O	
:S	F 3.3	SEX' INITIATION FIAG SSNR 4	
:			
	:0 :ON :Ju :REC : : : : : : : : : : : : : : : : : : :	<pre>:0 F 3.0 :ON F 3.0 :Ju FB121 :RECEIVE : KYO, 2 : KYO, 218 : FW41 : KSDB : KYO, 10 : KF+11 : KF+5 : FY45 : :0 F 42.3 :0 F 42.3 :0 F 45.0 :- Q 0.5 : : C DB10 :A D 11.8 := F 3.1 :R D 11.8 := F 3.1 :R D 11.8 : : S F 2.5 : S F 2.5 : S F 3.2 : S F 3.3</pre>	:0 F 3.0 :ON F 3.0 RLO = 1 :Ju FB121 :RECEIVE :RECEIVE INTERFACE NUMBER 2 :KY0, 218 JOB NUMBER 218 : FW41 CONDITION CODEWORD :KSDB DEST. IS A DATA BLOCK :KY0, 10 DATA BLOCK NUMBER 10 :KF+11 FROM DATA WORD 11 :KF+5 5 DATA WORDS :FY45 PARAMETER ASSIGNMENT ERROR BYTE :0 F 42.3 :0 F 45.0 :0 SET "READING SUCCESSFUL" BIT :R D 11.8 :F SET "READING SUCCESSFUL" BIT :R D 11.8 :AN I 0.5 EDGE EVALUATION: :R F 2.5 :AN F 2.5 FROM 0 TO 1 :S F 3.2 :S F 3.2 :S F 3.3

*) Already programed in Section 11.1

	:				
	:A	F 3.2	IF INITIATION FLAG SET AND		
	:A	F 3.1	DATE/TIME READ. SET SLAVE		
	·.111	FB120	CLOCK SSNR 0		
T 47	·SEND		CLOCK SSAR O		
COND		, 7200 0			
A ND		VVA 910	TOD NUMBER 0		
ADTER	•	KIU, 210	JOB NUMBER 210		
ANZW	:	FMOT	CONDITION CODEWORD		
QTYP	•	KSDB	SOURCE IS A DATA BLOCK		
DBNR	:	KY0,10	DATA BLOCK NUMBER 10		
QANF	:	KF+11	FROM DATA WORD 11		
QLAE	:	KF+5	5 DATA WORDS		
PAFE	:	FY55	PARAMETER ASSIGNMENT ERROR BYTE		
	:A	F 52.2	SET SSNR O COMPLETE WITHOUT ERROR,		
	:R	F 3.2	THEN RESET INITIATION FLAG		
	:A	F 3.3	IF INITIATION FIAG SET AND		
	:A	F 3.1	DATE/TIME READ, SET SLAVE		
			•		
	:JU	FB120	CLOCK SSNR 4		
NAME	: JU : SEN	FB120	CLOCK SSNR 4		
NAME SSNR	: JU :SENI :	FB120) KY0.4	CLOCK SSNR 4 INTERFACE NUMBER 4		
NAME SSNR A-NR	: JU :SENI :	FB120 KYO,4 KYO,218	CLOCK SSNR 4 INTERFACE NUMBER 4 JOB NUMBER 218		
NAME SSNR A-NR ANZW	: JU : SENI : :	FB120 KY0,4 KY0, 218 EW56	CLOCK SSNR 4 INTERFACE NUMBER 4 JOB NUMBER 218 CONDITION CODEWORD		
NAME SSNR A-NR ANZW OTYP	: JU : SEN : : : :	FB120 KY0,4 KY0,218 EW56 KSDB	CLOCK SSNR 4 INTERFACE NUMBER 4 JOB NUMBER 218 CONDITION CODEWORD SOURCE IS A DATA BLOCK		
NAME SSNR A-NR ANZW QTYP DBNR	: JU :SENI : : : :	FB120 KY0,4 KY0,218 EW56 KSDB KY0,10	CLOCK SSNR 4 INTERFACE NUMBER 4 JOB NUMBER 218 CONDITION CODEWORD SOURCE IS A DATA BLOCK DATA BLOCK NUMBER 10		
NAME SSNR A-NR ANZW QTYP DBNR OANE	: JU : SENI : : : :	FB120 KY0,4 KY0,218 EW56 KSDB KY0,10 KF+11	CLOCK SSNR 4 INTERFACE NUMBER 4 JOB NUMBER 218 CONDITION CODEWORD SOURCE IS A DATA BLOCK DATA BLOCK NUMBER 10 FROM DATA WORD 11		
NAME SSNR A-NR ANZW QTYP DBNR QANF OLAF	: JU :SENI : : : : : :	FB120 KY0,4 KY0,218 EW56 KSDB KY0,10 KF+11 KF+5	CLOCK SSNR 4 INTERFACE NUMBER 4 JOB NUMBER 218 CONDITION CODEWORD SOURCE IS A DATA BLOCK DATA BLOCK NUMBER 10 FROM DATA WORD 11 5 DATA WORDS		
NAME SSNR A-NR ANZW QTYP DBNR QANF QLAE PAFF	:JU :SEN : : : : : :	FB120 KYO,4 KYO,218 EW56 KSDB KYO,10 KF+11 KF+5 FY60	CLOCK SSNR 4 INTERFACE NUMBER 4 JOB NUMBER 218 CONDITION CODEWORD SOURCE IS A DATA BLOCK DATA BLOCK NUMBER 10 FROM DATA WORD 11 5 DATA WORDS PARAMETER ASSIGNMENT ERROR BYTE		
NAME SSNR A-NR ANZW QTYP DBNR QANF QLAE PAFE	: JU : SENI : : : : : : :	FB120 KYO,4 KYO,218 EW56 KSDB KYO,10 KF+11 KF+5 FY60	CLOCK SSNR 4 INTERFACE NUMBER 4 JOB NUMBER 218 CONDITION CODEWORD SOURCE IS A DATA BLOCK DATA BLOCK NUMBER 10 FROM DATA WORD 11 5 DATA WORDS PARAMETER ASSIGNMENT ERROR BYTE		
NAME SSNR A-NR ANZW QTYP DBNR QANF QLAE PAFE	: JU : SENI : : : : : :	FB120 KYO,4 KYO,218 EW56 KSDB KYO,10 KF+11 KF+5 FY60	CLOCK SSNR 4 INTERFACE NUMBER 4 JOB NUMBER 218 CONDITION CODEWORD SOURCE IS A DATA BLOCK DATA BLOCK NUMBER 10 FROM DATA WORD 11 5 DATA WORDS PARAMETER ASSIGNMENT ERROR BYTE SET SSNB 4 COMPLETE WITHOUT EPPOP		
NAME SSNR A-NR ANZW QTYP DBNR QANF QLAE PAFE	: JU : SENI : : : : : : : : : : : : : : : : : : :	FB120 KY0,4 KY0,218 EW56 KSDB KY0,10 KF+11 KF+5 FY60 F 57.2 F 3.2	CLOCK SSNR 4 INTERFACE NUMBER 4 JOB NUMBER 218 CONDITION CODEWORD SOURCE IS A DATA BLOCK DATA BLOCK NUMBER 10 FROM DATA WORD 11 5 DATA WORDS PARAMETER ASSIGNMENT ERROR BYTE SET SSNR 4 COMPLETE WITHOUT ERROR, THEN DESET INITIATION FLAC		
NAME SSNR A-NR ANZW QTYP DBNR QANF QLAE PAFE	: JU : SENI : : : : : : : : : : : : : : : : : : :	FB120 KY0,4 KY0,218 EW56 KSDB KY0,10 KF+11 KF+5 FY60 F 57.2 F 3.3	CLOCK SSNR 4 INTERFACE NUMBER 4 JOB NUMBER 218 CONDITION CODEWORD SOURCE IS A DATA BLOCK DATA BLOCK NUMBER 10 FROM DATA WORD 11 5 DATA WORDS PARAMETER ASSIGNMENT ERROR BYTE SET SSNR 4 COMPLETE WITHOUT ERROR, THEN RESET INITIATION FLAG		
NAME SSNR A-NR ANZW QTYP DBNR QANF QLAE PAFE	:JU :SENI : : : : : : : : : : : : : : : : : : :	FB120 KY0,4 KY0,218 EW56 KSDB KY0,10 KF+11 KF+5 FY60 F 57.2 F 3.3 E 52.2	CLOCK SSNR 4 INTERFACE NUMBER 4 JOB NUMBER 218 CONDITION CODEWORD SOURCE IS A DATA BLOCK DATA BLOCK NUMBER 10 FROM DATA WORD 11 5 DATA WORDS PARAMETER ASSIGNMENT ERROR BYTE SET SSNR 4 COMPLETE WITHOUT ERROR, THEN RESET INITIATION FLAG		
NAME SSNR A-NR ANZW QTYP DBNR QANF QLAE PAFE	:JU :SENI : : : : : : : : : : : : : : : : : : :	FB120 KY0,4 KY0,218 EW56 KSDB KY0,10 KF+11 KF+5 FY60 F 57.2 F 3.3 F 52.3 F 52.3	CLOCK SSNR 4 INTERFACE NUMBER 4 JOB NUMBER 218 CONDITION CODEWORD SOURCE IS A DATA BLOCK DATA BLOCK NUMBER 10 FROM DATA WORD 11 5 DATA WORDS PARAMETER ASSIGNMENT ERROR BYTE SET SSNR 4 COMPLETE WITHOUT ERROR, THEN RESET INITIATION FLAG JOB SSNR 0 TERMINATED WITH ERROR		
NAME SSNR A-NR ANZW QTYP DBNR QANF QLAE PAFE	:JU :SENI : : : : : : : : : : : : : : : : : : :	FB120 KY0,4 KY0,218 EW56 KSDB KY0,10 KF+11 KF+5 FY60 F 57.2 F 3.3 F 52.3 F 55.0 F 55.0	CLOCK SSNR 4 INTERFACE NUMBER 4 JOB NUMBER 218 CONDITION CODEWORD SOURCE IS A DATA BLOCK DATA BLOCK NUMBER 10 FROM DATA WORD 11 5 DATA WORDS PARAMETER ASSIGNMENT ERROR BYTE SET SSNR 4 COMPLETE WITHOUT ERROR, THEN RESET INITIATION FLAG JOB SSNR 0 TERMINATED WITH ERROR OR PARAMETER ASSIGNMENT ERROR		
NAME SSNR A-NR ANZW QTYP DBNR QANF QLAE PAFE	:JU :SENI : : : : : : : : : : : : : : : : : : :	FB120 KY0,4 KY0,218 EW56 KSDB KY0,10 KF+11 KF+5 FY60 F 57.2 F 3.3 F 52.3 F 55.0 F 57.3 F 57.2	CLOCK SSNR 4 INTERFACE NUMBER 4 JOB NUMBER 218 CONDITION CODEWORD SOURCE IS A DATA BLOCK DATA BLOCK NUMBER 10 FROM DATA WORD 11 5 DATA WORDS PARAMETER ASSIGNMENT ERROR BYTE SET SSNR 4 COMPLETE WITHOUT ERROR, THEN RESET INITIATION FLAG JOB SSNR 0 TERMINATED WITH ERROR OR PARAMETER ASSIGNMENT ERROR JOB SSNR 4 TERMINATED WITH ERROR		
NAME SSNR A-NR ANZW QTYP DBNR QANF QLAE PAFE	:JU :SENI : : : : : : : : : : : : : : : : : : :	FB120 KY0,4 KY0,218 EW56 KSDB KY0,10 KF+11 KF+5 FY60 F 57.2 F 3.3 F 52.3 F 55.0 F 57.3 F 60.0	CLOCK SSNR 4 INTERFACE NUMBER 4 JOB NUMBER 218 CONDITION CODEWORD SOURCE IS A DATA BLOCK DATA BLOCK NUMBER 10 FROM DATA WORD 11 5 DATA WORDS PARAMETER ASSIGNMENT ERROR BYTE SET SSNR 4 COMPLETE WITHOUT ERROR, THEN RESET INITIATION FLAG JOB SSNR 0 TERMINATED WITH ERROR OR PARAMETER ASSIGNMENT ERROR JOB SSNR 4 TERMINATED WITH ERROR OR PARAMETER ASSIGNMENT ERROR		

The values set could also have been sent to interface number SSNR 1 and 5 since the CP 525 has only one hardware clock for both interfaces.

12 Info

Using the COM 525 programming software you can obtain information about user data which has already been programmed and have it displayed on the PG monitor.

Starting from the 'SELECTION' mask, press F4 to **call** the 'INFO mask. In this mask you can further decide whether you require information about user data on the **CP** or a program file (FD).

The data programmed on the $\ensuremath{\mathbf{CP}}$ for device interface 1 can be output on the monitor.

-> PROG. SELECTION	SIMATIC S5 / COM525		
	STORAGE MEDIUM: INTERFACE: COMPONENT:	C P 1 CL	
	Plant designation: Generated by: Generated on:	application example Fred 07.10.87	
F1 F2 BRIEF	F 3 F 4	F5 F6 JOB BLOCK	F7 F 8

- 1

Press $\ensuremath{\texttt{F1}}$ (BRIEF $\ensuremath{\texttt{DESC.}}$) to obtain a brief description of the $\ensuremath{\texttt{CP}}$ 525 interface 1.

-> PROG. SELECTION I N F O	SIMATIC S	5 / COM525		
	STORAGE MEDIUM: Interface Component	CP 1 CL		
	Plant designation: Generated by: Generated on:	application example Fred 07.10.87		
Data type Ele	ment name Number	Data type	Number	
Interpreter: RK5 Procedure: P39 Print para.: PRII	12 01 1 64R 01 1 NT PARA	Job block	1	
Total number Program length :	3 elements 5225 words			
F 1 F 2 BRIEF DESC.	F 3 F 4	F 5 F 6 JOB BLOCK	F 7 HELP	F 8 EXIT

In this **mask you can read** the interpreter and procedure names as well as the version number and see whether a job block has been **programmed or not. In addition you can see the program length** displayed in words.

Press F5 (JOB **BLOCK)** in the **'INFO' mask to** display a table of all the **programmed** PC job numbers. With **F1** (ON PUNTER) in the **'JOB BLOCK' mask you can have this information printed out on a PT88** or PT89 printer connected to the PG.

B8576540-03

Ē

··········

13 Program Documentation

ana waaddadaa gaalaana ahayyyaana

All the elements of a COM 525 program file can be documented in detail. The printouts can have either a header or a trailer programed that is output on a PT88 printer.

In the 'SELECTION' mask F6 (CONTINUE) and F3 (ASSIGN PRI. PAPA.) call up the following mask

	DRIVE:	в Р	ROGRAM:	CL512PC1	COMPONE	NT:
Printout header:						
SIEMENS SIMATIC S5 Exam COM525 - CP525/524	mple of appli S	cation CO 5-DOS	XM 52.5	- RK512	Page:	
Drive: B Program	: CL512PC1	la	st worke	d with:	07.10.87	
Γιαπι. Ελαπριε υι αρμ	nication	Ge	enerated	by: Free	1	
	olication	Ge	enerated	by: Free		I
Printout trailer:	on carion	Ge	enerated	by: Fred	1	I
Printout trailer: You can write any text	t here.	Ge	enerated	by: Fiel	1	I
Printout trailer: You can write any text	t here.	Ge	nerated	by: Free	1	
Printout trailer: You can write any text	t here.	Ge		- 6	F 7	

For the printout header and trailer you can program any two lines of text.

Ress F6 to save the printout header and trailer in your user program file CL512PC1.525.

Press F6 (CONTINUE) and F4 (LISTING) in the 'Selection mask to call the 'LISTING' mask. Using the keys

- F1 (TOTAL PROGRAM) - F2 (INTERPRT PROCEDURE)

- F4 (JOB BLOCK)

you can list the whole program or parts of it.

Try out the various options.

By pressing F4 the following list of all the programmed PC jobs will be printed out on the printer connected to the PG 685:

00110140-00

! SIEMENS S	CP525/524	Example OF RK	appli cati 512	on COM 525 P	10.30s7
! Dr ive: B !Plant: app]	Progr ication • .wr]	am: CL512PC	l Last Gene	worked with: rated by: Fr	10.30.87
JOB BL	0СК				
Name: JOB B	LOCK				
A-NR. JOP	JOE TYPE	CPU NO. DI	₿∕ sour	CE ADDR.DEST	. ADDR. COORD.

 Mol
 Send
 Data
 block
 011
 0004
 00004

 0D2
 Fetch
 Data
 block
 011
 0004
 00004

1 You can write any text bere

SIEMENS

We have checked the contents of this manual for agreement with the hardware and software described. Since deviations cannot be precluded ● ntirely, we can-not guarantee lull agreement. However, the data in this manual are reviewed regularly • nd any necessary cor-rections included in subsequent editions. Suggestions for improvement are welcomed. Technical data subject to change.

The reproduction, transmission or use of this document or its contents is not permitted without • xpress written authority.

Offenders will be liable for damages. All rights, in-cluding rights created by patent grant or registration of a utility model or design, are resewed

Copyright Siemens AG 1989 All Rights Reserved

Siemens Aktiengesellschaft

C79000-B8576-C540-05 Gerätewerk Karlsruhe Printed in the Federal Republic of Germany

SIEMENS

SIMATIC S5 CP 525: Event Output and Listing with the PT88/PT89 Printer

Example of Application

Order No. C79000-B8576-C542-05



Contents

1	Introduction	1
2	Aims	2
3	Hardware	3
3.1	Assigning Parameters to the Printer	4
4	COM 525 programing Software	7
4.1 4.2 4.3 4.3.1 4.3.2 4.3.3	Installing COM 525 on the PG 685 Calling COM 525 and Program Selection Interpreter and Procedure Copying from the Library into the User Program Parameter Assignment Transfer to the CP 525 User Memory	7 8 11 11 13 15
5	PC Job NEW PAGE	17
5.1 5.2 5.2.1 5.2.2 5.3	CP 525 User Program STEP 5 User Program Program Start-up Cyclic Program Test	17 20 20 21 23
б	ProcessStatusList (PSL)	24
6.1 6.2 6.3	CP 525 User Program STEP 5 User Program Test	24 33 36

Page

7	Sequential Message List (SML)	37
7.1	CP 525 User Program	37
7.1.1	PC Job SEQ. Message	37
7.1.2	Messages	40
7.2	STEP 5 User Program without COM PMC	45
7.2.1	Signaling Functions for Standard CPs	45
7.2.2	Assigning Parameters to the Signaling Function	15
	Blocks DB PMC	48
7.2.3	Determining the Origin of the Message, DB M-GEBER	50
7.2.4	Message Parameters DB M-PARS!	51
7.2.5	Assigning Parameters in DB PARA-R and DB PARA-S	51
	for Triggering Jobs	52
7,2.6	Handling Block Calls PB-HDB	55
7.2.7	Start-up Organization Blocks	58
7.2.8	Cyclic Program	60
7.2.9	Transmit Clock pulse	62
7.3	STEPS User Program with COM PMC	63
7.3.1	Installing COM PMC on the PC 685	65
7.3.2	Parameter Assignment with COM PMC	65
7.4	Test	73
7.5	STEPS User Program without "Signaling Functions	15
	for Standard CPs"	75
7.5.1	Program Start-up	75
7.5.2	Cyclic Program	70 77
7.5.3	Testing the Program	80
		00
8	Current Message List (CML)	81
0 1		
8.1	CP 525 User Program	82
8.2	STEP 5 User Program	83
8.3	l'est	85
0	_	
У	Frames	86
9.1	CP 525 User Program	86
9.2	Test	90 90
		50

-

ii

10	Date/Time PC Jobs	91
10.1 10.2 10.3	Setting the Date/Time Read Date/Time Synchronizing Several CP Hardware Clocks	91 93 94
11	Reading the Error Message Area in the SYSTAT	95
12	Information	98
13	Program Documentation	101

a personi (P

iii

s . . +\$

1 Introduction

This is **an example** of an **application** using the **communications** processor **CP** 525 for listing and monitoring technical processes. When you have worked through this **example** you will have a finished **CP** 525 user program in which all the **most** important functions are implemented.

......

To start with, the required hardware settings are explained. After this, not **only** the generation of a **CP** 525 user program with the **COM** 525 programing software but also the corresponding STEP 5 program in the CPU will be explained.

For this example you require the following hardware

one CP 525 module with RAM memory submodule

- one PC 685 programmer
- one cable connector PC <--> CP 525
- one **PT88** or **PT89** printer without memory expansion
- one cable connector PT88 <--> CP 525
- one S5-135U programmable controller
- one R processor with RAM memory submodule
- one cable connector PG <--> CPU

and if possible

- one digital input module (24 V, 16 channels)
- cme digital output module (24 V, 16 channels)
- one SIMATIC S5 simulator

the following software is also required

- the S5-DOS programming package COM 525
- the STEP 5 basic package for the PC 685 programmer (supplied with the programmer)
- handling blocks for the R processor
- standard function blocks 'signaling functions for standard CPs'

2 Aims

First of all, you will set up the hardware. After this, you will install the software necessary for writing your user programs.

. . .

The first step is to program a PC job for a form feed.

Based on the example of a process **status list (PSL)**, current process values will be output on the printer; all possible s5 formats will be used.

You will program 11 process messages to be output on the printer when a particular digital **input is** set. One message **will have no** process variables and the others will each have one. All possible S5 formats will be used. The messages will include the time and message status (coming or going) and will be output whenever the signal at the corresponding digital input changes (**sequential message list - SML**).

Using a current message list (CML) you can list all the messages on the printer which are still 'active at the time the CML is called.

The sequential message list, current message list and process status list may also have **frames**, i.e., header and trailer. The programming of frames is also explained based on an example.

How to set and read the CP 525 hardware clock and read the error message area in the SYSTAT will also be demonstrated.

Finally, the example covers the **information** and **documentation** facilities provided by the **COM** 525 programming software.

The texts of the messages of the process status list and the frame are intended to illustrate the **CP** 525 function. For this reason no terminology from process engineering w-ill be used.

3 Hardware

ه هو بي بو ما برمين الله المسالية . او

.

As the $\ensuremath{\text{programmable}}$ controller, an $\ensuremath{\text{S5-1350}}$ with R processor is used.

Fundamentally the procedure for the **S5-115U**, **S5-135U**, **S5-150U** and **S5-155U programmable controllers** is identical. Nothing changes in the **CP** 525 user program. The handling blocks and standard function blocks "signaling functions for standard **CPs"** have **differ**ent numbers for different **programmable** controllers but they are identical in terms of programming.

The CP 525-2 cannot be Used in conjunction with tile S5 1350 S processor.

Plug the R processor into one of the CPU slots (11, 19, 27 or 35) in the **S5-135U**. Carry out an overall reset of the CPU and switch the mode selector to "STOP".

On the **CP** 525, set the module address to zero; i.e. , no jumpers are plugged in at jumper block 16. All the **CP** 525 printer functions can be implemented without **IPC** flags. You should, therefore, inhibit all the **IPC** flags to avoid multiple addressing when using several CPs and/or a coordinator module. Remove all the jumpers **ON** jumper block 25. **Plug** the **CP** 525 into one of the slots (11, 19, 27, 35, 43, 51, 59 or 67) in your **S5-1350**. Switch the mode selector to "**PGR**" (programming) .

On the digital **input** module and the digital output module, set the **module** address to O. Input bytes **IB** O and **IB** 1 and output bytes QB O and QB 1 are then available. The ties can be plugged into any slot on the **S5-135U. Connect** the simulator to the 1/0 modules.

If you have no **I/O** modules or no simulator available, then instead of using IB O and **IB** 1, use flag bytes **FY** O and **FY** 1. **These** flags can be set and reset using the STEP 5 online function **CONTROL VARIABLE**.

Connect the printer to interface 1 (IF 1) of the CP 525.

B8576542-03

Connect the **PG 685** either to the programming interface (IF 2) of the **CP** 525 or to the programing interface of the R processor, (depending on which **module** you wish to program), using the correct cable connector. The **mode** selector on the **CP** 525 **must** be set **to PGR during programming**.

3.1 Assigning Parameters to the Printer

The coding switch cm the central controller of the printer is located **under** the front hinged cover of the printer in front of the platen. Here the following settings must be **made**:

	1	2	3	4	5	6	7	8	9	10	
ON OFF	x	x	x	X	x	x	x	x	x	x	P T 88

	1	2	3	4	5	6	7	8	9	10	
on OFF	x	X	х	X	Х	х	х	х	х	x	PT89

Among other things, the printer is then set for the German character set, even parity and 12" paper length. (For other character sets refer to the description of the PT88/PT89).

The printer has different interface adapter cards depending on whether it is intended for use **with TTY**, V.24 or **combined TTY/V.24** signals. To check the settings on these cards, you must completely **remove** the cover of the printer.

The mode switches on the interface adapter SAP-S1 (V.24/V.28) must be set as follows for this example:



. . . .

If your printer has the interface adapter SAP-S2 (TTY/20 mA) the mode switches must be set as follows:

Switch 1								5	wita	th 2							
	1	2	ŝ	4	5	6	7	8		1	2	3	4	5	6	7	8
on Off	X	X	X	x	x	X	x	x	on Off	x	x	x	x	x	x	X	X

With the mode switch S4 on the combined interface adapter **SAP-S3** (**V.24/TTY**), you can select between V.24 (switch position 1) and TTY (switch **position** 2). The settings for the mode switches S1, S2 and S3 must be as follows for this example:



5

.

The settings of switches ${\bf 5}\, {\rm and}\,\, {\bf 6}\, {\rm are}\,$ irrelevant.

In all three cases the printer is set to a data rate of $9600\;bps$ and to use the X-ON/X-OFF protocol.

The cable connector CP 525 <--> PT88/89 is always the same regardless of which interface adapter is used.

Example PT88/PT89

4 COM 525 Programming Software

4.1 Installing COM 525 on the PG 685

A **PG 685 programme**r must be available in which the STEP 5 basic package has already been installed,

Go to drive B: user number 0 by entering

0:

Copy the contents of the three COM 525 floppy disks with

PIP **B:=A:*.*[R** V]

Option V = verify copying R = copy SYS files

onto the hard disk, user number 0.

Assign the system and read only attributes to the files with

SET S5?EC?5X. CMD[SYS RO] SET S5PEP05X. CMO[SYS RO] SET COMLIB*. 525[SYS RO]

You can work with COM 525 on any user area.

Exit user number 0, which should be reserved for system files, with $% \left({{{\bf{x}}_{i}}} \right) = {{\bf{x}}_{i}} \left({{{\bf{x}}_{i}}} \right)$

n: (n = required user number).

4.2 Calling COM 525 and Program Selection

By entering

S5

you call the SIMATIC programing packages. Place the cursor in the line 'C 0 M 5 2 5 . . . 'and use the function key F1 (PACKAGE) to select the COM 525 programming software for the CP 525.

The COM 525 basic mask then appears as follows:

COPYRIGH BASIC	T (C) BY S II MASK	EMENS			SIMATIC S5 / COM525
	2222222	00000	юмм мм	55555555 22222	2 5555555
	cc	00	00 MMM MMM	55 22	22 55
	cc	m	OO MH MM MI	55 2	2 55
	сс	00	OO MM MM	5555555 22	5555555
	сс	00	00 MM MM	55 22	55
	сс	00	OO MM MM	55 22	55
	2222222	00000	NO MM MM	5555555 222222	22 5555555
	Program Version/iss	ning pac sue: Al	kage for the and the 04 se	communications pro communications pro rial no.: 79%-007	cessor CP 525 cessor CP 524 4-654321
F 1 SELECT PROGRAM	F 2 System Data	F 3	F 4	F5 F6	F7 F8 EXIT

 $Press\ F1$ (SELECT PROGRAM) to call the mask 'PROGRAM SELECTION' .

BASIC MASK -> PROGRAM S	ELECTION	SIMATIC S5 / C(YI525
	DRIVE: B PROGRAM MANE: PT88ABSP COMPONENT: P T	
	Plant designation: application example Generated by: Wally Generated on: 28.10.87	
	PG date/time: D_M_Y_H_M 28.10.87- 14:28	
F 1 F 2 SELECTION	F3 F4 F5 F6	F7F8 HELPEXIT

A program name (in this case "PT88ABSP") and the component "PT" for the event output and listing function must be entered. As plant designation enter "application example" and enter your own name beside 'Generated by".

17

In the 'SELECTION' mask, which you call with Fl (SELECT), you can select the basic COM 525 functions.

BASIC MASK->PROG. SELECTION	SELECTION ->	SIMATIC	S5 /	/ COM525
	DRIVE: B PROGRAMNAME: PT88A8SP COMPONENT: P T			
	Plant designation: application example Generated by: Wally Generation: 28.10.87			
51 52	= 2 = 4 = 5 = 4	E 7	I	го
PROGRAM USER DATA TRANSFER	DELETE INFO FUNCTION CONTINUE	г/		F 8 EXIT

F 1	F 2	F 3	F 4	F 5	F 6	F 7	F 8
ASSIGN INT.PARA.	PROC.PARA	ASSIGN PRI.PARA.	LISTING		CONTINUE		EXIT

With the COM 525 programming software the user program is first generated on floppy or hard disk (storage medium FD) and then transferred to the CP 525 user program where it is then tested.

The S5-DOS file on the storage medium FD automatically has the program name .525, in this case PT88ABSP. 525.

1.1

4.3 Interpreter and Procedure

4.3.1 Copying from the Library into the User Program

The interpreter and procedure must first be copied into **the** user **program. They are located in** the library **COMLIB.** n (n = **version** number), which is part of the **COM** 525 software (in drive B, user O).

Ress F2 (TRANSFER) and F5 (FD->FD) in the 'SELECTION' mask to call the function for transferring from FD to FD. Press F7 (HELP) in the 'TRANSFER' mask and enter drive "B" as the source and program name "COMLIBO1" (the library in our example has version number 01). The destination is automatically the program PT88ABSP selected in the 'PROGRAM SELECTION' mask.

ROG. SELECTION -> SEI A N S F E R	ECTION ->	SIMATIC S5 / CC		
	Source:	DeSt. :		
STORAGE MEDIUM: DRIVE:	m B	FD B		
INTERFACE NUMBER: PROGRAM NAME: COMPONENT:	COML IB01	PT88ABSP PT		
Plant designation: Generated by: Generated on:	standard 1 ibrary GW Kar Isruhe 01.10.87	application example Wally 28.10.87		
	2 57 55	- / - •		
TOTAL INTE PROGRAM PRET	S F 4 F 5 R- ASSIGN ER PROCEDURE PRI PARA	USER DATA HELP EXI		

e s e europe

•

Press F3 to **call** the **'INTERPRETER'** mask. By pressing F7 (HELP), in this mask the interpreter to be transferred **"PT88" (component** "PT") is selected. **F1** (TRANSFER) starts the transfer. The successful transfer is displayed with **'MESS.002** completed'.

B8576542-03

17

-> PRDG. SELECTION-> SI INTERPRETER	SIMATIC S5 / COM525	
	Source:	Dest.:
STORAGE MEDIUM: DRIVE:	F D B	FD B
INTERFACE NUMBER PROGRAM NAME: COMPONENT:	COMLIB01	PT88ASPB PT
Source: 0	COMPONENT NAME VERSION PT PT88 01	
F 1 F 2 TRANSFER	F3 F4 F5	F6 F7 F8 HELP EXIT \$

With F8 (EXIT) you return to the 'TRANSFER' mask.

With F4 (PROCEDURE) you call the 'PROCEDURE' mask. The procedure "LAUFPT88" is transferred in the same way as the interpreter.

Press F8 (EXIT) twice to **return** to the 'SELECTION* mask **from** the '**PROCEDURE' mask**.

12

4.3.2 Parameter Assignment

With F6 (CONTINUE) in the 'SELECTION' mask, call the second set of function keys .

F1 (ASSIGN INT. PARA.) branches to the 'ASSIGN INTERP. PARA. ' mask. In this mask, the interpreter belonging to program PT88ABSP is displayed. The mask already has standard default values entered. Most of these are used in this program.

The entry in the "printer type" field may have to be changed to "N" (needle head). The separator for the time should be changed to ":" and the character set to 'ENGLISH". All the entries can be made using F7 (HELP). Using F6 (SAVE) you write the interpreter to the hard disk. 'MESS. 003: saved ! ' indicates that this is complete.

• PROG. SELECTION SS I GN I N	•> SELECTION •> T ERP. PAR A.			SIMA	TIC S5 / COM525
	DRIVE	E: B I	PROGRAM: P	T88ABSP	COMPONENT: P T
Interpreter:	COMPONENT: P T	NAME: PT&	B VER	SION: 01	
		Presetti	ng for inte	erpreter	PT88:
Printer model: Printer type:	PT 8	8 PT 88 or [N≕needle	PT 89 head	I≃inkjet	
Lines per inch: Page length: Page width: Format (date) Format (time): Separator (date Separator (time)	6 72 080 0 M 1 H M S 2):) 2 lines) from 1 to Y D=day 3 H=hour	80 charact M=month M=minute	ers per l Y=Year S=seco	i ne nd
Time format: Character set:	ENGLISH AM/PM ENGLISH	GERMAN 24	H, EN	GLISH AH/	PM
F1 F2	F3 F	4 F 5	5 F 6 SAVE	F HEL	7 F 8 P EXIT

r

With F8 (EXIT) you return to the 'SELECTION' mask.

With F2 (ASSIGN PROC. PARA) in the second set of function keys, call the 'ASSIGN PROC. PARA' mask. In this mask the procedure and version number belonging to our program PT88ABSP are displayed. You must enter the same parameters that were set on the printer with the switches (baud rate "9600", "1" stop bit, "EVEN" parity). These are also the standard defaults in the mask.

		DRIVE: B	PR0	GRAM: Pt88ABSP	COMPONENT: P
Procedure:	COMPONENT: PT	NAME: L AU	FPT88	VERSI ON: 01	
Baud rate:	9600			Char. length:	7
Number of s	top bits: 1			Pri ori ty:	LOWER
Pari ty:	EVEN				
					_
F 1 F	2 F 3	F 4	F 5	F 6	F7 F8
				SAVE H	ELP EXIT

Ress F6 to **save** the procedure parameters on the hard disk and press F8 (EXIT) and F6 (CONTINUE), to return to the 'Selection mask.

4.3.3 Transfer to- CP 525 User Memory

The interpreter and procedure still must be transferred to the user memory of the CP 525. The **CP** 525 must be **connected** to the PC 685 and its mode selector must be set to "**PGR**" (programming).

S. 1990

Press F2 and F3 in the 'SELECTION' mask to call the function for transferring from FD to CP 525. In the 'TRANSFER' mask enter the CP 525 interface "1" as the destination. Make the transfer with F3 (INTERPRETER) and F4 (PROCEDURE).

PROG. Selection-> Sele A N S F E R	ECTION ->	SIMATIC S5 / COM52		
	Source:	DeSt. :		
STORAGE MEDIUM: F	O B	CP525		
INTERFACE NUMBER: PROGRAM NAME: COMPONENT:	PT88ABSP PT	1		
Plant designation: Generated by: Generated on:	application example Wally 28.10.87	application example Wally 28.10.87		
	3 F 4 F 5	F 6 F 7 F 8		
COLD TOTAL INTE START PROGRAM PRET	R- Er procedure para	USER DATA NELP EXIT		

'MESS.002 completed!' shows that the transfer has been completed.

IMPORTANT!: when the interpreter and procedure are transferred to the CP 525 memory you must cold restart the CP 525 with Fl (COLD RESTART). an and an an an and an and an and a strain and a strain and a strain and the strain and the

If you switch the CP 525 mode selector to "RUN", the red LED of interface 1 (IF 1) must go out.

All the preparations have now been made for programing the actual event output and listing functions.

.

Example PT88/PT89

5 PC Job NEW PAGE

Using the PC job NEW PAGE the CPU can trigger a form feed \boldsymbol{on} the printer.

- -

5.1 CP 525 User Program

Switch the mode selector on the CP 525 to "PGR".

Press **F1** (PROGRAM USER DATA) in the 'SELECTION' mask followed by F2 (PC JOB), F6 (CONTINUE) and F2 (NEW PAGE) to call the 'NEW PAGE mask. The job number must be entered with which the form feed will be initiated by the CPU. Select job number "1". The PC job is automatically named NEW PAGE.

Job name: NEW PAGE With the job no.: 1 the CP 525 will initiate a form feed on the PT88/PT89 printer.		DRIVE:	В	PROGRAM:	PT88ABSP	COMPONENT: P
With the job no.: 1 the CP 525 will initiate a form feed on the PT88/PT89 printer.	name: NEW PAGE					
	h the job no.: 1 CP 525 will initiate a nter.	form feed	on	the PT 88/ P	PT89	
•• • • • • • • • • • • • • • • • • • • •						
F1 F2 F3 F4 F5 F6 F7 F	F 2 F 3	F4	1	F 5 F	-6 F	7 F 8

÷, •

• 5 x ***

The PC job is saved on the hard disk with F6 (SAVE). Press F8 (EXIT) three times to return to the 'SELECTION' mask .

Press F2 (TRANSFER) and F3 (FD->CP) to transfer the PC job to the user memory of the CP 525. This is similar to transferring the interpreter and procedure (cf. Section 4.3 .3). Press F6 (USER DATA) in the 'TRANSFER' mask to display a new set of function keys .

F 1	F 2	F3 Pc	F 4	F 5	F 6	F 7	F 8
RESTART	MESSAGE	JOB	FRAME				EXIT

Press F3 (PC JOB) and press F7 (HELP) in the 'PC JOB to enter the name of the PC job NEW PAGE in the input column of the source. The PC job is transferred with F1 (SINGLE),

9

-> PROG. SELECTION -> SELECTION SIMATIC S5 / COM525 TRANSFER							
STORAGE MEDIUM: DRIVE: INTERFACE NO.: PROGRAM NAME: COMPONENT:	Source: F O B PT88ABSP PT	DeSt.: CP525 1 PT					
Name:	NEW PAGE	NEW PAGE					
FI F2 F ALL SINGLE WITH /	3 F 4 ACK.	F 5 F 6 TOTAL	F7 F8 HELP EXIT\$				

.

With F8 (EXIT) return to the 'TRANSFER' mask. Carry out a cold restart on the CP 525 with Fl.

Switch the CP 525 to the "RUN" mode. The red LED on interface 1 (IF 1) must now go off.
40 D. Fr. -

5.2 STEP 5 User Program

Call the **S5** package **LAD**, CSF, **STL**. Program in **STL** (statement list) and **call** your STEP **5** program f **ile** V: **PT88R1ST.S5D**[•]. **Trans**fer the handling blocks for the R processor into this program file. Connect the R processor to the **PG** 685.

whys.

The STEP 5 program must do the following:

- synchronize the CPU and CP 525 during the PC start-up
- trigger the form feed job.

5.2.1 Program Start-up

The start-up menintion blocks OB 20, OB 21 and OB 22 call the SYNCHRON handling block unconditionally.

	:JU	FB125	SYNCHRONIZATION CP <> CPU
NAME :	: SYN	CHRON	
SSNR	:	KYO , O	INTERFACE NUMBER O
BLCR	:	KYO , 6	FIELD LENGTH 6 (MAX. 512 BYTES)
PAFE	:	FY10	PARMETER ASSIGNMENT ERROR BYTE

:BE

The interface number **SSNR** for interface 1 (IF 1) is **the same** as the **module** address "O". Select the largest field length '6". This means that data can be sent in strings of up 51.2 bytes (one complete data block) at one time between the CPU and **CP**. Use flag byte **"FY10"** as the parameter assignment error byte **PAFE**.

A parameter assignment error is to be indicated at digital output Q 0.0. In the start-up organization blocks, there is no process image so no digital 1/0s are addressed. Therefore this indication can be made only in the cyclic program (see Section 5.2.2). Normally an error evaluation program is called during start-up. If a parameter assignment error occurs, check whether the interface number and module address are the same. If they are correct, a hardware fault is suggested.

5.2.2 Cyclic Program

11 •

الحرر بالمفاقعا الجاجا

.

If a parameter assignment error occurs during the start-up, it is indicated at digital output Q 0.1 at the beginning of the cyclic program (see Section 5.2.1).

A PC job is triggered by a SEND **DIRECT** call; i.e. the handling block SEND DIRECT is called with a job *number* other than 0. The job number for the SEND DIRECT and the number of the job on the **CP** 525 must be the same. Therefore you program job number **A-NR** '1". The job is performed if a positive going edge appears at digital input I 0.3.

The interface number SSNR is "O"; "FW11" is selected as the condition codeword ANZW and "FY15" as the parameter assignment byte PAFE. The condition codeword requires two words for a SEND DIRECT call. The job status and any error messages that occur are indicated in flag word FW 11.

Since for a form feed no data needs to be transferred from the CPU to the **CP** 525, no source needs to be specified. "NN" must then be **programme**d as the **Source** type **QTYP**, which means that the three following source parameters are not evaluated.

The job is triggered with the result of logic operation **(RLO)** 1. **If RLO** = 0 when the call is made, only the **condition** codeword will be updated.

sł.

B8576542-03

The termination of the job with an error and the occurrence of a parameter assignment error is indicated only at digital output Q 0.1. In this **situation** an error evaluation program tailored to the particular application is usually called. If an error occurs, evaluate the error **mumbers** in the condition codeword, the error message area of the **SYSTAT** (see Section 11) and **in** the parameter assignment error byte.

The corresponding program in the cyclic **organization block OB 1** is then as follows:

:A F 10	.0 IF SYNCHRON PARAMETER ASS . ERROR
:= Q 0.0	OCCURRED> SET OUTPUT
:	
:AN I 0.3	3 FLANK EVALUATION:
:R F 2.	3 THE RESULT OF LOGIC OPERATION
:A I 0.	3 (RLO) IS SET FOR ONE CYCLE IF THE
:AN F 2.3	3 SIGNAL CHANGES FROM O TO 1 AT
:S F 2.	3 INPUT I 0.3.
÷.	
:JU FB12	O TRIGGER PC JOB NEW PAGE
NAME : SEND	
SSNR : KYO,	O INTERFACE NUMBER O
A-NR : KYO,	1 JOB NUMBER 1
ANZW : FW11	CONDITION CODEWORD
QTYP : KSNN	NO SOURCE SPECIFIED
DBNR : KYO , o	NOT EVALUATED
QANF : KF+O	NOT EVALUATED
QLAE : KF+0	NOT EVALUATED
PAFE : FY15	PARAMETER ASS. ERROR BYTE
:	
: 0 F 1.2	.3 JOB 1 TERMINATED WITH ERROR
:0 F 15	.0 OR PAMMETER ASSIGNMENT ERROR
:= Q 0.1	L INDICATED AT CUTPUT Q 0.1
1	-
:BE	

Transfer the R processor handling blocks FB 120 to **FB** 127, **OB** 20, **OB** 21, **OB** 22 and **OB** 1 to **the** user **memory** of your R processor. Carry out a cold restart on the R processor.

22

. ۱۳۰۰ میری رام و ۲۰۰۰ ماری با ۲۰۰۰ ۱۹۰

,

5.3 Test

If you now switch the signal at **digital** input I 0.3 **from** zero to me, a new page c**ommand** is output to the printer.

!Note! The CP 525 does not check whether a form feed has been carried out or even if the printer has been switched on. To avoid paper being wasted, a form feed **only** takes place following a printout or after you press the **LF** key on the printer.

If you have no digital input module with a simulator available, use flag $F\,0.3$ instead of input I 0,3. Set (or reset) the flag with the PG 685 online function CONTROL VARIABLE. Observe the condition codeword FW U and the parameter assignment error byte FY 15 using CONTROL VARIABLE .

. . . .

6 Process Status List (PSL)

When a PSL is called the current process values sre printed out on the printer with static text previously programed on the CP 525.

.

6.1 CP 525 User Program

Switch the mode selector on the CP 525 to "PGR".

Ress F1 (PROGRAM USER DATA), F2 (PC JOB) and F4 (PSL) in the 'SELECTION' mask to call the following mask:

-> PROGRAM -> PC JOB .> PROCESS STATUS LIST SIMATIC S5 / COM525 PROGRAM : PT88ABSP COMPONENT: PT DRIVE: в Name of list: PSL-EXAMPLE Name to be stored: PSL-EXANPLE With the job no.: 2 the output of the process status list with the name selected above will be started. The dynamic data are on CPU no .: 1 in data block DB no.: 30 List header is frame: List trailer is frame: Output of messages possible while the PSL is being output (Y/N) N Form feed on PT88/PT89 at start of list (Y/N) Y F 1 F 2 F 3 F 4 F 5 F 6 F 7 F 8 DYN. STAT. COMPLETE PART FRAME DYN .PART PART SAVE HFI P EXIT

Select the list name "PSL-EXAMPLE" (this is automatically entered in the input field 'name to be stored") and the job numb& '2".

e 19

-. L

The CPU number is "1", since only one CPU is plugged into the ${\bf S5-}$ 135U. The data block for the process data is DB '30".

. . **.**

Initially do not enter a list header or trailer. The output of the **PSL** is not to be **interrupted** by process messages (entry: "N"). Before the list is output a form feed should be carried out (entry: Y'').

Press **F1** (STAT. PART) to select the **mask** 'STATIC **PART'**. In this mask you can enter the following attributes, static texts and reserve characters (#) for dynamic values:

Attribute BECU	GrElement name:	PSL-EXAMPLE	Insert	1.1	
	NO. Page Mich	ch: 080	OFF	Column no	.: 003 p.: 008
YYNY	Text ⊯i thwt	vari able			
YNNN	Date: #####	### Time:###	*****		
N N Y N	Binary value	:# Character	: ########## E	Byte: ###	
NNNN	16 bit BCD Va	ilue:	####		
8 8 8 8 8 8 9 8 8 8 9 8	32 hit pop va	d pint value:	###### ########		
NNNNI	32 bit fixed	pint value: #			
NNNN	32 bit float	ing point valu	e: ##########		
NNNN	Timer value:	#### Counter	value: ###		
YYYNI	Process stat	tus variable: #	******		

With the function keys F1 to F4 for editing **functions** the above function key set can be replaced by others.

The following pages explain the **most important** editing **functions**. A detailed **description** of the editor functions can lx-found in the **user's** guide for the programming package **COM** 525 in this manual.

ş.

When you use the cursor keys with single arrows, the cursor is roved character by character; when you use the double arrow keys it is **moved** word by word.

The character \mathbf{o} which the cursor is currently positioned can be deleted with **the** delete key (in the cursor key block).

If you called a new set of function keys using function keys **F1** to F4 the cursor cannot be moved nor can entries be made in the editing window. These keys are explained as follows:

F1 (DELETE) calls the delete menu as follows:

- F2 (WORD)

deletes all the characters from the current cursor position to the end of the word

- F3 (LINE) deletes the line in which the cursor is positioned
- F4 (BLOCK)

deletes the block of lines marked

F2 (BLOCK) calls the block handling menn as follows:

- F1 (SET START)

the start of a block is $\ensuremath{\mathtt{marked}}$ in the $\ensuremath{\mathtt{mask}}$ with S in the line in which the cursor is positioned

- F2 (SET END)

the end of a block is marked with ${\bf E}$ in the first column of the mask in the line in which the cursor is ${\bf located}$

– **F3 (COPY)**

a block of lines is copied in before the line \mbox{in} which the \mbox{cursor} is located

- F4 (HOVE)

a block of lines is roved in before the line in which the **cursor is located**

F3 (SEARCH) branches to the search menu as follows:

- F1 (SEARCH)

allows a string of up to 20 characters to be searched for - F2 (REPLACE)

allows a string of up to 20 characters to be replaced



- F1 (LEFT MARGIN)
 - to the beginning of the line
- F2 (RIGHT MARGIN)
- to the end of the line F3 (INPUT FIELDS)

into the input f ieMs for the attributes and for the group number. Using the **PG** 685 cursor keys you can now move from column to **column** and line to line within the attributes. With the function key F4 **(TO ED.VINDOW)** you return to the text editing window.

F5 (INSERT ON/OFF)

switches the INSERT writing mode on or off. Characters entered are either inserted before the current cursor position (ON) or they overwrite text that is already there (OFF).

In the input fields for the attributes, you fix the way **in which** static texts in a list line will be printed, You can select bold **print** (B), expanded print (E), compressed print (C), underlining (U) and acoustic signal (A) by entering **"Y"** (= yes) or **"N"** (= no).

Save the static part of the PSL with F6 (ENTER) in the **PG** 685 **RAM** and exit the mask 'STATIC PAN with F8 **(EXIT)**.

بجور والمتعيم محيا والحوار

.

•/ •

Press F2 (DYN. PART) in the *PROCESS STATUS LIST' mask to call the 'DYN. PART' mask. Using the function keys Fl to F5 you select the parameter you wish to specify. By pressing F5 (JUMP) you will be prompted to specify where you wish to jump ("jump to line ##, field ##"). Press F5 (JUMP) again to jump to the selected field. The specified variable is displayed in the text window inversely.

. . .

↔ PC JOB S T A T I (-> PSL -> C PAR1	r				SIM	ATIC S5	/ COM525
			DRIVE:	B P	ROGRAM:	PT88ABSP	COMPON	IENT: PT
Attribute B E C	es Gr Elem UANO.F	ent name: Page width	PSL-EXAN	IPLE		Line Col u	e no.: "man no.:	003 0 0 8
Y Y N Y N N N N Y N N N N N N N N N Y N N N N N N N N N	N Tex N Da N Bin N 16 N 16	xt without te: ###### hary value bit BCD v bit fixe	variable ### Tim :# Char alue: d point v	: ##### acter: value:	#### ##### ##### ######	## Byte: ##	!#	
Type o DATE	f dynamic is fetche	variables: ed fron	DATE CP 525 CLO order D M	ск ү		Attribu bold p expand under l i acousti	tes (Y/ rint led prin i ned ic signa	'N): Y t N N al N
F 1 UP	F2 Down	F 3 LEFT	F 4 RIGHT	F5 JUM	P EN	6 F TER HI	7 ELP	F 8 EXIT

The following presettings are shown: in the field 'type of dynamic variables" - PROCESS VARIABLE, "format" - 16BIT FXP, 'address" DW and as 'no. of characters" the number of reserve characters (#). All the attributes have the default "N". You can change the type of dynamic variables with F7 (HELP). The liner part of the mask then changes automatically. These defaults must be extended and possibly changed. The address, format and attributes can be entered using F7 (HELP).

Example PT88/PT89

The specification of a variable is stored in the PG 685 RAM with F6 (ENTER). The next variable is then selected with F1 to F5. This is then specified and entered. When all the variables are specified, you exit the 'DYN. PART' with F8 (EXIT). The whole PSL is saved with F6 (SAVE) on the hard disk in the 'PROCESS STATUS LIST' mask.

The date and time should be taken from the **CP** 525 hardware clock and printed out in bold print. Seconds should not be shown.

Type of dynamic variable	: DATE	Attributes (Y/N):
DÁTE is fetched from	CP 525 CLOCK	bold print Y
		expanded print N
	order DMY	underlined N
		acoustic signal N

Press F6 (ENTER) to store the specification in the **PG** 685 RAM. Using F4 (WRITE), select the next **input** field.

Type of dyna	mic variable:	ГІМЕ	Attributes (Y/N):
TÍME is fe	tched from	CP 525 CLOCK	bold print Y
			expanded print N
		order H M	underlined N
			acoustic signal N

Ress F6 (ENTER) to store the specification in the PG 685 RAM. Using F2 (DOWN), select the next input field,

All other **variables in** the **PSL** are of the **type** PROCESS **VARIABLE**. It has already been specified in the 'PROCESS STATUS **LIST'** mask that all the process variables will be fetched from CPU 1 **data** block DB30. The number of reserve characters for dynamic variables (#) is adopted as the number of characters (before the point).

- (ð

Program the type of access and representation as follows:

Type of o	dynamic	variable: PRDCESS	VARIABLE		Attributes (Y/N):
					bold print N
CPU no.	1	Format	BINAR	Y	expanded print N
DB no.	30	No. of	characters	1	tit-lined N
Address	BI 1.4				acoustic signal N

The specification is stored in the PG 685 RAM with F6 (ENTER). Select the next input field with F4 (WRITE) and continue specifying the following variables accordingly.

The table below shows all the variables of the type PROCESS **VARIABLE**, in the order **in** which they appear in the PSL.

Format	Address	No. of characters	Attributes
BINARY CHAR PROCESS STATUS BYTE 16BIT BCD 16BIT FXP 32BIT BCD 32BIT FXP 32BIT FLP TIMER COUNTER	BI 1.4 DL 2. DL 7. DR 7. DW 8. DW 9. DD 10. DD 12. DD 14. DW 16. DW 17,	1 10 8 3 4.0 6.0 8.0 10.0 5.4 3.1 3	underlined underlined

Select type PROCESS STATUS for the variable in the last line of the list. Ress F6 to branch to the PROCESS STATUS VARIABLE mask. Uptoeight different texts with individual attributes can be programmed in this mask. These texts cannot exceed the maximum number of reserved spaces (in this case eight characters).

Example PT88/PT89

Enter the texts as shown in the following mask:

> PSL '> ROCE	DYN.PART SS STA	TUS V	AR.				SIMA	TIC S5 / CO	M525
			DRIVE:	В	PROGRAM	M: PT88/	ABSP	COMPONENT:	ΡT
Element	name: PS	-EXAMPLE							
cpu No .	1	DB No. 30		Addre	ess: DL	7			
Stat. 0 1 2 3 4 5 6 7	Attributes B E U A Y N N N N N N N N Y N N N N N N N N N N Y Y N N Y N N Y N Y	Text "STOP" "START" OPERATION WARNING POW ON POW ON POW OFF FAULT "TEST"	I						
F 1	F 2	F 3	F 4	F	5	F 6 ENTER	F	7 F 8 P EXIT	1

Which text is printed out depends on the value of the left data byte DL 7.

Ress F6 to enter the stares variable in the PG memory and press F8 EXIT to return to the DYN. PART mask.

31

العوي وراداريس والعاليان ويو

,

.

B8576542-03

and the second states and the second

والمهيد يرد الدو

. . .

Ress F4 (COMPLETE DYN. PART) in the 'PROCESS STATUS LIST' mask to check whether every variable field marked **# in** the 'STATIC PART mask is also specified in the 'DYNAMIC PART' mask.

If all the variables have been specified, the message "MESS .017: All fields have already been programed!" appears above the function keys .

If not all **variables** are specified, the **'COMPLETE DYN**. PART' mask appears. Here you can specify the variable fields for which **you** have not yet **programme**d the format, address and representation.

The whole process status list is written to the hard disk with F6 $(\ensuremath{\texttt{SAVE}})$.

After saving the list, return to the 'SELECTION' mask by pressing F8 (EXIT) three times; then press F2 to branch to the $*_{TRANSFER'}$ mask. Transfer the PC job PSL-EXAMPLE in the same way that you transferred the NEW PACE PC job (see Section 5.1) to the CP 525 user memory.

Following each transfer, carry out a cold restart on the CP 525 with F1 and switch the CP 525 to "RUN". The red LED for interface 1 (IF 1) must go out.

B8576542-03

Example PT88/PT89

"6.2 STEP 5 User Program

The STEP 5 program must achieve the following:

- the job for the PSL mat be triggered

- the process data must be transferred from the CPU to the CP.

The triggering of the process status list is achieved in the same way as the triggering of the PC job for form feed (using a SEND DIRECT job). The only cliff erence is that a different job number and different addresses for the condition codeword (ANZW) and parameter assignment error word (PAFE) must be used. The job number A-NR "2" is identical with the number of the PC job progr-d on the CP. Use the digital input I 0.4 and start the PSL on a positive going signal edge.

The SEND DIRECT job is triggered with the result of logic **opera-tion (RLO)** 1. If the **RLO** is 0 when the call is made, only the condition codeword is updated.

With the SEND DIRECT call, the condition codeword requires two words. The job status and any error messages are indicated in flag word FW 16. The number of pieces of data transferred is indicated in flag word FW 18.

If the job is to be terminated with the status " job terminated with error" or if a parameter assignment error occurs, this is indicated at digital output Q 0.2. Normally, an error routine **tailored** to the particular application is called. If an error **occurs, evaluate** the error numbers in the condition codeword, in the error message area of the SYSTAT (see Section 11) and in the parameter assignment error byte.

The transfer of the process data is executed with a SERD ALL handling block call. The SEND ALL (also known as SEND 0) is a SEND call with the job number **A-NR** 'O". It is called unconditionally in every program cycle and checks whether the **CP (dual**port RAM) requires data. If it does, the SEND **ALL** fetches it from the source specified by the **CP** and sends it to the **CP**. If it does not, the function block is exited and the cyclic program continued.

Example PT88/PT89

The SEND AIL is **called** unconditionally **and** is executed when it is called regardless of the result of logic operation. Its condition codeword is only one word long. The number of the job is entered for which the SEND ALL trensfers **data** from the CPU to the CP.

OB 1 must be expsnded by the following statements:

	: AN	10.4	EDGE EVALUATION:
	R	F 2.4	THE RESULT OF LOGIC OPERATION
	:A	I 0.4	(RLO) IS SET FOR ONE CYCLE IF THE
	: AN	F 2.4	SIGNAL AT INPUT I 0.4 CHANCES
	:S	F 2.4	FROM o To 1.
	:		
	: JU	FB120	CALL PROCESS STATUS LIST
NAME	: SEN	D	
SSNR	:	IWO, o	INTERFACE NUMBER O
A-NR	:	KYO , 2	JOB NUMBER 2
ANZW	:	FW16	CONDITION CODEWORD
QTYP	:	KSNN	NO SOURCE SPECIFIED
DBNR	:	KYO, 0	NOT EVALUATED
QANF	:	KF+0	NOT EVALUATED
QLAE	:	KF+0	NOT EVALUATED
PAFE	:	FY20	PAMMETER ASS. ERROR BYTE
	1		
	:JU	FB126	TRANSFER PROCESS DATA TO CP
NAME	: SEN	D-A	
SSNR	:	KYO, O	INTERFACE NUMBER O
A-NR	:	KYO , 0	JOB NUMBER O
ANZW	:	FW21	CONDITION CODEWORD
PAFE	:	FY23	PARAMETER ASS. ERROR BYTE
	÷		
	:0	F 17.3	JOB 2 TERMINATED WITH ERROR
	:0	F 20.0	OR PARMETER ASSIGNMENT ERROR
	:= Q	0.2	INDICATED AT OUTPUT Q 0.2

Example PT88/PT89

.

The process data is located in **data** block **DB** 30. This must be programed with a length of at least 32 data words: it is then long enough for all our example programs. Assign data words **DW** 1 to **DW** 17 as follows:

0 :	KH⊨ 0000;	
1:	KM= 0000000000010000;	bit 4 = 1
2:	KS= EXAMP. TEXT	character string
7:	KY= 005,123;	process variable 5, byte variable 1.23
8:	KH= F123;	-123 in 16 bit BCD format
9:	la?= +12345;	+12345 in 16 bit fixed pt. format
10 :	KH= F123;	-1234567 in 32 bit BCD format
11 :	KH= 4567;	
12 :	KF= +00001;	+77881 = 12345 + 65536
13 :	KF= +12345;	in 32 bit fixed point format
14 :	KG=+1234567+02;	1.2.34567 floating point format
16 :	KT= 123.3;	timer value 1230 seconds
17 :	KC= 123;	counter value 1.23

Transfer DB 30 and the expanded \boldsymbol{OB} 1 to the user memory of your R processor.

6.3 Test

Carry out a cold restart on the R processor to synchronize the CPU and CP 525. When you switch the signal at digital input I 0.4 from zero to one, the PSL is output on the printer:

- **-**

Text without variable

Date: 01 .01.88 Time 00:00 am

Binary value: 1 Character: EXAMP.TEXT Byte: 123 16 bit BCD value: -123 16 bit fixet point value: 12345 32 bit BCD value: -1234567 32 fixed point value: 77881 32 bit floating point value: 12.3457

<u>Timer value: 123.3 Counter value: 123</u> Process status variable: POW OFF

If you have no digital input module with a simulator available, use flag F 0.4 instead of input I 0.4. Set (or reset) the flag with the PG 685 online function CONTROL VARIABLE. Observe condition codeword FW 16 and the parameter assignment error byte FY 20 with CONTROL VARIABLE.

Change the values in data block DB 30 and start the process status list again.

7 Sequential Message List (SML)

The **SML** is a series of single messages that are printed out immediately after an event has been detected.

First program the messages on the **CP** 525 (Section 7.1). You will thenbeshown the various ways of programming your **STEP5** program.

- Either use the standard function blocks "signaling functions for standard CPs" as follows:
 - without the S5-DOS programming package COM PMC (Section 7.2),
 - with the S5-DOS programming package COM PMC (Section 7.3).
- The other option is not to use the 'signaling functions for standard CPs" (Section 7.5).

7.1 CP 525 User Program

Switch the mode selector of the CP 525 to "PCR".

Before you can program process messages you must program the PC **job 'SML'. Among other** things the positions of the date, time and status in the message text are fixed for all the messages.

7.1.1 PC Job SEQ. MESSAGE

Press F1 (PROGRAM USER DATA) in the 'SELECTION' mask to call the 'PROGRAM' mask. Press F2 (PC JOB) in this **mask to obtain the 'PC** JOB' mask. In this mask, press **F1 (SML)** to call the 'SEQUENTIAL MESSAGE USC'. Some standard default values are already entered.

37

.

الجمي والمالة المرام الألاف

,

		DRIVE:	B PROGRAM: PT88ABSP COMPONENT: P
List name:	SEQ. MESS	AGE	
With the j	ob number: 3		
messages ar	e output as	events take pl	lace.
List header	is frame:		
List trailer	is frame:		
Form reed	on Pi88/Pi89 smoot of data ti	as start of lis	st (Y/N) N
	innat of uate, th	me, status is as	
	At	tributes Om	der
	At Column B	tributes On EUA	der
Date:	At Column B 1 N	tributes On EUA NNNDD.	der
Date: Time:	At Columan B 1 N 7 N	tributes Ord EUA NNNDD. NNN HH:	der MM MM:SS
Date: Time:	At Column B 1 N 7 N	tributes Ord EUA N N N DD. N N N HH:	HM HM:SS
Date: Time: Status:	At Columan B 1 N 7 N 65 Y	tributes Ord EUA NNN DD.J NNN HH: Tex	MM MM:SS rt (enclose in inverted comas)
Date: Time: Status: caning: going:	At Columan B 1 N 7 N 65 Y	tributes Ord EUA NNN DD. NNN HH: Tex NNN '1' NNN '0'	MM MM:SS It (enclose in inverted comas)
Date: Time: Status: caning: going: acknowl	At Column B 1 N 7 N 65 Y edged: N	tributes Ord EUA NNN DD.J NNN HH: Tex NNN '1' NNN '0' NNN ''	MM MM:SS rt (enclose in inverted comas)
Date: Time: Status: caning: going: acknowl	At Columan B 1 N 7 N 65 Y edged: N	tributes Ord EUA NNN DD.J NNN HH: Tex NNN '1' NNN '0' NNN ''	MM MM:SS (t (enclose in inverted comas)
Date: Time: Status: caning: going: acknowl F 1 F	At Column B 1 N 7 N 65 Y edged: N 2 F 3	tributes Ord EUA NNN DD.J NNN HH: Tex NNN '1' NNN '0' NNN ''	MM MM:SS (t (enclose in inverted comas)
Date: Time: Status: caning: going: acknowl F 1 F	At Column B 1 N 7 N 65 Y edged: N 2 F 3	tributes Ord EUA NNN DD. NNN HH: Tex NNN '1' NNN '0' NNN ''	MM MM:SS (t (enclose in inverted comas)

First of all, a job number must be assigned with which the CPU starts the SML. All PC jobs must have different job numbers. For this example select `3".

In this case do not enter anying the in the list header and trailer.

A form feed before the SML is printed out is not needed so "N" is entered.

The date is to appear in column "1" without the year being specified; the time should appear in column "7" and the status in column "65".

Messages which are "coming" (positive going edgeat digital input) should have the status '1"; messages which are "going" (falling edge) should have the status '0". The status is tobe printed in bold print: a "Y" must be entered in the first attribute column.

سجون والأراب الريوانة فالعالم والمواق

٠

 $\delta_{-\frac{1}{2}}(\delta_{1})$

Messages **canno**t be acknowledged using the **CP** 525 alone. For this reason we delete the text entered in this line.

The following format is therefore **determined** for all process messages:

Printer	column	1	7	15	65	80
	!		!	1	!	!
	DDI	DDD	(TTTT)		. xxxxxxSxxxxx	XXXXXX
	D X	- d	late still	T - time to program	S -stat	tus

The number of columns **reserved** for the status is determined **by** thelongest ofthethree status texts; **in this case one character.**

The page width of 80 characters and the separators for the date (.) and time (:) were determined when the **parame**ters were assigned to the interpreter (see Section 4.3.2).

In the input fields for attributes you can specify how the date, time and status are to be represented. You can select bold print (B), expanded print (E), underlining (U) and acoustic signal (A) by entering "Y" (yes) or "N" (no). The message status should be printed in bold print.

The SML PC job is written to the hard disk with F6.

Ress F8 (EXIT) twice to return to the 'PROGRAM' mask.

7.1.2 Messages

Press F1 (MESSAGE) in the 'PROGRAM' mask to call the 'MESSAGES ' mask. Enter the number of the first message; select '100' as the mumber for the first message.

Press Fl (STAT. PART) to call the 'STATIC PART' mask. Enter 11 messages.

-> PROGRAM -> MES S T A T I C P A	SAGES -> R T		SIMATIC S	5 / COM525
	DRIVE:	B PROGRAM:	PT88ABSP COMPO	NENT: PT
Ness attributes No. B E C U	Gr. A No. page width	Inse : 080 0	ert Line.no.: FF Column m	0001 .: 006
0 1 0 0 Y N N Y Y 0101 N N N N N 0 1 0 2 N N N N N 0 1 0 4 Y N N N N 0 1 0 4 Y N N N N 0 1 0 5 Y N N N N 0 1 0 5 Y N N N N 0 1 0 7 N N Y N N 0 1 0 8 N N Y N N 0 1 0 9 N N Y N 0 1 1 0 N N N Y N	00000 TITTITI DDDDD TITTITIT DDDDD TITTITIT DDDDD TITTITIT DDDDD TITTITIT DDDDD TITTITIT DDDDD TITTITIT 00000 TITTITIT DDDDD TITTITIT DDDDD TITTITT	TT MESSAGE 100 sta MESSAGE 101 bi HESSAGE 102 ch TT MESSAGE 103 MESSAGE 103 MESSAGE 105 1 MESSAGE 105 1 MESSAGE 10732 T MESSAGE 108 fi T MESSAGE 109 ti MESSAGE 110 co	atic text and ac. si nary value: laracter: byte variable: a 16-bit BCD number: 6-bit fixed pt. no.: 32-bit BCD number: 10-bit fixed pt. no.: loating point number mer value: unter value:	ign. char. # ### ##### ########################
F 1 F 2 Delete Block	F 3 F 4 Search/ Replace curs	4 F 5 INSERT OR ON	F6 F7 ENTER	f 8 Exit

The text editing window can be up to 136 columns wide. Of these, only 59 columns can be displayed on the PG screen. As soon as the cursor is moved beyond the margin of the editing window, the visible area is moved by 20 columns.

The character # is the reserve character for message parameters. Each message can have **one** parameter. Messages with a parameter must be numbered consecutively when the standard function blocks 'signaling functions for standard **CPs'** are used (see also Section 7.2.4).

In the input fields for the attributes you specify the way in which the static parts of a message will be represented. You can select bold print (B), expanded **print** (E), **compressed** print (C), underlining (U) and acoustic signal (A) by entering "Y" (yes) or "N" (no). Various combinations of attributes are programed. Do notenteranygroupnumbers.

By using the function keys $\ensuremath{\texttt{F1}}$ to F4 for editing functions the set of function keys previously described is replaced by a different set.

The operator environment of the message editor is the same as that for the PSL editor. The most **important** functions are explained below. A more detailed description of the message editor can be found **in** the user's guide for the **COM** 525 programming: package in Sections 5.3.2 and 5.3.3.

Using the cursor keys with single arrows moves the cursor character by character; the double arrow keys **moves the cursor** word **by** word.

The character \mathbf{on} which the cursor is currently positioned can be deleted with the delete key (in the cursor key block).

If you call anew set of function keys using function keys **F1** to F4, the cursor cannotbemovednorcanentriesbemadeinthe editing window. **The** following keys are defined:

F1 (DELETE) calls the delete menu:

- F2 (WORD)

deletes all the characters from the current cursor $\ensuremath{\text{position}}$ to the end of the word

- F3 (LINE) deletes the line in which the cursor is positioned
- F4 (BLOCK)

deletes the block of lines marked

and the second second

F2 (BLOCK) calls the **block** kindling menu:

- F1 (SET START)

the start of a block is $\ensuremath{\mathtt{marked}}$ in the mask with S $\ensuremath{\mathtt{in}}$ the line in which the cursor is positioned

- **F2** (SET **END)**

the end of a block is marked with ${\bf E}$ in the first column of the mask in the line in which the cursor is located

- F3 (COPY)

a block of lines are copied in before the line in which the cursor is located.

– F4 (MOVE)

a block of lines are moved \mbox{in} before the line \mbox{in} which the \mbox{cursor} is located.

F3 (SEARCH) branches to the search menu:

- F1 (SEARCH)
 - allows a string of up to 20 characters to be searched for

```
- F2 (REPLACE)
```

allows a string of up to 20 characters to be replaced

F4 (CURSOR) roves the cursor:

- F1 (LEFT MARGIN)

to the start of the message line

- **F2 (RIGHT** MARGIN)

to the end of the message line

- F3 (INPUT FIELDS)

into the input fields for the message **number**, the attributes and the message **group** number. Using the **PG** 685 cursor keys you **can** now **move** from **column** to **column** and line to line within the attributes. With the function key **F4** ('10 **ED.VINDOW)** you return to the text editing window.

F5 (INSERT ON/OFF)

switches the INSERT writing \mathbf{mode} on or off. Characters entered are either inserted before the current cursor position (ON) or they overwrite text that is already there (OFF).

F6 (ENTER) saves all the programmed messages on the hard disk.

F8 (EXIT) returns you to the 'MESSAGES ' mask. Press F2 (DYN. PART) in this mask to call the 'DYN. PART' mask.

	DRIVE	В	PROGRAM:	PT88ABSP	COMPONEN	IT: P
Attributes Gr No. m B E C U A No. Pa	ess.: 0011 ge width: 080			Co	lumnno.: 0	50
0101 N N N N N 0102 N N N N N 0103 N N N N N 0104 Y N N N N	DODOD TTTTTTTT I DODOD TTTTTTTTT DODOD TTTTTTTTT DODOD TTTTTTTT DODOD TTTTTTTT	HESSAGE 1 MESSAGE 1 MESSAGE 1 MESSAGE	01 binary 02 charac 103 byte 104 16-	value: cter: variable:	# # ###	;
0105 Y N N N N 0106 Y N N N N 0107 N N Y N N	DDDDD TTTTTTTT DDDDD TTTTTTTT DDDDD TTTTTTTT	MESSAGE MESSAG MESSAGE	105 16-b E 10632 107 32-b	bit fixed p -bit BCD bit fixed p	pt. no.: ### number: ### pt. no.: ###	# ### ####
0105 Y N N N N 0106 Y N N N N 0107 N N Y N N	DDDDD TTTTTTT DDDDD TTTTTTT DDDDD TTTTTTT DDDDD TTTTTTT	MESSAGE MESSAG MESSAGE	105 16-k E 10632 107 32-k	-bit fixed p -bit BCD it fixed p Attribu bold p expan underli acoust	number: ### number: ### pt. no.: ### utes (Y/N): print ded print N ined N ic signal M	** ***** *****
0105 Y N N N N 0106 Y N N N N 0107 N N Y N N	DDDDD TTTTTTTT DDDDD TTTTTTTT DDDDD TTTTTTTT	MESSAGE MESSAG MESSAGE	105 16-t E 10632 107 32-t	Attribu bold p expan underli acoust	pt. no.: ### pt. no.: ### pt. no.: ### utes (Y/N): print ded print N ined N ic signal N F 7 F	***** ******

In this mask only the type of representation (attributes and places before and after the point) of the parameters is determined. The number of characters resenred for message parameters (#) is adopted as the default for the number of places (before the point). The parameter type is determined only when you program the STEP 5 program (data block DB M-PARAM, see section 7.2.4).

With F1, F2 and F5 you select the parameter to be programed; it is displayed on the screen inversely. Then the attributes and number of characters are programmed. These are written into the PG RAM individually sing F6. The next parameter is then selected with F1, F2 or F5, programmed and saved. Continue in this way until all the parameters have been programmed. When you exit the mask with F8 (EXIT) all the dynamic parts are written to the hard disk.

 $\sim \infty$

Specify the following attributes (input ``Y") and ${\tt number}$ of ${\tt char}$ acters for the parameters:

Binary value:	BOLD	no. of char.: 1.
Character:	EXPANDED	no. of char.: 1.
Byte variable:	UNDERLINED	no. of char.: 3.
16-bit BCD number:	BOLD	no. of char.: 4.
16-bit fixed pt. no.:	EXPANDED	no, of char.: 6.
32-bit BCD number: U	JNDERLINED	no. of char.: 8.
32-bit fixed pt. no.:	BOLD	no. of char.: 10.
Floating point no.:	EXPANDED	no. of char.: 5.4
Timer Value:	UNDERLINED	no. of Char.: 3.1
Counter value:		no. of char.: 3.

Press F4 (COMPLETE DYN. PAM') in the 'MESSACES' mask to check whether all the variable fields in the 'STATIC PART' marked with # have been specified in the 'DYN. PART'.

If all the variables have been specified, the message 'MESS. 016: All fields from message 0100 to 0110 already **programmed!'** appears above the function keys.

If variables are still unspecified, the 'COMPLETE DYN. PART' mask appears. Here, you can specify the variable fields for which no type of representation has yet been programmed.

Transfer the process messages and the PC job **SEQ.MESSAGE** (sequential message list) to the user memory of your **CP** 525.

For the process messages, press the **function** keys F6 **(USER DATA)** and F2 **(MESSAGE)** in the 'TRANSFER' **mask**. Transfer all the process messages with F5 (TOTAL) in the 'MESSAGE' mask. The PC job **SEQ.MESSAGE is transferred** in the **same** way as the NEW PAGE PC job (see Section 5.1).

After the transfer, carry out a cold **restart** on the **CP** 525 by pressing **F1**(COLD RESTART) in the 'TRANSFER' mask. Then the red **LED** on interface 1 (X1? 1) must go out.

7.2 STEP 5 User Program without COM PMC

The listing of messages can be carried out in the CPU by **standard** function blocks. In addition to the **handling blocks**, the **signal**ling functions for standard CPs are then also required (see 'Ordering data in this manual).

You Can output messages on a printer without the 'signaling functions for standard **CPs'**. To do this, you must use the interface between the CPU and **CP** 525 as described in the user guide 'Event **output** and listing with the **PT88/PT89** printer'. (See also Section 7.5 in this example).

In this example of an application and in Section 7.3 the "signalling functions for standard CPs" are used. This means that the program runs completely differently from the STEP 5 programs you have seen up to now. You must therefore open a new STEP 5 program file, "PT88R2ST. S5D". Transfer the handling blocks, the function blocks, program and organization blocks (not the **data** blocks!) for "signaling functions for standard CPs" for the R processor to this program file.

7.2.1 Signalling Functions for Standard CPs

These function blocks, which must be ordered separately, detect changes in the message statuses and generate 6 or 8 data word long message records from these changes. The message records contain the **following**:

message number

- message status
- date and time of their occurrence
- parameter type
- **pa**rameter (if present)

They are transferred to the CP 525 in the data block DB M-SEND.

 \sim

The structure of a message record and of DB M-SEND can be found in Section 2.2 of the user's guide "Event output and listing with the **PT88/89** printer" and in Section 7.5 of this example.

From the event, i.e. the change of status in a **data**, input, output or flag word, the data block **DB** M-SEND is generated in a chain of function and data blocks as follows:



The function and data blocks are supplied as part of a simple example program, which must be adapted to your own particular situation. Block numbers and variables are for the most part retained in this example.

The programing or parameter assignment of these blocks can be carried out easily with the **programming package COM PMC, which** must be ordered separately in addition to the standard function blocks (see Section 7.3). **PMC** stands for "Process monitoring and control system".

The **following** sections describe the **STEP** 5 programming and parameter assignment of the blocks without **CCM PMC**. In this way you gain **a** better insight into the way in which the blocks function. The terms used come mainly from the description of the **"signalling functions** for standard **CPs"**.

The function blocks must be called in succession in a cyclic program. Program **block PB** 3 is used. The **PB** 3 supplied is reduced to the calls that are absolutely necessary.

: C DB6 SELECT DB PMC :JU FB151 NAME : SELECTION Ξ :JU FB152 NAME : EDGE :JU FB154 NAME :MESS :JU FB155 NAME : TRANS . MESSAGE ENABLE FOR CP 525 (COMMUNICATION **:A** D5.8 :JC FB141 & MONITORING DEVICE 1) NAME : PUT BLNR : **KF+5** TRANSFER DB M-SEND TO CP 525 :JU FB143 NAME :PUT-A SEND ALL FUNCTION :BE

The function blocks have parameters assigned indirectly in data **block DB PMC** (DB 6). **DB** 6 **must be open** before the function blocks are called.

The **CP** 525 is the only operator **communication** and **monitoring (C & M)** device. The transfer of the message **records** to the **CP** is achieved by calling the PUT function block; the handling block parameters for the transfer must be located in parameter field 5 of the data block **DB PARA-S** (see also section 7.2.5). The enabling of a message is controlled **automatically** by the **function** blocks using data bit 5.8 in data block **DB** PMC **DB** 6 (see also Section 7.2.2).

PB 3 is called unconditionally **in every program** cycle in OB 1.

المعتوي وداهدت الرقب المعتدان وبالمتعاود الاخاذ العايد

47

,

B8576542-04

Example PT88/PT89

With the S5-115U, the cycle time must be triggered extra in PB 3.

Select the following data block numbers and set up the data **blocks** with the following lengths:

 DB M-OLDNEW:
 DB 201
 255 data words

 DB M-ACKN1:
 DB 202
 79 data words only with S5-1500

 DB M-NUMBER:
 DB 204
 256 data words

 DB M-RECORD:
 DB 205
 256 data words

 DB M-SEND:
 DB206
 256 data words

المراجع المتعمية والتدارين والموام والتارين والمراجع ويتعددون التارين والوالي والمراجع

Assign the following data words in the data **block DB M-OLDNEW** DB 201:

253 :KF= +00100lower limit of message numbers254 :KF= +00001number of words occupied by
message bits in DB M-OLDNEW

The CP 525 requires this information for outputting the current message list (see also Section 8.2).

7.2.2 Assigning Parameters to the Signaling Function Blocks DB PMC

The assignment of parameters to the signaling function blocks is carried out indirectly in a data block, DB PMC. Select DB 6. The listing with the CP 525 is the only communication and monitoring function. Since you have not connected an operator-process communication device (e.g. CP 526), it is not possible to acknowledge (nevertheless, a DB M-ACKN1 - DB 202 will be required in the S5-150U).

Set up DB 6 in its full length (256 data words).

48

م^{یر}وری کر به دادهم به م به برید.

•

The following entries must be made in the 256 word long data block **DB PMC (DB** 6):

0 1	:	KM = 000000000000000000; KM = 0000000000000000000000;	Signaling package present The interface c (printer) is present. This must have params . assigned from DW 136 onwards.
4	:	кн = 0001;	The printer is C + M device 1. (No further C + M devices).
5 6	:	KM= 000000000000000000000000000000000000	Message enable C + M device 1. Device 1 has the master clock.
20	:	KS= DB	The condition codewords of the handling blocks are written into data blocks.
33	:	KF= +00208;	DB M-INPUT is DB 208.
38 39	:	KF= +00201; KF= +00000;	DB M-OLDNEW is DB 201. no DB M-ACKN1 (DB 202 with S5-150U)
41 42	:	KF= +00204; KF= +00205:	DB M-NUMBER is DB 204.
43	:	KF= +00206;	DB M-SEND is DB 206.
44	:	KF= +00099;	DB M-PARAM is DB 99.
45	:	KF= +00014;	DB PARA-R is DB 14.
40 47	•	RF = +00013; KF = +00100.	DB FAKA-5 18 DB 15. Inver limit of message numbers
48	:	KF= +00110;	Upper limit of message numbers
49	:	KF= +00101;	Mess. no. with lower limit param.
50	:	KF = +00110;	Mess no. with upper limit param.
90 91	:	KF= +00006; KF= +00007;	DB PMC is data block DB 6. PB-HDB is program block PB 7.
136	:	KF= +00000;	Interface number of the CP 525 Printer interface (IF 1) =0
137	:	K)?= +00006;	Block size = 512 bytes

Another data words can be preassigned with 0.

7.2.3 Determining the Origin of the Message, DB M-INPUT

The origin of the message is entered in **DB** M-INPUT according to type, area start and area length. Messages are tobe triggeredby input word **IW** 0. Select data block **DB** 208 as follows:

0:	KY= 001,000;	input word type
1:	KY= 000,001;	from input word O, one word
2:	KH= FFFF,FFFF;	end identifier (two data words)

With the DB M-INPUT the origin of a message is automatically assigned to a message number. If the cause of a message is located in a data block, bit 0 of the first word in area 1 controls the message with the number located in DB PMC (DB 6) DW 47 (upper limit of message numbers). The following bits control the messages with the following message numbers. The message numbers of the maximum 10 input areas follow on without gaps.

In contrast to data words, the flags, **inputs** and outputs are organised in bytes. When imaging the bytes on a message **origin** word, the byte with the number programmed in DB M-INPUT is imaged in the higher order part of the word. The byte with the next number is transferred to the lower order part of the word.



7.2.4 Message Parameters DB M-PARAM

The message parameters are programmed in data block **DB M-PARAM**. You specified the message range by means of **parame**ters in DB PMC (DB 6) in **DW** 49 and **DW** 50. Beginning with the lowest message number in **DW** 49 (in the example, message 101), the parameters for the **successive messages are** fixed. Select data block **DB** 99 as the DB **M-PARAM** data block.

O:	кн= 0004;	message 101: bit variable; bit 4 of
1:	KY= 030,001;	data block DB 30, data word DW 1
2:	KH= 0108;	message 102: char. variable ; left byte
3:	KY= 030, 002;	data block DB 30, data word DW 2
4:	KH= 0200;	message 103: byte variable; right byte
5:	KY= 030,007;	data block DB 30, data word DW 7
6:	KH= 0300;	message 104: 16-bit BCD value
7:	KY= 030,008;	data block DB 30, data word DW 8
8 :	KH= 0500;	message 105: 16-bit fixed point number
9:	KY= 030,009;	data block DB 30, data word DW 9
10 :	KH= 0400;	message 106: 32-bit BCD value
11 :	KY= 030,010;	data block DB 30, from DW 10
12:	KH= 0600;	message 107: 32-bit fixed point number
12 :		5 1
	KY= 030,012;	data block DB 30, from DW 12
14 :	KY= 030,012; KH= 0700;	data block DB 30, from DW 12 message 108: floating point number
14 : 15 :	KY= 030,012; KH= 0700; KY= 030,014;	data block DB 30, from DW 12 message 108: floating point number data block DB 30, from DW 14
14 : 15 : 16 :	KY= 030,012; KH= 0700; KY= 030,014; KH= 0800;	data block DB 30, from DW 12 message 108: floating point number data block DB 30, from DW 14 message 109: timer value
14 : 15 : 16 : 17 :	<pre>KY= 030,012; KH= 0700; KY= 030,014; KH= 0800; KY= 030,016;</pre>	data block DB 30, from DW 12 message 108: floating point number data block DB 30, from DW 14 message 109: timer value data block DB 30, data word DW 16
14 : 15 : 16 : 17 : 18 :	<pre>KY= 030,012; KH= 0700; KY= 030,014; KH= 0800; KY= 030,016; KH= 0900;</pre>	data block DB 30, from DW 12 message 108: floating point number data block DB 30, from DW 14 message 109: timer value data block DB 30, data word DW 16 message 110: counter value

If there is not sufficient room **in** the data block to write the parameters (more than 128 messages with parameters) you can **continue to write the** parameters **into the data block with the** next higher **data** block number. If there are more than 256 messages with parameters takethenext higher **data block and so** on.

7.2.5 Assigning Parameters in DB MBA-R and DB MBA-S for Triggering Jobs

PC jobs are triggered by the 'signaling functions for **standard CPs'** on the **CP** 525 with PUT or **GET** function block calls. The function blocks trigger the job and the data transfer with the help of handling blocks.

The function blocks PUT and GET have parameters assigned to them indirectly. The function block **parameter** is a field number. You must program the handling block parameters for the **PUT** call (GET call) in the parameter field of the data block **DB PARA-S** (DB **PARA-R**) with the seine number.

A **parame**ter field is 10 data words long. The **first** data word in the parameter field N has the number (N - 1) 10. The handling block parameters are programmed in the parameter field; the condition codeword and parameter assignment error byte of the corresponding handling block call are also indicated here (see also program examples).

The **PB-PUT** function block transfers data directly to the **CP** 525, or triggers a job which is completed with a PUT-A function. The **FB-PUT** monitors the job and repeats it up to three times if an error occurs. **FB-PUT** uses the SEND handling block for the data transfer.

The function block **FB-PUT** is called unconditionally, PUT jobs are triggered by the setting of bit 0 in the first data word of the corresponding parameter field of **DB** PAM-S for one cycle.

If no data is to be transferred to the **CP** with the PUT call, then bit 2 in the first data word of the parameter field **in** DB **PARA-S** must also be set before each PUT call. Instead of a **SEND DIRECT** with the source type XX, a SEND DIRECT with source type NN **is** called.

'Job running is indicated in bit 3 of the f **irst** data word, **'job** complete without errors in bit 4, **'job** aborted' in bit 5 and the number of repetitions (0 . . . 2) in data byte DL 0.

The **function** block **FB-GET** tranfers data directly from the **CP** 525 to the CPU. **FB-GET** uses the handling block **RECEIVE**. GET **calls** are required only for special jobs such as reading the date/time, **SYSTAT** or **SYSID**.

The actual handling block calls must take place in a program block, **PB HDB**. The program block number must be entered in **DB PMC**, data word **DW** 91 (see also Sections 7.2.2 and 7.2.6).

Parameter field 1 in **DB PARA-R** and parameter fields 2 and 5 in DB **PARA-S** are necessary for listing messages.

You can also program the triggering of other PC jobs such as form feed and the **output** of process **status** lists with PUT or GET calls. These jobs can be carried out immediately following a handling block call; they are not **automatically** repeated if an error occurs. To achieve a uniform structure for triggering PC jobs, all PC jobs are triggered by PUT and GET function blocks.

The parameter fields up to number 5 **in** DB **PARA-R** and up to and including number 13 in DB **PARA-S** are reserved for further **PMC** functions. The fields immediately following these fields are available for triggering additional PC jobs.

Set up data block **DB** 14 as **DB PARA-R** with a length of 10 data words and **data** block DB 15 as **DB PARA-S** with a length of 150 data words.

Data block DB PARA-R DB 14, field 1 (read time from the CP 525 hardware clock into DB PMC; see also Section 10,2)

O:	KM= 000000000000000;	job status
1 :	KF= +00000;	interface number 0
2:	KF= +00218;	job number 218
3:	KS= DB	destination type data type
4:	KF= +00006;	destination data block DB PMC DB 6
5:	KF= +00051;	destination start data word DW 51
6:	KF= +00005;	destination length 5 data words
7:	KM= 000000000000000;	condition codeword (reply)
8 :	KM= 0000000000000000;	condition codeword (reply)
9:	KH= 0000;	parameter assignment error (reply)

Data block DB PARA-S DB15 field 2, (set master identifier in CP 525 in FB-START FB 130; see Sections 10.1 and 10.3)

10	:	KM = 000000000000000;	job status
11	:	KF= +00000;	interface mmber 0
12	:	KF= +00218;	job number 218
13	:	KS= DB	source type data block
14	:	KF = +00006;	source data block DB PMC DB 6
15	:	KF= +00051;	source start data word DW51
16	:	KF= +00001;	source length 1 data word
17	:	KM= 000000000000000;	condition codeword (reply)
18	:	KM= 00000000000000;	condition codeword (reply)
19	:	KH= 0000;	parameter assignment error (reply)

Data block DB MBA-S, field 5 (trigger sequential message list)

40	:	KM= 0000000000000000;	job status
41	:	KF= +00000;	interface number O
42	:	m?= +00003;	job number 3
43	:	KS= DB	source type data block
44	:	KF= +00206;	source DB DB M-SEND DB 206
45	:	KF= +00000;	source start data word DW 0
46	:	KF = +00000;	source length is entered
			dynamically by FB-TRANS, FB 155
47	:	KM= 0000000000000000;	condition codeword (reply)
48	:	KM= 000000000000000;	condition codewor d (reply)
49	:	KH= 0000;	parameter assignment error (reply)

The PC jobs for a new page and for the **PSL which have up** to now been triggered by a SEND DIRECT call are now also triggered by PUT calls. The corresponding parameter fields must be programed as follows. Select the first completely free usable fields 14 and 15 in DB **PARA-S**:

Parameter field 14 for the PC job NEW PACE

130 : KM= 00000000)000000; job stares
131 : KF= +00000 ;	interface number O
132 : KF= +00001;	job number l
133 : KS= NN	source type NN (no data)
134 : KF= +00000;	source DB not relevant
135 : KF= +00000;	source start not relevant
136 : KF= +00000 ;	source length not relevant
137 : KM= 00000000	0000000; condition codeword (reply)
138 : KM= 00000000	000000; condition codeword (reply)
139 : KH= 0000 ;	para. assignment error (reply)

Parameter field 15 for the PC job PSL-EXAMPLE

140	:	KM= 00000000000000;	job status
141	:	KF= +00000;	interface number 0
142	:	KF = +00002;	job number 2
143	:	KS= NN	source type NN (no data)
144	:	KF= +00000;	source DB not relevant
145	:	KF= +00000;	source start not relevant
146	:	m?= +00000;	source length not relevant
147	:	KM= 00000000000000;	condition codeword (reply)
148	:	KM= 0000000000000000;	condition codeword (reply)
149	:	KH= 0000;	para. assignment error (reply)

7.2.6 Handling Block Calls PB-HDB

Handling blocks are called by the function blocks START-UP, PUT, GET, PUT-A and GET-A. The handling block calls mat take place in a program block whose block mumber is entered in data block DB PMC DB 6 data word DW 91. Select program block PB 7.

55
The program block PB-HDB PB 7 is supplied with the 'signaling functions for standard CPs".

:C DB6 **:A** D 92.8 :JC FB126 NAME : SEND-A :JC FB127 NAME : REC-A **:A** D 92.10 :JC FB123 NAME :CONTROL Ξ :JC FB120 NAME :SEND

 ANZW :
 FWO

 QTYP :
 KSXX

 DBNR :
 KYO, 6

 QANF :
 KF+22

 QLAE :
 KF+0

 PAFE :
 FY199

SELECT DB PMC SEND ALL ENABLED? SSNR :KY255,18INTERFACE NO. PROM DB 6 DW 18A-NR :KY0,0JOB NUMBER FROM DB 6 DW 19ANZW :FWOC. CODEWORD ADDR. DB 6 DW 20/21PAFE :FY199PARAMETER ASSIGNMENT ERROR BYTE :A D92.9 RECEIVE ALL ENABLED ? NAMEKEC-ASSNR :KY255,18A-NR :KY0,0JOB NUMBER FROM DB 6 DW 19ANZW :FW0C. CODEWORD ADDR. DB 6 DW 20/21PAFE :FY199PARAMETER ASSIGNMENT ERROR BYTE CONTROL ENABLED? SSNR :KY255 ,18INTERFACE NO. PROM DB 6 DW 18A-NR :KY0,0JOB NUMBER FROM DB 6 DW 19ANZW :FW0C. CODEWORD ADDR. DB 6 DW 20/21P m :FY199PARAMETER ASSIGNMENT ERROR BYTE :A D 92.11 SEND XX ENABLED? NAMME : SENDSSNR :KY255,18A-NR :KY0,0JOB NUMBER FROM DB 6 DW 19ANZW :FW0C. CODEWORD ADDR. DB 6 DW 20/21QTYP :KSXXINDIRECT PARAMETER ASSIGNMENT!DBNR :KY0,6QANF :KF+22QLAE :KF+0PAFE :FV100 PARAMETER ASSIGNMENT ERROR BYTE

¥7 ·

الحوي الوادي المراجعة فالمراجع والمحاج

.

Example PT88/PT89

Continuation PB-HTB: **:A** D 92.12 :JC FB120 NAME :SEND SSNR : KY255,18 A-NR : KYO, o ANZW : FWO M.2.K.2.K.3.DBNR :KY0, oQANF :KF+0 QLAE : KF+0 PAFE : FY199 :A D 92.13 RECEIVE ENABLED ? :JC FB121 NAME :RECEIVE ANZW : FWO ZTYP : KSXX ZBNR : KYO,6 ZLAE : KF+0 PAFE : FY199 **:A** D 92.14 :JC FB125 NAME :SYNCHRON SSNR : KY255,18 BLGR : PAFE : FY199 :L MB199 ÷Т DR92 :L KBO :T DL92 : BE

SEND NN ENABLED?

INTERFACE NO. FROM DB 6 DW 18 JOB NUMBER FROM **DB** 6 **DW** 19 C . CODEWORD **ADDR. DB** 6 **DW** 20/21

PARAMETER ASSIGNMENT ERROR BYTE

SSNR :KY255,18INTERFACE NO. FROM DB 6 DW 18A-NR :KY0, oJOB NUMBER FROM DB 6 DW 19ANZW :FWOC. CODEWORD ADDR. DB 6 DW 20/21ZTYP :KSXXINDIRECT PARMETER ASSIGNMENTZBNR :KY0,6DEST. PARAMETER IN DB PMC DB 6ZANF :KF+22FROM DATA WORD DW 22

PARMETER ASSIGNMENT ERROR BYTE

SYNCHRON ENABLED?

NIZDD,18INTERFACE NO. FROM DB 6 DW 18KY0, oFIELD LENGTH FROM DB 6 DW 137FY199PARAMETER ASSIGNMENT PARAMETER ASSIGNMENT ERROR BYTE

PARMETER ASSIGNMENT **ERROR BYTE RECOPYING** TO DB PMC DB 6 **DW** 92

RESET HANDLING BLOCK ENABLE BITS

The handling block calls are enabled by the corresponding bit in data byte DL 92 of DB PMC (DB 6) . The enable bits are set automatically by the signaling function blocks. Flag byte FY 199 is selected as the parameter assignment error byte. Before the program exits PB-HDB, FY 199 is copied by DB PMC (DB 6) into DR 92 and all the enable bits are reset.

The handling blocks fetch the parameters SSNR, A-NR, QTYP/ZTYP, QANF/ZANF and QLAE/ZLAE in DB PMC from DW 18 or DW22. They are written by the function blocks START-UP, PDT, GET and POT-A. You specify the parameters in DB PARA-S for the PUT triggers (SEND DIRECT jobs) and in DB PARA-R for the GET triggers (RECEIVE DIRECT jobs). The replies to the handling block calls (condition code ANZW and parameter assignment error byte PAFE) are also located there. You can use these to check the processing of the jobs. Their evaluation is explained in the description of the handling block.

7.2.7 Start-up Organization Blocks

Start-up identifiers must be set and the function block START-UP called **unconditionally** in the start-up **organization blocks OB 20,** OB 21 **and** OB 22. **FB-START-UP** sets defaults and synchronieses the printer interface (and other interfaces, if programed).

You specify the SYNCHRON parameters SSNR and BLGR in DB PMC (DB 6) in data words DW 136 and DW 137 (interface c printer). The parameter assignment error byte of the SYNCHRON appears in data word DW 138.

The start-up routine is different for a cold restart (OB 20) and a warm restart (OB 21 and OB 22):

organization block OB 20

:JU PB6 :BE Example PT88/PT89

a material states

Organization blocks OB 21 and OB 22 :**JU** PB8 : BE The cold **restart** program block **PB** 6 is supplied with the "signaling functions for standard CPs". :C DB6 SELECT DB PMC :L KBO : T DL8 RESET START-UP **IDENTIFIERS** :AN D 8.15 **:S** D 8.15 SET COLD RESTART IDENTIFIER :JU FB130 NAME :START-UP **:A** D 8.15 R D 8.15 RESET COLD BESTART IDENTIFIER :AN D 8.13 **:S** D 8.13 SET AUXILIARY IDENTIFIER : BE Warm restart program block PB 8 is also supplied :C DB6 SELECT DB PMC :г кво :T DL8 RESET START-UP **IDENTIFIERS** :AN D 8.14 **:S** D 8.14 SET WARM RESTART IDENTIFIER :JU FB130 NAME :START-UP **:A** D 8.14 RESET WARM RESTART IDENTIFIER R D 8.14 :AN D 8.13 **:S** D 8.13 SET AUXILIARY IDENTIFIER BE

FB 130 must also be called in organization block OB 1 at the beginning of the cyclic program (see Section 7.2.8).

7.2.8 Cyclic Program

The triggering of the PC jobs must be programmed in **organization** block **OB** 1. In addition, you must also call up the start-up function block FB 130 and the program block PB-MELD PB 3.

The digital output Q O .n (n = job number) *indicates* whether the job was aborted or not. Normally in this situation an error routine tailored to the particular application is called up. If an error occurs, evaluate the error numbers in the **condition** codeword, parameter assignment error byte (words 7 and 9 in *the* corresponding parameter field of the DB **PARA-S**) and the error message **area of** the **SYSTAT** (see Section 11).

	:C DB6	SELECT DB PMC
NAME	:JU FB130 START-UP	CALL START-UP FB
	:AN I 0.3 :R F 2.3 :A I 0.3 :AN F 2.3 :S F 2.3	EDGE EVALUATION: THE RESULT OF LOGIC OPERATION (RLO) IS SET FOR ONE CYCLE IF THE SIGNAL AT INPUT I 0.3 CHANGES FROM O TO 1.
	:C DB15 := D 130.0 :S D 130.2	SELECT DB PARA-S SET INITIATION BIT NO DATA TRANSFER
NAME	C DB6 :JU FB141 :PUT	SELECT DB PMC CALL PC JOB NEW PACE
BLNR	: KF+14 : :C DB15 :A D 130.5 := Q 0.1	FIELD NUMBER IN DB PARA-S SELECT DB PARA-S JOB ABORTED AFTER 3 ATTEMPTS, INDICATE AT OUTPUT Q 0.1

45

OB 1 continued: :AN 10.4 :R F 2.4 :A I 0.4 :AN F2.4 EDGE EVALUATION: THE RESULT OF LOGIC OPERATION (RLO) IS SET FOR ONE CYCLE IF THE SIGNAL AT INPUT I 0.4 CHANGES FROM O TO 1. **:S** F 2.4 := D 140.0 SET INITIATION BIT IN DB PARA-S := D 140.0 :S D 140.2 NO IMMEDIATE DATA TRANSFER SELECT DB PMC :C DB6 :JU FB141 CALL PC JOB PSL-EXAMPLE NAME :PUT BLAR : KF+15 FIELD NUMBER IN DB PARA-S :C DB6 SELECT DB **PMC** SEND ALL FUNCTION :JU FB143 NAME :PUT-A SELECT DE **PARA-S** JOB **ABORTED AFTER** 3 **ATTEMPTS**, INDICATE AT **OUTPUT** :C DB15 :A D 140.5 := Q 0.2 PB-MELD : SIGNALLING FUNCTION BLOCKS JU PB3 :C DB15 SELECT DB PARA-S :CDB15SELECT DB PARA-S:AD 40.5SML PC JOB ABORTED:=Q 0.3INDICATE AT OUTPUT Q 0.3 BE

بعريرتي فصيفت معاق

.

7.2.9 T-mit Clock Pulse

Message records will be transferred to the CP 525 if

- DB M-RECORD is 90% full of message records or - the transmit clock pulse (DB **PMC DW** 8 bit 8) is "1".

By inverting data bit D 8.8 in DB **PMC**, generate a transmit clock pulse of200ms in **organization block OB 13 as** foil-:

: C	DB6	SELECT	DB PMC		
: AN : -	D 8.8 D 8.8	INVERT	TRANSMIT	CLOCK PULSE	BIT
BE					

Transfer all the organization, program, **signalling** function and message data blocks to the user **memory of the CPU**. Copy data block DB 30 for the variables in the messages and the process status list (see Section 6.2) from the file **PT88R1ST.S5D** and transfer them to the CPU. Switch the CPU to **"RUN"**.

7.3 STEP5 User Program with COM PMC

In this section the same messages will be triggered as were programed with the "signaling functions for standard **CPs"** in Section 7.2, however, in this case with the **S5-DOS** programming package **COM PMC**.

With COM PMC, all the parameters for the "signaling functions for standard CPs" are assigned in user-friendly screen menus. COM PMC generates the data blocks required for the signaling functions and saves the parameters in these data blocks.

The **CP** 525 user program **remains** unchanged (see Section 4.3, 5.1, 6.1 and particularly 7.1).

The triggering of the messages is to be integrated into the already existing program **PT88R1ST.S5D** (see Section 5.2 and 6.2). Transfer all function, program, data and organization blocks **except OB 1** of the "signaling functions for standard **CPs" MELL35ST.S5D** for the Reprocessor to the file **PT88R1ST.S5D**.

The jobs for a new page and a process status list are already programed in OB 1 of **PT88R1ST.S5D**. These remain part of the program.

.

.

During the start-up, the **SYNCHRON** handling block is no longer called **immediately**, but rather the START-UP **function** block (see Section 7.2. 7). The parameter assignment error byte of the **SYNCHRON** for the printer interface is now in DB **PMC DB** 6 in data word 138.

Delete the first statement in organization block OB 1:

:A F 10.0

For message processing, **only** the function block START-UP and the **program block PB-MELD PB** 3 **must be** called **in** organization block OB 1. Digital output Q 0.3 indicates whether the **SML** PC job was aborted (see Section 7.2 .8).

	C DB6	SELECT DB PMC
W :	:JU FB130 START-UP	CALL START-UP FB
	:C DB6 :A D 138.0 := Q 0.0	SELECT DB PMC IF SYNCHRON PARA , ASS . ERROR OCCURRED> SET OUTPUT
	: JU PB3	PB-MELD: SIGNALING FUNCTION BLOCKS
	:C DB15 :A D 40.5 := Q 0.3	SELECT DB PARA-S SML PC JOB ABORTED INDICATE AT OUTPUT Q 0.3
	:Triggering PC job NEW	PAGE see Section 5.2
	: Triggering PC job PSL	EXAMPLE see Section 6.2
	:BE	



B8576542-04

7.3.1 Installing COM PMC on the PG 685

You *require* a **PG** 685 **progr**ammer, on which the **STEP5** basic package is already installed.

Go to drive B: user area O by inking the following entry:

0:

Copy the **contents** of the **COM PMC** diskette to the hard disk drive, user area 0 with the following **comman**d:

PIP **B:=A:*.*[R** V]

Option V = verify R = copy SYS files

Declare the files as write-protected system files with the following commands :

```
SET S5??C?4X. CMD[SYS RO]
SET S5??C?4X. DAT[SYS RO]
```

You can now work in any user area with COM PMC.

User area O should be **reserved** for system files. Exit this area with the **comman**d as follows:

n: n = **required** user area.

65

بې يو يې يې ^يې مړو ته يې يو يو. و

88576542-04

.

•• •

7.3.2 Parameter Assignment with COM PMC

Call the "PACKACE SELECTION" mask with the following command:

S5

Position the cursor on the "COM PMC ., ," line and select the COM PMC programing software with Fl (PACKACE).

The COM PMC logo mask appears.

Copyright (c) by SII PROCESS MONITORING	EMENS AG And Control System (PM	2)	SIMATIC S	5 / COM PMC
333333 55 55 55 55 55 55 55 55 55 55 55 55 55	OCOOOO MM MM co 00 MMM MMM m O MM MMM 00 00 MM MM 00 00 MM MM	PPPPPP MM PP PP MM PP PP MM PPPPPP MM PP MM PP MM	MM CCCCC MMM cc ci MM CC MM cc MM CC MM CC MM cc ci	C
VERSION	PROGRAMING PACK PROCESS MONITORING AND I: XXX XX. XX.XX SI	AGE FOR THE Control System Erial-No. : XXXX-	XXXX-XXXxXx	
F 1 F 2 Local Master System System	File: B: PT8	8R1ST. S5D F5 F6	F 7 HELP	F 8 Exit

Enter drive **"B"** and the program **name "PT88R1"ST. S5D"** in the input field. You can enter the file name with F7 (HELP).

r.

Listing belongs to the local system functions. Ress function key F1 (local system) to branch to the "select" mask.

This mask already has standard values as defaults that can be used for this example.

LOCAL OPERATOR SYSTEM SELECT	SIMATIC S5 / COM PMC File:A:PT88R1ST.S5D		
CONFIGURED ARE: O MASTER OPERATOR SYSTEM(S) 1 MONITOR(S) O MESSAGE PRINTER			
GENERAL PARAMETERS IN OB 6 (OB PMC) Receive parameters in dB 14 (DB para-R) Seno parameters in dB 15 (dB para-S)			
F1F2F3F4F5F6ONNECTIONMONITORSTANDARDMESSAGEENTERMASTERPRINTERDISPLAYSMESSAGEENTER	F7F8 HELPABORT		

This mask also shows that a monitor but no printer is planned.

Press F2 (monitor printer) to branch to the "monitor and printer" mask.

The messages are not required on the **monitor** via the CP 525 but only on the printer. Overwrite the interface **number** of monitor 1 with blanks and enter "**no**" in the "message" column on the same line.

In the "printer" line select "0" as the interface number (SSNR)
of the CP 525. This is the number of the upper interface 1 (IF 1)
set cm your module. Press F7 (HELP) to enter "yes" in the
"message" column. The job number for the messages must be the
same as the job number for the sequential message list on the CP
525. The preset job number 199 must therefore be changed to "3"
in the "A-NO" column. Enter the parameters with F6 (ENTER).

Local ope Monitor Ai	RATOR SYSTEM ND PRINTER					SIMATIC S5 File:A:PT8	/ COM PMC 38R1ST.S5D
		CP TYPE	SSNR	Monitor- ing	Message	A-No for Messages	-
	Monitor 1 Monitor 2 Printer	CP526 CP526 CP525	0	NO 	No No Yes	3	
		-			-		•
F 1	F 2	F 3	F 4	F 5	F 6 ENTER	F 7 NELP	F 8 ABORT

68

The next stage is to specify the message numbers and data blocks required by the "signaling functions for standard CPS" in the messages mask. Select the "messages" mask with F4 (message) in the "select" mask.

The "message numbers" must be overwritten. "100" is entered for thefirst message. The COM PMC package can only process the messages in words; therefore assign "115" as the last message although you have only programmed up to message 110. The messages with parameters are located in the area between 101 and 110; therefore enter "101" and "110" in the "messages with parameter" field. Do not enter anything in the status or system messages fields.

Enter the data block numbers as used in the file **MEL135ST. S5D** and add DB "99" for the message parameters. DB **M-ACKN1** and DB **M-ACKN2** are required since they are called in **PB MELD PB** 3 by **FB** QUIT FB 153.

LOCAL OPERATOR SYSTEM MESSAGES	SIMATIC S5 / COM PMC File:A:PT88RI ST. S5D
MESSAGE NUMBERS: 100 - MESSAGESWITH PARAMETER: 101 - 1 STATUS MESSAGES: SYSTEM MESSAGES: NO	¹¹⁵ 10
ADDRESSES OF INPUT FIELDSINDB208MESSAGE BITS FOR EDGE EVALUATIONINDB201MESSAGE AND ACKNOWLEDGEMENT NUMBERSINDB204MESSAGE SETSINDB205MESSAGE OUTPUT BUFFERINDB206ACKNOWLEDGEMT. RECEIVE BUFFER FORCP526/1 INDB202ACKNOWLEDGEMT. RECEIVE BUFFER FORCP526/2 INDB203DESCRIPTION OF MESSAGE PARAMETERSINDB1	(DB H-INPUT) (DB M-OLDNEW) (DB M NUMBER) (DB H-RECORD) (DB U-SEND) (DB M-ACKN1) (DB M-ACKN2) (DB M-PARAM)
F1F2F3F4F5F6INPUTMESSAGEENTERFIELDSPARASENTER	F7 F8 HELP ABORT

The input fields (the flag, data, input and/or **output areas) that** can trigger messages must now be specified.

Press **F1** (input fields) in the "messages" **mask** to branch to the "input fields" mask. In this mask, the causes of messages are specified according to type, start of area and length.

The messages are to be triggered by input word IW O. Press F7 (HELP) to enter "I" (input) in the first input field of the "1/O/F/D" column. The number of the first word "IW O" and the length "1" word are entered in the "word no. " and "length" fields.

Cal oper 10t fieli	ATOR SYSTE	M Message	S		S	SIMATIC S ile:A:PT8	5 / COM PM 88R1ST.S5D
MESSAG MESSAG	E NUMBERS: Es with Par	1 RAM: 1	00 - 115 01 - 110	STATU SYSTE	IS MESSAGES: IM MESSAGES:		
Ser. No	I/O/F/D	DB-No	Word No	Length	Message No	·	
1 2 3 4 5 6 7 8 9 10	I		IMO	1	100 - 115 116 -		
F1	F 2 INSERT LINE	F 3 ERASE LINE	F 4 Page Backwards	F 5 PAGE FORWARDS	F 6 ENTER	F 7 HELP	F8 ABORT

The message numbers are displayed automatically by COM PMC.

Ress F6 (enter) to enter the input fields and return to the "messages" mask.

You must now specify the message parameters. Press F2 (message parameters) in the "messages" mask to branch to the 'message parameters" mask.

Press F3 (enter parameter) to move to the 'parameter type" column and the first message with parameter; in this case message 101.

Press F7 (HELP) to enter the appropriate parameter type in the input field. Enter the parameter sources in the messages as follows :

MESSAG	GE NUMBE	RS:	100 -	115	MESS	AGES WI	th pai	ran:	101 -	110
Mess. No.	Bit Add I/O/F/D	ress DB-No	of message WORD BIT	•	Parameter type	Bit Add I/0/F/D	ress o DB-I	f mes: Io WORD	sage) BIT	
100	I		I 1.0		I					
101	1		11.1		BINARY	D	DB :	30 D	1.4	
102			1 1.2		CHAR	D	DB 3	SO DL	2	
103	i i		11.3		BYTE	D	DB 3	sol or	7	
104			1 1.4		16 BIT BCD	D	OB 3	SO DU	8	
105			1 1.5		16 BIT FXP	D	DB 3	sol dw	9	
106			I 1.6		32 BIT BCD	D	DB 3	SO 00	10	
107	1		I 1.7		32 BIT FXP	D	DB 3	so do	12	
108	1		1 0.0		32 BIT FLP	D	DE 3	SO DD	14	
109	I		I 0.1		TIMER	D	DB 3	50 DW	16	

Press F1 (enter mess. no.) to position the cursor in the 'mess. no." input field. Enter the required message number and press F2 or F3. The message then appears at the first position.

Press F2 (enter **bit-addr.)** to move the cursor to the "1/0/F/D" input field. **Enter** the required input and then the bit address in the next input fields and press **Flor** F3. Following **this**, **the** message appears at the first position.

Press F5 (page forwards) in the "message parameters" mask to obtain the higher message numbers.

Press F4 (page backwards) to obtain lower message numbers.

MESSA	ge numbe i	RS:	100 - 11	15 MESS	AGES WIT	h parai	M: 101 -
Ness. No.	Bit Ad	dress DB-Nc	f message IORD BIT	Parameter type	Bit Ado /0/F/D	lress ç DB-No	f message WORD BIT
108	I		I 0.0	32 BIT FLP	D	DB 30	Do 14
109	I		I 0.1	TIMER	D	DB 30	DW 16
110	1		10.2	COUNTER	D	DB 30	DW 17
111	I		I 0.3				
112			I 0.4				
113	1		1 0.5				
114							
115	I		I U.7				
				<u> </u>			

Ress F6 (enter) to return to the "messages" mask. Now press F6 (enter) again to branch to the "select" mask. The parameter assignments are saved on the hard disk in the file B:PT88R1ST.S5D by pressing F6 (enter) in the "select" mask. Data blocks that do not exist in this file are generated; in the example this applies to data block M-PARAM DB 99.

Transfer all the blocks in the user **memory of the CPU. Transfer** DB 30 from **Section** 6.3.1 to the **CPU** as source for the message variables.

switch the CPU to "RUN".

B8576542-03

يساغ

Example PT88/PT89

1

o

7.4 Test

Switch all the digital inputs which trigger messages from 0 to 1. All the messages will be printed with the status coming (1):

 12.31_11:58:42
 pm__MESSAGE_100_static_text_and acoustic_signal_1

 12.31_11:58:43
 pm_MESSAGE_101_binary_value:
 1

 12.31_11:58:44
 pm_MESSAGE_102_character
 A
 1

 12.31_11:58:45
 pm_MESSAGE_103_byte_variable:
 123
 1

 12.31_11:58:46
 pm_MESSAGE_104_16-bit_BCD_number:
 -123
 1

 12.31_11:58:46
 pm_MESSAGE_105_16-bit_fixed pt. no.:
 1234.55
 1

 12.31_11:58:46
 pm_MESSAGE_106_32-bit_BCD_number:
 -1234.55
 1

 12.31_11:58:46
 pm_MESSAGE_106_32-bit_BCD_number:
 -1234.55
 1

 12.31_11:58:57
 pm_MESSAGE_107_Z-bit_fixed pt. no.:
 1234.557_1
 1

 12.31_11:58:53
 pm_MESSAGE_109_tiner value
 @ J
 1

 12.31_11:58:53
 pm_MESSAGE_110_counter_value____123_____1
 1

If you now switch all the inputs which trigger messages from 1 to 0 the CP prints out all the messages with the status going (0):

D1.D1_00:00:13 pm_MESSAGE_100_static_text_and_acoustic_signa	1_0
D1.01 00:00:14 pm MESSAGE 101 binary value:	~ 0
01.01 00:00:15 pm MESSAGE 102 character 🛆	0
01.01 00:00:16 pm MESSAGE 103 byte variable: 123	Q
01.01 00:00:19 pm MESSAGE 104 16-bit BCD number: -123	D
01.01 00:00:20 pm MESSAGE 105 16-bit fixed pt. no.: 12345	
01.01 00:00:22 pm MESSAGE 106 32-bit BCD number: , <u>-1234567</u>	0
01.0100:00:24 pm MESSAGE 107 32-bit fixed pt. ea.: 77851	
01.01 00:00:29 pm MESSAGE 108 floating point number 12.3457 0	
01.01 00:00:30 pm MESSAGE 109 timer value 123.3 0	
D1.01_00:00:31 pm MESSAGE 110 counter value 123	• ••• •

73

,

فتهيئ فحاسي حاصر ماجعوا الديه

یہ ہے ہے۔ رضی تحصیر ا

B8576542-04

If you have not connected a simulator, select flag word 0 for triggering the messages. Switch the flag bits with the PG 685 online function CONTROL VARIABLE. Type 3 (flagword)mustbe programed in DB M-INPUT as follows:

DB M-INPUT DB 208 data word DUO: KY= 003,000;

In this case do not use any digital outputs. Observe the condition codewords (DU 47, DW 137 and DW 147 in **DB PARA-S DB** 15) belonging to the jobs and the **parameter** assignment error bytes (DW 49, DW 139 and DW 149 DB PARA-S DB 15) using CONTROL VARIABLE.

Assuming, for example, that messages 100 and 101 arrive in one program cycle, the data block DB M-SEND **DB** 206 appears as follows (see also "Interface between CPU and CP 525" in the instructions "Event output and logging with the **PT88/PT89** printer" in this manual):

0	:	KY=	051,014;	identifier 33H/body length 14 data words
1	:	ICY=	: 255,001;	no parameter / message coming
2	:	KY=	000,100;	disturbance message /message 100
3	:	KH=	3045	300 milliseconds / 45 seconds
4	:	KH=	3210	32 minutes / 10 hours
5	:	KH=	2810	28th October
6	:	KH=	8700	1987
7	:	KY=	000,001;	binary variable /message coming
8	:	KY=	000,101;	disturbance message /message 101
9	:	KH=	3045	300 milliseconds / 45 seconds
10	:	KH=	3210	32 minutes / 10 hours
11	:	KH=	2810	28th October
12	:	KH=	8700	1987
13	:	KM=	0010	binary variable data bit 4=1
14	:	KM=	0400	bitaddress4

If you transfer this data block using a SEND DIRECT job with job number A-NR 3, source tgpe QTYP DB, data block number DBNR 206, source start QANF 0 and source length QLAE 15 to the CP 525, the messages 100 and 101 are printed out as 'coming" (see Section 7.5).

 $r \sim r$

7.5 **STEP5** User Program without 'Signaling Functions for Standard **CPs**"

This section explains how to trigger the messages without "signalling functions for standard CPs". The simplest case is demonstrated in which the DB M SEND contains only one message record and not up to 31 or 42 (without parameter or with parameter) as when messages are triggered with the standard function blocks. This message record is transferred to the CP 525 by a handling block.

By varying this message record you can get to know how the interface between the CPU and **CP** 525 functions **when** messages are triggered (see also Section 2.2 in the user's guide 'Event output and listing with the **PT 88/PT** 89 printer" in this manual).

Call the S5 package IAI), **CSF, STL.** Program in **STL** (Statement List) and **call** your new **STEP5** program **"B:PT88R3ST.S5D".** Transfer the handling blocks for the R processor to this program file. Connect the Reprocessor with the **PG** 685.

The STEP5programmust

synchronize the $CPU \ and \ CP$ 525 during start-up and trigger the PC job for the output of messages.

7.5.1 Program Start-up

All the start-up organization blocks OB 20, OB 21 and OB 22 call the handling block SYNCHRON unconditionally.

	: JU	FB125	SYNCHRONIZATION CP <> CPU
NAME	: SY	INCHRON	
SSNR	:	KYO , O	INTERFACE NUMBER O
BLGR	:	KYO , 2	BLOCK SIZE 2 (MAX. 32 BYTES)
PAFE	:	FY1O	PARMETER ASSIGNMENT ERROR BYTE

:BE

The interface number SSNR for IF 1 of the CP 525 is the same as the set module address 'O". As block size select "2". This means that data can be transferred between the CPU and CP in groups of a maximum of 32 bytes. If there are **more** than 32 bytes to be transferred, the CP 525 requests the excess bytes with another SEND AIL. The block size is selected so that DB M-SEND can be transferred in one **operation**. Any other block size could be selected.

Use "FY10" as the parameter assignment error byte PAFE.

Any parameter assignment error is to be **indicated** at digital output Q 0.0. Since there is no process image in the start-up organization blocks and therefore no digital 1/0s can be addressed, this indication **can only** occur in the cyclic program (see Section 7.5. 2). Normally an error evaluation program is called during the start-up. If a **parameter** assignment error occurs during the start-up, check that the interface number and module address are the same ("0"). If both are set to '0", there is probably a hardware fault.

Example PT88/PT89

7.5.2 Cyclic Program

Any parameter assignment error detected during the start-up is indicated at digital output Q 0.0 at the beginning of the cyclic program (see Section 7.5 .1).

The PC job sequential message list is triggered by a SEND DIRECT; i.e. by calling a SEND handling block with a job number other than O. The job number of the SEND DIRECT and the number of the SML PC job on the CP 525 must be the same. **Program** the job number as **A-NR** "3". The job is executed if a positive-going edge is detected at input I 0.3.

The interface number is SSNR "0".Select condition codeword ANZW "FW11" and parameter assignment byte PAFE "FY15". The condition codeword occupies two words for a SEND DIRECT call. The job status and any error messages are indicated in flag word FW 11.

For each message output, data block M-SEND must be transferred from the CPU to the CP 525; select data block DB 20. "DB" must be programmed as the source type QTYP; the source parameters are DBNR "20", QANF "O" and QLAE "9". (If there are several message records of six or eight data words in DB M-SEND, the source length must be adjusted accordingly.) The SEND DIRECT call triggers the job; i.e. the job is entered in the job queue. Data M-SEND DB 20 is only requested by the CP 525 by a SEND ALL.

The job is triggered on result of logic operation **(RLO)** "1". If the **RLO** is "O" when the call occurs, only the condition codeword of the SEND **DIRECT** is updated.

If the job is completed with an **error** or if a parameter assignment error occurs, this is displayed at digital output Q 0.3. Normally, in this case, a special error routine is called. If an error occurs, evaluate the error **numbers in** the condition codeword in the error message area of the **SYSTAT** (see Section 11) and in the parameter assignment error byte.

50 5 W

The program in the cyclic organization block OB 1 is as follows:

	:A :=	F 10.0 Q 0.0	IF SYNCHRON PARAMETER ASSIGNMENT OCCURRED> SET OUTPUT
	:AN :R :A :AN :S	10.3 F 2.3 10.3 F 2.3 F 2.3	EWE EVALUATION THE RESULT OF LOGIC OPERATION (RLO) IS SET FOR ONE CYCLE, IF THE SIGNAL AT INPUT I 0.3 CHANGES FROM 0 TO 1
	:JU	FB120	TRIGGER SML PC JOB
NAME SSNR A-NR ANZW QTYP DBNR QANF QLAE PAFE	: SEN : : : : :	KYO, O KYO, 3 FW11 KCDB KYO, 20 KF+O KF+9 FY15	INTERFACE NUMBER O JOB NUMBER 3 CONDITION CODEWORD SOURCE M DB DATA BLOCK DB 20 SOURCE START DW O SOURCE LENGTH 9 DATA WORDS PARMETER ASSIGNMENT ERROR BYTE
NAME: SSNR A-NR ANZW PAFE	: JU SENI : :	FB126 D-A KYO ,0 KYO ,0 FW16 FY18	TRANSFER DB M-SEND TO CP 525 INTERFACE NUMBER O JOB NUMBER O CONDITION CODEWORD PARAMETER ASSIGNMENT ERROR BYTE
	:0 F :0 H := () :BE	1.2.3 7 15.0 2 0.3	JOB FINISHED WITH ERROR OR PARAMETER ASSIGNMENT ERROR INDICATE AT OUTPUT Q 0.3

in carr

Structure of DB M-SEND

	Left byte DL	Right byte DR
DUO	Message ID = 33H	Body length = 6 or 8 DW
DW 1	Parameter type	Message status 0 or 1
DW 2	Message type = 0	Message number
DW 3	100 ms	Seconds
DW 4	Minutes	Hours
DW 5	Day	Month
DW 6	Year	not used
DW 7	PARAMETER VALUE	
DW 8	PARAMETEF	R VALUE

DB M-SEND parameters for the messages from Section 7.1

Mess. no. (DW 2)	Parameter ty <u>r</u>	HEX (DL 1)	PARAN (DW 7)	4 E T E R (DW 8)	status (DR 1)
100 101 102 103 104 105 106 107 108 109 110	No. var. Binary Character Byte 16 bit bcd 16 bit fpt 32 bit fpt 32 bit fpt floating pt. timer counter	FF 00 01 02 03 05 04 06 07 08 09	KH= 0010 KS= BE KY= 0,123 KH= 0123 KF= +12345 KH= F123 KF= +00001 KG= +123456 KT= 123.3 KC= 123	KH= 0400 KH= 0800 KH= 0000 KH= 0000 KH= 0000 KH= 4567 KF= +12345 +02 KH= 0000 KH= 0000 KH= 0000	0 or 1 0 or 1

.

For parameter type **FF** (no parameter) the message record is only six (instead of eight) data words long; i.e. DR O = 6 (instead of **DW** O = 8).

You can read the **date** and time using the PC job RECEIVE DIRECT 218 (see Section 10. 2) **from** the **CP** and enter them in data words 3 to 6. Enter the value "O" for day so that the **CP** 525 reads the **date** and time directly from its hardware clock.

7.5.3 Testing the Program

Using the **PG** 685 online function **CONTROL** VARIABLE, supply the **data** block M-SEND DB 20 with data (see table). Monitor the **condi**tion codeword **FW 11** and the PAFE byte **FY** 15 using the same **CONTROL** VARIABLE function.

If you switch the signal at digital output I 0.3 fram 0 to 1, the printer will print the appropriate message.

8 Current Message List (CML)

Using the **CML** you can print out the disturbance status of a process at a particular time. All the messages that have cane and not yet gone again, i.e. are still "active" on the CPU at the time of the **CML** call are output to the printer by the **CP 525.**

The CP 525 initiates the printout by requesting the data block DB M-OLDNEW immediately after the PC job is triggered by the CPU. The CP outputs to the printer the message texts belonging to all the bits set in the new value area of the DB M-OLDNEW (see also Section 8.3 and the description of "The interface between CPU and CP 525" in the user's guide "Event output and listing with the PT88/PT89 printer" in this manual).

Instead of the dynamic data (date, the, message status and process variable) question marks are printed.

Note: if you use the "signaling functions for standard CPs", DB M-OLDNEW is generated and updated **automatically**. Without the "signaling functions for standard CPS" you must ensure that the required data block is available cm the CPU.

جهي برايا جاله المالة

فالري والالا المالية وال

,

8.1 CP 525 User Program

Switch the mode selector of the CP 525 to "P(2R".

Press F1 (PROGRAM USER DATA) in the 'SELECTION' mask to call the 'PROGRAM' mask and press F2 (PC JOB) to call the 'PC JOB' mask. F3 (CML) then calls the 'CURRENT MESSAGE LIST':

> PROGRAM ->PCJOB -> ∶URRENT MESSAGE	LIST		SIM	NATIC S5 / C	OM525
	DRIVE: B	PROGRAM:	PT88ABSP	COMPONENT	: P T
List name: CUR. MESSA	GE				
With the jodo no.: 4 the output of the current	t message list	is started.			
List header is frame: List trailer is franc:					
Output of messages possib Form feed on PT88/PT89 at	olewhileCML i start of list	is being out	put	(Y/N) N (Y/N) Y	
F1 F2 F3	F4	F 5 F	6 F	7 F 8	8
FRAME		S	AVE HE	ELP EX	IT

Enter the job number '4". Do not enter anything for list header and trailer. Block the output of process messages (enter "N") and program a form feed before each CML (enter "Y"). Save the Pc job CUR.MESSAGE on the hard disk with F6 (SAVE).

Press F8 (EXIT) three **times** to **return** to the **'SELECTION'** mask. Transfer the PC job CUR. **MESSAGE** in the same way that the PC job NEW PACE was transferred to the user memory of the **CP** 525 (see Section 5.1). After the transfer, carry out a cold restart on the **CP** 525 and switch the **CP** 525 to "RUN".

8.2 STEP 5 User Program

Trigger the current message list with a SEND DIRECT job with job number A-NR "4". The data block DB M-OLDNEW is transferred from the Reprocessor to the CP 525 by a SEND ALL call; the SEND ALL is already programmed for the process status list (see Section 6.2) and the sequential message list (see Section 7.2.1).

The termination of the job with an error is indicated at digital output Q 0.4. Normally in this situation an error evaluation program tailored to the particular application is called. If an error occurs, evaluate the error numbers in the condition codeword (FW 24), parameter assignment error byte (PY 28) and the error message area of the SYSTAT (see Section 11).



Add the statements to trigger the PC job to **OB** 1 as follows:

	:AN	I 0.5	EDGE EVALUATION:
	R	F 2.5	THE RESULT OF LOGIC OPERATION
	:A	I 0.5	(RLO) IS SET FOR ONE CYCLE IF THE
	: AN	F 2.5	SIGNAL AT I 0.5 CHANGES FROM
	:S	F 2.5	0 TO 1
	:		
	:JU	FB120	TRIGGER CURRENT MESSAGE LIST
NAME	:SEN	D	
SSNF	₹:	KYO, O	INTERFACE NUMBER O
A-NR	:	KYO, 4	JOB NUMBER 4
ANZW	:	FU24	CONDITION CODEWORD
QTYP	:	KCDB	SOURCE TYPE DATA block
DBNR	:	KYO, 201	SOURCE DATA BLOCK DB M-OLDNEW
QANF	1	KF+0	SOURCE START DATA WORD O
QLAE	:	KF+255	SOURCE LENGTH 255 DATA WORDS
PAFE	:	FY28	PARAMETER ASSIGNMENT ERROR BYTE
	÷		
	:0	F 25.3	JOB 4 FINISHED WITH ERROR
	:0	F 28.0	OR PAMMETER ASSIGNMENT ERROR
	:-	Q 0.4	INDICATE AT OUTPUT Q 0.4

8.3 Test

Switch on all the inputs which trigger a message (I 0.0 to I 0.2 and I 1.0 to I 1.7).

As soon as the signal at digital input I 0.5 is switched from zero to one, the following $C\!M\!L$ is printed out:

??.?? ??:??:?? MESSAGE 100 static text and acoustic signal	?	
??. ?7??: ??:?? MESSAGE 101 binary value: ?	?	
??.?? ??:??: MESSAGE 102 character ?	?	
??.?? ??:?? MESSAGE 103 byte variable: 222	?	
??.??????????? MESSAGE 104 16-bit BCD number: ????	?	
??.?? ??:??:?? MESSAGE 105 16-bit fixed pt. no.: ? ?'??'?		?
??.?? ??:??:?? MESSAGE 106 32-bit BCD number: <u>73777777</u>	?	
??.?? ???????? MESSAGE 107 32-bit fixed pt. no.: ????????????????????????????????????		
??.?? ??:??: ?? MESSAGE 108 floating point number ????????????????????????????????????		
??.?? ??:???? MESSAGE 109 timer value 27777 ?		
??.??_??:??:MESSAGE 110 counter value???/	?	

The data block **DB M-OLDNEW DB** 201 **may** appear as follows in this situation:

0: 1 : 2:	КН= 0001; КF= +xxxxx К = +xxxxx	new values from data word DW 128 write pointer not relevant readpoi.nternotrelevant
3: 4: :	KM= 0000010111111011; KM= 000000000000000;	old values
: 127 :	KM= 000000000000000;	end of old value area
128 : 1.29 :	KM = 0000011111111111; KM = 0000000000000000;	new values
252 :	KM= 000000000	end of new value area
253 : 254 :	KF= +00100 KF= +00001	lower limit of message numbers 1 word occupied with message bits

9 Frames

17

Each list can have a header and/or a trailer. In **SMLs** and **CMLs**, the header is printed out at the beginning of a list and at the beginning of each new page; the trailer is printed out at the end of a list and at the end of each page. In **PSLs**, they are printed out only at the beginning and end of the list. The list header and trailer are known as frames. They are programed as a **PSL**, with a maximum of 9 lines, and cannot be started separately by the CPU.

9.1 CP 525 user program

Press **F1** (PROGRAM USER DATA) and F3 (FRAME) in the 'SELECTION' mask to call the following mask:

-> SELECTION -> PROGRAM -> F R A M E				SI	MATIC S5 / C	XXM525
	DRIVE:	В	PROGRAM:	PT88ABSP	COMPONENT	1: PT
Frame name: HEADER	N	ame to	be stored:	HEADER		
The clynamic data are on C	PU no.: 1					
in data block	0B no.: 30					
F rames can be used as a status lists, sequential n⊄	header an essage lis	d/or tr ts and	ailer for p I currant	process message l i	sts.	
F1 F2 F3 STAT. DYN. PART PART	F 4 Comple t Dyn. Par	E T	5 I SA	F 6 AVE H	F7 F ELP EX	8 XIT

Select the **frame** name 'HEWER", CPU number "1" and data block DB "30" for the dynamic **variables**. "HEADER" is automatically the name for saving the frame unless you specify a different name.

Press **F1** (STAT. PART) to call the 'STATIC **PART' mask**. Program the following frame. The editing is performed **in** just the same **way as** for the PSL (see **Section** 6 .1).

-> PROGRAM -> FR/ STATIC PA	AME -> R T		SIMATIC S5 / COM525
	DRIVE:	B PROGRAM: PT88ABSF	COMPONENT: PT
Attributes Gr E B E C U A N	lement name: HEADER 10. Page wi dth:080	Insert OFF	Line no.: 001 Column no.:001
Y N N N N Y Y N N N N N N N N N N N N N N N N N	Printout header Date: ######## Static text 16-bit fixad point no.	Page:#### Tim: #######	
F 1 F 2 DELETE BLOCK	F 3 F 4 SEARCH K REPLACE CURSOR	F5F6 INSERT ONENTER	F 7 F 8 EXIT

Store the static part of the frame **on** the **PG** 685 RAM with F6 (ENTER) and exit the 'STATIC PART' mask with F8 (EXIT).

Press F2 (DYN. PART) in the 'FRAME' **mask** to call the 'DYN. PART' mask. The dynamic variables are programmed in just the same way as those for the **PSL** (see Section 6.1).

			DRIVE:	B PR	OGRAM:	PT88AI	BSP COMP	PONENT: PI
Attribu BECU	tes Gr El I A No . Pa	ement name: age wi dth:	HEADER 80				Line no.: Column no	002 D. : 037
Y N N N Y Y N Y N N N N	N N Pr N	rintout head	ler	 Pa	ige:###	 ¢		
N N N N N N N N N N Y Y N N N N	N 00 N S ⁻ N 10	ate: #### tatic text 6-bit fixed	### point ro. :	Time: #####	####	±###		
Type of	°dynamic	vari abl e: P	AGE NUMBER no. of	char.	4	Att: bol c expa unde acou	ibutes(Y/ I print andad pri erlined ustic sign	Y N): N N N al N
F 1	F 2	F 3	F4	F 5	F	6	F 7	F 8

Press F7 (HELP) to enter the variable type, e.g. "PAGE NUMBER". The CP counts the pages printed out up to page 9999. It then returns automatically to page 1. Using the PC job RESET PAGE NUMBER you can reset his counter to 1 at any time from the CPU.

ThedateandtimearetobetakenfromtheCP525hardware clock andprintedoutinboldprint.

Type of dynamic variable:	DATE		Attributes(Y/N):	
DATE is fetched from	CP 525 -	CLOCK	bold print Y	
			expanded print N	
	order	DMY	underlined N	
			acoustic signal N	

Store the specification on the PG 685 RAM with F6 (ENTER) and with F4 (RIGHT) select the next variable.

Type of dynamic variable: TIME is fetched from TIME CP 525 - **Clock**

order

Attributes(Y/N): bold print Y expanded print N underlined N acoustic signal N

Store the specification on the PG 685 RAM with F6 (ENTER) and with F2 (DOWN) select the next variable.

ΗMS

As specified in the 'FRAME' **mask**, process variables are fetched from CPU 1 data block **DB 30. The** number of reserve characters for dynamic **variables (#)** is taken as the default for the **number of** places.

Type of dynamic variables:	PROCESS VARI ABLE	Attributes(Y/N):
		bold print N
CPU No. 1	Format 16BIT FXP	expanded print N
DB No. 30	no. of char. 6.0	underlined N
Address DW 9.		acoustic signal N

Store the specification on the PG 685 RAM with F6 (ENTER) and exit the mask with F8 (EXIT). Save the static and dynamic part on the hard disk by pressing F6 (SAVE) in the 'FRAME' mask.

Program a purely static frame "TRAILER". Enter the frame name "TRAILER" in the 'FRAME' mask. The CPU number and data block number are not necessary since there are no dynamic fields in this frame.

The frame is to be printed out as follows:

,

Static text

The broken lines are to be printed out in bold print.

Ress F6 to save the static part, exit the 'STATIC PART' mask with F8 (EXIT) and save the frame on the hard disk with F6 $({\tt SAVE})$.

Call the PC job CML and enter "HEADER" as the list header and "TRAILER" as the list trailer in the 'CURRENT MESSAGE **LIST'** mask (see Section 8.1). Save the PC job CUR. **MESSAGE** and transfer **it** to

B8576542-03

the user memory of the CP 525. Overwrite the CML already in the CP 525 user memory, Fl (YES, in response to the prompt 'ACK. Olo: element already exits! - overwrite?').

Transfer both frames to the CP 525 user memory and perform a cold restart on the CP 525 with F1 in the `TRANSFER' mask.

9.2 Test

The test is performed in the same way as it is for the CML (see Section 8.3). The printout should be as below:

<u>Printout</u> be	ader		Page:	Z	
Date : 1 2.3 I. 87	T i me:	11 :59 : 30 am			
static_text 1. 6-b i. t f i xed po i ntno.:12345					
??.?? ??:??:?? MESSAGE ??. ?? ??:??:? MESSAGE	100 stat_ 101 b i nar	i_c_text_and acou v valu e :	sti <u>c</u> s_i_anal	 ↓ 7 ?	
??. ?? ??:??:?? MESSAGE ??. '????:?? MESSAGE ??. ????:?? MESSAGE	102 char = 103 byte 104 16-1	.cter vapiable: pit BCD number:	<u>5555</u> <u>555</u> <u>5</u>	; ; ?	
??.???????? MESSAGE ??.???????? MESSAGE	105 16-bit 106 `32-bi	fixed pk. no. : it BCD number:	<u> </u>	?	
??.?? ??.?? MESSAGE 105 flating point numb r????????????????????????????????????					
??.?? ??:??:?? MESSAGE	110 count	er value	???	?	
Static text					

?

90

مميايات رميوا والجام بالاليان

,

10 Date/Time PC Jobs

The **CP 525** has an internal hardware clock that is used as the calendar and clock for lists. This clock can be set, read and synchronized with the clocks of other CPs in the same PC.

The PC job number 218 on the CP 525 is reserved for reading and writing the date and time. These special jobs are carried out immediately; i.e. they are not entered in the ten-job queue of the CP 525. No jobs need to be programmed on the CP 525.

10.1 Setting the Date/Time

Enter the value 31st December 1987, 11 p.m., (23 hours) 58 minutes and 30 seconds in BCD format in data block DB 30 from data word DW 20. The CP 525 is to remain the master clock. For this reason, the master identifier (bit 8) must be set in data word DW 20.

20	:	KH= 0100	master identifier set
21	:	KH = 0030	1/10 s, 1/100 s / seconds
22	:	KH 5823	minutes / hours
23	:	KH= 3112	day / month
24	:	KH= 8700	year /

The set function is to be triggered when the signal at input I 0.6 changes from 0 to 1. The termination of the job is to be indicated at output Q 1.0.
Add the statements for triggering the PC job to **OB** 1 as follows:

	:AN	10.6	EDGE EVALUATION:
	R	F 2.6	THE RESULT OF LOGIC OPERATION
	:A	I 0.6	(RLO) IS SET FOR ONE CYCLE IF THE
	: AN	F2.6	SIGNAL AT INPUT I 0.6 CHANGES
	:S	F 2.6	FROM o to 1.
	:		
	:JU	FB120	TRIGGER SET DATE/TIME
NAME	E:SEN	D	
SSNR	:	KYO,o	INTERFACE NUMBER O
A-NR	:	KY0,218	JOB NUMBER 218
ANZW	:	FW29	CONDITION CODEWORD
QTYP	:	KCDB	SOURCE m data block
DBNR	:	KYO, 30	SOURCE DATA BLOCK DB 30
QANF		KF+20	SOURCE START DATA WORD 20
QLAE	:	KF+5	SOURCE length 5 data words
PAFE	:	FY33	PARAMETER ASSIGNMENT ERROR BYTE
	:0	F 30.3	JOB FINISHED WITH ERROR
	:0	F 33.0	OR PARAMETER ASSIGNMENT ERROR
	:= Q	1. O	INDICATE AT OUTPUT Q 1.0

Test

Switch the digital input I 0.6 from 0 to 1. The date and time of the CP 525 hardware clock are set to the value in data block DB 30. Check that the setting of the date/time was successful simply by reading the date and time on the message printout. After one and a half minutes the CP 525 indicates that the year is changed.

The CP 525 recognizes leap years and when necessary automatically counts 29 days in February.

10.2 Read Date/Time

The current time is to be transferred cyclically from the CP 525 hardware clock to data block DB 30 from data word DW 25. It is stored in the same format as the values programmedstartingat DW 20 (see Section 10.1).

The transfer is to take place in every cycle. If an error occurs when the job is being processed it is indicated at output Q 1.1. Add the following statements for triggering the PC job to organization block OB 1:

	:JU	FB121	TRIGGER READ DATE/TIME
NAME	:REC	EIVE	
SSNR	:	KYO, 0	INTERFACE NUMBER 0
A-NR	:	KY0,218	JOB NUMBER 218
ANZW	:	FW34	CONDITION CODEWORD
ZTYP	:	KCDB	DEST TYPE DATA BLOCK
DBNR	:	КҮ0,30	DEST DATA BLOCK DB 30
ZANF	:	KF+25	DEST START DATA WORD 25
ZLAE	:	KF+5	DEST LENGTH 5 DATA WORDS
PAFE	:	FY38	PARAMETER ASSIGNMENT ERROR BYTE
	:0	F 35.3	JOB FINISHED WITH ERROR
	:0	F 38.0	OR PARAMETER ASSIGNMENT ERROR
	:=	Q 1.1	INDICATE AT OUTPUT Q 1.1

The time **canbe** read only if the master identifier is set on the **CP** 525. After the **CP** 525 module has started up following power **up, the master** identifier is reset. In the **example** of **message** listing (see Sections 7.2 and 7.3), the rester identifier is set during the start-up. You established **in** the data **block DB PMC** DB 6 in data **word DW** 6 that the **CP** 525 is the master clock.

Test

Observe data block DB 30 from **DW** 25 to **DW** 29 with the **PG** 685 online **function** CONTROL **VARIABLE.** Tenths of **seconds**, seconds and minutes must change constantly.

1 · · · ·

10.3 Synchronizing Several CP Hardware Clocks

To **synchronize** the hardware clocks of several **CP** modules, the date and time must be transferred from a **CP** that is declared as time master to the other **communications** processors.

The time is read by a RECEIVE DIRECT 218 call or by a corresponding **GET** call (see Section 7.2. 5). 'l'he **time** is set by a SEND DIRECT 218 call or a **PUT** call (see Section 7.2. 5).

A **CP** is declared time master **with** a SEND DIRE(7I 218 if bit 8 is set in the **first** data vord transferred. If this bit is 0 the **CP** is declared time slave.

If only one data word is transferred with the SEND DIRECT 218 (**QIAE** = 1), only the **master** identifier is affected and **the** time is not set.

If five data words are transferred with the **SEND DIRECT** 218 (QLAE = 5), the master identifier, date and time are transferred. The hardware clock is set accordingly.

11 Reading the Error Message Area in the SYSTAT

The SYSTAT is a memory area on the CP 525 that contains info-tion separately for each interface.

Within the error message area of the **SYSTAT** the causes of errors are coded **in** detail. The **error** message area is 4 bytes = 2 words long and appears as below:



E - set, if error is entered in the SYSTAT

- **O** set, if there is an error overflow (more than **3** errors)
- **B** set, if there is a BREAK on the interface
- R set, if the printer is not ready to receive for longer than one minute (XOFF status)

The PC job RECEIVE DIRECT 200 is reserved for reading the error message area. This job is performed immediately, i.e. it **is** not entered in the 10-job queue of the **CP** 525, No job needs **programming on** the **CP** 525.

Transfer the error information in data block **DB** 30 starting at data word **DW** 30. The job is called unconditionally in every program **cycle** but is executed only if an error occurs.

When RESET DIRECT 200 is called, the error message area of the interface addressed in the SYSTAT (in our example SSNR = 0) is reset. The data words DW 30 and DW 31 in data block DB 30, into which the error messages are written by RECEIVE DIRECT 200, must be reset separately.

Reset the error message area with a positive going edge at digital input I 0.7.

Add the following statements to **organization** block **OB** 1.

:0 F 3.0	
:ON F 3.0	GENERATE RLO = 1
:	
:JU FB121	TRIGGER READ DATE/TIME
NAME :RECEIVE	
SSNR : KYO.O	INTERFACE NUMBER 0
A-NR : KY0.200	JOB NUMBER 200
ANZW : FW39	CONDITION CODEWORD
ZTYP: KCDB	DEST TYPE DATA BLOCK
DRNR : KYO 30	DEST DATA BLOCK DB 30
$7ANE \cdot KE+30$	DESTSTART DATAWORD30
$7IAF \cdot KF+2$	DEST LENGTH 2 DATA WORDS
	DAMMETTER ASSIGNMENT FRROR RYTE
PAPE · PIHS	IAMETER ADDIGNMENT ERROR DITE
:0 F40.3	JOB FINISHED WITH ERROR
:0 F 43.0	OR PAMMETER ASSIGNMENT ERROR
:= 0 1.2	INDICATE AT OUTPUT 0 1.2
:AN 10.7	EDGE EVALUATION
R F2.7	THE RESULT OF LOGIC OPERATION
A 10.7	(RLO) IS SET FOR ONE CYCLE.
:AN F2 7	TF THE STGNAL AT TNPUT I 0.7
·S F2 7	CHANGES FROM O TO 1
NAME OF COOMAG	REDEI SISIAI OF INIERFACE U
NAME :K-SISTAT	

Function block FB 100

NAME :R-SYSTAT

NAME	:JU :RES	FB124 ST	RESET SYSTAT
SSNR A-NR PAFE	: : :	KYO,0 KYO,200 FY198	INTERFACE NUMBER O JOB NUMBER 200 PARAMETER ASSIGNMENT ERROR BYTE
	:C :L :T :T :BE	DB30 KF+O DW30 DW31	IN DATA BLOCK DB 30 DELETE DATA WORDS DW 30 AND DW 31

Transfer organization block OB 1 and function block FB 100 to the user memory of the R processor. Carry out a cold restart. The green RUN LED on the R processor must light up.

Test

Observe data block DB 30, DW 30 and DW 31 with the PG 685 online function CONTROL VARIABLE and then unplug the printer cable at the CP 525.

KH= OAFF appears immediately in data word 30; i.e. the bits 'error' and 'BREAK' are set and the error FF'BREAK on the line' is displayed.

As soon as you trigger messages you obtain KH= OEFF in data word DW 30 and KH= FFFF in DW 31; further FF errors are entered and for this reason the 'overflow' bit is set in the first byte.

Plug in the printer cable again. The 'break bit in DL 30 is reset immediately. Reset the error message area of interface 1 in the SYSTAT, by switching input I 0.7 from zero to one.



12 Information

The COM 525 programming software displays information on the PC monitor about user data that is already programmed.

Press F4 in the 'SELECTION' mask to call the 'INFO' mask. In this mask you can decide whether you wish to have information about user data on the CP or from a program file on FD.

Press F1 (CP) to display the data programmed on the CP for interface 1.

-> PROG. SELECTION -> SELECTION -> I N F O					5 / COM525
		STORAGE MEDIUM: Interface: Component :	CP 1 PT		
		Plant designation: Generated by: Generated on:	applicatim example Wally 28.10.87		
F 1 Brief Desc.	F 2 Message	F3 F4 PCJOB FRAME	F 5 F 6	F 7 HELP	F 8 EXIT

98

وجوريا والأروا والمروا والو

.

.8,

Press Fl to display a submask with a brief description of interface 1 on the CP 525.

-> PROG. SELECTION INFO	-> SELECTION ->		SIMATIC S	5 / COM525
	STORAGE MEDIUM: Interface Component	CP 1 PT		
	Plant desi <u>q</u> nation: Generated by: Generated on:	application example Wally 28.10.87		
Data type Ele	ment name Number	Data type	Number	
Interpreter: Procedure: LAU Print para.: PRI	PT88 01 1 PT88 01 1 NT-PARA 1	Message: PC job: Franc:	11 4 2	
Total number Program length	20 elements 9800 words			
F1 F2 BRIEF DESC. MESSAGE	F3 F 4 PC J08 FRAME	F5 F6	F 7 HELP	F 8 EXI T

You can see the interpreter and procedure name in this mask and you can obtain a list of the existing elements for all types of user data. In addition, you see the total number of elements and the length of your program.

Call up detailed information about allthedatatypes.

B8576542-03

Press F3 in the **'INFO'** mask to display the names and job numbers for all PC jobs and the memory space **required** for each job **in** words.

-> PRDG. SELECTION -> SELI P C J O B	ection -> INFO ->	SIMATIC S5 / COM525
STORAGE MEDIUM: CP 525	INTERFACE: 1 PR	OGRAM NAME:
		Page 01 of 01
Length Element name A-Nr(words)	Length Element name A-Nr (word	Length s) Element name A-Nr (words)
PSL EXAMPLE 002 00299 New Page 001 00006	SEQ.MESSAGE 0 0 3 00028	CUR.MESSAGE 004 00018
F1 F2 F3 ON PRINTER	F 4 F 5 Page Backwards Fo	F6 F7 F8 PAGE RWARDS EXIT

 $\ensuremath{\texttt{Press}}\xspace{\texttt{F1}}$ (ON PRINTER) to have this information printed out on a $\ensuremath{\texttt{PT88}}\xspace{\texttt{Printer}}\xspace{\texttt{commetted}}\xspace{\texttt{to the PG}}.$

13 Program Documentation

AU the elements of the COM 525 program file can be documented in detail. The printouts are output with a printout header and, if programmed, with a printout trailer at the bottom of the page. To achieve this, the settings of the mode switches on the interface adaptor of your printer must be the same as the settings **ON** your **PG** (see **PG manual**). The settings on the coding switch of the central **controller** on the printer can be **taken** from **Section** 3.1

F6 (CONTINUE) and F3 (ASSIGN $\ensuremath{\text{PRI}}$, PARA.) in the 'SELECTION' mask call the mask below:

-> PROG. SELECT ION -> A S S I G N P R I. P	SELECTION -> A R A.		SIMATIC s5 / COM52
Printout header:			
SIEMENSSIMATIC S5 COM525 - CP525/524	Application e S5-D	xample COM 525 - PT 88 I OS	8 Page:
Drive: B Progr Plant: application	am PT 88AB SP example	Last worked with: Generated by: Wa	28. 10. 87 Il ly
Printout trai ler:			
You can write any te	ext here.		
F1 F2	F 3 F 4	F5 F6 SAVE	F 7 F 8 Exit

You can program two lines **of** text for the header and trailer. F6 saves the **header** and trailer in your user program file **PT88ABSP.** 525.

5

Section 2

Press F6 (CONTINUE) and F4 (LIST) in the 'SELECTION' mask to call the 'LISTING' mask. The softkeys

- F1 (TOTAL PROGRAM)
- F2 (INTERPRT / PROCEDURE)
- F3 (MESSAGES)
- F4 (PC JOB)
- F5 (FRAME)
- F6 (SUMMARY LISTING)

can be used to list the whole program or parts of it.

Try out the various options. As an example the 'FRAME mask is shown below:

BASIC MASK -> PR(F R A M E	SIMATIC S5	/ COM525		
	DRIVE: B Program name:Pt88A Component: P T	BSP		
	Plant designation: Generated by: Generated on:	application example Wally 28.10.87		
	Name : HEADER			
F 1 F 2 SINGLE LISTING	F 3 F 4 Total Listing	F 5 F 6	F 7 HELP	F 8 EXIT

With F3 (TOTAL LISTING) all the frames sre listed on the printer. To list the single frame HEADER, enter the name <code>'HEADER'</code> in the mask with F7 (HELP) and press F1 (SINGLE LISTING).

The following listing will be printed out on the printer connected to the \mathbf{PG} 685:

. Drive : 3	Program:	PTABASSP	Last wo:	wed with:	:0.30.8
, Plant: appl:	catim '2, ampl	•	Generat	ed by: Wa	11Y
F RAMES 2211111221111					
-		Namo	HEADER		
		CPU no.:	1		
		DÈ no.:	030		
Line: Ci		Group no.:	- .	Attributes:	: e ·
Dynamı c fiold	sinone				
line: 02		Group no.:	-	Attributes:	3 E ·
Printout heade	iد.	Page: ###			
Dynamıc fieldo	.: 01				
Field Variab	letype Sou	rce Address	For mai Order	t/ No of charg.	Attributes
Di Page nu	mber CP			63	3 E
Line: 03		Grou o no.	:- /	Attributes:	
Dynamic fields	t none				
Lina: 04		Grou p.no.	:- <i>'</i>	Attr ibu ter	5:
Date: #######	# Time	: ********			
Dynamic fields	. 02				
Field Variabl	e type Sou	rce Address	Format/ O rder	No. of chare.	Attributes
Ci Date	CP	clk -	DMY	08	B
UZ IIMO	CP	CTX -	HES .	08	в

Dynamic fields:none

You can write any tout here

SIEMENS SIMATIC 53 A	policatication ana 55-3	mole COM 525 -FTS8 Page: 2 , DOS LO. 30. a:,
3.,0 : 5 Prog . Plant: application a::	ram:PT88ABS P amolo	Lastworked with: 10.30. &?, Generated by, Wally
Line: 05	Graup na.:	- Attributos: CU -
Static test		
Dynamic fields: none		
Line: 07	Group no.:	- Attributes:
16-bit fired point no.:	*****	
Dynamicf t-la=: 01		
Field Variable type	Source Address	Format/ No. of Attributes Order chars.
01 Processivar .	DB DW 007	15-91 T FXP 06
Line : 08	Group no. :	- Attributes: B

Dynamic fields: none

Vou can write any text here

We have checked the contents of this manual for agreement with the hardware and software described. Since deviations cannot be precluded entirely, we can. not guarantee lull agreement. However, the data in this manual are reviewed regularly and any necessary corrections included in subsequent editions. Suggestions for improvement are welcomed. Technical data subject to change.

The reproduction. transmission or use of this document Or its contents is not permitted without express written authority.

Of tenders will be **liable for** damages. All **rights**, in. eluding rights **created** by **patent grant** or registration of a utility model or design, are reserved.

Copyright [©] Siemens AG 1989 All Rights Reserved

Siemens Aktiengesellschaft

7.

,

C79000-B8576-C542-05 Gerätewerk Karlsruhe Printed in the Federal Republic of Germany

SIEMENS

SIMATIC S5

Notes on the Operating Systems PCP/M-86 and **S5-DOS**

User's Guide

Order No. C79000-B8576-C543-03



** *

CONTENTS

-

1	What Does COM 525 Include?	1
1.1 1.2	COM 525 Parts Distribution on Floppy Disks	1 2
2	Making Back-up Copies	4
3	COM 525 Files	5
4	Working with Files under PCP/M-86	6
4.1 4.2 4.3	Directory (DIR) Erase (ERA) Copy (PIP)	6 7 7
5	Loading COM 525 with the PG 635 and PG 675	9
5.1 5.2	Loading PCP/M-86 Calling COM 525	9 9
б	Loading COM 525 with the PG 685 and PG 695	11
6.1 6.2 6.3 6.4 6.5	Loading PCP/M-86 Using S5-DOS for the First Time Setting up a Working Area Calling COM 525 COM 525 cm Several User Areas	11 11 12 12 13

i

1 What Does COM 525 Include?

1.1 COM 525 **Parts**

COM 525 includes the main COM program, five overlays, PROM 525 snd a standard library with interpreters and procedures. The following table shows the names of the overlays and the functions they contain.

Filename	Functions
S5PEC05X.CMD	Main program
S50ECP5X.CMD	Listing (of programs)
S50ECR5X.CMD	Computer link
S5OECD5X.CMD	Printer
S50ECM5X.CMD	Message editor
S50ECZ5X.CMD	List editor
S5PEPO5X.CMD	EPROM programmer PROM 525
COMLIBn. 525	Standard library (interpreters, procedures)

1

.

1.2 Distribution on Floppy Disks

The distribution of $COM\ 525\ on$ floppy disks depends on the programmer.

There are three floppy disks tf you are using any of the programmers **PG** 675, **PG** 685 and **PG** 695. The COM 525 files are distributed on these disks as follows:

Disk 1 (1/3)

- S50ECI	25X.CMD	overlay	' 'listing'	,	
- S50ECR	5X.CMD	overlay	*comput	<i>er</i> link	c'
- S5PEP()5X.CMD	EPROM p	rogramer	PROM	525
- S5PECO)5X.CMD	COM 525	main prog	gram	

Disk 2 (2/3)

-	S50ECD5X.CMD	overlay	`printer?
-	S50ECM5X.CMD	overlay	'message editor'
-	S50EC25X.CMD	overlay	`list editor'

Disk 3 (3/3)

-	COMLIBn. 525	standard	library
	(n = release)		

For programer **PG** 635, **JOM** 525 is distributed cm **two** disks as **follows:**

Disk 1 (1/2)

- SSOFC25X CMD overlay list editor!

Disk 2 (2/2)

....

- **COMLIBn.525** standard library (n= release)

Ļ

2 Making Back-up Copies

When you are using a programmer without a hard disk drive (PG 635 and **PG** 675) you should make **back-up** copies of the original diskettes as soon as you receive them. To do this, you must first format an empty floppy disk by calling the **DSKMAINT** program.

Press function key F7 to select the **formatting** function; with **F1** or F3 you can select the drive in which the floppy disk to be formatted has been inserted. You are prompted to **confirm** your **intention**.

The copying function is called with function key F3. Again F1 and F3 are used to specify the source and destination drive. When you are sure that you are going to copy in the correct direction, confirm your intention.

3 COM 525 Files

COM 525 initializes one file for each generated program. The name of this file is the program name you selected and the file is automatically given the file type ' .525'.

Example: program name: TURBINE file name: TURBINE.525

Files with the name COMLIB??.525 are recognized as libraries by COM 525. "?" stands for any alphanumeric character. The term "library" is explained in the section "Libraries" in the COM 525 instructions in this manual.

.....

والمحافظ والمراجع والمترج والمراجع والمراجع

•

4 Working with Files under PCP/M-86

The following sections briefly explain the PCP/M-86 functions, directory, erase and copy. These functions are also made **available** by COM 525; however, they are then restricted to COM 525 programs. To use these PCP/M-86 functions fully you should have some experience of this operating system.

4.1 Directory (DIR)

Using the DIR function, you can have all files orgroupsoffiles listed that have the DIR attribute. When specifying the files or groups of files, Y^{OU} can replace a single unknown character by ? and a group of unknown characters by * (? and * are known as wild cards).

Syntax:	DIR	d:filename[Options]	d:	drive

Example: DIR C: DIR *. 525	all files on drive C: all COM 525 files on the current drive
DIR S5?EC?5X.CHD	COM 525 main program and all overlays

Files with the **SYS** attribute can be listed if you use the function **DIRSYS**. You can also use wild cards with this function.

Example:	DIRSYS *.CMD	all <i>system</i> CMD	files	of	the	file type
	DIRSYS	all system drive	files	on	the	current

The permissible **options** and further examples can be found in the **PCP/M-86** description.

4.2 Erase (ERA)

The ERA function allows you to erase single file: or groups of files. You can also use wild cards with this function.

syntax:	ERA d:filename [opt	cions] d: drive
Example:	ERA FLANT.525 ERA B:*.525	deletes the COM 525 file PLANT deletes all COM 525 files on drive B:
	~*.*	after an acknowledgement is prompted all files on the current drive and current user area are erased
	ERA *.*[CONFIRM]	this option means that every erase job is comfirmed

Further options can be found in the PCP/M-86 description.

4.3 **Copy** (PIP)

The PIP function allows **you** to copy single files or groups of files. You also possible to rename individual files when they are copied.

The following options are available:

• Single file without renaming

PIP	d:	=	s:sourcefilename	d:	dest.	drive
				s:	source	drive

7

• Single file with renaming

PIP d:destfilename = s:sourcefilename

• Single file with renaming and specification of the USER area

PIP d:destfilename[Gn] = s:sourcefielname[Gn]

n: number of the user area of the destination or source

•Single system file with renaming

```
PIP d:destfile = s:sourcefile[R]
```

• Group of files (the **group** is selected using wild cards)

PIP d: = S: filegroup

• Group of files with **system files** and specification of the user areas

PIP d:[Gn] = s:filegroup[Gn R]

For further **information**, particularly about other options, refer to the **PCP/M-86** description.

5 Loading COM 525 with the PG 635 and PG 675

5.1 Loading PCP/M-86

After powering up, the programmer requests the **operating** system disk to be inserted in drive A:.

If **PCP/M-86 has** never been loaded in your **PG**, insert the **PG-TEST** disk in drive **A**:. You are, among **other** things, prompted to specify the memory capacity of your programmer by entering plus and minus characters. When this is completed, insert the operating system disk in drive A: and then **switch** the device off-and on. The operating system is then loaded.

5.2 Calling COM 525

Wait **until** the operating system has been completely loaded. Then **remove** the **PCP/M-86** disk and insert **the** following disks:

with **PG** 635

in drive A: disk 2/2 of the **S5-DOS** in drive B: disk 1/2 of **COM** 525

with **PG** 675

in drive A: disk 2/3 of S5-DOS in drive B: disk 1/3 of COM 525.

Enter

S5

to load the S5-DOS operating system. The S5-DOS command interpreter (S5-KOMI) then becomes active. S5-KOMI allows you to select the available packages. You can now call COM 525 by positioning the cursor in front of the COM 525 package and pressing the function key F1 (PACKAGE). If you wish to program an EPROM, position the cursor in front of the PROM 525 package and press F1.

Operating Systems PC. $^{\prime}/M-86$ and S5-DOS

بالوافح بن مام والواص

Once COM 525 has been called, you can replace the $\rm S5\text{-}DOS$ disk in driveA: with the disk ${}\cdot m$ which to generate your COM 525 programs.

6 Loading COM 525 with the PG 685 and PG 695

6.1 Loading PCP/M-86

Before you load COM, the PCP/M-86 and S5-DOS operating systems on the hard disk should be in user area O. The files for this operating system must have the attribute SYS (system). Check this with the **comman**d

d: hard disk drive
(with PG 685 always
B:; with PG 695,
usually C:)

After powering up, ignore the PG request to insert the operating system disk in drive A: and wait approximately 10 seconds. After this time the PG switches over to the hard disk drive and loads the operating system.

If PCP/M-86 has never been loaded on your PG, then follow the instructions in Section 5.1. After switching on the programmer, transfer the operating system to user area 0 on your hard disk. Enter the command:

PIP d:[G0] = A:*.*[G0 R V] d: hard disk drive

6.2 Using S5-DOS for the First Time

If the S5-DOS operating system is not yet cm user area 0 of your hard disk, transfer it to this area. Insert disk 2/2 of S5-DOS in drive A: and enter the following command:

PIP d: [GO] = A:*.*[GO R V] d: hard disk drive

The **files located on** disk 1/2 of **S5-DOS** are not required for working with **COM 525.** A **detailed** description can be found in the instructions for **S5-DOS**.

6.3 Setting up a Working Area

To generate programs with COM 525, select an area other than user area 0 (if possible select a user area that is empty or only contains COM 525 programs). Insert disk 1/3 in drive A: and copy the contents to the user area you selected on your hard disk.

PIP d: $[Gn] = A:*.*[GO \mathbb{R} \mathbb{V}]$	d:	hard disk drive
	n:	selected user area

Copy the contents of disks 2/3 and 3/3 in the same way.

6.4 Calling COM 525

Enter

d:

d: hard disk drive

to select the hard disk drive and follow this with

USER n	n:	user	area	on	which	COM	525	is
		locat						

to set the required user area. Load the operating system S5-DOS, by entering

S5

ю.

,

The S5-DOS command interpreter (S5-KOMI) is now called and you can select one of the packages available on this user area. You select a package by placing the cursor in front of the required package and pressing function key F1 (PACKAGE).

6.5 COM 525 on Several User Areas

If, to simplify matters, you want to work with COM 525 on several user areas, **a** different procedure is recommended.

In this case, copy disks 1/3, 2/3 and 3/3 of COM 525 to user area \boldsymbol{o} with

PIP d:[G0] = A:*.*[G0 R V] d: hard disk drive

and assign the system attribute to the files with

SET d:S5?EC?5X.CHD[SYS RO] SET d:S5PEP05X.CHD[SYS RO] SEX d:COHLID*.525[SYS RO]

d: hard disk drive

You can now work with **COM** 525 on every user area. **Area** O should, however, be reserved for system files.

SIEMENS

.

....

We have checked the contents of this manual for agreement with the hardware and software described. Since deviations cannot be precluded entirely, we cannot guarantee full agreement. However, the data in this manual are reviewed regularly and any necessary corrections included in subsequent editions. Suggestions for improvement are welcomed. Technical data subject to change. The reproduction, transmission or use of this documentor its contents is not permitted without express written authority.

Offenders will be liable for damages All rights. including rights created by patent grant or registration of a utility model or design, are reserved.

Copyright Siemens AC 1989 All Rights Reserved

Siemens Aktiengesellschaft

C79000-88576-C543-03 Gerätewerk Kartsruhe Printed in the Federal Republic of Germany

SIEMENS

SIMATIC S5

Programming Package COM 525

User's Guide

* × •

С79000-В6576-С544-07

معرورية الأرامة فالحصر والمعاق

1	The Programming Package COM 525	1
1.1 1.2 1.3 1.4 1.5 1.6 1.7 1,8	Introduction The Layout of the Masks with COM 525 Overview of the Masks File Organization Structure of a COM 525 Program Libraries Handling User Data Standard Outputs when Working with COM 525	1 4 8 12 13 14 17 18
2	Selecting the User Program	20
2.1 2.2 2.3 2.4 2.5	General Information Basic Mask SYSID Info program Selection Selection	20 22 23 25 27
3	Functions in the Main COM Section	30
3.1 3.2 3.2.1 3.2.2 3.2.3 3.3 3.3.1 3.3.2 3.4 3.4,1 3.4.2 3.4.3 3.4.4 3.4.4	General Information Transfer Transferring from CP 525 to FD Transferring from FD to CP 525 Transferring from FD to FD Delete Delete CP 525 Deleting on FD Info Brief Description Information about Messages Information about PC Jobs Information about Frames Information about a Job Block	30 31 33 37 39 41 43 47 48 51 53 54 54 54 54
3,4.6	Information about Interpreters/Procedures	55

Page

١

~

۰

Contents

i

Terms and Definitions

6

159

ii.

.....

1 The Programming Package COM 525

1.1 Introduction

The software package COM 525 supports the programming of the CP 525 and CP 524 communications processors,

When connected to the **PT88/PT89** printer, the **CP** 525 can list process statuses and messages originating in the process.

The messages acquired by the CPU of the **programmable** controller are transferred to the **CP** 525 in a data block **(DB)**; the **CP** 525 evaluates the DB and initiates the printing out of the messages on **the** printer.

The message text includes static texts and dynamic parts such as the date, time, message status (coming, going, acknowledged) and a process variable,

To highlight parts of **the** message texts, you can select attributes such as bold **print**, expanded print, compressed print or underlining. The messages can also be accompanied by an acoustic signal **on the printer**.

The static texts programmed with COM 525 are stored in the user memory of the $CP \; 525.$

List printout:

LIST HEADER				
Message te	xt 1			
Message te	xt 2			
Message te	xt 3			
Message text	n			
	LIST TRAILER			



Solomerane and the second second

Messages can be grouped together; for example, you could group together messages for a particular part of the plant. Each of the maximum 16 message groups can be inhibited or scanned.

All functions to be executed by the CP 525 (e.g. output of messages on the printer) can be programmed when you complete interactive masks on the PG 685 (also the PG 635 or a PG 7xx).

More information on the types of lists and functions can be found in the user's guide "Event output end listing with the PT88/PT89 printer" in this manual.

The computer link with RK 512 allows the data exchange between two programmable controllers or between a programmable controller and a central process computer (see user's guide "Computer link with RK 512" in this manual). Using the COM 525 software package you can generate the required CP 525 user program.

If you are not familiar with the terms or abbreviations used, refer to Section 6, "Terms and definitions" in this user guide.

The operating system expansion S5-DOS allows COM 525 to be read on several programmers without being adapted. Before using COM 525 you should get used to working with the S5-DOS command interpreter (KOMI). You require this function to be able to work with a STEP 5 package, Read through the section "Getting started" in the manual for your PO.

"÷.

.

If you have not worked with PCP/M, CP/M or a similar operating system, refer to the description of the operating system to get at least a working knowledge of the following utility programs:

- DSKMAINT (formatting ind copying disks)
- (copying single files) - PIP
- DIR (displaying the file directory)
- SET (setting file attributes)
- **erase** USER (erasing files)
- (changing the user area number)

1.2 The Layout of the Masks with COM 525

To work with COM 525 you do not need programming experience. All inputs are prompted by COM 525 in plain language in the masks; if you make errors a text is displayed to indicate the **nature** of the errors. The 'PROGRAM JOB' mask, for example, appears as below:

-> SELECTION -> JOB BLOO PROGRAM JOB	CK•>				SI	MATIC S5 / (COM525
		DRI VE:	\$	PROGRAM:	\$\$\$\$\$\$\$	COMPONENT:	SS
		J 0 E	3				
Job no.:	\$ss						
Job:	## <u>#</u> ##	¥					
Job type	####	;;;; ;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;	\ 				
CPu 170.:	#						
DB 110.:	###						
Dest word address:	#####	D 7	###H				
Ifrequired with coor	rdi na	tion flag	g: ###	.#			
F 1F 2FONPAGEPAPRINTERBACKWARDSFORW	F 3 AGE WARD:	F4	F DEL JC	5 ETE B	F 6 ENTER JOB H	F7 F	8 XI T

AU masks have basically the same layout. In the top left corner the "route" to the currently displayed mask is displayed. In the line below this (in **spiced** letters) is the name of the current mask, for example:

-> SELECTION -> JOB BLOCK -> P R O G R A M J O B

محمولي وملكم العالي الله. الا
You can press the function key F8 (EXIT) to follow the same route back; i.e. you return to the last mask displayed (this is shown farthest right in the first line).

In the right-hand top corner, **you see** the name **of** the programming package

SIMATIC S5 / COM525

In the lower part of the mask there is a **softkey** menu. This corresponds to the function keys F1 to F8 on your keyboard. The **labelling** of the keys on the screen corresponds to the current function of the individual keys, Using the function keys, you can call a follow-on mask **cr** second set of **softkey** functions or you can execute the function currently assigned to the key.

PRINTER BACKWARDS FORWARDS JOB JOB HELP EXIT	F 1	F 2 PAGE	F 3 PAGE	F 4	F 5	F 6	F 7	F 8
	PRINTER	BACKWARDS	FORWARDS		JOB	JOB	HELP	EXIT

The keys F6, F7 and **F8** have special functions that remain the **same in** all the masks as follows:

- F6 allows an alternative set of function keys to displayed if there are more than six functions available in one mask. This is displayed on the softkey by the word 'CONTINUE'. Press this key to obtain the second set of softkey functions. Press it again to return to the first set.
 F6 is also reserved for storing your input; e.g. 'SAVE' or in the example 'ENTER JOB'. If 'ENTER' is displayed, your input in the mask is transferred to the RAM of the programmer; and if 'SAVE' is displayed, to the storage medium (floppy or hard disk).
- F7 (HELP) allows you to make entries in certain input fields; (in the currently displayed mask you can enter either 'SEND' or 'FETCH' in the "Job" field with F7).

5

مانون رواند است. با م م

بقولو والانتانية فالمروا والالا

,

. .

•F8(EXIT) allows you return to the previous ma sk.

Messages, error displays and acknowledgement **prompts** are displayed in the line above the function key menu. If you press key F4 with the mask in the example, the error **message MF.001** 'illegal key' is displayed since this key has no function in this menu. The section "Messages" in this manual provides a list of all possible messages and an explanation of their causes.

The central section of the mask consists of input fields, explanatory texts and output fields.

Output fields contain specifications which **COM** 525 has taken over from previously completed masks. These are always represented as "\$\$\$\$\$\$" in the manual.

The cursor keys and the RETURN key can be used in all the masks for moving the cursor. They have the following significance:

└→	INSERTCHARACTER
Ø	DELETE CHARACTER
₽	(RETURN) completes input and has
	the same effect as FIELD REGISTER
***	RUB-OUT
+	CHARACTERRIGHT
╋	CHARACTER LEFT
Û	FIELDRIGHT
Û	FIELDLEFT
Û	FIELD DOWN
Û	FIELDUP

With all other keys the input depends on which characters are permissible for the particular field (no special characters, only numbers etc.). In file and element names, lower case characters are automatically converted to upper case. The **exact** specifications for each field **can** be found in the description of the particular mask (Sections 2 to 5 in this guide).

1.3 Overview of the Masks

The following overviews show the masks of COM 525 and the route you must follow to reach a particular mask. Please note that not all the functions, but only the COM 525 masks are listed. The route for the components printer (pT), computer link (CL) and for libraries (see 1.6) is identical, i.e. via the basic mask.

0.000



The program branches **tc** different masks from the SELECTION mask depending cm which component, "PT" or "CL" or a library (see Section 1.7) is specified. The shaded fields in the following **overviews** indicate the masks which can be branched to from the **SELECTION** mask.

Component, printer



*From these masks it is possible to branch to the FRAME mask ** From this mask it is possible to branch to the PROCESS SATUS LIST mask





* From these mask it is possible to branch to the MESSAGES, PC JOB and FRAME masks just as from the TRANSFER mask



م میں داخلہ جا جا

*



٠, ۰

1.4 File Organization

Most modern programmers are equipped with hard d .sk drives containing a capacity of 10 to 40 Mbytes, COM 525 uses a file format which is adapted to these capacities.

You can **divide** the hard (or floppy) disk into 16 **"USER** areas" by means of **USER** ? ("?" stands for a number between 0 and 15).

This allows a more orderly storage of files.

. . . .

You should reserve user O for the following:

- the operating system
- the operating system **stility** programs
- the COM and STEP 5 packages

(with the PG 685 the STEP 5 software is already loaded)

you assign system attributes to these files using the **comman**d SET *.*[SYS]. Then you can start all programs from any **USER area**.

When you generate programs with COM 525 the program name you select is automatically given the file type `.5251. This type allows you to **recognize** your COM 525 programs when you list the files using DIR.

and the state of the

1.5 Structure of a COM 525 Program

A program generated with **COM** 525 is structured hierarchically. It includes data types that are then subdivided into elements. A program always contains the following data types: an interpreter, a procedure and depending on the component, either user data (printer) or a job block (computer link).

Printer	Computer link
Interpreter Procedure Printer parameter assignment User data messages PC jobs sequential message list update sequential message list current message list process status list chained list update group inhibit bits new page page number = 1 Frames	Interpreter Procedure Printer parameter assignment Job block

The data type **PRINT_PARA**, i.e. the parameter assignment for the **printer** (specifying the **labelling** of the printout header or trailer for lists **and** listings with **COM** 525) can, but does **not** have **to be part of a program**.

1.6 Libraries

Libraries are files in which parts of programs are collected and from where they can be transferred to the individual programs. They are not assigned to any particular interface.

You should initialize a library to store **e.g** invariable parts of the program, which are required regularly, and which can then be transferred to the individual programs when required. This means that interpreters, procedures, printer parameter assignments, messages, PC jobs (same PC job several times), frames and job blocks can be stored in a library.

You can rename data **types** when they are transferred **from** a program to a library (and vice-versa) ; this does not, however, apply to interpreters and procedures named by **SIEMENS**. These cannot be changed.

The data type, printer parameter assignment, always has the **name PRINT-PARA in** a program. When you transfer it to a library it can, however, be changed. When this data type is transferred back to the program, **COM** 525 automatically changes it back to **PRINT-PARA**.

If you wish to establish a library you must give it the name COMLIB?? ("?" stands for any alphanumeric character). COM 525 then recognizes that this is a library.

For processing libraries, two "fail-safe" procedures have been
included as follows:

- when you specify a library name in the 'PROGRAM SELECTION' mask, COM 525 outputs the prompt "process library?". If you acknowledge positively, COM 525 then makes available masks and functions which are to some extent different from those for programs (see overview of masks at the end of this section. Remember that only the masks and not all the functions are listed.)

a library cannot be 'purged'. Purging is the deleting of parts of a user program which **are** no longer required. From the structure of a program it is possible to determine whether, for **example, frames** are present which are not used with any **list**. Since a library rarely contains a consistent program the whole library would be deleted if it were purged.

You should store libraries on user O and set them to **[SYS]**, to allow them to be accessed **from** every user area.

Also, you should assign the IRO] attribute (read only) to libraries to avoid them being changed accidentally.

The COM 525 programming package already contains a standard library with the name COMLIB02. This contains the interpreters and procedures you require togenerate a program.

*3·4 ··

Overview of masks, libraries



* From these masks it is possible to branch to the same masks as from TRANSFER

v •

1.7 Handling User Data

After you generate user data, you want to be **abl**e to see it and change it if necessary. **COM** 525 makes available the following functions in the **interactive** masks:

- deleting single elements
- transferring elements or whole data types between user programs and/or libraries
- transferring the whole program or parts of it from disk to disk or from disk to **CP** and vice-versa
- information about prcgrams on disk or CP
- listing programs or parts of programs
- purging (logical) unnecessary elements can be deleted
- condensing (physical) necessary because of the file organization and is carried out (particularly in conjunction with the delete function) without the user having to take any action
- programing (if you wish to change user data)

.

The corresponding **mask:** and programming options canbe found in Sections 2 to 5 in this guide.

1.8 Standard Outputs when Working with COM 525

overlays

After you select the function PROGRAM or LISTING in the 'SELEC-TION' mask the message 'Loading overlays!' is displayed. Certain COM 525 overlays must be loaded to allow COM 525 to run on programmers that have a memory capacity of only 384 Kbytes. If you select a function that is one of the overlays and has already been loaded, it will not be loaded again.

COM 525 includes a main program and 5 overlays:

S5PEC05X.CMD	•	•			•	•	main program
S50ECD5X.CMD	•	•	•	•	•		programming frames and
\$50FCM5Y CMD							PC JODS
SSOECTSX CMD	•	•	•	·	·		programing the static parts
byohozyk, ord	•	•	•	•	•		of PSLs and frames
S50ECR5X.CMD	•	•	•	•	•		programming computer link
GEOFORER OF							(JOD block)
SOULCPOX.CMD	•	•	•	·	·		listing functions

Since loading overlays takes time, the functions have been distributed so that the loading of overlays is restricted as **much** as possible. On devices that have no hard disk, you may need to change the floppy disk to load an overlay. If an incorrect floppy disk is inserted or if an incorrect COM 525 version (e.g. German) is on the floppy disk, then an error message is output.

If you intend to use $COM\ 525$ very often, you should use a hard disk $(PG\ 685\ \text{or}\ PG\ 695).$

'ACTIVE' message

Whenever you process **user data on** a floppy disk with **COM** 525 or when data is exchanged in any way with the **CP** 525, the **'ACTIVE' message is displayed.**

Data transferred from and to the CP 525

If the data transfer **berween** the **PG** and **CP** is **not** functioning correctly and one of the actions affecting the **C**? is triggered (TRANSFER, **DELETE** or INFO), an associated error message is displayed.

In this case check the ${\tt :able}$ connection and that the ${\tt CP}$ 525 is switched on.

and the second second

2 Selecting the User Program

The masks described in this section (except **SYSI**) info) lead to the 'SELECTION' mask, From this mask you can **select** all the other functions.

2.1 General Information

File names (called program names in the masks)

File names must correspond to the $\ensuremath{\texttt{PCP/M}}$ conventions for file names.

Lower case letters, the characters < > =, ! * ? / \$ () . : ; and spaces and square brackets must **not** be used in file names.

Letters separated $\boldsymbol{b}\boldsymbol{y}$ a space in a file name are automatically moved together.

Names of data types

Messages, PC jobs, frames etc. are known as data types.

Within this name only single spaces are allowed. If several spaces appear between characters they are automatically reduced to **one** space.

Messages do not use a tame but rather a 4-digit **message** number (including initial **zercs)**.

Note

If only upper case letters are permitted, then lower case letters that are input are automatically converted.

B8576544-03

If there is no other specification, then:

• F7 (HELP) - you call the help mode to allow various entries to be made. In the tables describing the fields in the individual masks, the fields marked with '*' can have an entry made using F7 (HELP)

• F8 (EXIT) - you return to the previous mask. This function key can be used (for example when transferring or deleting several elements) as an abort key.

The '-> acknowledgement' prompt means that the following **acknow**ledgement menu will be displayed:

F 1	F 2	F 3	F 4	F 5	F 6	F 7	F 8	
YES		NO						

When this menu is displayed you must answer a prompt in the message line. Generally the answer 'YES' means-an action is to be executed, while 'NO' simply returns you to the previous menu,

B8576544-03

1 -

COM 525

2.2 Basic Mask

Call the 'BASIC MASK' with F1 (PACKAGE) or the enter key in the S5 command interpreter

COPYRIGH BASIC	HT (C) BY SIE C MASK	EMENS			SIMATIC S5 / COM5
	222222	000000	MM MM	55555555 22222	2 55555555
	сс	00 00	MMM 1444	55 22	22 55
	СC	0 0 00	MM MM MM	55 22	2 55
	СС	00 00	MM MM	5555555 22	5555555
	СС	00 00	MM MM	55 22	5 5
	сс	00 DO	MM MM	55	55
		00000	MM MM	5555555 2 % 2 2 2 2	2 2 555555
	Program Versi on/Iss	ming packa sue: A04	ige for the and the c Ser	communications pro communications proc ial no.: 7994-007	cessor CP 525 :essor CP 524 4-654321

- F1: change to the 'program selection' rode; branch to the 'PROGRAM SELECTION' mask (see 2.4)
- F2: change to the 'SYSID info' mode; read SYSID from the CP 525 (see 2.3); the CP must be connected
- F8: return to the S5 command interpreter

alant, at N W

. А

2.3 SYSID Info

The $\ensuremath{\text{SYSID}}$ contains the total system data required to identify a module and a program.

Call the <code>'SYSID INFO'</code> mask with F2 (SYSTEM DATA) in the <code>'BASIC MASK'</code>.

Accule: 12: Firmware version: 13: Plant: 14: Senerated on: 15: Symbolic address: 16: Ymbolic address: 16: Ymbolic address: 17: Slave no. on PG bus: 18: 'assword: 19: YDM version: 20:	Men. submodule:	11:	
Firmware version: 13: Plant: 14: Senerated 0n: Symbolic address: Symbolic address: Sind. bus Jave no. on PG Dus: 18: Password: 19: XOM version: ZOC: 20:	Module:	12:	
Plant: 14: Generated on: 15: Symbolic address: 16: Ind. bus address: 17: Glave no. on PG bus: 18: Password: 19: XOM version: 20: ROM version: 21:	Firmware version:	13:	
Symbolic address: 16: Symbolic address: 17: Slave no. on PG bus: 18: Password: 19: XOM version: 20:	Plant:	14:	
Ind. bus address: 17: Slave no. on PG bus: 18: Password: 19: XOM version: 20: ROM version: 21:	Symbol ic address:	16:	
Slave no. on PG bus: 18: Password: 19: DM version: 20: ROM version: 21:	Ind. bus address:	17:	
Password: 19: DM version: 20: PROM version: 21:	Slave no. on PG bus:	18:	
VROM version: 20:	Password:	19:	
	DUM version:	20:	
	FROM VEISION.	21:	

F1: SYSIDinfo isoutputagain.

B8576544-03

- Symbolic address
- Ind. bus address
- Slave no. on **PG** bus
- Password
- COM version

If present, the following data are output in the **numbered** fields on the right-hand side of the mask:

Device interface 1

Field U: component Field 12: name of the **interpreter** Field 13: version of the interpreter Field 14: name oftheprocedure Field 15: version of the procedure

Device interface 2

Field 16: component Field 17: name of the interpreter Field 18: version of the interpreter Field 19: name of the procedure Field 20: version of the procedure

....

,

2.4 Program Selection

All the data entered **here** defines a program and is part of the program (except the time).

If the program already exists, the program name can be entered with F7; COM 525 then outputs the component, plant designation, who generated it and when.

Call the 'PROGRAM **SELECTION'** mask with **F1 (SELECT PROGRAM) in** the 'BASIC MASK'.

BASIC MASK -> PROGRAM SE	LECTION	SIMATIC S5 / COM525
	DRIVE: # PROGRAM NAME: ####### COMPONENT: ## Plant designation: ####################################	
	PGdate - time: OMYHM ##.##.## ##:##	
F 1 F 2 SELECTION	F 3 F 4 F 5 F 6	F7F8 , HELPEXIT

If a library name is specified in the PROGRAM NAME field then an '-> acknowledgement prompt' appears.

When generating anewprogramyoumust specify in the COMPONENT field whether a printer program (PT) or a computer link program (CL) is to be generated. If the program already exists the field will be written by COM 525.



F1: branch to the 'SELECTION' mask (see 2.5).

Field		Field type keys allowed	Limit value	Alternatives
DRIVE		upper case letters		A,B,C,D,E,F, dependent on device type *
PROGRAM		file name (see 3.1)		*
COMPONENT		upper case letters		PT,CL
Plant designatic	n	any (must b; entered)		
Generated	by	any (must be entered)		
Generated	on	OUTPUT FIELD		
PG date	D M Y	numbers numbers numbers	1 - 31 1 - 12	
PG time	H M	numbers numbers	0 - 23 0 - 59	

F7 (HELP) can be used to make an entry in the fields marked '*'.

26

معہوری است کے بریاد میں د چ ъ.

In this mask you can select the function with which a program is
to be processed. There are some hierarchical menus (TRANSFER,
DELETE, INFO); i.e. after you press the corresponding function
key the mask remains the same but a second menu is output.

The fields DRIVE, PROGRAM, COMPONENT, plant designation, generated by, generated on are output fields and are taken from the **'PROGRAM SELECTION'** mask.

Call the 'SELECTION' mask with F1 (SELECTION) in the 'PROGRAM SELECTION' mask,

BASIC MASK ->PROG. S E L E C T I O N	SELECTION ->		SIMATIC S5 / COM525
	DRIVE: \$ PROGRAM NAME: \$\$\$ COMPONENT: \$\$	\$\$\$\$\$	
	Plant designation: Generated by: Generated on:	\$\$\$\$\$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$\$\$ \$8588\$\$\$ \$8588\$\$\$	
F 1 F 2 PROGRAM USER DATA TRANSFER	F 3 F 4 DELETE INFO	F5 F6 Special Function Continue	F7 F8 EXIT

For component = CL:

B8576544-03

F1: load the specific overlay and branch to the 'PROGRAM' mask (PT, see 5.2) or 'JOB BLOCK' mask (RK, see 4.1)

F2: branch to the 'TRANSFER' mask (see 3.2)

F3: branch to the 'DELETE' mask (see 3.3)

F4: branch to the 'INFO' mask (see 3.4)

F5: branch to the 'SPECIAL FUNCTIONS' mask (see 3.5)

Press F6 to obtain the following menu:

1							
F 1	F 2	F 3	F 4	F 5	F 6	F 7	F 8
ASSIGN	ASSIGN	ASSIGN					
INT. PARA	PROC_PARA	PRI.PARA	LISTING		CONTINUE		EXIT
		•			•		

- El: branch to the 'ASSIGN INT. PARA' mask (only with PT component, see3.6)
- F2: branch to the 'ASSIGN PROC. PARA' mask
 (see 3.7)
- F3: branch to the 'ASSIGN PRI. PARA' mask (see 3.8)
- F4: branch to the 'LISTING' mask
 (see 3.9)

Press F6 to obtain the original menu again.

·····

With a library, the **following** menu is output in the 'SELECTION' mask:

na. Tarata

F 1	F 2	F 3	F4	F 5	F 6	F 7	F 8
LISTING	TRANSFER	DELETE	1 NFO	CONDENSE			EXIT

Fl: branch to the `LISTING' mask
 (see 3.9)

- F2: branch to the 'TRANSFER' mask (see 3.2)
- F3: branch to the 'DELETE' mask (see 3.3)
- F4: **branch to the'INFO'** mask (see 3.4)
- **P5:** physical condensing of the library (gaps in the files are closed)

COM 525

3 Functions in the Main COM Section

3.1 General Information

This section provides an overview of the standard **functions** of COM 525 (transfer, delete, info, special functions, interpreter, assign parameters to the procedure and printer and list).

The '-> acknowledgement prompt' means that COM 525 displays a prompt and the acknowledgement menu as follows:

a) with TRANSFER, the destination element exists with the same name but with different contents. The following acknowledgement prompt is displayed:

ACK.010: Element already exists! - overwrite?

b) for the TRANSFEROf a whole program, the prompt appears:

ACK.023: Transfer all elements?

c) to DELETE a whole program or an interface, or to delete all the elements of a data type you must answer the following prompt:

ACK.023: Delete all elements?

The prompts are acknowledged in the following menu:

F 1	F 2	F 3	F4	F 5	F 6	F 7	F 8
YES		NO					

Press F1 (YES) to execute the action described; press F3 (NO) to **return** to the original menu.

The fields marked with '*' in the masks in this **manual** can have entries made using F7 (HELP).

a sana an

3.2 Transfer

The transfer function allows the transfer of programs or parts of programs both on the hard disk and floppy disk as well as between the hard disk (or floppy) and the $C\!P$ (in both directions).

The 'TRANSFER' mask is called with F2 (TRANSFER) in the 'SELECTION' mask.

BASIC MASK -> PROG S E L E C T I O N	. SELECTION ->		SIMATIC	S5 / COM525
	DRIVE: \$ PROGRAM NAME:SSSS COMPONENT: \$\$	S\$\$S		
	Plant designation: Generated by: Generated on:	\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$ \$858\$\$58\$\$\$ \$\$\$\$\$\$\$\$		
F1 F2 PROGRAM	F 3 F 4	F 5 F 6 SPECIAL	F 7	F8
USER DATA TRANSFER	DELETE INFO	FUNCTION CONTINUE		EXIT

	_						
F 1 CP->	F 2	F 3 FD ->	F 4	F 5 m ->	F 6	F 7	F 8
m		CP		m			EXIT

,. ¹.

به مرین تو ۲۰۰ منوف ماند. و $\mathbf{The}\xspace$ 'working program' is the program specified in the 'SELECTION' mask .

This is the SOURCE for transferring FD->CP, but DESTINATION for transferring CP->FD and FD->FD

(FD is used for both floppy and hard disk).

The working program can be changed only in the 'SELECTION' mask, not within the 'transfer' function.

F1: change to the 'transfer from CP 525 to FD' mode

- F3: change to the `transfer from FD to CP 525' mode
 (see 3.2.2)
- F5: change to the `transfer from FD to FD' mode (see 3.2.3)

1 -

3.2.1 Transferring from CP 525 to FD

Call this menu with F1 (CP->FD) in the 'TRANSFER' mask.

->PROG. SELECTION -> SEL T R A N S F E R	ECTION ·>	SIMATIC S5 / COM525
	Source:	Dest.:
STORAGE MEDIUM: DRIVE: INTERFACE NUMBER: PROGRAM NAME: COMPONENT:	CP525 # \$\$	FD \$ 1 \$\$\$\$s\$\$\$ \$\$
Plant designation: Generated by: Generated on:	\$5.\$5\$\$\$\$\$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$	\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$\$
F1 F2 F Cold total inte Restart program pret	3 F 4 F 5 R- PRINT ER PROCEDURE PARA.	F6F7F8 USER DATA HELP EXIT

For COMPONENT = CL:

F 5	F 6
	JOB
	BLOCK

- F1: cold restart of the selected device interface on the CP 525 (must be carried out after every transfer to the CP)
- F2: transfer total program, -> acknowledgement prompt
- F3: transfer interpreter, if destination element already exists -> acknowledgementt prompt

- -> acknowledgement prompt
- F5: transfer print parameters
 -> acknowledgement prompt
 (With this function the specifications made in the 'ASSIGN
 PRI. PARA.' mask (see 3.8) are transferred).

F6: case 1: component = PT

The user data is transferred, thefollowingmenuisoutput:

	F 1	F 2	F 3	F 4	F 5	F6	F 7	F 8
	COLD RESTART	MESSAGE	Рс JOB	FRAME				EXIT
I								

M1: cold restart of the selected device interface on the CP 525

F2: transfer messages (see following page)

F3: transfer PC jobs (see following page)

F4: transfer frames (see following page)

F8: return to original menu

Case 2: component = CL

The job block is transferred.

All the jobs are always transferred, i.e. not only new or changed jobs. If the destination already exists, the destination is compared with the source to establish whether they are identical.

1 -

The following mask is used with the **PT** component for transferring the element types MESSAGE, PC JOB **and FRAME** (in each case with the corresponding designation in the mask header):

-> PROG. SELECTION -> SELE M E S S A G E	Ection •> Transf	ER ->	SIMATIC S	S5 / COM525
	Source:	Dest.:		
STORAGE MEDIUM: DRIVE: INTERFACE NUMBER: PROGRAM NAME: COMPONIENT:	CP525 \$	FD \$ \$\$\$\$\$\$\$\$		
Names:	35 ####################################	33 ###################################		
F1 F2 F3 All Single WITH A	3 F 4	F 5 F 6	F 7 HELP	F 8 Exit

F1: transfer single elements according to the entries in the 'names' fields.

If the **destination** element already exists (with different contents), -> acknowledgement prompt.

- F3: transfer all elements. If a destination element already exists (with different contents), -> acknowledgement prompt.
- **15**: transfer all elements without acknowledgement prompt (existing elements are overwritten).

35

B8576544-03

COM 525

مر_{ا الا} مرادر المراجعة الم

٠

Field	Field cype keys allowed	Limit value	Alternatives
DRIVE	OUTPUT FIELD		
INTERFACE NUMBER	numbers	1-2	1 _* 2
PROGRAM NAME	OUTPUT FIELD		
COMPONENT	OUTPUT FIELD		
Name	alphanumeric	characters -	* (in source)

F7 (HELP) can be used to make an entry in the fields marked $^{\prime}\star^{\prime}.$

3.2.2 Transferring from FD to CP 525

Call this menu with F3 (FD->CP) in the 'TRANSFER' mask.

The description of the fields (except the output fields "plant designation", 'generated by" and "generated on") can be found in Section 3.2.1

PROG. SELECTION-> SELE A N S F E R	SIMATIC S5 / COM5			
	Source:	Dest.:		
STORAGE MEDIUM:	FD	CP525		
DRIVE:		\$		
INTERFACE NUMBER:	#	1		
PRUGRAM NAME:	* *	\$\$\$\$\$ \$		
	2	\$ \$		
Plant designation:	\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$	\$\$\$\$\$\$\$\$\$\$\$\$\$		
Generated by:	\$SS\$\$s\$sS\$\$	\$\$s\$\$\$\$\$\$\$		
Generated on:	\$\$s\$\$ss\$	\$S\$S\$\$\$\$		
F1 F2 F	3 F4 F5	F6 F7 F8		
COLD TOTAL INTE	R- PRINT	USER		
START PROGRAM PRET	ER PROCEDURE PARA.	DATA HELP EXIT		

For COMPONENT = CL:

1	1 1
F 5	F 6
	JOB
	BLOCK

- Fl: cold restart of the selected device interface on the CP 525
 (must be carried out after every transfer)

- F3: transfer interpreter, -> acknowledgement prompt
- F4: transfer procedure, -> acknowledgement prompt
- F5: transfer print parameters
 -> acknowledgement prompt
 (With this function, the specifications made in the `ASSIGN
 PRI. PARA.' mask (see 3.8) are transferred).

F6: see 3.2.1.

- . ⁻

3.2.3 Transferring from FD to FD

Call this menu with F5 (FD->FD) in the `TRANSFER' mask.

...

ANSFER			
	Source:	Dest.:	
STORAGE MEDIUM:	FD	FD	
	#	\$	
PROGRAM NAME:	#	\$\$\$\$\$\$\$	
COMPONENT:	\$\$	\$\$	
Plant designation: Generated by:	\$\$s\$s\$\$s\$\$\$\$\$ \$\$\$\$\$\$\$\$\$\$\$	\$\$\$\$\$\$\$\$\$\$\$\$\$ \$ \$ \$ \$ \$ \$ \$ 3 \$ \$ \$	
Generated on:	\$\$ \$\$\$\$\$\$	\$\$\$\$\$\$\$	
	3 F 4 F 5	F 6 F ? F 8	
PROGRAM PRETE	R PROCEDURE PARA.	OATA HELP EXI	

For COMPONENT = CL:

F 6 JOB BLOCK

F 5

- F2: transfer total program -> acknowledgement prompt
- F3: transfer interpreter, -> acknowledgement prompt
- F4: transfer procedure,
 -> acknowledgement prompt

-

F5: transfer print parameters
 -> acknowledgement prompt.

F6: see 3.2.1.

Field	Field type keys allowed	Limit value	Alternatives
DRIVE (source)	alphamm. characters		A,B,C,D,,J
DRIVE (dest.)	OUTPUT FIELD (from the PROGRAM SELECTION mask)		
PROGRAM NAME (source)	alphamm. characters		*
PROGRAM NAME (dest.)	OUTPUT FIELD		
COMPONENT	OUTPUT FIELD		
Plant designation	OUTPUT FIELD		
Generated by	OUTPUT FIELD		
Generated on	OUTPUT FIELD		

F7 (HELP) can be used to make an entry in the fields marked '*' .
3.3 Delete

This function allows programs or parts of programs to be deleted on the CP, hard disk or floppy disk.

Call the **`DELETEI** mask with F3 (DELETE) in the `Selection mask (see **2.4**).

PROG. SELECTION -> D E L E T E	SELECTION ->		SIMATIC	S5 / COM52
	DRIVE: \$ PROGRAM NAME: \$\$\$\$ COMPONENT: \$ \$	SSSS		
	Plant designation: Generated by: Generated on:	\$\$\$\$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$ \$\$\$\$\$\$\$		
F1 F2	F3 F4	F5 F6	F 7	F 8
PROGRAM USER DATA, TRANSFER	DELETE INFO	SPECIAL FUNCTION CONTINUE		EXIT

ll V

F 1	F 2	F 3	F 4	F 5	F 6	F 7	F 8
СР		FD					EXIT

Fl: change to `delete CP 525¹ mode (see 3 .3.1)

F3: change to `delete **FD'** mode (floppy disk or hard disk, see 3.3.2)

3.3.1 Delete **CP** 525

Call the 'DELETE' mask (CP 525) with F1 (CP) in the delete menu.

STORAGE MEDIUM: CP		
COMPONENT: \$\$		
Plant designation: \$\$\$\$\$\$\$\$\$\$\$\$\$\$\$ Generated by: \$\$\$\$\$\$\$\$\$\$ Generated on: \$\$\$\$\$\$\$\$		
F 1F 2F 3F 4F 5F 6COLDBOTHASSIGNUSERRESTARTINTERFACEINTFACESPRI.PARADATA	F 7 HELP	F 8 EXIT

For COMPONENT = CL:

- **fl:** cold restart on the selected device interface on the **CP** 525 (must be carried out following each deletion)
- P2: delete selected interface
 -> acknowledgement prompt
- F3: delete whole user RAM
 -> acknowledgement prompt

مري منځمن دن و

- F5: delete print parameters (with this function, the specifications made in the 'ASSIGN PRI. PARA' mask (see 3.8) will be deleted).
- F6 : case 1: component = PT

Delete user data, the following menu is displayed:

F1 F2	F 3	F 4	F 5	F 6	F 7	F 8
COLD	Pc					
RESTART MESSAGE	J08	FRAME				EXIT
	1					

- F1: cold restart on the selected device interface on the CP 525
 (must be carried out following each deletion)
- F2: delete messages (see following page)
- **F3**: delete PC jobs (see following page)
- F4: delete frames (see following page)
- **F8:** return to original menu

Case 2: component = CL

Delete job block (all jobs are deleted)

44

.................

,

1 -

The following mask is output with the **PT** component to DELETE the element types MESSAGE, PC JOB and FRAME (in each case with the corresponding designation in the mask header):

-> PROG. SELECTION M E S S A G E	-> SELECTION -> DELE	ete •>	SIMATIC S5 / COM525
	STORAGE MEDIUM: INTERFACE: COMPONENT:	FD # \$\$	
	Plant designation: Generated by: Generated on:	\$\$\$\$\$\$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$	
	Name : ###################################	1### 1### 1### 1###	
F 1 F 2 DELETE SI NGLE	F 3 F 4 DELETE ALL	F5 F6	F 7 F 8 HELP EXI T

P3: delete all elements

-> acknowledgement prompt

B8576544-03

COM 525

Field	Field type keys allowed	Limit value	Alternatives
INTERFACE NUMBER	numbers	1-2	1, 2 *
PROGRAM NAME	OUTPUT FIELD		
COMPONENT	OUTPUT FIELD		
Plant designation	OUTPUT FIELD		
Generated by	OUTPUT FIELD		
Generated on	OUTPUT FIELD		
Name	alphanum. characte:	rs	*
F7 (HELP) can 1	be used to make an en	try in the fi	lelds marked '*'.

. -

and a second second

3.3.2 Deleting on FD

Call the 'DELETE' mask (FD) with the function key F3 (FD) in the delete menu.

المالية المعاجرة معمدته فلالار

-> PROG. SELECTION • D E L E T E	-> SELECTION		SIMATIC S5	/ COM525
	STORAGE MEDIUM: DRIVE: PROGRAM NAME: COMPONENT: Plant designation:	FD \$ \$\$\$\$\$\$\$\$\$ \$\$ \$		
	Generated by: Generated cm:	\$\$\$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$\$		
F1 F2 TOTAL PROGRAM	F 3 F 4 INTER- PRETER PROCEDURE	F 5 F 6 PRINT USER PARA DATA	F 7 HELP	F 8 EXIT

For COMPONENT = CL:	F 5	F 6 JOB
FOR COMPONENT = CL:		
	•	· DLUUK

- F2: delete total program -> acknowledgement prompt
- F3: delete interpreter
- F4: delete procedure
- F5: delete print parameters

F6: see 3.3.1, but without F1 (COLD RESTART)

COM 525

3.4 Info

In this mask, you can call up information about parts of the
program (e.g. the names of all the PC jobs and their lengths). It .
is also possible to call up an overview of a whole program with
F1 (BRIEF DESC). You can also have this information printed out
(see 3.4.2 to 3.4.5), If you require information about further
programs enter their names in the 'PROGRAM NAME' field.

κ.,

Call the 'INFO' mask with F4 (INFO) and F1 (CP) or F3 (FD) in the 'SELECTION' mask.

->PROG.SEL	ECTION -> S	ELECTION -	>			SIMATIC	S5 / COM525
	ST(DR) PR(COI	DRAGE MEDIU (VE: DGRAM NAME: MPONENT:	M: F # \$ \$	D + ***********************************	₲₲₵₲₲₲₲₲₲		
	Ge	nerated by: nerated on:	\$ \$	********** is.\$s\$ssss: \$\$\$\$\$\$\$	*****	>	
ra l	I		1		1 - /	1	
BRIEF DESC.	F 2 IESSAGE	F 3 Pc JOB FI	F 4 Rame	F 5	F 6	F 7 HELP	F 8 EXIT

For COMPONENT = CL:

F 5 J09 BLOCK

F1: brief description of a user program

F2: info about messages (only for component PT)

F3: info about PC jobs (only for component PT)

F4: info about frames (only for component PT)

F5: info about a job block (only for component CL, in this case keys F2 to F4 have no function)

If the name of a library is specified as the program name the function keys F2 to F4 and F5 are assigned functions and F6 has the function 'CONTINUE'.

F6: output of a second menu:

.							
F1	F 2	F 3	F 4	F 5	F 6	F 7	F 8
BRIEF		INTER-		PRINT			
DESC.		PRETER	PROCEDURE	PARA	CONTINUE		EXIT
'			1		I		I

M1: brief description of the library

F3: info **about** interpreters

F4: info about procedures

F5: info about parameters

F6: returntofirstmenu

Field	Field type keys allowed	Limit value	Alternatives
DRIVE	alphamm. characters (only for FD info)	A - J	A,B,C,D,J
INTERFACE NUMBER	numbers (only for CP info)	1-2	1, 2 *
PROGRAM NAME	alphamm. characters (only for FD info)		*
COMPONENT	OUTPUT FIELD		
Plant designation	OUTPUT FIELD		
Generated by	OUTPUT FIELD		
Generated on	OUTPUT FIELD		

F7 (HELP) can be used to make an entry in the fields marked $'\star'$.

3.4.1 Brief Description

?ress F1(BRIEF DESC.) in the 'INFO mask to call up the brief description submask.

PROG. SELECTION N F O	SELECTION ·>		SIMATIC S5 / C	DM525
	STORAGE MEDIUM: DRIVE: PROGRAM NAME: COMPONENT :	FD # ######### \$\$		
	Plant designation: Generated by: Generated on:	\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$		
Data type Ele	nent name Number	Data type	Number	
Interpreter: SSS Procedure: SSS Print para.: PRIN	33555 SS 8 35555 SS 8 T -Para S	Message: PC job: Frame:	\$\$\$\$\$ \$\$\$\$\$ \$\$\$\$\$	
Total number : Program length :	SSSSS elements SSSSSS words			
F 1 F 2 BR IEF DES: . MESSAGE	F3 F4 PC JOB FRAME	F 5 F 6	F 7 F HELP EX	8 (17

The names of the interpreted and procedure are displayed (in each case with version number) and the number of elements. In addition, COM 525 displays how much memory space is occupied by the whole program in the CP 525.

Note: the specified program length **always** includes 2 x 4 Kwords (2 x 8 Kbytes) for interpreters and procedures of both device interfaces (regardless of whether one or both interfaces are being used). The program length also includes the memory required for the memory manager. In the 'program length' field, the length of your program is specified, but not the memory space actually taken up; this is always a multiple of 8 Kbytes.

51

* • • • • •

The name for print parameters is always <code>PRINT-PARA</code> in <code>COM</code> 525 programs (this is automatically assigned by <code>COM</code> 525). It can only be renamed when it is being transferred to libraries.

If **the name** of a library is specified as the program name the brief description appears as follows:

-> PROG. SELECTION I N F O	> SELECTION ->	SI	MATIC S5 / COM525
	STORAGE MEDIUM: DRIVE: PROGRAM NAME: COMPONENT:	FD # ########## \$\$	
	Plant designation: Generated by: Generated on:	\$SS\$\$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$\$\$ \$Ss\$ss\$s	
Data type Numb	er	Data type Number	
Interpreter: \$\$\$\$ Procedure: \$\$\$\$ Print para.: \$S\$\$	\$ \$ S	Message: \$\$\$\$\$ PC job: \$\$\$\$\$ Frame: \$\$\$\$\$ Job block: \$\$\$\$\$	
Total number :	SS\$\$S elements		
F1 F2 BRIEF DESC. MESSAGE	F 3 F 4 PC JOB FRAME	F5 F6 Job Block Continue Hi	F7 F8 ELP EXIT

In addition to the number of data types, the number of interpreters, procedures and print parameters contained in the library are also displayed.

52

معمورة بعالما لالتحال

,

3.4.2 Information about Messages

÷

Call the 'MESSAGE' mask with F2 (MESSAGE) in the 'INFO' mask.

The functions paging forwards and paging backwards will be required if there are more elements than can be displayed.

RAGE MEDIUM: FD	DRIVE:	<pre>\$ PROGRAM NAME: \$\$\$\$\$\$\$\$ Page \$\$of \$</pre>
Len Element name (wo	yth 'ds) Element name	Length Length e(words) Element name(words)

F1: output the displayed information on the printer

F5: page backwards

F6: page forwards

pet Br /

3.4.3 Information about PC Jobs

Call the 'PC JOB' mask with F3 (PC JOB) in the 'INFO' mask.

For the function key assignment see 3.4.2.

In this mask the job number belonging to each element is also output.

3.4.4 Information about Frames

Call the 'FRAME' mask with F4 (FRAME) in the 'INFO' mask.

For the function key assignment see 3.4.2.

3.4.5 Information about a Job Block

Call the 'JOB BLOCK' mask with F5 (JOB BLOCK) in the 'INFO' mask. (Only for component CL).

For the function key assignment see 3.4.2.

The numbers of all programmed jobs will be displayed.

Call this mask with F6 (CONTINUE) and F3 (INTERPRETER) or F4 (PROCEDURE) in the **'INFO'** mask. This information function is only available with libraries. It provides the following mask:

with hitsen

'ORAG	E MEDIL	JM: FD		DRIV	E:		\$	PROG	ram na	ME: \$	\$\$\$SSS	SS
										Pag	e \$\$	of \$\$
emp.	name	Vers	Length .(words)	Comp.	n a	me	Leng Vers.(th words)	Comp.	name	Vers.	Length (words)
								1				
F I		F 2	F 3		F 4		F 5 PAGE	F	6	F	,	F 8
INTE	R					В	ACKWARDS	FORWAR	RDS			EXIT

This mask provides information about ${\tt components}~({\tt PT}~{\tt or}~{\tt CL})\,,$ names, version and length of the interpreters or procedures contained in the library.

٠.

3.5 Special Functions

Call the 'SPECIAL FUNCTIONS' mask with the function key F5 (SPECIAL FUNCTIONS) in the 'SELECTION' mask.

-> PROG. SELECTION -> SELECTION -> SIMATIC S5 / COM525 SPECIAL FUNCTIONS DRIVE: : S PROGRAM: \$\$\$\$\$\$ COMPONENT: \$S PURGE: logical 'tidying up' of a program. All the elements not used in the program are made available for deletion. CONDENSE: the program is physically condensed. Under no circumstances must the storage medium be removed from the drive, in order to break off the function. CHECK JOB NUMBERS: all PC jobs with numbers used more than once will be listed. F 1 F 2 F 5 F 8 F 3 F 4 F 6 F 7 CHECK PURGE CONDENSE JOB NOS. EXIT

F1: branch to the 'PURGE' mask (logical tidying up)

F3: condense a file (physically)

F5: check the PC jobs for numbers used more than once

3.5.1 Purge

 \mbox{Call} the 'PURGE' mask with \mbox{Fl} (PURGE) in the 'SPECIAL FUNCTIONS' mask.

The functions page forwards and backwards are required if there are **more** elements present than can be displayed,

		DRI VE:	\$	PROGRA	M: \$SS	\$\$\$S\$	COMPONENT	: S\$
The following f during purging	rames are	not used a	nd car	n be dele	eted			
Element name	Length (words)	Element	name	Length (words)	Elemer	nt name	Length (words)	
•••	<u> </u>		1		<u> </u>	1 -	'	_
PI F2 DN PAGE RINTER BACKWA I	PAGE FORWAR	DELETE DS ALL		F 5	F 6 DELETE SI NGLE	F	? F i EXI	5 T

M1: output of the displayed information on the printer

F2: page backwards

- F3: page forwards
- F4: delete all the elements made available -> acknowledgement prompt

3.5.2 Check Job Numbers

 \mbox{Call} the $'\mbox{CHECK}$ JOB NOS. ' mask with F5 (CHECK JOB NOS.) in the 'SPECIAL FUNCTIONS' mask.

The functions page forwards and page **backwards** are **required** if there are more elements present than can be displayed.

->Рі С Н	ROG.SELECTION -> SELEC ECK JOB NOS.	TION -> SPECIAL FUNCT	> SIMATIC S5 / COM525
STO	RAGE MEDIUM:\$\$\$\$\$	\$s\$\$ss\$\$sss\$	PROGRAM NAME: \$\$\$SSS\$
			Page \$\$ of S\$
	Job Element name nummber	Job Element name number	Job Element name number
PRI	NTER	F 4 F 5 PAGE BACKWARDS	PAGE FORWARDS EXIT

This mask displays all the PC jobs which (incorrectly) have the same job **number**, arranged according to job numbers **and** in alphabetical order.

M: output of the displayed information on the printer

F5: page backwards

F6: page forwards

جرز راما و جاما ما جاما و

•

3.6 Assign Interpreter Parameters

This function is available only for programs with the **PT compo**nent; this component is used to assign parameters to the printer (should not be **confused** with the function 'assign printer parameters' for listings which are output on the printer connected to the **PG**). The input fields on the left-hand side of the mask contain the default **values** of the parameters for the interpreter present in this program. The alternatives are displayed on the right-hand side of the mask.

4

The defaults for the assignment of date and time parameters corresponds to the German format.

If you require the English format, the specification should appear as below:

format (date): M D Y
format (time): H M S
separator (date): /
separator (timme): :
time format: ENGLISH AM/PM

Example:

date / time German: 30.06.87 14.30.22

date / time English: 06/30/87 02:30:22 PM

Call the *ASSIGN INTERP. PARA. ' mask with F6 (CONTINUE) and F1 (ASSIGN INT. PARA.) in the 'SELECTION' mask.

PROG. SELECTIONS S I G N 1 N	ON → SELECTI TERP.PA	ON •> A R A.			SIMATIC S	5 / Com52
		DRI VE:	S PROGR	RAM: S\$S	SSSS\$ COM	PONENT:
Interpreter:	COMPONENT:	\$S NAM	E: \$S\$SS	SSS VERS	ION: \$\$	
		Pr	esetting	for interp	oreter PT88	:
Printer model Printer type:	:	PT ## PT # N≕	88 or F reedle hea	РТ 89 d I :	=inkjet	
Lines per inc Page length: Page width: Format (date Format (time Separator (da Separator (ti Time format: Char. set:	h: :):): te): me>: ####################################	# ### li ### fr # # # E # # # # # # # # # # # # # # # #	nes mi 1 to SSI iday Mi ihour Mi MAN 24 H,	characte ⊫ month ⊫ minute ENGL	rs per lin Y=Year S=second ISH AM/PM	e
F1 F2	F 3	F 4	F 5	F 6 SAVE	F 7 HELP	F 8 EXI T

F6: save the interpreter parameter assignment

F8: does not save (if data have been changed without saving them an acknowledgement prompt appears); return to the 'SELECTION[®] mask

Field	Field type keys allowed	Limit value	Alternatives
COMPONENT	OUTPUT FIELD		
NAME	OUTPUT FIELD		
VERSION	OUTPUT FIELD		
Printer model	numbers		88, 89 *
Printer type	letters		T, N *
Lines per inch	numbers		6, 4, 3 *
Page length	numbers	dependent on `lines per inch' 1)	*
Page width	numbers	1 - 80 (for PT88)	*
		1 - 136 (for PT89)	
Format (date)	letters		D, M, Y *

F7 (HELP) can be used to make an entry in the fields marked '*'.

1)6	lines/inch	1 .	•	٠	1 -	99	(default	withheld:	72)
4	lines/inch				1-	66	(default	with HELP:	48)
3	lines/inch				1 -	49	(default	withheld:	36)

61

•

```
B8576544-03
```

COM 525

م با مرار با ماره ده مار ماند ه

Field	Field type keys allowed	Limit value	Alternatives
Format (time)	letters		H, M, S *
Separator (date, time)	letters		* + . , ' / : ; *
Time format	alphanumeric characters	C I	GERMAN 24 H ENGLISH AM/PM *
Char . set	letters	C F F S S F F I	SERMAN, ENGLISH, SCII, SCII, SRENCH, SPANISH, SORWEG./DANISH, SWEDISH, 'INNISH, INTERNATIONAL *

F7(HELP)can be used to make an entry in the fields marked ¹*¹. In the 'separator' field the character '*' is **also one** of **the** alternatives.

3.7 Assigning Procedure Parameters

Just as with the function 'assigning interpreter parameters' the parameters for the procedure can also be changed. The input fields contain defaults and display the parameters for the **proce**-dureintheprogram named in the mask.

Call the 'ASSIGN **PROC. PARA'** mask with F6 (CONTINUE) and F2 (ASSIGN **PROC. PARA)** in the 'SELECTION' mask.

	DRIVE: \$	PROGI	RAM: SS\$SS\$SS	COMPONENT: \$
Procedure: COMPONENT: \$	\$ NAME: \$SS	S\$SS\$	VERSION: \$\$	
Baud rate: ###	##		Char. length:	#
Number of stop bits: ##	##		Priority: ##	t#####
Pari ty: ######	##			
F1 F2 F3	F 4	F 5	F 6	F7 F8
			SAVE H	FIP FXIT

F6: save the procedure parameter assignment

F8: does not save (if data have been changed, without saving them an acknowledgement prompt appears), return to the 'SELECTION' mask

• 3

Field	Field type keys allowed	Limit	value	Alternatives
COMPONENT	OUTPUT FIELD (ace. to prog . name)			
NAME	OUTPUT FIELD			
VERSION	OUTPUT FIELD			
Data rate	numbers			9600,4800,2400, 1200,600,300, 200,150,110, 100,75,50 *1)
Char. length	numbers			8, 7, 6, ⁵ ₁₎
Number of stop bits	numbers			1, 1,5, 2 * 1)
Priority	letters			HIGHER, LOWER $\frac{1}{2}$
Parity	letters			EVEN, ODD * 1)

F7(HELP) can be used to make an entry in the fields marked '*'.

 The options depend on the procedure. Only those parameters which can be changed in the particular procedure aremade available for changing.

3.8 Assigning Printer Parameters

You can use this function to determine the printout header and trailer of listings to be output on the printer connected to the PG. This has nothing to do with the function 'event output and listing on the PT88/PT89 printer'. After this mask is saved, all the lists are printed out with the specifications made in this mask (see 3.4, 3.5, 3.9 and 4.2).

Call the 'ASSIGN **PRI. PARA.**' mask with F6 (CONTINUE) and F3 (ASSIGN **PRI. PARA.**) in the 'SELECTION' mask.

			DRIVE:	\$	PROGRAM :	\$\$\$\$\$\$	S COMPONI	ENT: \$
rintout hea	ader:							
SIEMENS S	IMATIC S5 #	**************************************	*****			## Page:	22222	
Dri ve:	\$ Progra	am: \$\$\$\$\$	\$\$\$\$	la	st worked	with:	\$\$\$\$\$\$\$	
Plant: \$	\$\$\$\$s\$S\$\$\$	\$\$\$\$\$\$\$		Gei	nerated b	y: \$sS\$\$	\$\$\$\$\$\$\$\$	
Plant: \$	\$\$\$\$\$\$\$\$\$ 	\$\$\$\$\$\$\$		Gei	nerated b	y: \$sS\$\$	\$\$\$\$\$\$\$	
Plant: \$	3\$\$\$\$\$\$\$\$\$ ailer:	\$\$\$\$\$\$ \$		Gei	nerated b	y: \$sS\$\$	\$\$\$\$\$\$ <u></u>	
Plant: \$ rintout tra	\$\$\$\$\$\$\$\$ ailer:	\$SS\$S%\$		Gei	nerated b	y: \$sS\$\$	\$\$\$\$\$\$\$\$ #########	
Plant: \$	\$\$\$\$\$\$\$\$\$ ailer: ####################################	\$\$\$\$\$\$ ###############################	****	Gei ######## #########	nerated b	y: \$sS\$\$	######################################	
Plant: \$	\$\$\$\$\$\$\$ ailer: ####################################	\$\$\$\$\$\$\$ ***********	F 4	Ger	nerated b	y: \$s\$\$\$ #################################	F 7	F8

F6: save the printer parameter assignment

F8: abort (if data is changed without saving it, an acknowledgement prompt appears); return to the 'SELECTION' mask

Field	Field type keys allowed	Limit value	Alternatives
Printout header	ASCII upper/lower case		
Printout trailer	ASCII upper/lower case		
Drive	OUTPUT FIELD (defaults from the PROGRAM SELECTION mag	sk)	
Program	OUTPUT FIELD (defaults from the PROGRAM SELECTION mas	sk)	
Component	OUTPUT FIELD (defaults from the PROGRAM SELECTION ma	sk)	
Iast worked with	OUTPUT FIELD (defaults from the PROGRAM SELECTION ma	sk)	
Plant	OUTPUT FIELD (defaults from the PROGRAM SELECTION ma	sk)	
Generated by	OUTPUT FIELD (defaults from the PROGRAM SELECTION ma	lsk)	

÷

. .

Using the listing function, you can have a **summary of** your program printed out. This shows the **structure** of the program and how the individual parts of the program access each other.

When parts of programs are listed the corresponding elements are listed one after the other.

Remember that these listings are printed out on a printer connected to the **PG**, i.e. not on the listing printer.

The pages of all listings have a printout header and trailer that is specified with the 'assign printer parameters' function (see 3.8) and that can be saved on floppy or hard disk. The print parameters of a **PT** program can also be stored on the **CP** 525 (CP 524).

The page width of a listing is always 72 characters. Message or PSL lines which are longer than this are broken up arbitrarily after 72 characters.

All the listing functions can be aborted with F8 (EXIT).

Call the **'LISTING'** mask with F6 (CONTINUE) -> F4 **(LISTING)** in the 'SELECTION' mask.

BASIC MASK ->PROG. LISTING	SELECTION -> SELEC	CTION ·>	SIMATIC S5 / COM525
	DRIVE: \$ PROGRAM NAME: SSS: COMPONENT: \$ S	\$\$\$\$\$	
	Plant designation: Generated by: Generated on:	\$\$\$\$\$\$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$ \$\$\$\$\$\$\$\$\$	
F1 F2	F3 F4	F5 F6	F 7 F 8
TOTAL INT. PRT. PROGRAM PROCEDURE	MESSAGES JOB	FRAME LISTING	EXIT

- **F1:** listthetotalprogramonthepri.nter
- **F2:** print out the parameter assignments for the interpreter and procedure
- **P3:** print out all messages

ه يې د ^{ير} د خانه خانه . و

This starts with the printout of the parameters for the PC job, sequential message list (SML). These parameters determine the position am-structure of the date, time and status for all messages. The messages are then printed out in ascending order with their static and dynamic parts.

F4: with component PT: a new mask is displayed:

Fl: all the PC jobs whose names have been entered manually or with HELP are printed out.

F3: printout of all the PC jobs. The PC jobs are printed out in the order in which they were made available during programming (see 5.4),

For component CL: printout of the job block.

F5: a new mask is displayed:

F1: all the frames whose names have been entered manually or with HELP are printed out.F3: all frames are printed out,

- F6: printout of a summary listing. In additicm to the interpreter and procedure parameters, a table listing the PC jobs is also printed out.
- F8: abort the current listing or return to the 'SELECTION' mask.

If the program name is a library then $\ensuremath{\texttt{Fl}}$ (LISTING) calls the menu shown as follows:

	1	1	1	1	1		
F 1	F 2	F 3	F 4	F 5	F 6	F 7	F 8
INTER -			Pc		JOB		
PRETER	PROCEDURE	MESSAGES	JOB	FRAME	BLOCK		EXIT
	I	I	I	I	1	1	

- F1: change to the mode 'listing of the interpreters contained in the library'. Analogous to the 'DELETE' function (see 3.3.1) in the follow-on menu, F1 (SINGLE LISTING) is used to list single interpreters or F3 (TOTAL LISTING) for all interpreters.
- F2: changetothemode `listing of the procedures contained in the library', Branch see F1.
- F3: printout of all messages (see previous page).
- F4: printout of all PC jobs (see previous page).
- F5: printout of all frames (see previous page).
- F6: change to mode `listing the job blocks contained in the library'. Branch see F1.
- F8: abort the current listing or return to the 'SELECTION' mask.

B8576544-03

4 Functions with the Computer Link Component

For further information about the computer link component refer to the user's guide "Computer link with **RK** 512" in this manual.

4.1 Programming the Job Block

Call the 'JOB **BLOCK'** mask with **F1** (PROGRAM USER DATA) in the 'SELECTION' mask.

->PROG. SELECTION -> SELECTION -> JOB BLOCK					IMATIC S5 /	COM52
	DRIVE:	\$	PROGRAM:	\$\$\$\$\$\$\$\$	COMPONENT:	\$\$
	1 O L	В				
Job number: # # #						
F1 F2 F	3 F 4		F 5 PROGRAM	F 6 STORE	F 7 F	8
		.	10B 10	DB BLOCK	E	XIT

F5: change to the `program job' mode; branchtothe `PROGRAM JOB' mask

F6: save a job block

71

جعاري فارتبا فتهام حصود الاناب

,

.

Field	Field type keys allowed	Limit value	Alternatives
Job number	numbers	1 - 189	

4.2 Programming a Job

Call the 'PROGRAM JOB' **mesk with** F5 (PROGRAM JOB) in the 'JOB BLOCK' mask.

	DRIVE:	\$	PROGRAM:	\$\$\$\$\$\$\$\$	COMPONENT	: \$\$
		JOB	5			
Job no.:	\$\$\$					
Job:	#####					
Job type	######################################	######				
CPU no.:	#					
DB no. :	###					
Source - word a	ddress: ##### D	####	н			
If required wi	th coordination fla	ag: ##	#.#			
F 1 F 2	F3 F4		F 5	F 6	F 7 F	8
ON PAGE RINTER BACKWAR	PAGE DS FORWARDS	DE		ENTER JOB	HELP E	кіт

F1: the job is output on the printer

- **F2**: if it exists, the job with the next lower job number is displayed on the screen
- **F3**: if it exists, the job with the next higher job number is displayed on the screen
- **F5**: delete the current job
- F6: enter the current job

Field Field type keys allowed		Limit value	Alternatives
Job number	numbers	1 - 189	
Job	upper case letters		SEND FETCH *
Job type (see Table 1)	upper case letters		*
CPU no.	numbers	1 - 4	1, 2, 3, 4 *
DB no.	numbers	3 - 255	
Address	numbers / hex. numbers	(see Table	1)
Coordination flag	numbers	bytes: 1 - 223 bits: o - 7	

F7 (HELP) can be used to make an entry in the fields marked `*'.

COM	525

- - .

JOB	JOB TYPE	max. ADDRESS	ADDR. TYPE
s E N D	DATA BLOCK SYSTEM DATA ABSOLUTE ADDRESSES EXT. DATA BLOCK	255 511 65535 255	WORDS WORDS WORDS WORDS
F E T C H	DATA BLOCK FLAGS INPUTS OUTPUTS 1/0'S TIMER LOCATIONS COUNTER LOCATIONS SYSTEM DATA ABSOLUTE ADDRESSES EXT. DATA BLOCK EXT. 1/0' S	255 255 127 255 255 255 511 65535 255 255 255	WORM BYTES BYTES BYTES WORDS WORDS WORDS WORDS WORDS BYTES

Table 1: Job types

If the specified limit values for addresses are exceeded then a warning is output. The jobs will nevertheless be accepted if the address is less than or equal to **FFFFH**.

5 Functions with the Printer Component

5.1 General Information

At the top right of all the masks the three output fields "DRIVE", "PROGRAM" and "COMPONENT" appear. They always contain the defaults specified in the 'PROGRAM SELECTION mask.

If a data element already exists, you are prompted to decide whether it should be overwritten and the **acknowledgement** menu is displayed.

ACK.010: Element already exists! - overwrite?

F 1	F 2	F 3	F 4	F 5	F 6	F 7	F 8
YES		NO					

If you exit a mask using F8 (EXIT) in which fields have been filled in without having entered the data, then an acknow-ledgement is also **prompted**.

ACK.002: Loss of data - exit mask?

F 1	F 2	F 3	F4	F 5	F 6	F 7	F 8
YES		NO					

If you answer 'YES' the action described is carried out, if you answer 'NO' you return to the original menu.

The fields in the masks marked with '*' in this manual can have inputs made using F7 (HELP).
5.2 Programming

Call the 'PROGRAM' mask with $\ensuremath{\textbf{Fl}}$ (PROGRAM USER DATA) in the 'SELECTION' mask.

In the 'PROGRAM' mask you select what you wish to program:

- messages
- PC jobs
- frames

-> PROG. SELECTION P R O G R A M	-> SELECTION ->	SIMATIC S5 / COM525
	DRIVE: SPROGRAM: \$\$S\$SS	SS\$ COMPONENT: S\$
MESSAGE:	programing messages	
PC JOB:	programing jobs to be initiated from the P	C
FRAME:	programing list headers / list trailers	
F 1 F 2 PC MESSAGE JOB	F 3 F 4 F 5 F 6 FRAME	F7 F8 EXIT

F1: change to 'programming messages' mode; branch to 'MESSAGES' mask

> **!Important!** before programming messages forthefirst tine, the PC_job that determines the position and **structure of the date, time** and **status** for all messages must be generated (see 5.3 and 5.4).

الجرور ومردية

- F2: change to the 'program PC jobs' mode; branch to the 'PC JOB' mask
- F3: change to the 'program frames' mode; branch to the 'FRAME' mask

78

•

·. ·

5.3 MESSAGES

Messages are made up of the following:

- static parts, i.e. fixed texts
- 'fixed' dynamic data (date, time, message status)
- -dynamic parts, **i.e.** variable data which can be inserted at any point between the static parts.

static parts

The structure and contents of the static parts are determined when you program them on the $\ensuremath{\text{PG}}$ using the text editor.

'Fixed' dynamical at a

The position and structure of the 'fixed' dynamic data are the same for all messages. They must be specified globally before you program the messages. You do this when you program the PC job 'sequential message list' (SML, see Section 5.4.1). Before you program the first messages, you must therefore program the PC job SML.

The specifications for

- the date (position, order, print attributes)
- the time (position, order, **print** attributes)
- the message status (position, length, texts, print attributes)

apply for all messages in a **program** and can be modified only in the 'SEQ.MESSAGE LIST' mask, i.e. they cannot be changed by the editor.

In the editing window of the ${\bf P}\!G$ screen, the corresponding fields are displayed inversely.

Dynamic part

Each message can have a maximum of one dynamic parameter. When programming the static parts, you can decide where you **wish** to insert this parameter.

The number of **reserve** characters inserted in the static text for a dynamic field serves as the default for the number of characters when you program the dynamic part.

You can select the following features for each dynamic field:

number of characters in the list
- print attributes (bold, expanded, underlined, acoustic signal)
- position in the list

All other data (data type, source) are supplied to the CP by the CPU. They are therefore not prompted during programming.

Messages are programmed in two steps as follows:

- 1. The static parts are programmed first. Press F1 (STAT. PART) in the 'MESSAGES' mask you call the 'STATIC PART' mask.
- The dynamic part is then specified for the static parts which have already been programmed. This is achieved by branching to the 'DYN. PART' mask using F2 (DYN. PART) in the 'MESSAGES' mask.

WithF4 (COMPLETE DYN. PART) in the 'MESSAGES' mask you can select the 'COMPLETE DYN. PART' mask. This function automatically searches for dynamic fields which have not yet been programmed although the static parts are already specified.

Notes on programming messages

To reduce the processing time of the editor to a minimum, up to 100 messages (**static** and dynamic parts) can be buffered in the RAM memory of the **PG** (called editing buffer below). Transfers from and to the mass storage of the PG (floppy or hard disk) are performed only at the beginning and end of an editing session:

at the beginning when you start the editor in the 'MESSAGES' mask with F1 (STAT.PART), F2 (DYN. PART) and F4 (COMPLETE DYN. PART) at the end of the session in the masks 'STAT. PART', 'DYN. PART' or 'COMPLETE DYN PART' with F8 (EXIT).

In these cases the time required for the transfer must be taken into account. The amount of time depends on the following:

the number of messages programed, i.e. the more messages to be transferred the longer the transfer time the speed of the external. $\ensuremath{\texttt{PG}}$ memory (hard disk is faster than floppy)

It is possible to optimise the editing and transfer **times** by following certain procedures, as shown in the following examples.

Example 1: programming a large number of new messages

Read in only a few (e.g. one) of the already programmed messages.

- Program your new messages (up to 99) in one session.
- Save the programmed message.

Advantages:

- -Atthe beginning there **are almost no transfer times**, since **You only** read in a few messages.
- You can then edit up to 99 messages without being disturbed (i.e. without losing time through transfers), before you complete the session and have to save the messages.

Disadvantages:

- At the end of the editing session the transfer time is particularly long. Depending on the number of messages and the type of external storage on the **PG** the time may exceed 10 minutes.

Example 2: programming a few new messages

- Read in only a few (e.g. one) of the already programmed messages.
- Program your new messages.
- Save the programmed messages.

Advantages:

- At the beginning there are almost no transfer times since you only read in a few messages.
- At the end there are also only short transfer times since only a few messages have been programed.

Example 3: mining individual messages

- Read in the appropriate message singly.
- Change this message.
- Save the modified message.

Advantages:

- At the beginning there are almost no transfer times since onlyonemessagehasbeenread in.
- At the end there are only short transfer times since only one message must be saved.

Example 4: changing a lot of messages

- If the messages are consecutive, follow the procedure as described in Example 1.
- If the messages are separated, follow the procedure as described in Example 3.:

You call the 'MESSAGES' mask with the function key \mathtt{Fl} (MESSAGE) in the 'PROGRAM' $\mathtt{mask}.$

SELECTION → PROO ESSAGES	GRAM ·>		SI	MATIC S5 / COM525
	DRI VE: \$	PROGRAM:	\$\$\$\$S\$SS	COMPONENT: \$\$
STAT.PART:	programming the stat	ic parts of	messages	
DYN. PART:	programing the dyna static parts alread	umic parts o y programmed	f messages v 1	wi th
COMPLETE DYN. PART:	programing dynam have been omitted,	ic parts (static part	of message s already p	es which may programed)
Program frc	m message no.:	#### (1-	2047)	
Number of m	essages to be loade	d: #### (ma ma	x. 70 for ax. 1000 for	static dynamic)
F 2 T. DYN. PART	F 3 F 4 COMPLETE DYN. PART	F 5	F6	F7 F8 EXIT

The field "program from message no." has the default 'l' or the number of the first message programmed. If you specify the "number of messages to **be loaded**" (default 'l2') you can select a specific number of messages.

- F1: change to 'generation of static text' mode; branch to 'STATIC PART' mask.
- F2: change to 'generation of dynamic text' mode; branch to 'DYN. PART' mask.
- F4: change to *completion of dynamic text' mode; branch to 'COMPLETE DYN. PART' mask.

B8576544-03

ar second a second a second a second second

COM 525

Field	Field type keys all<i>o</i>wed	Limit value	Alternatives
Program from message no.	numbers	1 - 2047	
Number of messages to be loaded 1)	numbers	1 - 70 or 1 - 1000	

والمتحج ومحافظة فتنقد والمناف والمتعاصية والمتعاوي والمتعاور

1) By 'messages to be loaded' the following is meant:

- -STAT. PART: number of messages to be read into the editing buffer,
- **DYN.** PART or COMPLETE **DYN.** PART: number of messages whose dynamic fields (or not yet programmed dynamic fields) are to be checked. Only the messages which have a dynamic field (or a not yet programed dynamic field) are read into the buffer.

,

5.3.1 Static Part of Messages

Before you can branch to the 'STAT. PART' mask, you must complete the following fields in the 'MESSAGES' mask:

- 'Program from message no.: #/## (1-2047)'
 - Read in areas with a maximum of 70 successive messages. Specify the first message number of the area to be read in.
- 'Number of messages to be loaded: ## (max. 70)'

Here you enter the number of messages to be read in from the already specified message number. The same number of already programmed messages will be read in as you have specified. There can be gaps between the message numbers. The messages will be read in in ascending order of the message numbers. If there are less messages present thin you specified then only those present will be read in.

You call the 'STATIC PART' mask with the function key **F1** (STAT. PART) in the 'MESSAGES' mask.

-> PROGRAM -> MESSAGE STATIC PART	S ->		SIMATIC S5 / COM525
	DRI VE:	\$ PROGRAM: \$\$S\$S\$	S\$ COMPONENT: \$\$
Mess attributes Gr No. B E CU A No.	page width: \$	Insert \$ OFF	Line no.: \$S\$\$ Column no.: \$.\$\$
# # # # # # # # # # # # # # # # # # # #	######################################	g editing window with messages in ascerdin fields for date, time a are displayed as outpu	####################################
FI F2 DELETE BLOCK	F 3 F 4 SEARCH/ REPLACE CURSOR	F5 F6 INSERT ON ENTER	F7 F8 EXIT

5.3.2 **Description** oft&? Editor

In COM 525 a text editor is available for the programming of messages. An editing window is displayed on the PG screen, 12 messages long and 80 characters wide.

When you call the **mask** it may appear as follows

.

1.40

۰.

- •It is empty if no message has been programmed in the selected area.
- •It contains the first 12 messages in the specified area, or all the programmed messages if fewer than 12 have been programmed.

86

..

~

If you reach the **margin** (**f** the editing window, the screen is automatically 'scrolled', i.e. the editing window is moved vertically by one message, **1.e.** if the upper edge is reached by one message upwards and if the lower edge is reached **ty** one message downwards.

Remember the following **points** when you edit the static parts:

- Edit the static texts in the way in which you want them printing out on the PT88 or **PT89**.
- Mark the field within the static text of a message line in which the dynamic parameter is to be printed out with a string of # or \$ to **reserve** the required number of characters.

!Important!

When marking a dynamic field use only # or \$, but not both.

Correct: ######## Wrong: ####\$\$\$\$

The # and \$ characters must not be used for other purposes, i.e. they must not be part of your message.

- Mark a maximum of one dynamic field in each message line.
- Assign the print attributes individually for each message if you wish. These then apply for **all** static parts of this message.

The attributes for the dynamic part are set when this is programmed (see 5.3.2).

- Allocate each message line to one of 16 message **groups if you** wish.

87

جرير والالمراص والمراج

B8576544-03

5.3.3 Editor Functions

The message editor makes available a number of functions to make editing easier. These are called up using function keys. Depending on which function is called, **submenus** are displayed with which you can start the subfunctions.

والاستهاد معرا والمرار والمرار

The following functions are available:

o **F1** in the 'STATIC PART' mask:

editor function DELETE; branch to a **submenu**:

F 1	F 2	F 3	F4	F 5	F 6	F 7	F 8
	WORD	LINE	BLOCK				ASORT

Words, lines or even whole blocks of your edited text can be deleted.

The following subfunctions **can be** started using the submenu displayed:

F2: WORD

The word on which the cursor is positioned is deleted.

F3: LINE

The line on which the cursor is positioned is deleted.

F4: BLOCK

The block is deleted which you markedby setting startandendmarkers.

88

ستان با

•F2 in the 'STAT. PART' mask:

editor function BLOCK, branch to a **submenu**:

FI	F_2	F 3	F 4	F 5	F 6	F 7	F 8
SET START	SE I END	СОРҮ	MOVE				ABORT

You can define one or more successive message lines as a block. You can then manipulate this block as a unit, i.e. copy the **whole** block or delete the whole block etc.

The following subfunctions can be started using the displayed submenu:

F1: SET START

The start of the block is marked. The start of the block is always the start of the message in which the cursor is currently located. The cursor itself need not be at the start of the message. The start of the block is marked to the left of the message line with the letter S on the screen.

F2: SET END

The end of the block is marked. The end of the block is always the end of the message on which the cursor is currently positioned. The cursor itself need not be at the end of the message. The **block** end is marked by the letter E to the left of the message line on the screen.

- !Important! You can only mark block end after you have marked the start of the block.
 - The start and end of the block can be **marked on** the same message. The block then simply **consists** of one message.

F3: COPY

The marked block as described above is copied in before the line in which the cursor is currently positioned. The numbers of the messages copied are set as follows:

- The first copied message receives the number of the first non-programmed message after the cursor position.
- The second message copied receives the number of the second non-programmed message after the cursor position etc.

The original block is not deleted. The block *now* exists **twice**.

If the cursor is positioned on a message which has not yet been programmed (empty except for the masked fields for date and/or time and/or **status**) the first message copied will receive the number of this non-programmed message.

F4: MOVE

The marked block as described above is moved to the position immediately before the line on which the cursor is positioned. The original block is deleted. The block only exists once. The message numbers will have changed as described under 'F3: COPY'.

Sector March 1977

•F3 in the 'STATIC PART' mask:

editor function SEARCH branch to a submenu.

1							
F 1	F 2	F 3	F 4	F 5	F 6	F 7	F 8
SEARCH	REPLACE						ABORT

You can search for characters or strings of characters in the edited text and replace **these** by other characters or strings of characters. The following subfunctions can be started from the submenu displayed:

F1: SEARCH

The 'search **line'** is displayed above the editing window. Here you can enter:

- A maximul 12-character long string which is to be searched for in the text.
- Whether you wish to search forwards or backwards from the cursor.

When you exit the search line, e.g. after it is completed, the editor searches for the string of characters you entered. If the string exists in the text the cursor is positioned on the character stringthathasbeen found. If the string is not contained in the text, an error message is displayed. The cursor is then positioned in the original position.

F2: REPLACE

Above the editing window the 'replace line' is displayed. Here you can enter:

- A maximum 12-character long string of characters to be searched for in the text,
- A maximum 12-character long string to be inserted in place of the string which has been searched for.
- Whether the string is to be searched forwards or backwards from the cursor.

- The number of times the string is to be searched for and replaced.
- Whether an operator acknowledgement is required before the string is replaced.

When you exit the 'replace line' e.g. after it is complete, the editor searches for the specified **string** of characters in the text and replace this with the second string of characters. Depending on what has been specified, this may occur several times.

F4 in the 'STATIC PART' mask:

editor function CURSOR, branch to a submenu:

F 1 LEFT	F 2 RIGHT	F3 INPUT	F4	F 5 MESSAGE	F 6	F ?	F 8
MARGIN	MARGIN	FIELDS		NUMBER	CONTINUE		ABORT

With these functions you can position the cursor at any point in the text.

The following subfunctions can be started using the displayed submenu:

F1: LEFT MARGIN

The cursor is positioned on the left margin of the line in which it was previously **located.**

F2: RIGHT MARGIN

The cursor is positioned on the right margin of the line in which it was previously located.

F3: INPUT FIELDS

A switchover is made so that yu can enter the message number, attributes and group number. The cursor is positioned on the message number in the line in

which **it was** previously located. You can now program only in these input fields. To return to the editing window press F4 (TO ED. WINDOW).

F5: MESSAGE NUMBER

An input field is displayed above the editing window in which you can enter the number of the message on which the cursor is to be positioned. The cursor is then positioned at the start of the specified message, if the message is in the editing buffer. Otherwise an error message is output.

F6: CONTINUE

A **switchover** is made to the second submenu. Here further cursor functions are made available:

		1	1	1	I = / I		
F 1 BLOCK	F 2 BLOCK	F3 TEXT		F 5	F 6	F /	F 8
START	END	START	END		CONTINUE		ABORT
	I	1		•	•		I

F1: BLOCK START

The cursor moves to the start of the block. This must already have been marked.

F2: BLOCK END

The cursor moves to the end of the block. This **must** alxeadyhavebeenmarked.

F3: TEXT START

The cursor **moves to the start** of the text in the editing buffer.

F4: **TEXT** END

The cursor movestotheendof **the text in the** editing buffer.

•F3 in the 'STATIC PART' mask:

editor function SEARCH branchtoasubmenu.

F 1	F 2	F 3	F4	F 5	F 6	F 7	F 8
SEARCH	REPLACE						ASORT

You can search for characters or strings of characters in the edited text and replace these by other characters or strings of characters. The following subfunctions can be started from the submenu displayed:

F1: SEARCH

The 'search **line'** is displayed above the editing window. Here you can enter:

- A **maximul** 12-character long string which is to be searched for in the text.

erix

- Whether you wish to search forwards or backwards from the cursor.

When you exit the search line, e.g. after it is completed, the editor searches for the string of characters you entered. If the string exists in the text the cursor is positioned on the character stringthathasbeen found. If the string is not contained in the text, an error message is displayed. The cursor is then positioned in the original position.

F2: REPLACE

Above the editing window the 'replace line' is displayed. Here you can enter:

- A maximum 12-character long string of characters to be searched for in the text,
- A maximum 12-character long string to be inserted in place of the string which has been searched for.
- Whether the string is to be searched forwar& or backwards from the cursor.



- The number of times the string is to be searched for and replaced.

والمحادث والمحاد المحاط الأرجاع والمراجع

- Whether an operator acknowledgement is required before the string is replaced.

When you exit the 'replace **line'** e.g. after it **is complete**, the editor searches for the specified string of characters in the text and replace this with the second string of characters. Depending on what has been specified, this may occur several times.

F4 in the 'STATIC PART' mask:

editor function CURSOR, branch to a submenu:

FI LEFT R MARGIN N	F 2 F3 Right input Margin Fields	F3 F4 F4 ME INPUT ME FIELDS NU	F5 F6 SSAGE MBER CONTINUE	F 7	F 8 ABORT

With these functions you can position the cursor at any point in the text.

The following subfunctions **can be** started using the displayed submenu:

F1: LEFT MARGIN

The cursor is positioned **on** the left margin of the line in which it was previously located.

F2: RIGHT MARGIN

The cursor is positioned on the right margin of the line in which it was previously located.

F3: INPUT FIELDS

A **switchover** is **made so that yu can** enter the message number, attributes and group number. The cursor is positioned on the message number in the line in

ويعام بيري والاستار فالمعام والاستار المالية

.

which it **was** previously located. You can now program only in these input fields. To return to the editing window press F4 (TO ED. **WINDOW**).

F5: MESSAGE NUMBER

An input field is displayed above the editing window in which you can enter the number of the message on which the cursor is to be positioned. The cursor is then positioned at the start of the specified message, if the message is in the editing buffer. Otherwise an error message is output.

F6: CONTINUE

A **switchover** is made to the second submenu. Here further cursor functions are made available:

F 1	F 2	F3	F 4	F 5	F 6	F 7	F 8
BLOCK	BLOCK	TEKT	I TEXT				
START	END	START	END		CONTINUE		ABORT
	1	1		I	1 1		I

F1: BLOCK START

The cursor moves to the start of the block. This must already have been marked.

F2: BLOCK END

The cursor moves to the end of the block. This must **already have been marked**.

F3: TEXT START

The cursor moves to hestart of the text in the editing buffer.

F4: **TEXT** END

The cursor moves to heen dofthetext in the editing buffer.

F6: CONTINUE You return to the first submenu.

•F5 in the 'STATIC **PART'** mask:

editor function INSERT (ON/OFF); press this function key to switch the editor from insert on to insert off and vice-versa, so that

edited characters are inserted in the existing text (INSERT ON)

- edited characters overwrite existing characters in the text (INSERT OFF)

Default setting: 'INSERT ON'

•F6 in the 'STATIC PART' mask:

saves the static parts of the message lines which have been edited or changed.

The total edited text (contents of the editing buffer with maximum 99 messages) is entered so that all edited or changed messages are transferred to the external storage.

•F8 in the 'STATIC PART' mask:

exits the editing **mask. You** return to the basic menu. If you have not already transferred the edited or changed messages to the external storage with F6 (**EXTER**) an acknowledgement **prompt will** be displayed.

.

Field	Field type keys allowed	Limit value	Alternatives
Page width	OUTPUT FIELD (from interpreter parameter assignment, see 3.6)	,	
Column no.	OUTPUT FIELD		
Mess. no.	numbers	1 - 2047	*
Attributes	Letters B=bold , E=expanded, C=compressed , U=underlined , A=acoustic signal		Ϋ, Ν
Group number	numbers	o - 15	

Description of the fields in the `STATIC PART mask (5.3.1)

CONTRACTOR & R.

F7 (HELP) can be used to make an entry in the fields marked '*'

- and the second se

5.3.4 The Dynamic Part of Messages

You must have already completed the following fields in the 'MESSAGES' mask:

والمتحد والمؤجري المحافظ فراريا والمحاور والمحاور والمحاور والمحاور والمحاور

• 'Program from message no.: ##### (1-2047)'

Specify the first message number of the area to be read in. You can read in areas with a maximum of 1000 successive messages.

• 'Number of messages to be loaded: ## (max. 1000)¹

Enter the number of messages to be read in starting from the already specified message number. The same number of already programmed messages are read in that you specified, however, only those which include a dynamic field marked with reserve characters, There can be gaps between the message numbers. The messages are read in in ascending order of the message numbers. If there are fewer messages present than you specified, then only those present are read in.

ж<u>ж</u>е 5

Call the 'DYN. PART' mask with the function key F2 (DYN. PART) in the 'MESSAGES' mask.

Depending on the type of function ' $\text{UP}/\text{DOWN}^{\text{r}}$ or 'JUMP the mask appears as follows:

UP/DOWN

	DRIVE: \$	PROGRAM: S	SSSSSSS COMPONENT: \$
Mess Attributes Gr No. BECUA No	No. mess.: \$\$\$\$ • Page width: \$\$\$		Column no.: \$\$\$
# # # # # # # # # # # # # # # # # # # #	######################################	window in whigh displayed. rently being yed inversely.	######################################
# # # # # # # # # # #	######################################	***********	######################################
############# DB No. \$\$\$	#####################################	######################################	Attributes (Y/N): bold print # expanded print # underlined # acoustic signal #
############# DB No. \$\$\$ F1 F2	#####################################	######################################	Attributes (Y/N): bold print # expanded print # underlined # acoustic signal # F7 F8

An editing window 7 lines long and 80 characters wide is displayed on the **PG** screen. It **appears** as follows after the mask has been called:

It displays the first seven messages which have already been programmed in the selected area or less than seven if a shorter area has been selected **or** if the **area** contains less than seven messages.

It inversely displays the dynamic field.

- Below the 7-line editing window, it also displays a **submask**. Here you can specify the number of characters and the print attributes for the current (inversely displayed) field.

!Important!

After you program a field, you must enter the values with F6 (ENTER). Otherwise **a warning** is displayed that you must acknowledge.

JUMP

L

•> PROGRAM - > MESSAG D Y N . P A R T	·S->	SIMATIC S5 / COM525
	DRIVE: \$ PROGRAM: \$\$S\$S	SS\$ COMPONENT: \$ \$
Mess AttributesGr No. B E C UA No	No. mess.: S\$\$\$. Page width: \$SS	Column no.: \$\$\$
# # # # # # # # # # # # # \ # # # # # #	######################################	######################################
Jump to message no.	: ####	
F1 F2 UP DOWN	F3 F4 F5 F6 JUMP	F7 F8 EXIT

With the following keys you can jump to fields in neighboring messages:

K1 (UP) to the field in the previous message K2 (DOWN) to the field in the next message,

With F5 (JUMP) you can also jump to any field in any message. **COM** 525 prompts you to enter the required message number and field number. Specify these and then press F5 (JUMP) again.

F1: the field in the message before is processed.

K2: the field in the message following is processed.

F5: the **submask 'JUMP'** is processed or the field in the specified message is processed,

F6: the dynamic part is saved on disk along with the static part

F8: abort: the data are not entered, and you return to the 'MESSAGES' mask.

99

بعاري والالتان فالماية

•

B8576544-03

1.8.1.1.1.1.1.Mg

COM 525

Field	Field type keys allowed	Limit value	Alternatives
No. mess.	OUTPUT FIELD	1 - 1000	
Page width	OUTPUT FIELD (from interpreter parameter assignment, see 3.6)	,	
Column no.	OUTPUT FIELD		
Mess. no.	numbers	1 - 2047	
Attributes	Letters B=bold, E=expanded, C=compressed, U=underlined, A=acoustic signal		Y, N *
Gr. no.	numbers	0 -15	
Bold, expanded, underlined, acou	letters stic signal		Y, N *
No. of chars.	numbers	01.0 - 80.9	
Jump to message no.	numbers	1 - 2047	

-

F7(HELP) can be used to make an entry in the fields marked '*'.

100

میری د د مراج اس • 1.1

an an an an an Ar

5.3.5 Completing the Dynamic Parts of Messages

Call the 'COMPLETE DYN. PART mask with the function key F4 (COMPLETE DYN. PART) in the 'MESSAGES mask.

> PROGRAM -> MESSAGE COMPLETE DY	S-> N. PART	SIMATIC S5 / COM525
	DRIVE: \$ PROGRAM:	\$sss%\$s\$ COMPONENT: \$S
Mess Attributes Gr No. B E C U A No	No. mess.: \$\$\$\$. Page width: \$\$S	Column no.: \$ss
# A # # # # # # # # # # # # #### # # # # # # # # Y # # # # # # # # # # # # # # # # # # #	######################################	######################################
	No. of chars. ##.#	Attributes (Y/N): bold print # expanded print # underlined # acoustic signal #
F1 F2 PREVIOUS FIELD	F3 F4 F5 NEXT FIELD ENT	F6 F7 F8 TER HELP EXIT

The layout and distribution of the screen as well as the submasks for the different types of data correspond to the 'DYN. PART' mask.

The mask is, however, different since not every dynamic field can be programmed but only the fields for which no dynamic part has as yet been programmed.

Only messages that have not had their dynamic fields programmed are read in and displayed in the editing window.

Once you have programmed such a field you must enter the input with F6 (ENTER) .

Once all the fields have been programmed, COM 525 indicates this with a message; it does not then indicate any further fields.

- **F1:** the previous non-programmed field is completed
- **F3**: the next non-programmed field is completed
- F6: the dynamic part is saved on disk along with the static part
- F8: abort:

the **data** is not entered, return to the **'MESSAGES'** mask B8576544-03

Field	Field type keys allowed	Limit value	Alternatives
Page width	OUTPUT FIELD (from interpreter parameter assignment see 3.6)	,	
Column no.	OUTPUT FIELD		
Mess, no.	numbers	1 - 2047	
Attributes	Letters B=bold, E=expanded, C=compressed, U=underlined, A=acoustic signal		Y, N *
Gr. no.	numbers	o -15	
Bold, expanded, underlined, aco	letters ustic signal		Y , N *
No. of chars.	numbers	01.0 - 80.9	

.

F7(HELP) can be used to make an entry in the fields marked '*'.

5.4 **PC JOB**

Call the 'PC JOB' mask with the function key F2 (PC job) in the 'PROGRAM' mask.

In the 'PC JOB' **mask**, you select which PC job you wish to **program**. The list of programmable PC jobs can be seen in the following mask. Depending on which selection is made the program will branch to different masks.

-> SELECTION -> PC JOB	PROGRAM ->				SIM	ATIC S5 / C	OM525
		DRI VE:	\$	PROGRAM:	\$ss\$\$ss\$	COMPONENT	: \$\$
SML: UPDATE SML:	Sequential Mess continuous outp updating of dyn	age List ut of the amic data	mess in t	ages (wi he frames	th dynamic s of the SM	parts) L.	
CML:	Current Message	List		unnontlu /	on the CDU		
PSL:	Process Status	List	yes c	th commo			
CL:	Chained List	ss status	es wi	in comme	nts.		
UPDATE GIB:	updating of grou	up inhibit	t bit	s.	us fists (PSL).	
NEW PAGE:	for form feed O	n PT88/PT	89 pr	inter.			
PAGE NO.=1:	set page number	ing on PT&	38/PT8	9 printe	r to 1.		
••		- (-	- 1	- / -	- -	~
PI FI UPOA	Z F3 TE	F 4	F	5	-6 F	/ F	8
SML SML		PSL	CL	CON	NTI NUE	EX	ΙT

F1: programming a PC job for output of the sequential message list:

branch to the $\ensuremath{^{\prime}SEQ}$ MESSAGE $\ensuremath{LIST'}$ mask

!Important! Before messages are programed for the first time (see 5.3), the **SML** (position and structure of date, time and status for all messages) must be specified.

- F2: programming a PC job for updating the frames of the sequential message list, branch to the 'UPDATE SML' mask (see 5.4,2),
- F3: programming a PC job for output of the current message list, branch to the 'CUR .MESSAGE LIST ' mask (see 5.4.3).
- F4: programming a PC job for output of a process status list branch to the 'PROCESS STATUS LIST' mask (see 5.4.4).
- F5: programmingg a PC job for output of several chained process
 status lists
 branch to the 'CHAINED LISTS ' mask
 (see 5.4.5).
- F6: output of a menu with further PC jobs:

ł				_ /			_	I
	UPDATE	F 2 NEW	F 3 PAGE NO.	r 4	F 5	F 6	F 7	F 8
	GIB	PAGE	= 1			CONTINUE		EXIT
I		•	•					•

- F1: programming a PC job for updating the group inhibit bits branch to the *UPDATE GIB' mask (see 5.4.6).
- F2: programming a Pc job to initiate a form feed branch to the 'NEW PAGE' mask (see 5.4.7).
- F3: programming a PC job to reset the page numbering branch to the 'RESET PAGE NO. ' mask (see 5.4.8).

•,

F6: return to the previous menu

F8: abort:
 return to the 'PROGRAM' mask

Note: If several PC jobs were programmed with the same job number only the last job stored can be executed. With F5 (check job number) in the SPECIAL FUNCTIONS mask, you can check whether job numbers have been used more than **once**.

5.4.1 Sequential Message List (SML)

Call the 'SEQ. MESSAGE **LIST'** mask with the function key **F1** (sML) in the '**PC** JOB' mask.

Before you can program a message, the global data valid for all messages must be specified as follows:

- Number of the $P\!C$ job, with which the CPU is to send the message records to the $C\!P$ 525.
- Name of the list header and/or list trailer which is to be printed out at the beginning or end of a page filled with messages.
- The position, status and structure of the date, time and message status for all messages. **These** specifications **determine** the position and length of the masked fields that cannot be changedwhenprogr**amming** the individual messages (see Section 5.3).

			DRIVE:	\$ F	PROGRAM:	\$\$\$\$\$\$\$\$	COMPONENT:	\$
List na With the message List h List th Form f Positio Date: Time:	ame: SE e job numbe es are outpu eader is f railer is f feed on PT88/ n and forma Colu	Q.MESSAGE r: ### it as even rame: #### pTay as t of date, Attrii mn B E L # # # # #	ts take pi ######### ######### start of time, sta butes Ord J A # # ## \$	iace. ist (Y/ atus is er ##\$##	N)# as belo s	∤ for all r	nessages:	
Statu: co go ac	s: ## ming: ing: knowledged:	# # # + # # -	Tex # # ### # # ###	t (encl ####### ########	ose in i ### ###	nverted cor	onas)	
E 1	F 2	F 3	F 4	F 5	5 F	6 F	? F 8	

a construction of general provide the state of the second s

1 Section of the

F3: generate a frame. You branch directly to the mask for programming frames (see 5.5). You can then program or change the frames (list header and/or list trailer) that are to be printed out with the sequential message list (you can program other frames at this point).

- F6: save the PC job on disk
- F8: abort:

the data is not entered; return to the 'PC JOB' mask

....

1

Field	Field type keys allowed	Limit value	Alternatives
Job no.	numbers	l - 199	
List header/ list trailer	element name (see 5,1)		*
Form feed	letters (default 'Y')		Υ, N *
Column	numbers	1- 136	
Attributes	letters B=bold E=expanded, U=underlined, A=acoustic signal		Υ, Ν *
Order - date	letters (default 'DD.MM.YY')		DD, MM, YY *
Order - time	letters (default 'HH:MM:SS')		HH,MM,SS
Text	any (defaults)		

F7 (HELP) can be used to make an entry in the fields marked $'\star'.$

109

•
A.

B8576544-03

5.4.2 Update SML

Call the *UPDATE SML' mask with the function key F2 (UPDATE SML) in the 'PC JOB' mask.

•> PROGRAM •> PC JO8 •>
U P D A T E SML SIMATIC S5 / COM525 DRIVE : \$ PROGRAM: \$\$\$\$SSS\$ COMPONENT: \$\$ Job name: UPDATE SML. With job no.: ### the **CP** 525 **will update** the dynamic data for the header/trailer of the **SML**. F 6 F 1 F 2 FЗ F 4 F 5 F 7 F 8 SAVE EXI T

F6: the PC job is savec on the storage medium

F8: abort:

return to the 'PC JOB' mask

Field	Field type keys allowed	Limit	value	Alternatives
Job no.	numbers	1 -	199	

110

مريز والمرام والمرام والم

•

5.4.3 Current Message List (CML)

Call the *CURRENT MESSAGE LIST' mask with the function key F3 (CML) in the 'PC JOB' mask.

SIMATIC S5 / COM525 -> PROGRAM -> PC JOB -> CURRENT MESSAGE LIST DRIVE: \$ PROGRAM: \$ss\$ss\$\$ COMPONENT: S\$ List name: CUR.MESSAGE With the job no.: ### the output of the current message list is started. Output of messages possible while CML is being output Form feed on P188/P189 at start of list (Y/N) # (Y/N) #F 5 F 6 F 7 F 4 F 1 F 2 F 3 F 8 FRAME SAVE HELP EXI T

F3: generate a frame.

You branch directly to the mask for programming frames (see 5.5). You can then program or change the frames (list header and/or list trailer) which are to be printed out with the sequential message list (you can program other frames at this point).

- F6: save the PC job on disk.
- F8: abort:

the data is not entered; return to the 'PC JOB' mask

Field	Field type keys allowed	Limit value	Alternatives
Job no.	numbers	1 - 199	
List header/ list trailer	element name (see 5.1)		*
Output of messages. , .	letters (default 'Y')		¥, N *
Form feed	letters (default 'Y')		Y, N *

F7 (HELP) can be used to make an entry in the fields marked '*'.

5.4.4 Process status List (PSL)

Process status lists include the following:

- static parts, i.e. fixed texts
'dynamic parts, i.e. variable data, which can be distributed at
any point within the static parts

and the second second

Static parts

You determine the structure and contents of the static parts when programming on the PG with the text editor. They can then no longer be changed on **the** CP 525,

Dynamic Parts

When programming the static parts you specify where the dynamic parts are to be inserted as follows:

- per list line maximum 40 dynamic fields
- per PSL maximum; approximately 1400 dynamic fields (dependent on data type)

The number of **reserve** characters inserted in the static text for a **dynamic** field is used as the default for the **number** of characters when programming**g the dynamic part**.

The following information can be freely programmed for the dynamic data:

- source (address in the data block of the CPU)
- data type (bit, byte, characters, 16-bit FXP etc.)
- number of characters in the list
- printer attributes (e.g. bold, expanded etc.)
- position in list

Process status lists and frames are programmed in two steps:

- 1. First the static parts are programmed. Press Fl (STAT. PART) in the 'PROCESS STATUS LIST'tocallthe'STATIC PART' mask.
- 2. The dynamic parts for the static parts which have already been programmed are specified. Press F2 (DYN. PART) in the 'PROCESS STATUS LIST' to branch to the 'DYN. PART' mask.

In addition, you can call the 'COMPLETE DYN. PART' mask with F4 (COMPLETE DYN. PART) in the 'PROCESS STATUS LIST' mask. When you use this function, dynamic fields that have not yet been **pro-**grammed within the already programed static parts are searched for.

Call the 'PROCESS STATUS LIST' mask with the function key F4 (PSL) in the 'PC JOB' mask,

-> PROGRAM -> PC JOB -> SIMATIC S5 / COM525 PROCESS STATUS LIST \$ PROGRAM : \$\$\$\$\$ COMPONENT: \$S DRI VE With the job no.: #### the output of the process status list with the name selected above will be started. The dynamic data are on CPU no.: # in data **block** DB no.: ### Output of messages possible **while** the **PSL** is being output Form feed on **PT88/PT89** at start of list (Y/N) # (Y/N) # F 1 F 2 F 3 F 4 F 8 F 5 F 6 F 7 COMPLETE STAT. DYN. PART PART FRAME DYN.PART SAVE HFLP EXIT

- F1: change to the mode 'generating the static part of a process status list'; branch to the 'STATIC PART' mask (see a) in this section)
- F2: change to the mode "generating the dynamic part of a process status list'; branch to the 'DYN. PART' mask (see b) in this section)
- F3: change to the 'programming frames' mode; branch to the 'FRAME' mask (see 5.5)

B8576544-03

a an an an an an Alain Alain an an an an an an an Alain an an Alain.

F4: change to the mode 'completing the dynamic part of a process
status list';
branch to the 'COMPLETE DYN. PART' mask;
(see c) in this section)

F6: store the PC job on disk

F8: abort:

د ا بر بر ا بر ا بر ا بر ا بر ا مر ا م ا ا data is not entered; return to the 'PC JOB' mask.

ىقى مەرىپ يەرىپ يېرىكى بىرىكى تەرىپىرىكى تەرىپىرىكى تەرىپىرىكى تەرىپىرىكى تەرىپىرىكى تەرىپىرىكى تەرىپىرىكى تەر

Field	Field Sype keys allowed	Limit value	Alternatives
List name	element name (see 5.1)		*
Name to be stored	element name (see 5.1)		
Job no,	numbers	1 - 199	
CPU no.	numbers	1 - 4	*
DB no.	numbers	<pre>1 - 255 (the DB numbers 1 and 2 are reserved for the system)</pre>	
List header/ list trailer	element name (see 5.1)		*
Output of messages	letters (defatlt 'Y')		Ү, N *
Form feed	letters (default 'Y')		¥, N *

F7(HELP) can be used to make an entry in the fields marked '*'.

ţ,л.

a) Static part of a process status list

Call the 'STATIC PART' mask with the function key Fl (STAT. PART) in the 'PROCESS STATUS LIST' mask.

ne: \$\$\$\$\$SS th: \$S\$ ###################################	S\$\$\$\$ In 	sert OFF ##################################	Line no. Column no.	: \$\$\$: \$\$\$
\		<i></i>		
				*######## ########
######################################	* ################	######################################	######################################	*****
list lines	which have	al ready be	een ##### #	
generated. #			####### #######	**************************************
*********		******	****	******
**************************************	• ~~~~ ~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	******	*******	**************************************
######################################		********	\############ ~~~~~~~~	*******
F 4	F 5	F6	F7	F 8
1/	I NSERT	ENTER		
	12 line la list lines generated. ####################################	12 line long editing list lines which have generated. f	12 line long editing window with list lines which have already be generated. Image: state s	12 line long editing window with ###### list lines which have already been ##################################

An editing window that is 12 lines long and 80 characters wide ${f is}$ displayed on the screen of the programmer.

When the mask is displayed, it can appear as follows:

- it is empty, if a new **PSL** is edited
- it **contains** the first **12** lines of a **PSL** which has already been programmed or less if the **PSL** is less than 12 lines long

You can edit a maximum of 99 list lines each with a maximum of 136 characters.

- 24

If the cursor reaches the margin of the editing window the screen is automatically 'scrolled'; i.e. the editing window is scrolled vertically, one message upwards if the lower **margin** has been reached and one message downwards if the upper margin is reached.

The same thing happens if the *cursor* reaches the right or left margin of the editing window (scrolls one word to the right or one word to the left).

You edit the static texts as you wish to have them output on the PT88 or PT89.

•Fields within the static text in which you want *to* print out dynamic data have their positions and lengths marked with the characters # or \$ to reserve space for them.

!Important!

- One field must contain only # or \$ but not both

Correct: ######## Incorrect: ####\$\$\$\$

- Dynamic fields which are immediately adjacent to each other (i.e. there is no gap between them) must have different reserve characters, either # or \$.

- The characters #and \$ must not be used for other purposes; i.e. they cannot be part of a PSL.
- A maximum of 40 dynamic fields can be marked in each line; however, there must not be more than approximately 1400 dynamic fields in any PSL. The maximum value depends on the type of variables pro-grammed for the fields, If status variables are used, the number of dynamic fields which can be programmed is considerably reduced since these can involve up to eight statuses. Such a variable is pro-grammed eight times and therefore occupies eight times more memory space.
- The **print** attributes can be set individually for each line. They then apply for all static parts in that line.

119

• رز د د ـ • • • •

B8576544-03

- The attributes for the dynamic parts are set when they are programmed (see b) in this section).
- Each list line can be assigned to one of 16 message groups.

Editor functions

A text editor is available in COM 525 for programming process status lists.

The editor makes available a number of functions in order to make editing easier. These are called up using function keys. Depending on which function is called, submenus will be displayed with which you can start the subfunctions.

The **followin**g functions are available:

•F1 in the 'STATIC PART' mask:

editor function DELETE, branch to a submenu:

FΙ	F 2	F 3	F 4	F 5	F 6	F 7	F 8
	WORD	LINE	BLOCK				ABORT
							•

Words, lines or even whole blocks of **your** edited text can be **deleted.**

The following subfunctions can be started using the submenu displayed:

F2: WORD The word on which the cursor is positioned is deleted.

F3: LINE

The line $\boldsymbol{\mathfrak{I}}\!n$ which the cursor is positioned is deleted.

F4: BLOCK

The block that you marked by setting start and end markers is deleted.

•F2 in the 'STATIC PART' mask:

editor function **BLOCK**, branch to a submenu:

F 1	F 2 SET	F 3	F 4	F 5	F 6	F 7	F 8
START	END	COPY	MOVE				ABORT

You can define one or more successive message lines as a block. You **can** then manipulate this block as a unit, i.e. copy the whole block or delete the whole block etc.

The **followin**g subfunctions can be started using the displayed submenu:

F1: SET START

The start of the block is marked. The start of the block is always the start of the text line in which the cursor is currently located. The cursor itself need not be at the start of the line. The start of the block is marked to the left of the text line with the letter S on the screen,

F2: SET END

The end of the block is marked. The end of the block is always the end of the text line on which the cursor is currently positioned. The cursor itself need not be at the end of the line. The block end is marked by the letter E to the left of the text line on the screen.

!Important! - You can only mark block end after you have marked the start of the block.

- The start and end of the block can be marked on the same text line. The block then simply consists of one line.

F3: COPY

The block marked as described above is copied in before the line in which the cursor is currently positioned. The original block is not deleted. The block now exists **twice**.

F4: MOVE

The block marked as described above is moved to the position immediately before the line on which the cursor is positioned. The original block is deleted, The block only exists once.

B8576544-03

COM 525

•F3 in the 'STATIC PART' mask: editor function SEARCH branch to a submenu:

F 1	F 2	F 3	F 4	F 5	F 6	F 7	F 8
SEARCH	REPLACE						ABORT

You can search for characters or strings of characters in the edited text and replace these by other characters or strings of characters. The following subfunctions can be started from the **submenu** displayed:

F1: SEARCH

The `search line' is displayed above the editing window. Here **you can** enter:

- a maximum 12-character long string which is to be searched for in the text
- whether you wish to search forwards or backwards from **the** cursor

When you exit the search line, e.g. after it is completed, the editor searches for the string of characters entered. If the string exists in the text, the cursor will be positioned after the character string which has been found. If the string is not contained in the text an error message is displayed. The cursor is then positioned in the original position.

F2: REPLACE

Above the editing window the `replace **line'** is displayed. Here you can enter:

- a max. 12-character long string **of** characters to be searched for in the text,
- a max. 12-character long string to be inserted in place of the string which has been searched for.
- Whether the string is to be searched forwards or backwards from the cursor.

B8576544-03

- How often the string is to be searched for and replaced.
- Whether an operator **acknowledgement** is required before the string is replaced.

When you exit the 'replace line' e.g. after it is complete, the editor searches for the specified string of characters in the text and **replacees** this with the second string of characters. Depending **on** what has been specified, this may take place several times.

•F4 in the 'STATIC **PART'** mask:

editor function CURSOR, branch to a submenu:

ΕI	F 2	F 3	F 4	F 5	F 6	F 7	F 8
LEFT	RIGHT	INPUT		LINE			
MARGIN	MARGIN	FIELDS		NUMBER	CONTINUE		ABORT
		I		1			I

With these functions you can position the cursor at any point in the text.

The following subfunctions can be started using the displayed **submenu:**

F1: LEFT MARGIN

The cursor is positioned on **the** left margin of the **line** in which it was previously located.

F2: RIGHT MARGIN

The cursor is **positioned** on the right margin of the line in which it was preciously located.

1.24

B8576544-03

F3: INPUT FIELDS

A switchover is made so that you can enter the attributes and group number. The cursor is positioned on the first attribute in the line in which it was previously located. You can now program only in these input fields. To return to the editing window press F4 (TO ED. WINDOW).

and the second second second second second

F5: LINE NUMBER

An input field is displayed above the editing window in which the number of the text line can be entered on which the cursor is to positioned. The cursor is then positioned at the start of the specified line, providing this is in the editing buffer. Otherwise an error message is output.

F6: CONTINUE

A switchover is made to the second submenu. Here further cursor functions are made available:

F 1	F 2	F 3		F 5	F 6	F 7	F 8
START	END	START	END		CONTINUE		ABORT
	•	•	•				•

F1: BLOCK START

The cursor moves to the start of the block. This must alreadyhave been marked.

F2: BLOCK END

The cursor **moves** to the end of the **block. This must** already have been marked.

F3: TEXT START

·. ·.

The cursor moves to the start of the process status list.

F4: m END
The cursor moves to the end of the process status
list.

10 mil 11

- F6: CONTINUE You return to the first submenu.
- •F5 in the 'STATIC PART' mask:

editor function INSERT (ON/OFF); press this function key to switch the editor from insert on to insert off and vice-versa, so that

- edited characters are inserted in the existing text (INSERT ON)
- edited characters overwrite existing characters in the text (INSERT **OFF**)

Default setting: 'INSERT ON'

م م به م بن بر ام م ه ه

- •F6 in the 'STATIC **PART'** mask: saves the static parts of the text lines which have been edited or changed. The total edited text (static part of the edited **PSL)** is entered. This means that all **PSLs** are transferred to the external storage.
- •F8 in the 'STATIC PART' risk: exits the editing mask. You return to the basic menu. If you have not already transferred the edited PSL to the external storage with F6 (ENDER), an acknowledgement prompt will be displayed.

9. L

Field	Field type keys allowed	Limit value	Alternatives
Attributes	letters B=bold, E=expanded, C=compressed, U=underlined, A=acoustic signal		Y, N *
Gr no.	numbers	0 -15	
Line no.	OUTPUT FIELD		
Page width	OUTPUT FIELD (from interprete parameter assignment see 3.6)	r -	
Column no.	OUTPUT FIELD		

Description of the fields in the 'STATIC PART' mask

F7 (HELP) can be used to make an entry in the fields marked `*'.

ь.

b) Dynamic part of a process status list

You call the 'DYN. PART' mask with the function key F2 (DYN. PART) in the 'PROCESS STATUS LIST' mask.

An editing window is displayed on the screen of **the** PG. This is 7 lines long and 80 characters wide and appears as follows after the mask has been called.

It displays the first seven lines of a PSL which has already been programed or less if the PSL has less than seven lines.

The first dynamic field is displayed inversely.

Below the seven-line editing window, a **submask** is **displayed that** is laid out depending on the type of dynamic variable. There are various **submasks** for:

- process variable

(always displayed **if** there are fields which have not yet been programmed)

- date/time
- (source is the **CP** 525 clock) date/time
- (source is the &tab Lock)
- process stares
- page number

!Important!When you have completely **programme**d a field you must enter the values with F6 (ENTER). Otherwise **a warning is dis**played which you must acknowledge.

You can jump in any direction to any neighbouring field with

F1 (UP) to the first field in the line before

F2 (DOWN) to the first field in the next line

F3 (IEFT) to the next field to the left in the same line or to the last field in the **previous** line

F4 (RIGHT) to the next field to the right inthesamelineor to the firstfield in the next line

مري ۾ جو جا جا ماند ه

If there are no **more** fields in the direction you selected, this is indicated by a message.

By using F5 (JUMP) you can also jump to any field in any line. COM 525 then prompts you to specify the required line number and field number. Input these numbers and press F5 (JUMP) again,

Depending on the type of dynamic variables you selected the mask appears as one of the following masks:

PROCESS VARIABLE

1-

≻PCJOB YN.F	•> PSL •> • A R T					SIMATIC	S5 / COM5
		I	DRIVE: \$	B PROGRAM	M: \$\$\$\$\$	\$\$ COMPO	NENT: \$\$
Attribut B E C	es GrElen U A No.	ent name: page widt	\$\$\$\$\$\$\$\$ h: \$\$\$	\$\$\$		Line no Columm n	0.: \$\$\$ 0.: \$\$\$
# # # # # # # # # # # # # # # # # # # # # # # # # # # # # # # # # # # #	# ### ### # ### ### # # # # # # # # # # # # # # # # # # # # # # # # # # # # # #	######################################	line long list lines field cu ith is disp	window are displ urrently b played inve	in which r ayed. eing wor ersely	######################################	
Type of	dynamic \	variables: P	ROCESS V	ARIABLE	At	tributes (Y/N): #
CPU No DB No. Address). \$ \$\$\$ ## ###.#	¥	Form No.o	nat ###### of chars. #	#### ex ##.# un ac	panded pr derlined oustic sigi	int # # nal #
F 1	F 2	F 3	F4	F 5	F 6	F 7	F 8
UP	DOWN	LEFT	RIGHT	JUMP	ENTER	HELP	EXIT

DATE / TIME

You can select the location from which the date and time are to be fetched:

** ***

- the data block - the internal **CP** 525 clock

Depending on the selected source, the mask appears as **below**:

DATE / TIME from the **CP** 525 clock **(example** for date)

-> PC JOB D Y N. P	-> PC JOB -> PSL -> SIMATIC S5 / COM525 D Y N. P A R T								
			DRIVE:	\$ I	PROGRAM	1: \$\$\$\$s\$\$	\$ COMPO	ONENT: \$S	
Attributes Gr Element name: \$\$\$\$\$\$\$Line no.: \$\$\$B E C U A No. page width: \$\$\$Column no.: \$\$\$									
<pre># # # # ## # # # # ## # # # # # ## # # # # # # ## # # # # # # ## # # # # # # ## # # # # # # ## # # # # # # ## # # # # # # ## # # # # # # ## # # # # # # ######</pre>									
Type o DATE is	Type of dynamic variables: DATE Attributes (Y/N): DATE is fetched from CP 525 - CLOCK bold print # Order # # # underlined # acoustic signal #								
F 1 UP	F 2 DCUN	F 3 LEFT	F 4 RIGHT	JL	F5 JMP	F 6 ENTER	F 7 HELP	F 8 Exit	

Г

DATE / **TIME** from the data block (example for time)

			DRI VE:	\$ PROGRAM	l: \$\$\$\$\$ss	\$ COMPO)NENT: \$\$
Attribu [.] B E C l	tes Gr El J A No. p	ement name bage width	e: \$S\$SSS\$ n: \$\$\$	\$\$\$\$		Line no Column n	0.: \$\$\$ 0.: \$\$\$
# # # # # # # # # # # # # # # # # # # # # # # # # # # # # # # #	# # # # # # # # # # # # # # # # # # # #	######################################	######################################	######################################	which ma ayed. being worl	######################################	\\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\
# # # #	### #	##W### W !\ ####################################	3 h is disp ####################################	layed inve ####################################	ersel y #############	##### ###########	******
# # # # Type of TIME is CPU No DB No . Address	# # # # # f dynamic s fetched c. \$ \$\$\$ s DW ###	vari abl es:	TIME DATA BLOC	layed inve ####################################	ersely ####################################	###### ributes (d print banded pr erlined ustic sig	######################################
# # # # # Type of TIME is CPU No DB No . Address F?	# # # # # f dynamic s fetched c. \$ \$\$\$ s DW ### F 2	vari abl es: from	TIME DATA BLOC Order # #	ayed inve ####################################	Att bold exp und aco	ributes (d print anded pr erlined ustic sig	Y/N): # int # mal #

.

.

PROCESS STATUS

In the 'PROCESS STATUS' mask the function key F6 is renamed 'PROCESS sTATus'.

			DRI VE:	\$ PROGRAM	: \$\$\$\$\$ss\$	COMPON	IENT: \$\$
Attribu B E C	tes Gr Ele UA No. p	ment name: age width:	\$S\$S\$SSS \$. S\$	S\$S\$		Line no. Column no	: \$\$\$ 0.: \$\$\$
# # # # # # # # # # # # # # # # # # # #	# ### # # # # #	######################################	//////////////////////////////////////	g window s are disp currently b played inve ####################################	/#####################################	(#####################################	
Type o The pr	f dynamic ocess stat	variables: us is prog	PROCESS S	TATUS ng key F6.	Attr bol c expa unde acou	ibutes (Y I print anded pri erlined ustic sigr	'/N): # nt # # nal #
F 1	F 2	F 3	F 4	F 5	F 6 PROCESS	F7	F 8

F6: branch to the 'PROCESS STATUS VAR.' mask

1 -

Process status - programming the text

You call the 'PROCESS STATUS VAR' mask by entering ' PROCESS STATUS in the 'type of dynamic variables field and with the function key F6 (PROCESS STATUS) in the 'DYN. PART mask.

The length of the input fields below the ' text corresponds to the number of characters reserved with \$ or #, that you programed for this dynamic field in the static part of the PSL.

-> PROCESS STATUS LOG -> DYN. PART -> PROCESS STATUS VAR.				SIM	ATIC S5 / COM525		
			DRIVE:	\$	PROGRAM :	\$\$\$\$\$\$\$\$	COMPONENT: \$ \$
Elemen	t name: \$\$\$	\$\$\$\$\$\$\$\$\$					
CPU n	o. \$	DB no. \$\$\$	Addı	ess:	## ###		
Stat. o 1 2 3 4 5 6 7	Attributes B E U A # # # # # # # #	Text ####################################	######################################	###### ###### ###### ###### ###### #####	/############ ########################	****************** *******************	######################################
F 1	F 2	2 F 3	F 4		F 5 E	F 6 F NTER HE	7 F 8 LP EXIT

F8: abort:

the data is not entered; return to the 'DYN. PART' mask.

PACE UMBER

The page number is entered in this field when the lists are output by the $\ensuremath{\text{CP}}$ 525,

ти, г <i>и</i> к I	DRIVE: \$ PROGRAM	: \$\$\$\$\$\$	\$\$\$ COMPOI	NENT: \$
Attributes Gr E B E C U A No.	Element name: \$SS\$\$SSS\$ page width: \$\$\$		Line no. Column no	: \$\$\$\$ D.: \$\$\$
# #	#######7 line long window in ####################################	whi ch yed.	max. ###### ###### #######	
# # # # # # # # # # # # # # # # # # # #	######################################	eing wo ~sely ########	orked ###### ##############################	********** ************
# # # # # # # # # # # # # # Type of dynami	The field currently b ####################################	eing wo rsely A A b e u a	orked ###### ##############################	///N): # nt # # nal #
<pre># # # # # # # # # # # # # # Type of dynami F1 F2</pre>	F3 F4 F5	F 6	brked ###### ##############################	////): # nt # nal #

and the second second

,

Jump to any field in a PSL

Press F5 (JUMP) in the 'DYN. PART' mask.

			DRIVE:	\$ PROGR	AM : \$\$\$	\$ \$\$\$ CON	IPONENT: \$\$
Attribu BECUA	tes Gr E No.	lement name page wid	: \$S\$\$\$S\$: th: \$\$\$	\$SS. S\$		Line r Column	no.: \$\$\$\$ no.: \$\$\$
##### ###### ###### #####	## ## ## ## ##	######### ######## ####### ##########	Hine long list line The field	g window s are dis currently	//////////////////////////////////////	######################################	
# # # # # # # #	# # # # # #	######################################	/ith is di ###############	spl ayed in	versel y	#### #############	
##### ####	# # # # # # o line ##	######################################	/ith is di ####################################	splayed in (######### ##	versel y ###########	#### Attributes bold print expanded p underlined acoustic si	(Y/N): # print # ignal #
#### Jump to	# # # # # # D line ##	######################################	vith is di ####################################	splayed in	versel y	Attributes bold print expanded p underlined acoustic si	(Y/N): # print # ignal # F 8

~

F5: a jump is made to the field specified in the fields 'line' and 'field'.

If this field does not exist (line and/or field not present) an error message is displayed.

significance of the function keys in the 'DYN. PART' masks

M: the first field of the previous line is processed.

- F2: the first field of the next line is processed.
- **F3**: the field to the left of the current field is processed (this can also be the last field of the line before).
- F4: the field to the right of the current field is processed (this can **also be** the first field of the following line).
- F5: processing of the submask 'jump to any field' or the field specified in the specified line will be processed.
- F6: the dynamic part is entered, but saved on disk with the static part only if you press F6 (SAVE) in the 'PROCESS STATUS LIST' mask or if you program a process status variable. Branch to the 'PROCESS STATUS VAR. ' mask.
- F8: abort: the data is not entered, return to the `PROCESS STATUS LIST' mask.

136

جهيرة الأناف فالمريد فرق

.

Field	Field type keys allowed	Limit value	Alternatives
Attributes	letters (default 'N')		Y , N
Gr no.	numbers	o - 15	
Element name	OUTPUT FIELD		
Line no.	OUTPUT FIELD		
Page width	OUTPUT FIELD (from interprete parameter assign see 3 .6)	er nment	
Column no.	OUTPUT FIELD		
CPU no.	OUTPUT FIELD	1 - 4	
Format	letters (default '16BIT FXP')	(see Table 2)	
DB no.	OUTPUT FIELD	1 - 255	
No. of chars	numbers (default: no . of reserve characters)	01.0 - 80.9	

Description of the fields in the 'DYN. PART' masks and the 'PROCESS STATUS VAR. ' mask.

F7 (HELP) can be used to make an entry in the fields marked $`\star'.$

areas of

ē\$

Field	Field type keys allowed	Limit	value	Alternatives
Address = memory access mode and DB address	letters (default 'DW') numbers	000.00 -	255.15	DW, DL, DR, DD, BI *
bold/expanded underlined/ acoustic signal	letters (default `N')			Υ, Ν *
DATE/TIME is fetched from	letters (default 'CP 525 CLOCK')			CP 525 CLOCK DATA BLOCK *
Order	letters (default 'DMY' or 'HMS')			D, M, Y, or H, M, S
Type of dyn. variables	letters (default 'PROCESS VARIABLE	')		DATE , TIME , PROCESS VARIABLE, PROCESS STATUS, PAGE NUMBER *

Text any

F? (HELP) can be used to make an entry in the fields marked '*'.

B8576544-03

Field	Field type keys allowed	Limit value	Alternatives
Jump to line	numbers	1 - 99 (last list line)
Field	numbers	1 - 40	

Note on the defaults: if the **submask** of the selected type has already been edited this data will be entered as defaults. An **exception** to this is the 'number of characters', which always has the number of reserve characters.

Note on 'number of characters' in the 'PAGE NUMBER' mask: if a number of characters greater than four is selected, the **maximum** four-digit page number is output right-justified in a field with the width 'number of characters'.

COM 525

X

Format	Si gni fi cance	Range of values	For memory access mode
31 NARY	1 bit binary	0.1	BI
CHAR	Character	ASCI I	DR, DL
BYTE	1 byte binary	0 to 255	DR, OL
16BIT BCD	16 bit in BCD cede	-999 to +999	DW
16BIT FXT	16 bit in fixed point format	-32768 to +32767	DW
32BIT BCD	32 bit in BCD code	-99999999 to +99999999	DD
32BIT FXT	32 bit in fixed point format	-2 ³¹ to 2 ³¹ -1	DD
32BIT FLP	32 bit in floating point format	$-9999999 * 10^{9} + 10^{-9}$ to $-9999999' = 10^{-9}$ to $+9999999 * 10^{-9} + 9^{-9}$ $+9999999 * 10^{-9}$	OD
TIMER	16 bit BCD code + time base (example: 154.2)	o to +999 o to 3	DW
COUNTER	16 bit BCD code	o to +999	DW

· •• •

Table 2: Format (all S5 formats can be specified)

When you enter the data with F6 (ENTER), COM 525 checks whether the format specified is compatible with the mode of memory access.

1 -

c) Completing the dynamic parts of a process status list

Call the 'COMPLETE **DYN. PART'** mask with the function key F4 (COMPLETE **DYN.** PART) in the 'PROCESS STATUS LIST' mask.

•> PC JOB •> PSL •> Complete dyn.	. PART		S	IMATIC S5 / COM525
	DRI VE:	\$ PROGRAM:	SSS\$S\$S\$	COMPONENT: S\$
Attributes Gr Element B E C U A No. pag	: name: \$\$\$\$\$\$\$ e width: \$\$\$	SSS\$	L Co	ine no.: \$\$\$ blumn no.: \$.\$\$
# # # # # ## ##### # # # # # # # ##### # # # # # # # # ###### # # # # # # # # ####################################	//////////////////////////////////////	ng window i are displa currently b blayed inver	n which max yed. eing worked sely	**************************************
Type of dynamic vari DATE is fetched fra	ables: DATE an CP 525 CL Order # #	.0CK 4 #	Attrik bold p expan underl acoust	outes (Y/N): orint # ded print # ined # tic signal #
F1 F2 PREVIOUS FIELD	F3 F4 Next Field	F 5	F 6 ENTER H	F7 F8 HELP EXIT

The layout and distribution of the screen as well as the **submasks** for the various data types correspond to the 'DYN. PART' mask for a process status list (see **b**).

There is, however, one major difference - you cannot **program** every dynamic field, but only those for which no dynamic part has yet been programmed.

The dynamic fields which have not yet been programmed are made available for programming automatically (displayed inversely) and you can position the cursor on these fields with

- F1 (PREVIOUS FIELD)

- F3 (NEXT FIELD)

and a second second stand

When you program one of these fields you must enter the specification with F6 (ENTER). Only then can you have the next field made available.

When all these fields are programmed, COM 525 outputs a confirming message. It then makes no further fields available.

F1: the last non-programmed field is processed

F3: the next non-programmed field is processed

- 159- -

1999 - 1997 - 1988 - 1988 - 1988 - 1988 - 1988 - 1988 - 1988 - 1988 - 1988 - 1988 - 1988 - 1988 - 1988 - 1988 -

5.4.5 Chained List (CL)

Call the 'CHAINED **LISTS'** mask with the function key F4 (CL) in the 'PC JOB' mask. You can have a **maximum** of eight chained process lists output on the printer.

If a PSL was stored under a specified name, COM 525 displays the following programmed parameters for this PSL in the column next to the PSL name.

- Name of the list header (if no list header has been specified the field remains empty).
- Name of the list trailer (if no list trailer has been specified the field remains empty).
- Number of the PC job with which this **PSL** is to be started by the CPU (if no job number has been programmed **for** this **PSL** the field remains empty).

The only lines in the mask that are completed are those for which you entered the name of a PSL. All other lines remain empty.

You can read in existing **chained** lists. The name of the chained list must be specified in the 'list name' field. You can modify this chained list and store it again under a different name. This new name must be entered in the field 'name to be stored'.

The 'name to be stored' field always has as a default the name specified in the 'list name' field.

• • •

	DRI VE :	\$ PROGRAM: \$	\$\$\$S\$\$\$	COMPONENT:	\$5
Name of list:	############	Name to bestow	ed: ######	#######	
With the job the following c	no.: ### hained PSL's will be	e output.			
PSL name	List header	List trailer	Job no.		
######################################	\$55\$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$\$	\$\$\$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$ \$\$\$\$	\$\$\$ \$\$\$ \$\$\$ \$\$\$ \$\$\$ \$\$\$ \$\$\$ \$\$\$		
F1 F2	F 3 F 4 PSL	F5 F6	F 7 /E HELP	F 8 EXIT	

F3: generate a PSL.

You branch directly to the mask for programming a $\ensuremath{\text{PSL}}$ (see 5.4.4).

This allows you to program or modify process status lists which you wish to have output as a chained list. Exiting this mask you return to the 'chained lists' mask.

- **F6**: save the PC job on the storage medium.
- **F8:** abort:

the data is not entered; return to the 'PC JOB' mask.

Field	Field type keys allowed	Limit value	Alternatives
Name of list	element name (see 5.1)		*
Name to be stored	element name (see 5.1)		
Job no.	numbers	1 - 199	
PSL name	element name (see 5.1)		*
List header	OUTPUT FIELD (as for corres. PSL)		
List trailer	OUTPUT FIELD (as for corres. PSL)		
Job no.	OUTPUT FIELD (as for corres. PSL)		

F7 (HELP) can be used to make an entry in the fields marked '*'.

145

العول و الا و الا مو و المادية. 19
5.4.6 Updating the Group Inhibit Bits

Call the 'UPDATE **GIB'** mask with the function key F6 (CONTINUE) and **F1** (UPDATE **GIB)** in the '**PC** JOB' mask.

-> PROGRAM > PC JOB > U P D A T E G I B				5	SIMATIC S	5 / COM525
	DRI VE:	\$	PROGRAM:	\$\$\$\$\$\$	\$\$ COMF	ONENT: SS
Job name: ########	;#####					
With this job no. the data word will be transferred from the data block on the CPU to the CP 525.	Job CPU a DB	no.: ## no.: no.: ## DW : ##	+# # ## ##			
The 16 bits of this data by the CP 525 as group in	word wil hibit bit	be in s.	terpreted			
Bit number in data word = Bit = 0 : group release Bit = 1 : group inhibit	= group n ed :ed	umber				
F1 F2 F3	F 4	F	5	F 6 SAVE	F 7 HELP	F 8 EXI T

F6: the PC job is saved on the storage medium

F8: abort:

return to the 'PC JOB' mask

,

Field	Field type keys allowed	Limit value	Alternatives
Job name	element name (see 5.1)		*
Job no.	numbers	1 - 199	
CPU no.	numbers	1 - 4	*
DBno.	numbers	1 - 255 (DB numbers 1 and are reserved for system)	d 2 the
DW	numbers	0 - 255	

F7 (HELP) can be used to make an entry in the fields marked $^{\prime\ast\prime}.$

5.4.7 New Page

Call the 'NEW PAGE mask with the function key F6 (CONTINUE) and F2 (NEW PAGE) in the 'PC JOB' mask.

You can only program one 'NEW PAGE' PC job in a user program. The name NEW PAGE is assigned by COM 525 and cannot be changed.

SIMATIC S5 / COM525 -> PROGRAM -> PC JOB -> NEW PAGE DRIVE: B PROGRAM: COMPONENT: PT PT88ABSP NEW PAGE Job name: ### With the **job** no.: the CP 525 will initiate a form feed on the PT88/PT89 printer. F 2 F 4 F 5 F 6 FΙ F 3 F 7 F 8 SAVE EXI T

F6: save the PCjobon thedatamedium

F8: abort: return to the 'PC JOB' mask

B8576544-03

COM 525

I

Field	Field type keys allowed	Limit value	Alternatives
Job no.	numbers	1 - 199	

and a second second

and the second second

5.4.8 Reset Page Number

Call the 'RESET PAGE NO. ' mask with the function key F6 (CONTINUE) and F3 (PAGE NO. = 1) in the 'PC **JOB'** mask.

You can program only one 'RESET PAGE **NUMBER'** PC job in a user program. The name PAGE NO. = 1 is assigned by COM 525 and cannot be changed.

•> PROGRAM > PC JOB > RESET PAGENO.			SIM	ATIC S5 / COM	525
	DRI VE:	\$ PROGRAM:	\$\$\$\$S\$S\$	COMPONENT:	\$\$
Job name: PAGE NO.=1 With the job number: ### the CP 525 will set the output on the PT88/PT89 pr	numbering of inter to 1	f the list pag	jes		
F1 F2 F3	F 4	F 5	F 6 F SAVE	7 F 8 EXIT	

F6: save PC job on the storage medium

- F8: abort:
 - return to the 'PC JOB' mask

Field	Field type keys allowed	Limit value	Alternatives
Job number	numbers	1 - 199	

5.5 FRAMES

Call the 'FRAME' mask with the function key F3 (FRAME) in the following masks

- 'PROGRAM'

+ م ن د ا^ر د + س -و

- 'PROCESS STATUS LIST'
- 'CUR. MESSAGE LIST'
- 'SEQ. MESSAGE LIST'

Frames are processed in the same way that process status lists are, i.e. with the same text editor (see 5.4.4). There are, however, certain restrictions as follows:

- frames can be only maximum nine lines long
- the dynamic **data** in a frame is expected in a data block, which the **CP** 525 automatically requests from the CPU. This block must not exceed 128 bytes.

You can enter the CPU number and the number of the data block in the mask.

It is possible to read in existing frames. You specify the name of the frame in the 'frame name' field. You **can** modify the frame which has been read in and save it again under **a** different name. The new name must be entered in the 'name to be stored' field.

The 'name to be stored' field initially always has the name specified in the 'frame name' field as a default.

> SELECTION •> PROGRAM •> ⁼ R A M E				SIMATIC	S5 / COM52
	DRIVE: \$	PROG	RAM: \$\$\$\$	\$\$\$\$ CON	IPONENT: \$
Frame name: ####################################	Name	to be s	stored: ##	*#####################################	
The dynamic data are on CP	U no.: #				
in data block DB	no.: ###				
Frames can be used as a head status lists, sequential ma lists.	der and/or tr ssage lists a	ailer 1 Ind curi	for process rent messag	s ge	
F1 F2 F3	F4 COMPLETE	F 5	F 6	F 7	F8
PART PART	DYN. PART		SAVE	HELP	EXI T

- F1: generate the static part, branch to the `STATIC PART' mask
- F2: generate the dynamic part, branch to the 'DYN. PART' mask
- F4: complete the dynamic part, branch to the 'COMPLETE DYN. PART' mask
- **F6**: save the frame on disk

F8: return to the mask: 'PROGRAM' or 'PROCESS STATUS LIST' or 'CUR.MESSAGE LIST' or 'SEQ.MESSAGE LIST'

Field	Field type keys allowed	Limit value	Alternatives
Frame name	element name (see 5.1)		*
Name to be stored	element name (see 5.1)		
CPU no.	numbers	1 - 4	*
DB no.	numbers	1- 255 (the DB numbers and 2 are reser for the system)	1 ved

- - -

F7 (HELP) can be used to make an entry in the fields marked `*'.

5.5.1 Static Part of a Frame

Call the 'STATIC PART' mask with the function key F1 (STAT. PART) in the 'FRAME' mask.

-> PROGRAM •> S T A T I C I	• FRAME -> P A R T	SIMATIC S5 / COM525
	DRIVE: \$ PROGRAM:	: \$\$\$\$\$\$\$ COMPONENT: S\$
Attributes C BECU	۲۲ Element name: \$\$\$\$\$\$\$\$\$\$\$ In: A No. page width: \$\$\$	sert Line no.: \$\$\$ \$\$\$ Column no.: \$\$\$
# # # # # # # # # # # # # # # # # # # #	<pre>## ##################################</pre>	Mindow with ####################################
F 1 DELETE	F 2 F3 F 4 SEARCH/ INSERT BLOCK REPLACE CURSOR \$S\$	F5 F6 F7 F8 ENTER EXIT

F1: editor function DELETE (see 5.3.1)

F2: editor function BLOCK (see 5.3.1)

F3: editor function SEARCH (see 5.3.1)

F4: editor function CURSOR (see 5.3.1)

F5: editor function INSERT (ON/OFF), switches the insert mode on or off

155

الأجيراني فاليت المراجعين

,

, -,

- F6: enter the static part of a frame
- F8: abort: the data is not saved; return to the 'FRAME' mask

Field	Field type keys allowed	Limit value	Alternatives
Attributes	letters B=bold, E=expanded, C=compressed, U=Underlined, A=acoustic signal		¥, N *
Or no.	numbers	0 - 15	
Element name	OUTPUT FIELD		
he no.	OUTPUT FIELD		
Page width	OUTPUT FIELD (from interpreter parameter assignment see 3.6)		
Column no.	OUTPUT FIELD		

л н та**л**ат

F7 (HELP) can be used to make an entry in the fields marked $^{\prime\ast\prime}.$

1 -

5.5.2 Dynamic Part of a Frame

Call the 'DYN. PART' mask with the function key F2 (DYN. PART) in the 'FRAME' mask.

			DRI VE:	\$ PROGRAM	: \$\$\$\$s\$	\$\$ COMPO	NENT: \$S
Attri bu BECUA	ites Gr Elei No. paç	ment name ge width	: \$S\$SSSSS : SS\$	SSS\$		Line no. Column no	. : \$\$\$\$ D. : \$\$\$
##### ##### #####	* ## # # ## # * ## ## * ## #	######################################	ne long list lines	window in are displ	which m ayed.	######################################	
##### # # # # # # # #	* ## # ### #	########### ##########################	The field th is disp	currently b blayed inve ####################################	oeing wor ersely ####################################	ked	********** ***************************
######################################	# # # # ## # # # # # of dynamic lo. \$ \$\$\$:s ## ###.#	######################################	The field th is disp the state of the state PROCESS V Form No. c	currently B blayed inve ####################################	being wor ersely ####################################	ked ###### ##############################	//////////////////////////////////////
######################################	# ## #################################	######################################	The field th is disp market PROCESS V Form No. c F 4	Currently B Diayed inve ####################################	being wor ersely ################### At bo #### ex ###.# ac F 6	ked ###### tributes (Y Id print panded pri derlined oustic sigr	//////////////////////////////////////

The function and processing of this mask corresponds to the 'DYN. PART' mask of a process status list (see 5.4.4 b).

The limit values for the DB address and line are, however, different as follows:

DB address
0 - 126
0 -127
0.0 - 127.15

a) Frame - programming the text for the process status

The function and processing of this mask corresponds to the 'PROCESS STATUS VAR. 'mask (see 5.4.4 b).

The limit values for the DB address are, however, different:

memory access mode DB address DR,DL DB address

5.5.3 Complete 1b Dynamic Parts of a Frame

The function and processing of this mask corresponds to the 'COMPUTE DYN. PART' mask in a PSL (see 5.4.4 c).

The limit values for the DB address are, however, different:

memory	access	mode	DB	address
D	D		0 ·	- 126
D	R,DL		0 ·	- 127
В	I		0.0	127.15

6 Terms and Definitions

COH 525

Programming package for the ${\tt commun}$ ications processors ${\tt CP}$ 525 and ${\tt CP}$ 524

Component

Indicates the function of a CP 525 interface:

PT = printer CL = computer link

CP 525/CP 524 communications processors

Data type

The user program is divided into data types (message, PC job, frame or job block).

Element

Single partofadatatype

Function keys

Keys that have a specific significance in each mask. The **labelling** of the key indicates which action will be executed **if** the key is pressed.

Identification header

Header that contains the designation of the plant, who generated the **program and when**. The identification header is **included on** the disk and in the **EPROM** and is used to distinguish between programs with the same name on different storage media (volumes).

Interpreter

interchangeable interface driver that defines the characteristics of the component

KOMI

S5-DOS command interpreter: see description of the operating system **S5-DOS**

Library

Collection of user data which can be transfer-red to several user programs. Various interpreters and procedures can also be collected here. A library is identified by the name COMLIB?? ("?" = any alphanumeric character). A detailed description of these can be found in the section "Libraries" in this user's guide.

Listing

Overview of a user program output on a printer by ${\rm COM}$ 525 (not to be confused with current message list and sequential message list).

Mask

Template displayed on the PG monitor. COM 525 prompts inputs in the masks and enters defaults from previous masks.

Menu

I'Unction keys F1 to F8 with the corresponding labelling.

Module (software) See **element**.

overlay

Part of a program which **cannot** run independently. Is only **loaded** inthememorywhenit is required.

PCP/M

Operating system for your programmer (pG). Is loaded when you switch on the $\ensuremath{\text{PG}}$ and insert the $\ensuremath{\text{PCP/M}}$ floppy disk.

Procedure

Interchangeable software driver, which implements the transmission procedure on the line.

S5-DOS

Operating system extension for $\ensuremath{\texttt{PCP/M}}$ to standardize all the software packages running on $\ensuremath{\texttt{programmers}}$.

user area

A hard or floppy disk can be divided into 16 user areas; brings order to large numbers of files.

user data

User program

Total of all the data generated by the user with COM 525; required to operate a CP 525/CP 524 interface.

ماري ريس م م

SIEMENS

We have checked the contents of this manual for agreement with the hardware and software described. Since deviations cannot b-s precluded entirely, we cannotguarantee full agreement. However, the data in this manual are reviewed regularly and any necessary corrections included in subsequent editions. Suggestions for improvement are welcomed. Technical data subject 10 change. The reproduction, transmission or use of this document or its contents is not permitted without express written authority.

written authority. Offenders will be liable for damages. All rights, ineluding rights created by patentgrant or registration of a utility model or design, are reserved

Copyright [©] Siemens AG 1989 All Rights Resewed

Siemens Aktiengesellschaft

C79000-B8576-C544-06 Gerätewerk Karlsruhe Printed in the Federal Republic of Germany

الري الانها بالعالية الحار

,

SIEMENS

SIMATIC S5 COM 525 Messages

e e messag

Reference Manual

Order No. C79000-B8576-C545-03



Contents

1	Introduction	1
2	Messages Common to All Masks	2
3	General Situations	3
3.1 3.2 3.3 3.4	General Error Messages General Warnings General Prompts General Messages	3 11 13 15
4	Editing	18
4.1 4.2 4.3	Errors during Editing Warnings during Editing Prompts during Editing	18 22 25
5	Deleting	28
5.1 5.2	Warnings when Deleting Prompts when Deleting	28 29
б	Transfer	30
6.1 6.2 6.3 6.4	General Transfer Errors Transfer Errors, Floppy and Hard Disk Errors when Transferring Between COM 525 and the CP 525 Prompts during Transfer	30 32 35 37
7	Saving	38
7.1 7.2	Errors when Saving PromptswhenSaving	38 39

i

8	Internal	Errors

9 Operating System Errors

42

40

ii

1 Introduction

When you are working with the COM 525, messages are output in the message line (the line **above** the function key menu) under the following conditions:

- -whenanerrorhasoc curred (error message)
- to avoid errors (warnings)

as information for the user, for example to show that **an** action has been carried out successfully (general message)

- to ask whether the user "really" intends to carry out an action, for example when deleting (prompts).

A message includes an identifier (e.g. MF, ERR), a number and a short text.

1

The foilowing section lists and describes all the COM 525 messages.

Message number	Description / Remedy
MF.001	<pre>Illegal key! - Illegal characters have been entered in the field in which the cursor is positioned. Refer to Sections 2 to 5 of the COM 525 user's guide in this manual to find the permissible characters.</pre>
MF.002	<pre>Illegal entry! - The entry which has been made in the field on which the cursor is located does not correspond to any of the permissible alternatives, or it exceeds or falls below the permissible limit values. Refer to Sections 2 to 5 of the COM 525 user's guide in this manual for the limit values and the permissible alternatives.</pre>

2 Messages Common to All Masks

· ·

3 General Situations

3.1 General Error Messages

Message number	Description / Remedy
ERR.001	Overlay has wrong version number! - Use the correct COM 525 system diskette.
ERR.004	<pre>Program not found - The overlay cannot be found (see also Section 'Standard outputs when working with COM 525' in the COM 525 user's guide in this manual).</pre>
ERR.009	<pre>Element destroyed! - When reading in an element from the file the length was found to be wrong. Delete the element! The error can also occur when the 'already existing" destination element is destroyed during transfer. (This is read in the acknowledgement mode in order to achieve a comparison.)</pre>
ERR.010	Error in user program The file with the user data has been corrupted and can no longer be interpreted.

Message number	Description / Remedy
ERR.011	User data generated with old COM version - no longer valid.
ERR. 012	<pre>Error in SYSID - An error in the memory submodule of the CP 525. With the EPROM: the memory submodule must be replaced. With the RAM: if a cold restart on the CP 525 does not bring about an improvement, the memory module should be replaced.</pre>
ERR.013	NotaCP525!
ERR.014	<pre>Interpreter not present - fetch from library - See Sections `Structure of a COM 525 program' and 'Libraries' in the COM 525 user's guide in this manual.</pre>
ERR.015	Procedure not present - fetch from library - See ERR.014.

•

Message number	Description / Remedy
ERR.020	Reserved name not allowed here - The following PC jobs can only exist once per program and therefore have resewed names: SEQ.MESSAGE UPDATE SML CUR.MESSAGE NEW PAGE RESET PAGE
ERR.021	Nameis illegal or reserved! - See ERR.020.
ERR.024	<pre>CP not programmed - interpreter missing! - Neither interface 1 nor interface 2 has been programmed, The CP 525 does not contain an interpreter, procedure or user data. Transfer the interpreter!</pre>
ERR.025	Interface does not exist - Only with the CP 525 can two interfaces be programmed.
ERR.026	Interface is not programmed - No interpreter has as yet been transferred to the selected CP 525 interface.

Message number	Description / Remedy
ERR.030	At x lines per inch: page length of y to z lines possible!
ERR.101	First generate static part => Fl
ERR.102	No dynamic fields in static part
ERR.103	There are no dynamic fields in messages x to y
ERR.104	Name already assigned
ERR.105	Illegal column number - The column number entered exceeds the maximum defined during interpreter parameter assignment.
ERR.107	<pre>Illegal message number! - The message number must be between 1 and 2047. When making a jump: the specified message does not exist.</pre>

•

Message number	Description / Remedy
ERR.108	<pre>Illegal line number -The line number must be between 1 and 2047. When making a jump: thespecifiedline number does not exist.</pre>
ERR.109	Illegal field number - The specified field number is greater than the number of existing fields.
ERR.110	First generate sequential message list => F2
ERR.111	Text must be enclosed in inverted commas!
ERR.112	Illegal entry: fields overlap
ERR.113	Bit not specified!

Message number	Description / Remedy
ERR. 114	<pre>Combination of memory access and format illegal - See Section 'Process status list', Table 2 in the COM 525 user's guide in this manual.</pre>
ERR. 115	Different types of memory access! - See Section `Process status list', Table 2 in the COM 525 user's guide in this manual.
ERR.116	Value too high - cannot be represented!
ERR.117	Job not saved - The job can only be printed out after it has been entered. First enter the job with F6!
ERR.118	Line is too long - The required structure for the date, time orstatusstarting at the specified column exceeds the maximum length of a line (136 characters per line). Change the structure or position accordingly!

Message number	Description / Remedy
ERR.119	DB boundary exceeded (No. of characters)!
ERR.120	No more fields in this direction!
ERR.121	All fields already programmed in this direction!
ERR.122	No more messages loaded in this direction!
ERR.123	All loaded messages programmed in this direction!
ERR.124	Message specified is not loaded!
ERR.125	No field in the line specified!

Message number	Description / Remedy
ERR.126	Field specified not in this line!
ERR.127	No field in the message specified!

10

3.2 General Warnings

With warnings the data input will generally be entered or saved.

Message number	Description / Remedy
WARN.001	Source/dest. address cannot be addressed - See user's guide `Computer link with RK 512' in this manual.
warn.002	Coord . flag too high as partner for CP 525
WARN.003	DB number used by system - Depending on the CPU, DB 1 and possibly DB 2 may be reserved for the system.
WARN.004	Less characters than number of positions reserved
WARN.005	More characters than number of positions reserved
WARN.006	Field too long - max. 60 characters used!

Message number	Description / Remedy
WARN.007	Only 80 characters will be printed!
WARN.013	From message x to message y read in! The remainder was not loaded!

3.3 General Prompts

Prompts must be answered with $\ensuremath{\texttt{Fl}}$ (YES) or F3 (NO) in the acknowledgement menu.

Message number	Description / Remedy
QUIT.001	Terminate program? - `YES' = COM 525 returns to the basic mask. A program which contains no more data will be deleted as a file.
QUIT.002	<pre>Loss of data - exit mask? - 'YES' = the entries made in the current mask will not be entered. 'NO' = the data can be entered with F6 (ENTER or SAVE).</pre>
QUIT.003	Process library? Libraries do not have to contain consistent programs; they are used to collect different parts of programs (see also Section 'Libraries' in the COM 525 user's guide in this manual).
QUIT.004	Loss of data - exit field? - As for QUIT.002

•

Message number	Description / Remedy
QUIT.010	Element already exists! - overwrite?
QUIT.011	Job block already exists - overwrite?
QUIT.012	No comparison of elements possible!Overwrite?There is insufficient memory to compare the contents of the source and destination element.

3.4 General Messages

Message number	Description / Remedy
MESS.001	 Active! This message is output while a function is being executed, when COM 525 user data are being processed on disk or data are being exchanged with the CP 525.
MESS.002	Completed!
MESS.003	Saved!
MESS.004	Active - abort with F8 - The function can be aborted with F8.
MESS.005	Deleted!
MESS.006	Element being processed:
MESS.007	Entered!

Message number	Description / Remedy
MESS.008	Aborted!
MESS.009	Loading overlays! - Is output while COM 525 overlays are being loaded.
MESS.010	<pre>Program being condensed - A program is automatically condensed when it is t∞ "broken up" by deleting and storing.</pre>
MESS.011	Program already purged
MESS.012	First page! - When trying to page backwards from the first page.
MESS.013	Last page! -When trying topageforwardsfromthe last page.
MESS.014	Number of messages to be read from FD:
Message number	Description / Remedy
----------------	---
MESS.015	Number of messages to write/delete on FD
MESS.016	All fields from message x to y already programmed!
MESS.017	All fields have already been programmed!
MESS.018	<pre>The fields cannot be programmed from message x to y! - All fields which appear in the specified messages are located after the endofthelineowing tochangesin the sequential message list and cannot, therefore, be programmed. The fields can be programmed again if the sequential message list is changed again (e.g. reducing the length of the status text).</pre>
MESS.019	<pre>Active - deleting may take several minutes! - If there is a large number of elements (up to 1000) on the CP 525, deleting messages can take several minutes - the elements must be searched in the RAM.</pre>

4 Editing

4.1 Errors During Editing

Message number	Description / Remedy
ERR.100	Data cannot be interpreted - The code of the stored data has an error and cannot therefore be re- translated. This data can only be deleted.
ERR.106	Max. 9 (frame) or99 (PSL) lines possible.
ERR.130	<pre>Macro too long -> changes will not be entered - A maximum of 35000 bytes are available for programming a process status list. Once this number has been reached (too many process statuses programed) no further lines can be acceptedly the editor.</pre>
ERR.131	Macro too long, changes in the line will not be entered - See ERR.130.

Message number	Description / Remedy
ERR.132	Too many dynamic fields A maximum of 40 dynamic fields per line are possible with the process status list and with frames. For each message only one dynamic field is permitted,
ERR.133	Start of block not set
ERR.134	Start of block after end of block
ERR.135	Block not set
ERR.136	No line saved yet
ERR.137	Destination for block transfer is within the block

Message number	Description / Remedy
ERR.138	Block cannot be saved - macro too long - A maximum of 35000 bytes are available for programming a process status list. Once this number has been reached (too many process statuses programmed) no further lines can be accepted by the editor.
ERR.139	Block cannot be saved - too many lines
ERR.140	No more found - press ENTER key
ERR.141	Reserve characters not allowed - The characters used to reserve positions for dynamic fields must not be searched for using the 'search' and 'search/replace' functions.
ERR.142	No reserve characters allowed in word to be deleted
ERR.143	Insertion of a line not possible

Message number	Description / Remedy
ERR.144	Insertion only possible when cursor is at start or end of line
ERR.146	Message number already assigned
ERR.148	Message number already assigned in area read in -> HELP
ERR.150	<pre>Max. number of messages already programmed! -Amaximum of 1000 messages (from 1 - 2047) can be programmed.</pre>
ERR.151	Message number assigned outside area read in -> HELP
ERR.152	Message number not in area read in!
ERR.153	 Block will not be copied - too many messages! Max. 99 messages can be entered in the editing buffer of the PG. By copying blocks this number has been exceeded.

4.2 Warnings during Editing

Message number	Description / Remedy
WARN.006	Field too long - max. 60 characters used!
warn.008	<pre>Lines longer than page width - If, when editing a process status list a frame or a message there are more characters edited per line than determined in the 'ASSIGN INTERP. PARA.' mask as the page width, the characters which extend beyond the end of the line will be printed in the next line. Remedy: increase the page width in the 'ASSIGN INTERP. PARA.'</pre>
WARN.009	Masked area changed - dyn. part has been deleted - cf. WARN.012
WARN. 010	Message numbers higher than 2000 are system messages!
WARN.011	Copied messages have system message numbers higher than 2000!

Message number	Description / Remedy
WARN.012	<pre>SML was modified! Inconsistency of data in messages with ">" - By changing the parameter assignment for the sequential message list when messages already exist it is possible that the following inconsistencies may occur (such messages are marked with ">")</pre>
	1) If messages were programmed with DATE, TIME or STATUS and the jobs were later deleted in the sequential message list, this warning points out that the jobs in these messages no longer exist (displayed by a narrow inverse field with the identifier D, T or S). They will not, however, be triggered by the CP. Remedy: change the parameters in the sequential message list.
	2) If messages have been programmed with DATE, TIME and/or STATUS and later the date, time or status field was extended in the sequential message list, it is possible that a message line which has already been programmed will be longer than 136 characters. The characters which would exceed this if anything is changed in the message are truncated. (It is possible that a dynamic field which has already been programmed may be lost). Remedy:change the parameters in the sequential message list (to see the original text of the truncated message, the date, time or status fields in the SML must be reduced again).

Message number	Description / Remedy
WARN.013	From message x to message y read in! The remainder was not loaded!
WARN.015	Dyn. field in line x too long (max. 80 res. pos. per field)!
WARN.016	Dyn. field in message x <i>too long (max.</i> 80 res. pos. per field)!

Message number	Description / Remedy
QUIT.040	Dyn . part already exists, delete?
QUIT.041	 Field with a dyn. part in the last column, delete? In the insert mode or when replacing a word a line may sometimes become longer than 136 characters. These characters and any programmed dynamic fields would be lost during the insertion. The characters may possibly be located outside the currently visible characters on the screen.
QUIT.042	Delete last character in the line? - See QUIT.041
QUIT.043	REPLACE?
QUIT.044	Line too long - replace?

4.3 Prompts during Editing

Message number	Description / Remedy
QUIT.045	 Field with dyn. part in column before masked field, delete? Inhibited fields are fields which cannot be changed for the date, time and status in the editing window for messages.
QUIT.046	Delete character before masked field?
QUIT.047	New text exceeds masked area, replace? - When replacing a word it is possible that the following text would go beyond the date, time or status field, since the text following the word to be replaced is shifted only as far as the next date, time or status field. These characters may possibly be outside the visible characters displayed on the screen.
QUIT.050	Save?
QUIT.051	Changed messages marked ">" will be truncated! ENTER? - See WARN.012

Message number	Description / Remedy
QUIT.052	Abort reading in of messages from FD?
QUIT.053	Abort saving of messages on FD?

5 Deleting

5.1 Warnings when Deleting

Message number	Description / Remedy
WARN.014	<pre>Before deleting SEQ.MESSAGE: check whether messages exist for it! - Do not forget that the position of the date, time and status is fixed with the PC job SML (sequential message list). If the SML is missing no messages can be output even if they have been programmed and stored on the CP 525/ 524.</pre>

Message number	Description / Remedy
QUIT.020	Delete both interfaces?
QUIT.021	Delete all elements?
QUIT.030	Delete element:
QUIT.031	No element will be deleted! - abort function?
QUIT.032	Delete a ll elements?

5.2 Prompts when Deleting

6 Transfer

6.1 General Transfer Errors

Message number	Description / Remedy
ERR.016	Function illegal: component wrong - An attempt has been made to transfer a PT element to a CL program or a CL element to a PT program.
ERR.022	Source =destination?
ERR.023	Different components in source and destination - See ERR.016
ERR.024	<pre>CP not programmed - interpreter missing! - The CP 525 has no interpreter, procedure or user data on interface 1 and interface 2.</pre>
ERR.025	<pre>Interface does not exist - Two interfaces can be programmed only on the CP 525-2, with other CP types only one can be programmed.</pre>

Message number	Description / Remedy
ERR.026	Interface is not programmed - No interpreter has been transferred to the selected CP 525 interface.
ERR.027	EPROM plugged in - No data can be written to the EPROM on the CP 525! Plug in a RAM card.

31

Message number	Description / Remedy
ERR.701	Drive cannot be addressed! - Check whether the drive compartment is closed!
ERR.702	Element directory not present - No element exists for the data type selected!
ERR.703	Element not present
ERR.704	FD is write-protected With a floppy disk: remove the write- protect tab. With hard disk: see description of the operating system PCP/M.
ERR.706	Maximum number of elements exceeded - No more elements of the selected data type can be stored.
ERR.707	File not present

6.2 Transfer Errors, Floppy and Hard Disk

.

Message number	Description / Remedy
ERR.708	FD external full No more entries can be made in the file directory. Check whether files can be deleted or use a new floppy disk,
ERR.709	Working diskette/CP memory full With FD: use a new floppy disk. With CP: transfer the new data to floppy or hard disk and use a larger RAM submodule.
ERR.710	Element already exists
ERR.712	 FD has been changed - program no longer available The floppy disk was changed while processing a program. Insert the correct floppy disk again!
ERR. 713	File is write-protected - Remove the write-protect attribute (see descriptionof the operating system PCP/M).

Message number	Description / Remedy
ERR. 714	Not a COM 525 file - The file was not generated by this COM 525.
ERR.715	<pre>Illegal file name - The file name specified does not correspond to the PCP/M conventions. See Section 'Selecting the user program' in the COM 525 user's guide in this manual.</pre>
ERR.716	File already exists
ERR. 717	Invalid date - The entry in the "date" field is illegal.

Message number	Description / Remedy
ERR.718	No RAM card inserted!
ERR.720	<pre>SYSID block does not match specification Error in memory submodule. With EPROM: the memory submodule must be replaced. With RAM: if a cold restart on the CP 525 does not bring improvement, the memory submodule must be replaced.</pre>
ERR.721	CP switch set to STOP - Switch the mode selector on the CP to RUN.
ERR.722	Transfer error - Error or fault on the CP 525 or the link between the CP 525 and PG which cannot be identified more exactly,
ERR. 723	Initiation of function negatively acknowledged - Please check whether there is. a connection to the CP 525.

6.3 Errors when Transferring Between COM 525 and the CP 525

Message number	Description / Remedy
ERR.724	Interpreter/procedure not present - Before data can be transferred the interpreter must first be transferred to the CP 525.
ERR.725	Cable not connected/wrong baud rate
ERR.727	Job number already exists on CP! -A PCjobwiththesame number has already been transferred to the CP 525.
ERR.728	<pre>Interface cannot be deactivated at present! - Repeat the transfer. When transferring between the CP 525 and a partner follow the rule of thumb: large volume of data -> low data rate.</pre>
ERR.729	<pre>Error in interpreter/procedure - Following this message, the interface on the CP 525 is completely deleted. Possible causes: error in the source file (interpreter or procedure on the PG) - error during transfer from PG to CP 525 error in the memory submodule of the CP 525</pre>

6.4	Prompts	during	Transfer
-----	---------	--------	----------

Message number	Description / Remedy
QUIT.022	<pre>Delete interface on CP and start program transfer? - 'YES' = before the program transfer the interface will be completely deleted.</pre>
QUIT.023	Transfer all elements?

7 Saving

7.1 Errors when Saving

Message number	Description / Remedy
ERR.130	<pre>Macro too long -> changes will not be entered - A maximum of 35000 bytes are available for programming a process status list. Once this number has been reached (too many process statuses programmed) no further lines can be accepted by the editor.</pre>
ERR.131	Macro too long, changes in the line will not be entered - See ERR.130

7.2 Prompts when Saving

Message number	Description / Remedy
QUIT.010	Element already exists overwrite?
QUIT.011	Job block already exists - overwrite?
QUIT.051	Changed messages marked 'Y' will be truncated! ENTER? - See WARN.012

8 Internal Errors

Notes **on** these errors

These errors can be caused by a corrupted COM 525. If, despite reloading **from** a back-up disk the problem cannot be remedied, please send a description of the problem to the nearest **SIEMENS** service department or technical office.

Message number	Description		
IF.001	Mask not present!		
IF.002	Field not present		
IF.003	Illegal MAINT call		
IF.004	Overlay identifier is illegal!		
)2?.005	Buffer not long enough!		
IF.006	Data type riot defined!		

Message number	Description \ Remedy
IF.007	CP function unknown Wrong function call caused by transfer error. Repeat the function call. If the error message occurs again see 'Notes on these errors' on the previous page.

9 Operating System Errors

Message number	Description			
ERR.002	Program cannot be closed - Internal error (see Section 8).			
ERR.003	Program cannot be executed			
ERR.004	Programnotfound			
ERR.005	Error loading program - not enough memory - See ERR.007			
ERR.006	Program cannot be removed - Internal error (see Section 8).			
ERR.007	<pre>Not enough memory for buffer - press ENTER key - Attempt to restart the system (RESET or switch off and then on). If the error message appears again there is insufficient memory capacity.</pre>			

We have checked the contents of this manual for agreement with the hardware and software described. Since deviations cannot be precluded entirely, we cannot guarantee Iull agreement. However, the data in this manual are reviewed regularly and any necessary co... rections included in subsequent editions. Suggestions for improvement are welcomed Technical data subjectto change.

Siemens Aktiengesellschaft

C79000-B8576-C545 -03 Gerätewerk Karlsruhe Printed in the Federal Republic of Germany

SIEMENS

SIMATIC S5

PROM 525

User's Guide

C79000-B6576-C546-04

1	Introduction	
2	Example of Programming an EPROM	3
3	Reference Section	16
3.1	General Notes	16
3.2	Overview of the Masks	17
3.3	Basic Mask	18
3.4	Transfer FD->EPROM	19
3.5	programing the EPROM	22
3.6	Transferring EPROM->FD	27
3,7	Empty Test	30
3.8	INFO EPROM	35
3.9		55
4	PROM 525 Messages	37
4,1	Messages Common to All Masks	38
4.2	Messages	39
4.3	Acknowledgement Prompts	41
4.4	Error Messages	43
4.5	Internal Errors	

Page

.....¹.

A . March

1 Introduction

The program PROM 525 supplements the programming package COM 525. It is used to transfer COM 525 programs, stored on floppy or hard disk, to an EPROM. With PROM 525 the transfer in the opposite direction is also possible, i.e. transferring a program back from an EPROM to a floppy or hard disk.

Using PROM 525 you can test whether a aubmodule has already been written to or whether it is empty. PROM 525 also has two information functions. The first one outputs en identification header for the programs located on the floppy or hard disk. The second function provides you with information about EPROMs which have already been written to.

You can have the length of your program displayed, and therefore check whether the au**bmodules** you want to program still have sufficient free memory capacity. PROM 525 informs you of the length of time required for programming without any transfer being made. While you program an EPROM, the programming time still left to run is displayed.

Before you start to work with PROM 525 you have probably already generated programs with COM 525 end know about the masks and menus. If you are unsure about using the masks and menus or are still unfamiliar with terms or abbreviations that are used in this part of the manual please refer to 'Terms and definitions in the user's guide to COM 525.

With the PG 695 it is possible to connect an external programmer (prommer) to the device. More information can be found in the installation guide and instructions supplied with the device. The initialisation (the setting up of the communications link) is carried out by PROM 525 (even with the integrated unit). This takes approximately 60 to 70 seconds.

المراجع من الم المراجع المراجع المراجع المراجع الم المراجع المراجع المراجع المراجع المراجع المراجع ا

,

A program generated with **COM** 525 is initially stored either on a floppy or hard disk. The following EPROM **submodules** can be programmed with PROM 525:

6ES5 373-0AA41 16 Kwords 6ES5 373-0&461 32 Kwords 6ES5 373-0AA81 64 Kwords 6ES5 373-0AA42 16 Kwords 6ES5 373-0AA42 16 Kwords 6ES5 373-0AA62 32 Kwords 6ES5 373-1AA41 16 Kwords 6ES5 373-1AA61 32 Kwords 6ES5 373-1AA81 64 Kwords *	EPROM submodule number	capacity
6ES5 373-0&461 32 Kwords 6ES5 373-0AA81 64 Kwords * 6ES5 373-0AA62 16 Kwords * 6ES5 373-0AA62 32 Kwords * 6ES5 373-1AA41 16 Kwords * 6ES5 373-1AA61 32 Kwords * 6ES5 373-1AA81 64 Kwords *	6ES5 373-0AA41	16 Kwords
6ES5 373-0AA81 64 Kwords * 6ES5 373-0AA42 16 Kwords * 6ES5 373-0AA62 32 Kwords * 6ES5 373-1AA41 16 Kwords * 6ES5 373-1AA61 32 Kwords * 6ES5 373-1AA81 64 Kwords *	6ES5 373-0&461	32 Kwords
6ES5 373-0AA42 16 Kwords 6ES5 373-0AA62 32 Kwords 6ES5 373-1AA41 16 Kwords 6ES5 373-1AA61 32 Kwords 6ES5 373-1AA81 64 Kwords	6ES5 373-0AA81	64 Kwords *
6ES5 373-0AA62 32 Kwords 6ES5 373-1AA41 16 Kwords * 6ES5 373-1AA61 32 Kwords * 6ES5 373-1AA81 64 Kwords *	6ES5 373-0AA42	16 Kwords
6ES5 373-1AA41 16 Kwords * 6ES5 373-1AA61 32 Kwords * 6ES5 373-1AA81 64 Kwords *	6ES5 373-0AA62	32 Kwords
6ES5 373-1AA61 32 Kwords * 6ES5 373-1AA81 64 Kwords *	6ES5 373-1AA41	16 Kwords *
6ES5 373-1AA81 64 Kwords *	6ES5 373-1AA61	32 Kwords *
	6ES5 373-1AA81	64 Kwords *

* With **PG** 685 **RELEASE < 8 only** possible with **MEP** adapter.

2 Example of Programming an EPROM

In this section you are introduced to the fictions of the PROM 525 step by step. You can implement the example on your OWN screen. If you find that an explanation is missing at any point (e.g. the explanation of the function key) you can find this in the reference section of this user's guide.

The PROM 525 program is loaded when you enter the co-d "S5" to call the S5 command interpreter.

Position the cursor on the programming package COM 525. Then press the function key F1 (PACKAGE) to call the 'BASIC MASK' of the PROM 525.

ASIC M	BY SIEMENS A S K	A G				SIMATIC	S5 / PROM
рррррр	RRRRR	00000	мм	MM	55555555	22222	5555555
PP PP	RR RI	r 00	CO	MMM MMM	55	22 22	2 55
PPPPPP	RR RR	00 CO	MM	M M MM	5555555	2	2 555555
PP	RRRRRR	00 00	MH	MM MM	55	22	55
PP	RR RR	00	00 MM	MM	55	22	55
PP	RR RR	000	X00 MM	MM	5555555	222222	555555
EPROM	programing	software	for ti and ti	ne communi ne communi	cations pro cations proc	cessor CP C essor CP	525 5 2 4
EPROM Version	programing /Issue:	software A04	for ti and ti	ne communi ne communi Seria	cations pro cations prod l no.: 7994	cessor CP c essor cp -0074-654	525 5 2 4 5 321
EPROM Version/	programing /Issue: F2 F	software A04 3	for th and th F4	ne communi ne communi Seria F5	cations pro cations prod i no.: 7994	cessor СР cessor ср - 0074-654 F 7	525 524 321 F8
EPROM Version F1	programing /Issue: F 2 F NSFER EMP	A04	for ti and ti F 4 NFO	ne communi ne communi Seria F5 INF0	cations pro cations prod t no.: 7994 F 6	Cessor CP C essor CP - 0074-654 F 7	525 5 2 4 321 F 8

~~______}

Use the function keys F1 to F5 to branch to the PROM 525 followon masks. Using F8 (EXIT) you always return to the previous mask, or from the 'BASIC MASK' to the S5 command interpreter.

For the example, press F5 (INFO **FD**). PROM 525 then outputs the 'INFO **FD** ' mask.

BASIC MASK-> INFOFD			SIMATIC S5 / PROM525
	Drive: Program:	B Test 1	
	Plant designation: Generated by: Generated on:	PLA.GL2000XA Tommy 22.01.87	
	Component: COM525 version:	РТ АО 4	
F 1 F	2 F 3 F 4	F 5 F 6	F7 F8 Help Exit

This mask displays information about your program,

The 'drive" field shows the drive in which your' 'PROM 525 program-~ package is located and needs to be changed only if for example the package is on the hard disk, end your program is on a floppy disk.

The cursor is positioned on the "program" field. You can enter the name of the file about which you wish to obtain information (for the example, take the program to be transferred to an

al antisa antisa

يركونها يتحافروا

محمورين الأ

.....

化磷酸基氟酸基脂酸 网络小说 建金属石 化分析器 化分析器

 $\sim 10^{\circ}$

EPROM). This can also be achieved with F7 (HELP), which you already know through working with COM 525. In the reference section of this user's guide you can see which fields can be completed using F7 (HELP). If the COM 525 file you have, specified is present, the identification header of the program is now displayed.

If you have programed both interfaces, i.e. you wish to transfer two programs, you can also have the identification header of the second program displayed. Press F8 (EXIT) to return to the basic mask.

As the next step, check whether the EPROM you wish to program is empty. With F3 (EMPTY TEST) select the 'EMPTY TEST' mask and plug in one of the EPROMs which has already been written to in the receptacle (in the example an EPROM with the EPROM number 6ES5 373-1AA81 is used).

BASIC MASK -> EMPTY TEST	SIMATIC S5 / PROM525
MLF8-number:6ES5 373- 1AA81Submodule	D: 463
F1 F2 F3 F4 F5 F	6 F 7 F 8
TEST	HELP EXIT

·

When pressing F7 (HELP) the following **mask** appears with a selection list.

BASIC NASK-> EMPTY TEST	SIMATIC S5 / PROM525
MLFB-number: 6ES5 373-1AA81 Su	PLEASE SELECT: 6ES5 373-1AA41 414 6ES5 373-0AA41 14 6ES5 373-0AA42 114 6ES5 373-0AA42 114 6ES5 373-0AA61 60 6ES5 373-0AA62 160 6ES5 373-1AA81 463 6ES5 373-0AA81 163
<u>,</u> 2,3,4,	5 F6 J7 F8 ENTER EXIT

Using the arrow keys you can select an MLFB number and acknowledge it by pressing F6 ($_{\tt ENTER}).$

If the number of the EPROM you have plugged in is not the same as the default displayed, enter the correct number with F7 (HELP) and press F1 (TEST). In the field "current address" which now appears on the screen you can follow all the addresses being searched until the following message is output in the message line:

'EPROM is empty! '

If the EPROM has already been written to, the search stops at the f irst address which is not empty and the message

'EPROM not empty!

is output.

Once you are sure that your EPROM is empty, exit the 'EMPTY TEST' mask with F8 (EXIT). You can now program the EPROM.

Press F1 (TRANSFER FD->EPROM) in the 'BASIC MASK' to call the 'TRANSFER FD - EPROM' mask.

Mask with the CP 525-2

BASIC MASK -> TRANSFER FD	- E P R O M	SIMATIC S5 / PROM525
	Module: 0	P525-2
	Source	
	Interface 1	Interface 2
Drive: Program:	B Test 1	B TEST2
Plant designation: Generated by:	PLA.GL2000XA TOMMY	PLA.GL2000XA TOMMY
Generated on: Component:	22.01.87 PT	22.01.87 CL
	D M Y Date: 23.01.87	H M Time: 11.39
F1 F2 PROGRAM REQUIRED EPROM STORMS	F 3 F 4 F 5	F 6 F 7 F 8 HELP EXIT
Mask with CP 524

BASIC MASK -> TRANSFER FD	- E P R O M	SIMATIC S5 / PROM525
	Module: CP524	
Drive: Program: Plant designation: Generated by: Generated on: Component:	Source Interface 1 B TEST1 PLA.GL2000XA TOMNY 22.01.87 PT	и
F1 F2 Program Rewired EPROM Storage	Бнт н Date: 23.01.87 Time: 11. F3 F4 F5 F6	7 7 7 7 7 7 7 7 7 7

The "module" field on which the cursor is Positioned has the default 'CP 525-2', If you are using a CP 524 module enter this in this field with F7 (HELP) and continue reading at b) below.

a) **CP** 525-2

The fields "drive" below "interface **1**" and 'interface **2**" already have the default of the **drive** for the PROM 525 as seen in the 'INFO FD' mask. After you leave the "module" field the cursor **is** positioned on the 'program" field. If you want to transfer one or two programs (for both interfaces) when leaving the 'module' field, enter one program name or two program names one after the other using F7 (HELP). The plant designation, generator, **date** of generation and component for the programs are **then** displayed.

b) CP 524

ومعري والمشتقة فتعمله البارج مالوا التيارية

•

No second interface is available for this module. **Enter** the name of your program with F7(HELP)-the "drive" field has as a default the **drive** with the PROM 525 package. The fields 'plant designation", 'generated byⁿ, 'generated **on**" end "component" now appear.

If the entries **in** the fields "**date**" and "**time**" are not up to date you should correct them (**date** in the form DD **MM YY** and the time **HH** MM).

Press F2 (REQUIRED **STORAGE)** to display the minimum storage **re**quired to store the specified program **in** the message line. You can then check whether the **submodule** you have selected has suff **icient** capacity.

Press F1 (PROGRAM EPROM) to branch to the 'PROGRAM EPROM' mask.

B8576546-04

PROM 525

e

12-1-200

s + 70

	Module:	CP525-2
	Source	2
	Interface 1	Interface 2
Drive:	В	В
Program:	TEST I	TEST2
	Dest i	nat ion
	EPROM: Plant designation:	6ES5 373-00AA41 PLA.GL2000XA
	EPROM capacity: Program length: Progr. takes approx.: Current address	016384 words 010344 words 018 min 00000000
F1 F2 Program	F 3 F 4 F	5 F6 F7 F8

: v

F 1	F 2	F 3	F4	F 5	F 6	F 7	F 8
							ABORT

All the entries above 'destination" have been **taken** over from the **previous** mask (i.e. are different forCP 525-2 and **CP 524)**.

If you change the 'EPROM" field using F7(HELP), notice that a different value is output in the "EPROM capacity" field; this is because you plugged in a larger or smaller submodule. The field "plant designation" has defaults entered from the previous mask but can still be changed.

Now press F1 (PROGRAM EPROM). PROM 525 sets up the directories of the programs which are to be transferred. Following this, the storage, required for your program, the programing time end the current address are displayed in a submask.

Now check to see that your EPROM is still plugged in and that it is the right one.

PROM 525 first checks that the EPROM is empty and only then starts the programming. From this point forward yOU **can still** interrupt the progreining using **F8** (ABORT). (The programming is not aborted immediately but only after the transfer of the element currently **being** processed).

During the whole programming process the remaining programing time and the current address are displayed. The message **line** displays the name **of** the element that is currently being processed. Once the programming of the EPROM is completed, the message 'completed' will be displayed.

After programming the EPROM, you can press **F8** (EXIT) (**PROM 525** will prompt you to confirm that you wish to leave the mask) to return to the '**BASIC MASK'** via the 'TRANSFER **FD** - **EPROM'** mask in which you once again press **F8**,

Pressing F4 (INFO EPROM) to branch to the **'INFO EPROM'** mask. Now you can obtain information about the EPROM you have just written to.

. ..

	EPROM: 6ES5 373-0A	A41
EPROM: Nodul e:	6ES5 373-0AA41 CP525-2	11: PT 12: PT88
Plant:	PLA. GL2000XA	13: 63 14: LAUFPT88
Generated on:	23. 01. 87	15: 03 16 [.] Cl
Ind. bus address:		17: RK512
Password:	•	19: P3964
COM version: PROM version:	A04 A04	20: 81 21:
·		1

Enter the EPROM number in this mask using F7 (HELP). Press F1 (INFO) and the PROM 525 outputs the EPROM and module parameters (SYSID) for the EPROM. The significance of the numbered fields can be found in the reference section of these instructions.

Return to the 'BASIC **MASK'**. There you see that only the **function** 'transfer EPROM -> **FD'** is now **missing**. You can **use** this function to transfer the contents of an EPROM-to a floppy or hard disk. **In** this case you will transfer the program which is now loaded in the EPROM to a floppy disk. **Press** F2 (TRANSFER **EPROM** ->J?D).

ASORT

BASIC MASK -> TRANSFER EPROM - FD	SIMATIC S5 / PROW!
EDDOM-	Source
Interface number:	1
Plant desi gnati on: Generated by: Generated on: Component:	PLA. GL2000XA TOMMY 23.01.87 PT
	Destination
Drive: Program:	A TEST 1
F1 F2 F3 F4 TRANSFER	F5 F6 F7 F8 HELP EXIT
: V	
F1 F2 F3 F4	F5 F6 F7 F8

If necessary, enter the EPROM tier in the 'EPROM' field using F7 (HELP). If the EPROM belongs to a module of the type CP 525-2, the field 'interface number' appears below this. This is assigned a default (when both or only the first interface is programed = "1"; only interface 2 is programmed = "2"). You can change the default with F7 (HELP), At the same time the plant designation, generated by, generated on and component fields are output (also applies to the CP 524),

In the example you will transfer the **contents** of the EPROM to a floppy disk. Insert a **formatted** disk in drive A: and enter 'A' in **the 'drive**" field using F7 (HELP). Write the name you wish to give your program on the floppy disk in the 'program" field (F7 cannot be used here) and then press F3 (ENTER). At this point the PROM **535 prompts you to confirm whether** you wish tostartthe transfer or not. With "YES" you start the **transfer**. The second menu now appears on the screen with which it is possible to stop the transfer using F8 (ABORT). The name of the element currently being transfer will also be displayed.

This completes the example. You can exit the PROM 525 program with F8 (EXIT) and return to the 'BASIC MASK' and from there to the S5 command interpreter.

1.5

3 Reference Section

This section contains all the masks displayed by PROM 525. There is also a description of the function keys and fields in the masks.

3.1 General Notes

The input fields (displayed inversely on the screen) are represented **in** the **manual** as **'***HHHHHHH***'** and the output fields as '\$\$\$\$\$\$\$\$.

Before aborting a transfer and before overwriting existing parts of the program **PROM 525** outputs a request or prompt and the following **acknowledgement** menu:

F 1	F 2	F 3	F4	F 5	F 6	F 7	F 8
YES		NO					

PROM 525 requires a positive or **negative** acknowledgement before **it** continues the program.

Some fields can have entries made **using** F7 (HEW); these fields are marked with **'*' in** the tables describing the fields.

.

. ¥.4

3.2 Overview of the Masks

The following overview shows you how to call each mask:



3.3 Basic Mask

Os11 the 'BASIC MASK' (after positioning the cursor on the line 'PROM 525 EPROM programing CP 525/524*) with the function key F1 (PACKAGE) in the S5 command interpreter.

BASIC MA	BY SIEMENS \SK	AG			SIMATIC S	95 / PROM52
рррррр РР РР РРРРР	RRRRRR RR RR RR RR ()	00000 MM 00 00 0 00 MM	MM MMM MMM M M MM	55555555 55 55555555555555555555555555	22222 22 22 22 22	5555555 55 555555
PP	RRRRRR OC	00 MH	MH MM	55	22	55
PP	RR RR	00 00	MM MM	55	22	55
PP	RR RR	00000 HM	MM	5555555	222222	555555
EPROM pr	ogramming sof	tware for the and th	e communic ne communic	ations proc cations proc	essor CP ! :essor CP !	525 524
EPROM pr Version/	rogranning sof Issue:	tware for the and the A04	e communic ne communic Serial	ations proc cations proc no.: 7994	essor CP : cessor CP 5 •0074-6543	525 524 21
EPROM pr Version/	rogramming sof Tissue:	tware for the and the A04	e communic ne communic Serial F5	ations proc cations proc no.: 7994 F6	essor CP : :essor CP : :0074-6543 F 7	525 524 21 F8
EPROM pr Version/ F 1 FA TRANSFER TRANSFER	Cogramming sof	A04 F 4 F 4 INFO	e communic ne communic Serial F5 INFO	ations proc cations proc no.: 7994 F6	essor CP : :essor CP : :0074-6543 F 7	525 524 21 F 8

F1: branch to the 'TRANSFER FD - EPROM' mask.

F2: branch to the 'TRANSFER EPROM - FD' mask.

F3: branch to the 'EMPTYTEST'mask.

F4: branch to the 'INFO EPROM' mask.

F5: branch to the 'INFO FD' mask.

.

F8: return to the S5 command interpreter.

3.4 Transfer FD->EPROM

Call the 'TRANSFER FD - EPROM' mask with the function key **F1** (**TRANSFER FD->EPROM) in** the 'BASIC MASK'.

Mask for CP 525-2

	Module: ####	***
	Source Interface 1	Interface 2
Drive: Program:	# #############	# #########
Plant designation: Generated by: Generated on: Component:	\$\$5555555555555555555 \$\$\$\$\$\$\$\$\$ \$\$\$\$\$\$\$ \$\$	55555555555555555555555555555555555555
	D M Y Date: ##.##.##	H M Time:##.##
F 1 F 2 Program Required FPROM Storage	F 3 F 4 F 5	F6 F7 F8

19

Mask for CP 524

BASIC MASK -> TRANSFER FD	SIMATIC S5 / PROM525 - EPROM
	Module: #######
Drive: Program:	Source Interface 1 Interface 2 Not present #
Plant designation: Generated by: Generated on: Component:	55 55 55 55 55 55 55
	DNY Date:##.##.## Time:&;
F 1 F 2 PROGRAM REQUIRED EPROM STORAGE	F3 F4 F5 F6 F7 F8 HELP EXIT

F1: branch to the 'PROGRAM EPROM' mask.

- **P2** display of the minimum memory requirement (program length) of the specified program or the two programs if both interfaces are used,
 - Note: the specified program length always contains 2 x 8 **Kbytes** for the interpreters and procedures of both interfaces. The actual memory space **required** by the program is always rounded up to a whole multiple of 8 **Kbytes** for each interface, since the memory is divided into memory pages of 8 **Kbytes**.
- **F8:** return to the *BASIC MASK'.

ς.

Field		Field type keys allowed	Unlit value	Alternatives
Module		any		CP524/ CP525-2 *
Drive ¹	1)	upper case letters		dependent on device type *
Program	n 1)	letters/ numbers		*
Plant designa	ation 1)	OUTPUT FIELD		
Genera	ted by 1)	OUTPUT FIELD	ti	
Generat	ted on 1)	OUTPUT FIELD		
Compone	nt 1)	OUTPUT FIELD		
Date	D M Y	numbers numbers numbers	1 - 31 1 - 12	
Time	H S	numbers numbers	0 - 23 0 - 59	

F7 (HELP) can be used to make an entry in the fields marked '*'.

1) If two interfaces are being used (CP 525-2) these specifications apply to the fields under 'interface 1' and under ' interface 2.

21

المعربي والأراقين مريح المرتبانية

•

.

- -

4

....

3.5 Programming the **EPROM**

Call the 'PROGRAM EPROM' mask with the function key F1 (PROGRAM EPROM) in the 'TRANSFER FD - EPROM' mask.

The floppy disk must not be changed while this mask is displayed!

Mask for CP 525-2

PROGRAM	E P R O M	
	Module: \$\$\$	3555
	Sourc Interface 1	e Interface 2
Drive: Program:	s s	s \$\$\$\$\$\$\$\$
	Destine EPROM: Plant designation:	nation 62s5 #n####w ###############################
	EPROM capacity:	\$\$\$\$\$\$ words
_		
F1 F2 Program Eprom	F 3 F 4 F	5 F 6 F 7 F 8 HELP EXIT

: v

					_		
F 1	F 2	F 3	F 4	F 5	F 6	F 7	F 8
	- 1						
	1		1 1				ABORT
-	1				1	8	

Mask for COM 524

BASIC MASK -> FD P R O G R A M	-EPROM -> SIMATIC S5 / PROM525 E P R O M
	Module: CP524
Dri ve: Program:	Source Interface 1 Interface 2 Not present MDY C RK1
	Destination HLFB-number: 6ES5 373-1AA81 Submodule ID: 463 Plant designation: test ii EPROM capacity: 065536 words
F1 F2 PROGRAM EPROM	F3 F4 F5 F6 F7 F8 HELP EXIT

When pressing $\ensuremath{\textbf{F7}}$ (HELP) the following mask appears with a selection list.

BASIC MASK ->F P R O G R A M	D-EPROM E P R O M	SIMATIC S5 / PROM525
Drive:	Module Sou Interface 1	PLEASE SELECT: 6ES5 373-1AA41 414 6 2 s 5 373-0AA41 14 6ES5 373-0AA42 114
Program:	GG Destination MLFB-number: 6ES5 373-1AA8 Plant designation: EPROM capacity:	6ES5 373-1AA61 4 6 6 6ES5 373-0AA61 6 6 6ES5 373-0AA62 160 - 6ES5 373- 1AA81 4 63 6ES5 373-0AA81 163 -
FI F2	F3 F4 F5	F 6 F 7 F 8 ENTER EXIT

Using the arrow keys you can select en MLFB number and acknowledge it by pressing F6 (ENTER).

: F 1 F 2 / / / / F 8 ABBRUCH

First menu:

F1: setting up the directory of the user **programs** to be transferred.

A submask is displayed that indicates the memory required, the programing time end the current address. A check is carried out to establish whether en EPROM is

plugged in end whether the EPROM type 5.s permissible. A further check establishes whether the EPROM is empty.

The **programming is** started.

The second menu is displayed.

During the entire programming time, the **remaining** programming time is displayed as well as the current address and name of the element currently being processed.

The message 'completed' is then output.

Note: the actual memory space occupied is displayed. This includes the memory required for the manager and the actual program. The memory required always includes 2 x 8 Kbytes for the interpreter and procedures of both interfaces.

/

25

F8: return to the 'TRANSFER FD - EPROM' mask.

Second menu:

F8: after the current element is programmed, the transfer is aborted the next time the hard or floppy disk is accessed.

PROM 525

Field	Field type keys allowed	Limit value Alternatives
Module	OUTPUT FIELD	(default from the 'TRANS- FER FD – EPROM' mask)
Drive 1)	OUTPUT FIELD	(default from the 'TRANS- FER FD - EPROM' mask)
Program ¹⁾	OUTPUT FIELD	(default from the 'TRANS- FER FD – EPROM' mask)
Programing number	numbers	
MLFB number	OUTPUT FIELD	
Plant designation	any	
EPROM capacity	OUTPUT FIELD	
Required storage	OUTPUT FIELD	
Programming time	OUTPUT FIELD	
current address	OUTPUT FIELD	-

1) If two interfaces are being used, the specifications apply to the fields under 'interface 1 ' end under 'interface 2'.

م ہے ہے کہ مرکز کا کا ہو ہے کہ جات 19

3.6 Transferring **EPROM-->FD**

Call the **'TRANSFER** EPROM - **FD'** mask with the function key **F2** (TRANSFER EPROM->FD) in the **'BASIC MASK'**.

BASIC MASK -> TRANSFER EPRON - FD		SIMATIC SS	/ PROM52
	Source		
EPROM: \$ \$\$\$\$\$	6ES5 ######## s		
Plant designation: Generated by: Generated on: Component:	\$5555555555555555555555555555555555555		
	Dest inat ion		
Drive: Program:	L		
F1 F2 F3 F4 TRANSFER	F 5 F 6	F 7 HELP	F8 Exit

: v

F 1	F 2	F 3	F 4	F 5	F 6	F 7	F 8
							ASORT

First menu:

F3: checks whether an EPROM is plugged in and whether the EPROM type is permissible, end whether the EPROM contains a COM 525 program. The prompt "Start transfer EPROM->FD?" is displayed as well as the acknowledgement menu, With a positive acknowledgement: the transfer is started the second menu is output. During the transfer: the name of the element currently being processed is displayed. The message 'completed' is output. With negative acknowledgement: the first menu is displayed.

F8: return to the 'BASIC MASK'.

Second menu:

F8: abort the transfer,

ware and see in the second

B8576546-04

 $\mathcal{M} \in$

PROM 525

Field	Field type keys allowed	Limit	value	Alternatives
Programming number	numbers		-	-
MLFB number	OUTPUT FIELD			
Interface mumber (only with CP 525-2)	numbers			1/2
Plant designation	OUTPUT FIELD			
Generated by	OUTPUT FIELD			
Generated on	OUTPUT FIELD			
Component	OUTPUT FIELD			
Drive	upper case letters			dependent on device type *
Program	letters/ numbers			

F7 (HELP) can be used to make an entry in the fields marked $^{\prime\ast\prime}.$

3.7 Empty Test

Call the 'EMPTY TEST' mask with the function key F3 (EMPTY TEST) in the 'BASIC MASK'.

BASIC MASK -> E M P T Y T	EST						SIMATIC	S5 / prom525
	MLFB	-number:	6ES5 3	73-1AA	81 Subm	octule ID	: 463	
F 1 TEST	F 2	F 3	F	•	F 5	F 6	F ? HELP	F 8 Exit

F1: a check is made as to whether an EPROM is plugged in and whether the EPROM type is permissible, The empty test is started. During the empty test: the current address is output. The message 'EPROM is empty! ' or'EPROM not empty! ' is output at the end of the empty test,

F8: return to the 'BASIC MASK'.

B8576546-04

х.)

PROM 525

.

Field	Field type keys allowed	Limit	value	Alternatives
Programming number	numbers		-	-
MLFB number	OUTPUT FIELD			
Current address	OUTPUTFIELD			

مية فريد

1. Kan

3.8 INFO EPROM

Call the 'INFO EPROM' mask with the function key F4 (INFO EPROM) in the 'BASIC MASK'.

5 2 Y 4

14 LFB-number:	6ES5 373-1AA81 Submodule	D: 463
EPROM:	\$ s S S \$ S S \$ S	11: \$\$\$\$\$\$\$\$\$\$\$\$\$
Module:	S\$S\$\$Sss	12: \$\$\$\$\$\$\$\$\$\$\$\$\$\$\$
Firmware version	: SSS\$SSS	13: \$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$
Flant: Generated on	44444444444444444444 22222222	14: 333333333333333333333333333333333 15: seccessessessesses
Symbolic address:	4333333	16: SSSSSSSSSSSSSSSSSSSSSSSSSSSSSSSSSSSS
Ind. bus address:		17: S\$\$\$\$S\$S\$S'\$S\$\$SS\$S
Slave no. on PG bu	IS:	18: SSSSSSSSSSSSSSSSSSS
Password:	******	19: 5555555555555555
PROM version.	2222222	20: \$\$\$5555555555555555555555555555555555

- F1: a check is made as to whether an EPROM is plugged in and whether the EPROM type is permissible. The EPROM and module parameters (SYSID) of the EPROM are output.
- F8: return to the 'BASIC MASK' .

203

Field	Field type keys allowed	Limit value	Alternatives
EPROM (central field at top)	any		
EPROM (field on left)	OUTPUT FIELD		
Module	OUTPUT FIELD		
Firmware version	OUTPUT FIELD		
Plant	OUTPUT FIELD		
Generated on	OUTPUT FIELD		
symbolic address	OUTPUT FIELD		
Ind. bus address:	OUTPUT FIELD		
Slave no.	OUTPUT FIELD		
COM version	OUTPUT FIELD		
PROM version	OUTFUT FIELD		

```
B8576546-04
```

PROM 525

Field	Field type keys allowed	Limit value	Alternatives
Data for addition	onal identification:		
<pre>11 (component/ interface 1)</pre>	OUTPUT field	-	-
12(name of the interpreter)	OUTPUT FIELD	-	-
13 (version of the interpreter)	OUTPUT FIELD	-	-
14 (name of the procedure)	OUTPUT FIELD	-	-
15 (version of the procedure)	OUTPUT FIELD	-	-
<pre>16 (component/ interface 2)</pre>	OUTPUT FIELD	-	
17 (name of the interpreter)	output field	-	
18 (version of the interpreter)	OUTPUT FIELD	-	
19 (name of the procedure)	output field	-	
20 (version of the procedure)	OUTPUT FIELD	-	
21 ¹⁾	OUTPUT FIELD	-	

1) This field is not used with the current version of PROM 5250

٠

-.

.!!

3.9 INFO FD

Call the 'INFO FD' mask with the function key F5 (INFO FL)) in the *BASIC MASK'.

BASIC MASK -> INFOFD			SIMATIC 55	/ PROM525
	Drive: Program:	# #########		
	Plant designation: Generated by: Generated On:	\$		
	Component: COM525 version:	\$\$ \$\$\$\$\$		
F 1 F	2 F 3 F 4	F5 F6	F 7 HELP	F 8 Exit

F8: return to the 'BASIC MASK'.

Field	Field type keys allowed	Limit value	Alternatives
Drive	upper case letters		dependent on device type *
Program	letters/ numbers		*
Plant designation	OUTPUT FIELD		
Generated by	OUTPUT FIELD		
Generated on	OUTPUT FIELD		
Component	OUTPUT FIELD		
COM525 version	OUTPUT FIELD		

F7 (HELP) can be used to make en entry in the fields marked `*',

.

.

-

4 PROM 525 Messages

When you work with the PROM 525, messages are output in the message line (the line above the fiction key menu) for the following reasons:

-when an error has occurred (error message)

as information for the user, for **example** to **show** that an action has been carried out successfully (messages)

to prompt the user to confirm "yes" or "no" in the acknowledgement menu (acknowledgement prompts)

A message includes an identifier (e.g. $\ensuremath{\text{MF}}$, MESS), a number and a short text.

The following section lists and describes all the PROM 525 messages.



Message number	Description / Remedy
MF.001	 illegal key! Illegal characters have been entered in the field in which the cursor is positioned. Refer to the reference section of this user's guide to find the permissible characters.
MF.002	 illegal entry! The entry which has been made in the field on which the cursor is located does not correspond to any of the permissible alternatives, or it exceeds or falls below the permissible limit values. Refer to the reference section of this user's guide for the limit values and the permissible alternatives.

4.1 Messages Common to All Masks



اجي و الايانة جم ماياني. •

PROM 525

4.2 Messages

Message number	Description
MESS .001	Completed - The end of the transfer is displayed.
MESS .002	Element being processed: - The element named is currently being transferred.
MESS ,003	Aborted - The transfer FD->EPROM has been aborted as required.
mess .101	EPROM is empty! - Result of the empty test.
MISS .201	HELP will not help here! - Here only the name of an existing program should be specified, if it is to be overwritten.

. .

Message number	Description
MESS.202	<pre>EPROM->FD transfer aborted / file deleted! - `l'he transfer was aborted as requested, the file which was initialized will be deleted.</pre>
MESS.301	<pre>Memory required for #,#; # words - The length of the specified programs will be displayed.</pre>
MESS.302	Directories being established ,
MESS . 303	Memory required for #: # words - The length of the specified program will be displayed.
MESS.401	Transfer link being established (takes approx. 70 see)!
MESS.402	External EPROM programmer is ready!

4.3 Acknowledgement Prompts

Acknowledgement prompts are always displayed with the acknowledgement menu; they must be answered with Fl(YES) or F3 (NO), before PROM 525 will continue.

Acknowledgement prompt number	Description
QUIT.001	Exit mask?
QUIT .002	Exit PROM 525?
QUIT .003	FD has been changed - correct FD inserted again?
QUIT.004	Plug in correct EPROM or remove adapter!
QUIT. 005	Please plug in adapter and confirm!
QUIT. 006	Please check EPROM and confirm!

PROM 525

and a street is well the set of the

an an an an tha tha start an the start of th

۸.

Acknowledgement prompt number	Description
QUIT. 101	EPROM has been changed - correct EPROM plugged in?
QUIT. 102	EPROM not erased - empty EPROM plugged in?
QUIT.103	Break off programing?
QUIT.201	Program already exists - delete before transfer?
QUIT.202	Start trsnsf er EFROM->FD?

4.4 Error Messages

Error message number	Description
ERROR 001	No program name specified - The name of the program must be entered in the field on which the cursor is located.
ERROR 002	Invalid date - time specification! Refer to the reference section of this user's guide for the permissible entries.
ERROR 003	Illegal entry! - Refer to the reference section of this user's guide for the permissible entries.
ERROR 004	No drive specified! - The drive identifier must be entered in the field on which the cursor is currently located.
ERROR 005	Not a COM525 file! - The file specified does not contain a COM 525 program.

,, •.

التعمد بارم

-

Error message number	Description
ERROR 006	Program does not exist! - No program exists under the program name specified.
error 007	Library must not be processed - The name specified is not the name of a program but the name of a library.
error 008	Illegal key! - See MF.001
ERROR 101 ERROR 102	<pre>Drive cannot be addressed! - Possible causes: - the floppy disk has not been inserted in the specified drive - the drive has not been closed - the drive or the hard disk is defect.</pre>
ERROR 103	Aborted! - The floppy disk was changed during the programing.
Error message number	Description
-------------------------------	---
ERROR 104	Drive is write-protected! - Please remove the write-protect from the drive or from the floppy disk.
ERROR 105	File is write-protected!
ERROR 106 ERROR 107	FD is full! There is not sufficient space on the floppy or hard disk.
ERROR 108	EPROM not empty! - Result of the empty test.
ERROR 109	<pre>EPROM - read after write error! - The element which has just been programed does not retch the element in the file. Possible causes: - the EPROM submodule is defect - the EPROM interface is defect - the device is defect.</pre>

Error message number	Description	
ERROR 110	Wrong EPROM plugged in! - The EPROM number of the submodule plugged in does not match the entry in the mask,	
ERROR 111	Adapter not plugged in! - In order to use the 64 Kword sub- module, an MEP adapter must be plugged in.	
ERROR 112	Remove adapter !	
ERROR 113	<pre>S5-DOS diskette removed! - The floppy disk with the S5 operating system should have been removed (see Section 'Notes on the operating systems PCP/M-86 and S5 DOS' in this manual) .</pre>	
ERROR 114	 S5-DOS - drive cannot be addressed! Applies to the floppy or hard disk with the operating system, see ERROR 101. 	
ERROR 115	Hardware error! - EPROM submodule or device is defect.	

••••

Error message number	Description
ERROR 116	No readable program! - The SYSID of the element is not readable
ERROR 117	<pre>Interface not programed! - Enter a different interface number with F7 (HELP).</pre>
ERROR 118	 No space for interface directory! The capacity of the submodule is not sufficient for the interface directory.
ERROR 119 ERROR 120	EPROM does not have enough capacity! The capacity of the submodule is not sufficient for the specified program.
ERROR 121	<pre>EPROM type not intended for programing! - The submodule plugged in is not one of the three permitted module types (see Section `Introduction' in this user's guide).</pre>

47

,

** *

B8576546-04

,

Error message number	Description
ERROR 1.22	EPROM exchanged! - After reading SYSID a different EPROM submodule was plugged in.
ERROR 201	No program specified! - The name of the program must be entered in the field on which the cursor is positioned.
ERROR 202	<pre>Invalid component specified! - The program contains neither the component PT nor the component CL. Check (and if necessary change) the specification in COM 525.</pre>
ERROR 203	<pre># missing on interface #! - The program is not complete. The element named in the message is not present.</pre>
ERROR 204	No plant designation specified! - The plant designation must be entered in the field on which the cursor is positioned.

1

Error message number	Description	
ERROR 401	<pre>Error in transfer to external EPROM programmer! - Possible causes: - the device was switched off after initialization - the connecting cable is not plugged in correctly - the connecting cable is defect.</pre>	
ERROR 402	 Fault on external EPROM programmer! Possible causes: the device is not switched on the connecting cable is not plugged in correctly the connecting cable is defect. 	
ERROR 403	Error in transfer to external EPROM programmer ! an error has occurred in the transfer of the S5 files to the external programmer,	
ERROR 404	External EPROM programmer not ready! The transfer link could not be established. 	

B8576546-04

PROM 525

Error message number	Description	
ERROR 405 ERROR 406	<pre>S5 DOS not addressable! - One or more \$5-DOS files are not present. Reload the operating system \$5-DOS!</pre>	

PROM 525

an an an an an an an an Arthread an Art

4.5 Internal Errors

Error message

INTERNAL ERROR 301 to INTERNAL ERROR 336

If PROM 525 displays one of these error messages please contact **your** nearest **SIEMENS** service department or nearest technical office and inform them of the error message.

SIEMENS

We have checked the contents of 11118 manual for © graamwu with the hardware © nd software described. Since deviations cannot be precluded enlirely, we can, not guarantee full agreement. However, the data in this manual are reviewed regularly and © ny necessary cormotions included in subsequent editions. Suggestions for imp rovement are welcomed. Technical data subject to change.

Siemens Aktiengesellschaft

The reproduction, transmission or use 01 this document or its contents is not permitted without express written \blacksquare ulhorily.

Offenders will be liable for damages. All rights, including rights created by patent grant or registration of e utility model or design, are reserved.

Copy fight
Siemens AG 1989 All Rights Reserved

C79000-B8576-C546 Gerätawerk Karlsruhe Printed in the Federal Republic of Germany

SIEMENS

÷

SIMATIC S5 Using the Handling Blocks

Description

Order No. C79000-B8576-C547-06



Contents

i i

1	Introduction	1
2	The Handling Blocks with the Computer Link	2
2.1 2.2 2.3 2.4 2.5 2.6 2.7 2.7.1 2.7.2 2.8 E	Overview SYNCHRON SEND DIRECT n and FETCH DIRECT n SEND ALL and RECEIVE ALL CONTROL RECEIVE DIRECT n and RESET DIRECT n Evaluation The Parameter Assignment Error Byte (PAFE Byte) The Condition Codeword (ANZW) xample, Using a Coordination Flag	2 3 5 9 12 13 16 16 16 19 22
3	The Handling Blocks with the Printer	24
3.1 3.2 3.3 3.4 3.5 3.6 3.7 3.8 3.8.1 3.8.2	Overview SYNCHRON SEND DIRECT n SEND ALL CONTROL RECEIVE DIRECT n and RESET DIRECT 200 Examples - HDBs with Parameters Assigned Evaluation The Parameter Assignment Error Byte (PAFE Byte) The Condition Codeword (ANZW)	24 25 27 30 32 33 36 39 39 41

Page

یو بر راین کاهیو به ۲۰۰۰ نوان و بر ۱۹

,

i

1 Introduction

Handling blocks (HDBs) are standard function blocks in the STEP 5 user **program. HDBs** trigger jobs and carry out the data exchange between the CPU and the **CP** 525.

"Using the handling blocks" is intended as an **overview** and as a brief introduction to the functions available with the HDBs. It in no way **claims** to be comprehensive. The handling blocks are described in detail in their own instruction manuals (see /1/ in the List of further **relevant** documentation),

Please remember that you cannot use the CP 525 in association with the CPU 921 (S processor) of the S5-1350.

Style 8

2 The Handling Blocks with the Computer Link

ан алаан таралыкан алаан талаа талаа талаа калаа ка

2.1 Overview

You can call handling blocks in your STEP 5 program (mode of representation STL) . The following handling blocks are used with the computer link:

HDB	Function	S5-1150	S5-1350	S5-1500
SYNCHRON	synchronises CP and CPU	FB248	FB125	FB185
SEND DIRECT n	starts the SEND job with job number n	FB244	FB120	FB180
FETCH DIRECT n	starts the FETCH job with job number n	FB246	FB122	FB182
SEND ALL	transfers the data from the CPU to the CP	FB244 A-Nr. =0	FB120 A-Nr.=0 or FB126	FB180 A-Nr. =0
RECEIVE ALL	transfers the data from the CP to the CPU	FB245 A-Nr =0	FB121 A-Nr=0 or FB127	FB181 A-Nr . =0
CONTROL n	copies the job status of a job into the specified condition codeword	FB247	FB123	FB184
RECEIVE DI- RECT 200/ 218/221/223	only for special jobs	FB245	FB121	FB181
RESET DIRECT 200	resets the error entries in the SYSTAT	FB248	FB124	FB183

n = job number (value range 1 to 223)

n = job number (value range 1 to 223)
The HDBs are located as follows:

•S5-115U and S5-135U/CP 922 (R processor) and CPU 928

The HDBs are in the operating system; with the S5-115U they even have the block headers. With the S5-135U/CPU 922 (R processor) the block headers are on floppy disk and must be loaded in the PC.

•S5-150U and S5-155U:

The **HDBs** are available as a STEP 5 **program on** floppy disk and must be loaded in the PC.

2.2 SYNCHRON

This HDB synchronizes the interface between the CPU and CP during cold restart (OB 20), during a manual warm restart (OB 21) or in an automatic warm restart following power failure (OB 22). The SYNCHRON block must be called up for every CP interface in the start-up organization block of the CPU (OB 20, OB 21 and OB 22). During the synchronization, the maximum field length for the data transfer between the CPU and CP 525 is set.

Note on multiprocessor operation: it is sufficient when the **SYNCHRON** is programmed for each interface in the start-up organization blocks **(OB** 20, OB 21 and OB 22) of **one CPU**.

Tobesure that the interface **CPU/CP really** is synchronized, **observe the PAFE byte** of the **SYNCHRON (PAFE byte - see** Section 2.7.1).

Assigning parameters to SYNCHRON:

The following table shows a description of the parameters that you must specify when the **HDB** is called. The form and order of these parameters are the same that you see when **programming** on the'programmer. The letters x and y are variables to be replaced by values when you are programming. For the FB number (PB 125) the *number* of the CPU 922 (R **processor)/S5-135U** is used in this **example**.

Further information canbefoundinthe instruction manuals for **the handling blocks** (see /l/ in List of further **relevant documen**-tation).

SSNR:	<pre>KYx,y ;y= number of the required interface according to the jumper setting for the interface number on the CP (for interface number - see CP 525 instructions in this manual) value range: x > 0 : indirect parameter assignment (see /l/ in list of documentation)</pre>
	x = 0: y = interface number values from 0 to 255
BLGR:	<pre>KY0,y ;field length for data transfer between CP and CPU value range: : y = o to 255 significance: y= 0 field length (BLGR) - see</pre>
D	= 255:field length 51.2 bytes *)
*) mea	ans: differfrominf ormation in the instruction manuals for

*) means: differirominf ormation in the instruction manuals for the handling blocks: with the computer link max. 256 bytes are transferred.

2.3 SEND DIRECT n and FETCH DIRECT n

SEND DIRECT n (FETCH DIRECT n) triggers the **job with the job** number 'n' (n = 1 to 223). Jobs for triggering the processing of a job are termed **DIRECT jobs**.

For every DIRECT job with the job number 'n' (except special jobs) that you programed in the STEP 5 program, there must be a job with the same job number 'n' stored in the user memory of the **CP**. You program these jobs with the programming package COM 525.

You start a DIRECT job by calling SEND DIRECT n **or** FETCH DIRECT n. The job is then entered in the internal job queue in the **CP**.

Queue

The **CP** has an internal queue for each interface in which the maximum 10 DIRECT jobs (SEND DIRECT and **FETCH** DIRECT jobs) can be entered. The **CP** notes the order in which the jobs are initiated and entered in the queue, the corresponding handling block parameter and sets bit 1 "job running" in the condition codeword for each of these jobs.

The **CP** processes the jobs in the order in which they are entered in the queue. For SEND jobs and PSEUDO READ/WRITE jobs the **CP** then requests all the data from the CPU using a SEND ALL. When the job is completed, bit 2 "job finished without error" is set in the condition codeword, otherwise bit 3 "job finished with error" is set. If an error occurs, an error number is entered in bits 8 to 11 of the condition codeword and in the error message area of **SYSTAT**.

If there are already 10 jobs in the queue, every further job is rejected with the error number **7H** in the condition codeword, **15H** in the error message area of **SYSTAT** and **CH** in the parameter **assignment** error byte.

A job can only occur once in the queue, since there is only one job status maintained on the **CP** for each job number.

The CPU cannot query the number of jobs in the queue and jobs cannot be deleted from the queue. **After** a cold restart or when the **CP** is synchronized, all entries in the queue are deleted.

If a job depends on another job being processed without errors, you must include an interlock in your **STEP5** user program.

Jobs sent by the partner are processed outside the queue,

Special jobs are processed directly and are not entered in the queue. The **followin**g jobs are special jobs:

RESET DIRECT 200 RECEIVE DIRECT 200 SEND DIRECT 218 RECEIVE DIRECT 218 RECEIVE **DIRECT** 221 RECEIVE **DIRECT** 223

Jobs the partner sends are processed outside the queue,

Assigning parameters to SEND DIRECT n and FETCH DIRECT n

The following table shows description of the parameters that you must specify when the HDB is called. The **form** and order of these parameters are the same as those you see when you are programming on the programer. The letters x and y are variables to be replaced by values when programming. For the FB numbers the numbers of the CPU 922 (**R** processor) /S5-135U are used in this example.

Further **information** can be found in the instruction manuals for the handling blocks (see /l/ in List of further **relevan**t documentation).

```
B8576547-03
```

 ${\rm ed}_{\mathbf{a}}$

_ مقوم 1987 کې وې د ۲۰۱۶ - د ۲۰۱۹ کې د ۲۰۱۹ کې د ۲۰۱۹ کې د ۲۰۰ و. د ۲۰۰ و.

SEND DIRECT n: FB120 SSNR: KYx, y ; interface number value range: x > 0: indirect para. assignment see /1/ in list of documentation **x = 0:** y = interface number **values** from O to 255 (see instructions CP 525) ;mmber of the direct job A-NR: KY0, xvalue range: x=lto223 ;condition codeword (a doubleword is occupied) ANZW: XXY value range see /1/ in list of documentation xx = flag word (FW) or data word (DW) (caution: with DW, DB/ DX must be called first) y= number (dependent on xx) QTYP: KSyy ;source type value range yy = DB,DX,CB,TB,BS,AS,FY,QB,IB,OB or **PB** (only these permitted) XX indirect para. assignment see /1/ in the list of documentation (Note: with PSEUDO READ/WRITE function A-Nr. 190 to 199 - DB or DX allowed. See also special jobs) DBNR: KYo,x ;source data block no. with QTYP DB, DX, (XX see value range: x=3t0255 above) with other QTYP parameter assignment irrelevant QANF: KF+x ;startaddress of the source value range: x=dependentonQTYP and PC type; see job tables in the user's guide "Computer link with RK 512" QLAE: KF+x ;source length - numberofdatatobetrsnsferred value range: x = dependent on QTYP und PC type; see job tables in the user's guide "Computer link with RK 512" ;parameter assignment error PAFE: xxy value range: see /1/ in list of documentation

FETCH DIRECT n: FB122

SSNR: KYx, y ; interface number value range: x > 0: indirect pars. assignment see /1/ in list of documentation $\mathbf{x} = 0$: $\mathbf{y} = \text{interface number values}$ from O to 255 (see instructions CP 525) A-NR: KYO,x ;number of the direct job value range: x = 1 to-223 ANZW: XXY ; condition codeword (a **doubleword** is occupied) value range see /1/ in list of documentation xx = flag word (FW) or data word (DW) (caution: with DW, DB/ DX must be called first) y = number (dependent on xx) ZTYP: KSyy ;destination type value range yy = DB,DX,RS,AS XX indirect para. assignment see /1/ in the list of documentation (Note: with PSEUDO READ/WRITE function (A-Nr. 190 to 199) only DB or DX allowed. See also special jobs) DBNR: KYO, x ; dest. data block no. with dest. type DB ,DX (XX see above) value range: X = 3 to 255 with other ZTYP parameter assignment irrelevant ZANF: KF+x ; start address of the destination **value** range: x = dependent **on ZTYP** and PC **type**; see job tables in the user's guide "Computer link with RK 51.2" ZLAE: KF+x ;number of data to be transferred value range: x . dependent on ZTYP und PC type; see job tables in the user's guide "Computer link with RK 512" PAFE: xxy ;parameter assignment error value range: see /1/ in list of documentation

and the second second and the second second

2.4 SEND ALL and RECEIVE ALL

1917 - Construction of the State of the Stat

These jobs are responsible for the data exchange between the CP 525 and the CPU. For ALL functions, the HDBs are assigned the job number O.

First, the HDB SEND DIRECT n or FETCH DIRECT n triggers a job on the CP 525. When the job is triggered, the CP 525 stores the source or destination parameters you assigned to the SEND DIRECT n or FETCH DIRECT n.

Then the CP 525 performs the job, i.e. it **sends** an **"ALL request"** to the CPU. It informs the CPU of the required data source or data destination. The SEND ALL then transfers the requested data from the CPU (source) to the dual-port RAM of the CP. The RECEIVE ALL transfers the data **from** the dual-port **RAM** to the CPU (destination). With larger amounts of data, several ALL functions may be necessary to transport the data (see field length).

The source/destination parameters that you must specify for the **direct** functions are **irrelevant** for **ALL** functions (any values can be entered here).

Note: if you have a long cycle time with your STEP 5 program, it may be a&sable to call up SEND ALL and RECEIVE **ALL** several times per cycle. SEND ALL and RECEIVE ALL calls extend your cycle time only when there is actually data to be transferred.

The SEND ALL and RECEIVE ALL HDB's in association with partner jobs

These HDBs are used not only to transport data for jobs from their own CPU, but when working with the computer link, they also transfer **data** the partner is sending or fetching **on** its own initiative.

If the partner sends a telegram, the **RK** 512 interpreter analyses it. If it recognises that the partner wishes to send or fetch data, it extracts the source and destination **parame**ters **contained** in the telegram header. **The** interpreter sends an 'ALL request" to the CPU so that the information data can be transferred.

.

4

The ALL HDB then executes the data exchange between the CPU and dual-pm% RAM.

Assigning parameters to SEND ALL and RECEIVE ALL:

The following tables contain a description of the parameters that you must specify when the HDB is called. The foxm and order of these parameters are the same that you see when programming on the PG. The letters x and y are variables that must be replaced by values when you program. For the FB numbers the numbers of the CPU 922 (R processor)/S5-135U are used in this example.

Further information can be found in the instruction manuals for the handling blocks (see /1/ in List of further relevant documentation).

SEND ALL: FB126

(with the **S5-115U/150U** the call of the SEND DIRECT **n with** job number 0 is used. The specifications for the source parameters are then unnecessary).

SSNR:	KYx,y	; interface number
		value range: x > 0: indirect para . assignment
		see /1/ in list of
		documentation
		x = 0: $y = interface$ number values
		from O to 255
A-NR:	KY0,0	;mmber of the job
ANZW:	xxy	;condition codeword value range: see /1/ in list of documentation
		xx = flag word (FW) or data word
		(DW) (caution: with DW, DB/
		DX must be called first!)
		y =number (dependent on xx)
PAFE:	xxy	;parameter assignment error
		value range: see /1/ in list of d ocumentation

9.

RECEIVE ALL: FB127

(with the S5-1150/1500, the call of the RECEIVE DIRECT n with job number 0 is used. The specifications for the source parameters are then unnecessary).

SSNR:	KYx,y	;interface number
		value range: x > 0: indirect para. assignment
		documentation
		x = 0: y = interface number values
		from 0 to 255 (see in- structions CP 525)
A-NR:	KY0,0	;number of the job
ANZW:	xxy	;condition codeword
		value range: see /l $\$ in list of documentation
		xx = flag word (FW) or data word
		(DW) (caution: with DW, DB/
		DX must be called first.)
		y =number (dependent on xx)
PAFE:	xxy	;parameter assignment error
		value range: see /1/ in list of documentation

*

A > 1

Incorrection a

2.5 CONTROL

The CONTROL HDB copies **the** job status of a job into the **specified condition** codeword. The job status provides information about operating or processing statuses.

CONTROL can be called at any point in the program.

_ parameters to CONTROL:

The following table shows a description of the parameters that you must specify when the HDB is called, The form and order of these parameters are the same that you will see when you program on the PG. The letters x and y are variables that must be replaced by values when you program. For the FB number (FB123), the number of the CPU 922 (R processor) /S5-135U is used in this example.

Further information can be found in the instruction manuals for the handling blocks (see /1/ in List of further relevant documentation) .

SSNR:	KYx, y	; interface number
		<pre>value range: x > 0: indirect para. assignment</pre>
		<pre>x = 0: y = interface number values from 0 to 255 (see in- structions CP 525)</pre>
A-NR:	KYO , x	;mmber of the job to be controlled
		value range: x = 1 to 223
		<pre>x = 0 number of the job currently being processed is copied into the ANZW</pre>
ANZW:	жху	;condition codeword value range: see /1/ in list of documentation
		🞞 = flag word (FW) or data word
		(DW) (caution: with DW, DB/

DX must be called first!)
y = number (dependent on xx)

B8576547-03

1.1. **. . .**

ag v

2.6 RECEIVE DIRECT n and RESET DIRECT 200

These are used only with special jobs. The following special functions are carried out with RECEIVE DIRECT n:

reading the error message area in SYSTAT
reading the whole status area SYSTAT
reading the identification area. SYSTAT
reading the date and time

The special job RESET DIRECT n allows you to reset the error entries in **SYSTAT**.

The **description** of the special jobs can be found in the user's guide "Computer link with **RK 512**" in this manual.

Assigning parameters to RECEIVE DIRECT n and RESET DIRECT n:

The following two tables give a description of the parameters that you must specify when the HDB is called. The form and order of these parameters are the same as those you see when you program on the programmer. The letters x and y are variables that must be replaced by values when you program. For the FB numbers the **numbers** of the CPU 922 (R processor) /S5-135U are used in this example.

Further information can be found in the instruction manuals for the handling blocks (see /1/ in List of further relevant documentation).

13

، الإين - ي د المدينة محمد محمد الداري

,

B8576547-03

والمراجع والمراجع والمحادث

14

RECEIVE DIRECT n: FB121 - only for special jobs SSNR: KYx,y ;interface number value range: x > 0: indirect para. assignment see /1/ in list of documentation x · 0: y = interface number values from O to 255 (see instructions CP 525) A-NR: KY0,x ;numberof the direct job value range: x = 200, 218, 221, 223 (callable special jobs) ANZW: xxy ;condition codeword (a doubleword is occupied!) value range see /1/ in list of documentation xx = flag word (FW) or data word (DW) (caution: with DW, DB/ DX must be called first) y = number (dependent on xx) ZTYP: KSyy ;destination type value range yy = DB orDX XX indirect para. assignment see /1/ in the list of documentation (special jobs) DBNR: KY0,x ;dest. **DB** no. with **ZTYP** DB, DX; (XX see above) value range: x = 3t0255ZANF: KF+x ;start address of destination **value** range: **x** = 0 to length of dest. **type** minus ZLAE ZLAE: KF+x ;number of data to be transferred value **range:** x= dependent on special job PAFE: xxy ;parameter assignment error value *range: see /l/ in* the list of **docum**.

en an attact and an

.

њ.,

. Here is a presentation of the second state of the second second second states ~ 200 km s ~ 200 km s ~ 200

 DIRECT 200: FB124 - only for special jobs
 SSNR: KYx,y ;interface number value range: x > 0: indirect para. assignment see /1/ in list of documentation x = 0: y = interface number values from 0 to 255 (see instructions CP 525)
 A-NR: KY0,200 ;number of the direct job (special job)
 PAFE: xxy ;parameter assignment error

value range: see /1/ in the list of docum.

· .

.

1418

2.7 Evaluation

The following sections contain information about how to evaluate the parameter assignment error byte (PAFE byte) and the condition codeword ANZW.

A. 4

2.7.1 The Parameter Assignment Error Byte (PAFE Byte)

When using the handling blocks SEND DIRECT n, FETCH DIRECT n, RECEIVE DIRECT n, **RESET** DIRECT 200, SEND ALL, RECEIVE **ALL** and **SYNCHRON**, you must specify a **parameter assignment error byte** (**PAFE-BYTE**).

If the HDB recognizes an error, it writes the corresponding error number into the **PAFE** byte. If the block runs through without errors, the HDB writes OOH into the PAFE byte.

In addition to the errors that can be attributed directly to the parameter assignment, the PAFE byte also indicates errors that occur during the communication CPU/CP 525. If such errors occur you should immediately remedy the situation to avoid other messages appearing during operation.

In certain exceptional cases, errors are indicated in the **PAFE** byte during **operation**. These errors are caused by the software of the **CP** 525 in special situations as follows:

if more than 10 DIRECT jobs are processed in the cycle, i.e. 10 jobs are already signalled as "running" and you want to start an eleventh job

if the STEP 5 program sets the $\ensuremath{\text{IPC}}$ (coordination flag) flag when the CPhas just received a telegram for this coordination flag

if the whole or part of the **CP** 525 user program is transferred to the **CP** 525 with **COM** 525 during normal operation

if a SEND or FETCH job is started before the SYNCHRON job has run through once without errors

During multiprocessor operation you can also receive a PAFE message if several CPUs access the same CP 525 interface at the same time. The access is prevented (with a PAFE message) until the current CPU/CP communication is completed. This PAFE message does not indicate an error but shows that access is momentarily blocked.

The following list provides a brief explanation of the **PAFE** error messages. Note that the significance of each error number depends on the PC' type. Check the **instructions** for the handling blocks for your PC (see /1/ in the list of further relevant documentation).

PAFE	no.	Significance
------	-----	--------------

- 11H Wrong ORG format: source/destination parameter type wrong, area (start address, length) not permissible.
- 21H DB/DX or memory area not present,
- **31H** Source/destination too small.
- **41H** Source/destination area not present (not plugged in) Timeout (QVZ) from this area,
- 51H Error in condition codeword.
- 61H No source/destination parameters with SEND/RECEIVE ALL.
- 71H Interface does not exist (QVZ for the CP 525). Check the jumper settings for the interface number on the CP 525.
- 81H Interface no: ready or not synchronized. Check that the SYNCHRON HDB is programmed in the start-up OBS and runs through without errors.
- **91H** Interface overload; part of a program is being transferred from the PC to the **CP** 525.

- 91H Interface overload; part of a program is being transferred from the PC to the CP 525.
- AlH CP interface being used by a different CPU in multiprocessor operation (S5-135U and S5 155U) when several CPUs access one CP interface.
- BlH Job number too high.
- ClH Error in handshake with the CP 525:
 - CP does not reply within the monitoring time
 - or CP rejects handshake, e.g. because more than 10 jobs were triggered simultaneously (when 10 jobs are running, bit 1 is set in the ANZW) or
 - the coordination flag was set when a telegram was received for this coordination flag
- D1H Field length (SYNCHRON) wrong and group error for other handshake errors.
- ElH DB call missing with indirect parameter assignment and group error for HDB software error.
- FlH Double block call (when a block can be interrupted at command boundaries).

2.7.2 The Condition Codeword (ANZW)

The condition codeword with direct jobs:

The first part of the condition codeword contains **information** about the status of the job processing - the second part contains **information about** the length of the data transferred.

If the direct jobs SEND DIRECT n, FETCH **DIRECT** n, RECEIVE DIRECT n are being used, the **condition** codeword requires a **doubleword**.

Significance of the bits in the condition codeword of a DIRECT job:

- Bit 0 : RECEIVE job **ready** (handshake acceptable) O = RECEIVE disabled 1 = RECEIVE enabled
- Bit 1 : job **running** O = SEND/FETCH enabled **1** = SEND/FETCH disabled
- Bit 2 : job complete without error
- Bit 3 : job complete with error
- Bit 4 : data transfer/data reception running
- Bit 5 : data transfer **complete** Bit 6 : data reception complete
- Bit 7 : 0 = data transfer/reception enabled 1 = data transfer/reception disabled
- Bit 8-11 : error messages see the error tables in the **user's** guide 'Computer link with **RK** 512" in this manual
- Bit **12-15** : not used

^{سو}می پر استان دانده مراجع دارد. ا

,

B8576547-03

and and a second se

In the second half of the condition code **doubleword**, the SEND **ALL** and RECEIVE ALL enter the. length of the transferred data. SEND ALL enters the number of pieces of data of the current **trans**mission, and RECEIVE ALL adds the current number of pieces of data to the number of pieces of data already received. If several active **ALL** functions are required for a RECEIVE ALL **to carry out** the **direct** job, the value in the second part of the condition code **doubleword** increases cent**inuously**.

Every direct job should have its own condition code **doubleword** reserved. The following areas can be used:

- the flag area FW 0 to 252 or
- the data words O to 254 in the currently open data block (DB or DX).

To **ensure** reliable data exchange it is sufficient with the CPU 922/S5 135U when the STEP 5 program **evaluates** bits 1, 2 and 3 of the **ANZW**:

- bit 1 indicates that the job was accepted for processing by the **CP**. The job can be triggered again only after bit 1 has been reset
- bits 2 and 3 indicate that the CP completed processing the job.

On completion of the job without errors (bit 2 = 1), new information data can be prepared for the next transfer or the data arriving can be processed.

If the job is terminated with an error (bit 3 = 1), the error number is entered in bits 8 to 11 of the ANZW. The list of error numbers can be found in the user's guide "Computer link with RK 512" in this manual.

20

يىغ مۇرى كەرى كەرى بەرىي بىرىيە « بەرە» - يەرە

The condition codeword with ALL jobs:

In **ALL** jobs, RECEIVE **ALL** or SEND ALL, the condition codeword requires only **one** data word or flag word.

While the ALL function is **running**, the number of the DIRECT **job** for which the data is being transferred is indicated in the **ANZW**. The value O is entered if the function is run through without any data transfer.

If the ALL function is active for its own CPU, it also updates the condition codeword of the corresponding direct job.

If the ALL function is active for a partner job it indicates the byte number of the coordination flag specified in the telegram in the AM?. This number is treated as a job number by the HDB and must therefore have values only between 1 and 223. If no coordination flag was specified (or the number is too high) FFH is entered in the ANZW.

If several ALL functions are required for a partner job with a coordination flag, the byte **number** of the coordination flag is entered in the **ANZW** only with the last data transfer. From this it is possible to recognise that the partner job is complete.

Then you can set the **IPC** flag (which corresponds to the coordination flag) with the STEP 5 program and process the information data that has arrived (if the partner **activated** a SEND job) or make new information **data** available (if the partner has activated a FETCH job). The set coordination flag prevents the partner accessing the data area again.

When the IPC flag is reset, this data area can be accessed tmce again.

2.8 Example, Using a Coordination Flag

A SEND job, sent by the Partner is to be protected against overwriting at the receiver by coordination flag 101.0 in the STEP 5 program as follows:

 The IPC flag must be enabled on the receiver CP by means of jumper settings, i.e. in this example jumper 5-12 cm jumper block number 25 cm the CP must be inserted. Correspondingly, the flag byte 101 in DB 1 of the S5-115U, S5-135U and S5-155U or with standard FB 186 in the S5-150U must be defined as an IPC output flag in the STEP 5 program.

2. Evaluation in the PC cycle of the receiver:

e.g. : FB 1 NAME :AUSWERT	FB to be written to by CP 525
:.	any program
:JU FB127 NAME :REC-ALL	RECEIVE ALL call
ANZW: FW12	pa rameter assignment of condition codeword
:L FY13 :L KF+101 :!=F	load low byte of condition codeword load byte number of coordination flag corresponding telegram received?
: S F 101.0	set coordination flag
:.	evaluation of the data
: A F 1 . O :R F101. O : .	all data evaluated i.e. enable coordination flag
: . :BE	program continues

The procedure is similar to prevent data being prepared for a FETCH job from being accessed. In the programmable controller from which the data is to be fetched, the coordination flag must be evaluated in the ANZW of the SEND ALL. The FETCH job of the partner does not receive any data as long as the coordination flag is set.

Lan

and and an and a second se

NUMBER AND TRANSPORTED AND A DESCRIPTION OF A DESCRIPTION OF

4. 2023

3 The Handling Blocks with the Printer

ويصفوننا فيادين والدار والات

3.1 Overview

You CSN call handling blocks in your STEP 5 program (mode of representation STL). The following handling blocks are used with the printer:

HDB	Function	S5-11 <i>5</i> 0	S5-1350 S5-1550	S5-1500
SYNCHRON	synchronises CP and CPU	FB248	FB125	FB185
SEND DIRECT n	starts the SEND job with job number n	FB244	FB120	FB180
SEND ALL	trsnsfers the data from the CPU to the CP	FB244 A-Nr. =0	FB120 A-Nr.=0 >r FB126	FB180 A-Nr. =0
CONTROL n	copies the job status of a job into the specifies condition codewors	FB247	FB123	FB184
RECEIVE DI- RECT 200/ 218/221/223	only for special jobs	FB245	FB121	FB181
RESET DIRECT 200	200 resets the error entries in the SYSTAT		FB124	FB183

n = job number (value range 1 to 223)

The HDBs are located as follows:

S5-115U and S5-135U/CPU 922 (R processor) and CPU 928:

The HDBs are in the operating system; in the S5-115U they have block headers. With the S5-135U/CPU 922 (Reprocessor) the block headers are on floppy disk and must be loaded in the Pc.

S5-150U and **S5-155U**:

The **HDBs** are available as a STEP 5 **program on** floppy disk and must be **loaded** in the PC.

3.2 SYNCHRON

This HDB synchronizes the interface between the CPU and CP during cold restart (OB 20), during a manual warm restart (OB 21) or in an automatic warm restart following power failure (OB 22). The SYNCHRON block must be called up for every CP interface in the start-up organisation block of the CPU (OB 20, OB 21 and OB 22). During the synchronization, the maximum field length for the data transfer between the CPU and CP 525 is set.

Note on multiprocessor operation: it is sufficient when the SYNCHRON is programmed for each interface in the start-up organisation blocks (OB 20, OB 21 and OB 22) of one CPu.

To be sure **that the** interface **CPU/CP** really is synchronised, observe the **PAFE** byte of the **SYNCHRON** (**PAFE** byte - see Section 3.8. 1) .

Assigning parameters to SYNCHRON:

The following table contains a description of the parameters you must specify when the HDB is called. The form and order of these parameters are the same as those you see when you program on the PG. The letters x and y are variables which must be **replaced by** values when you program. For the FB number (FB 125) the **number** of the CPU 922 (R **processor)/S5-135U** is used in this example.

Further information can be found in the instruction manuals for the handling blocks (see /1/ in List of further relevant documentation).

SSNR: KYx,y ;y =number of the required interface according to the jumper setting for the interface number on the CP (for interface number - see CP 525 instructions in this manual) value range: x > 0: indirect parameter assignment (see /1/ in list of documentation) $\mathbf{x} =$ **0:** y = interface number values from O to 255 (see instructions CP 525) BLGR: KYx,y ;field length for data transfer between CP and CPU value range:: x= 0 y = 0 to 255 significance: y = 0 field length (BLGR) - see /1/ in list of docum. = 1: field length 16 bytes = 2: field length 32 bytes · 3: field length 64 bytes • 4: field length 128 bytes 5: field length 256 bytes = 6: field length 512 bytes **7** to 254: field length 512 bytes = 255:field length 512 bytes PAFE: xxy ;parameter assignment error - see /1/ in list of &~.
3.3 SEND DIRECT n

•/ --

حويري بالانتراجية فالعام والمروان

.

SEND DIRECT n triggers the job with the job number 'n' (n = 1 to 223), The job for triggering the processing of a job is termed **DIRECT job**.

For every DIRECT job with the job number 'n' (except special jobs) you programmed in the STEP 5 program, there must be a job with the same job number 'n' stored in the user memory of the CP. These jobs are programmed with the programming package COM 525.

You start a DIRECT job by calling SEND DIRECT n. The job is then entered in the internal job queue in the **CP**.

Queue

The **CP** has an internal queue for each interface in which the maximum **10** DIRECT jobs (SEND DIRECT jobs) can be entered. The **CP** notes the order in which the jobs are initiated and entered in the queue, the corresponding handling block parameter and sets bit 1 "job running" in the condition codeword for each of these jobs.

The **CP** processes the jobs in the order in which they are entered in the queue. With jobs for sequential message logs and current message logs the **CP** then requests all the data from the CPU using a SEND AIL. When the job is completed, bit 2 "job finished without **error"** is set in the condition codeword, otherwise bit 3 "job finished with error" is set. If an error occurs, an error **number** is entered in bits 8 to 11 of the condition codeword and in the error message area of **SYSTAT**.

If there are already 10 jobs in the queue, every further job is rejected with the error number 7H in the condition codeword, **15H** in the error message area of SYSTAT and **CH** in the **parameter** assignment error byte.

A job can only occur **once** in the queue, since there is only one job status maintained on the **CP** for each job number.

The CPU cannot query the number of jobs in the queue and jobs cannot be deleted from the queue. After a cold restart or when the CP is synchronized, all entries in the queue are deleted.

If a job depends on another job being processed without errors, you must include an interlock in your **STEP5** user program.

Special jobs are processed directly and are not entered in the queue. The following jobs are special jobs:

RESET DIRECT 200 RECEIVEDIRECT200 SEND DIRECT 218 RECEIVE DIRECT 218 RECEIVE DIRECT 221 RECEIVEDIRECT223

Assigning parameters to SEND DIRECT n

The following table contains a description of the parameters **that** you must specify when the HDB is called. The form and order of these parameters are the same that you see when you program on **'the** PG. The letters xandy are variables that must be replaced by values when you program. For the FB number (**FB120**) the number of the CPU 922 (R **processor)/S5-135U** has been used in this **example**.

Further information can be found in the instruction manuals for the handling blocks (see /l/ in List of further relevant documentation).

28

متحوري وادحاد رامواد والمعتمين والمحوو حاجات والا

,

SEND DIRECT n: FB120 SSNR: KYx,y ;interface number value range: x > 0: indirect para. assignment see /1/ in list of documentation x = 0: y = interface number values from O to 255 (see instructions CP 525) A-NR: KYO,x ;number of the direct job value range: x = 1 to 223 ANZW: xxy ;condition codeword (a doubleword is occupied) value range see /1/ in list of documentation xx = flag word (FW) or data word (DW) (caution: with DW, DB/ DX must be called first) y = number (dependent on xx) QTYP: KSyy ;source type value range yy= DB (only data block permitted) NN (only job initiation, no data transfer) XX indirect **para**. assignment see /1/ in the list of documentation DBNR: KYO,x ;source data block no. value range: $x = 3 \pm 0.255$ with other $\ensuremath{\ensuremath{\mathsf{QTYP}}}$ parameter assignment irrelevant QANF: KF+x ;start address of the source value range: x=Odatatransfer always from DWO QLAE: KF+x ;source length - number of data to be transferred value range: $x = 1 \pm 0.256$ PAFE: xxy ;parameter assignment error

29

value range: see /1/ in list of documentation

.

3.4 SEND ALL

This is **responsible** for the **data** exchange between the **CP** 525 and **the CPU.** For the ALL function, the **HDB** is assigned the job number 0.

First, the **HDB** SEND DIRECT n triggers a job on **the CP** 525. When the job is triggered, the **CP** 525 stores the source parameters which you have assigned to the SEND **DIRECT** n.

Following this, the **CP** 525 carries out the job, i.e. it sends an "ALL request" to the CPU. It informs the CPU of the required data source. The SEND ALL then transfers the requested data from the CPU (source) to the dual-port RAM of the CP. With larger amounts of data, several ALL functions may be necessary in order to transport the data (see field length).

The source parameters which you must specify for **the** direct functions are irrelevant for ALL functions (any values can be entered here).

Note: if you have a long cycle time with your STEP 5 program it may be advisable to call up SEND **ALL** several times per cycle. SEND ALL calls only extend your cycle time when **there is** actually data to be transferred.

Assigning parameters to SEND ALL:

Below, there is a description of the parameters that you must specify when the HDB is called. The form and order of these parameters are the same as those you see when you program on the PG. The letters x and y are variables to be replaced by values when you program. For the FB number (FB 126) the number of the CPU 922 (R processor)/S5 135U is used in this example.

Further information can be found in the instruction manuals for the handling blocks (see /1/ in List of further relevant documentation).

SEND ALL: FB126

(With the S5-115U/150U the call of the SEND DIRECT $n\ with$ job number 0 is used. The specifications for the source parameters are then unnecessary.)

SSNR:	;interface number	
		value range: x > 0: indirect para. assignment
		see /1/ in list of
		documentation
		x = 0: $y = interface$ number values
		from O to 255 (see in-
		structions CP 525)
A-NR:	KYO, x	;number of the job
		value range: x = 0 for ALL function
ANZW:	xxy	;condition codeword
		value range: see /1/ in list of documentation
		xx = flag word (FW) or data word
		(DW) (caution: with DW, DB/
		DX must be called first!)
		y = number (dependent on xx)
PAFE:	жху	;parameter assignment error
		value range. see /1/ in list of documen tation

600

3.5 CONTROL

The CONTROL HDB copies the job status of a job into the specified condition codeword. The job status provides information about operating or processing statuses.

CONTROL can be called at any point in the program.

Assigning parameters to CONTROL:

The following table contains a description of the parameters that you must specify when the HDB is called. The form and order of these parameters is the same that you see when you program on the PG. The letters x and y are variables that must be replaced by values when you program. For the FB number (FB123) the number of the CPU 922 (R processor)/S5-135U has been used in this example.

Further information can be found intheinstruction manuals for the handling blocks (see /1/ in List of further relevant documentation).

SSNR:	SNR: KYx,y ;interface number			
		<pre>value range: x > 0: indirect para. assignment</pre>		
		<pre>x = 0: y = interface number values from 0 to 255 (see in- structions CP 525)</pre>		
A-NR: KYO,x ;mumber of the job to be control.		;mmber of the job to be controlled		
		value range: x=lto223		
		x=0 numberofthejob currently		
		being processed is copied		
		into the ANZW		
ANZW: xxy ; condition codeword		;condition codeword		
		value range: see /1/ in list of documentation		
		x = flag word (FW) or data word		
		(DW) (caution: with DW, DB/		
		DX must be called first)		
		y =number (dependent on xx)		
		The condition code of the job to be		
controlled must be specified.				

3.6 RECEIVE DIRECT n and RESET DIRECT 200

These are only used with special jobs. The following special functions are carried out with RECEIVE DIRECT n:

reading the error message area in SYSTAT
reading the whole status area SYSTAT
reading the identification area SYSTAT
reading the date and time

The special job **RESET** DIRECT n allows you to:

- reset the error entries in SYSTAT

The description of the special jobs can be found in the instructions "Event output and listing with the PT88/PT89 printer" in this manual.

Assigning parameters to RECEIVE DIRECT n and RESET DIRECT n:

The following table contains a description of the parameters that you must specify when the HDB is called. The form and order of these parameters are the same that you see when you program on the PG. The letters x and y are variables that must be replaced by values when you program. For the FB numbers the numbers of the CPU 922 (R processor)/S5-135U have been used in this example.

Further information can be found in the instruction manuals for the handling blocks (see /1/ in List of further relevant documentation).

. ...

والمنطاعي بين بها فالرقيل فالماط مام بما المربو و

•

RECEIVE DIRECT n: FB121 - only for special jobs! SSNR: KYx, y ; interface number value range: x > 0: indirect para. assignment see /1/ in list of documentation x = 0: y = interface number values from O to 255 (see instructions CP 525) A-NR: KY0, x ;number of the direct job value range: x = 200, 218, 221, 223 (callable special jobs) ANZW: xxy ;condition codeword (a doubleword is occupied) value range see /1/ in list of documentation xx = flag word (FW) or data word (DW) (caution: with DW, DB/ DX must be called first) y = number (dependent on xx) ZTYP: KSyy ;destination type valuerangeyy=DB XX indirect para. assignment see /1/ in the list of documentation (special jobs) DBNR: KYO,x ;dest. DB no, with ZTYP DB **value** range: x = 3 to 255 ZANF: KF+x ;start address of destination value range: x = 0 to length of dest. type minus ZLAE ZLAE: KF+x ;number of data to be transferred **value range: x** = dependent on special job PAFE: xxy ;parameter assignment error

34

value range: see /1/ in the list of docum.

RESET DIRECT 200: **FB124** - only for special jobs

SSNR: KYx,y ;interface number value range: x > 0: indirect para. assignment see /1/ in list of documentation x = 0: y = interface number values from 0 to 255 (see instructions CP 525)
A-NR: KY0,x ;number of the direct job value range: x = 200 (special job)
PAFE: xxy ;parameter assignment error

value range: see /1/ in the list of docum.

3.7 Examples - HDBs with Parameters Assigned

You start a job by calling the HDB SEND DIRECT n cn the CPU (in your STEP 5 user program). The parameters to be assigned to the HDB SEND DIRECT n depend on the type of job to be started.

Jobs without data transfer

This group includes the following PC jobs

- new page
- page no. = 1

The HDB SEND DIRECT n must have parameters assigned as follows:

: JU FB120 NAME :SEND SSNR : KYO,0 A-NR : KYO,10 ANZW : FW6 QTYP : KSNN DBNR : KYO,0 QANF : KF+0 QLAE : KF+0 PAFE : FY199

.

GENERATE RLO = 1 !!

CP 525 INTERFACE 1 JOB NUMBER 10 NO DATA TRANSFER

IRRELEVANT SINCE NO DATA IRRELEVANT SINCE NO DATA IRRELEVANT SINCE NO DATA PARAM. ASS. ERR. N IN FY199

Under the job **number above** (here: 10) you must program the **required** job **belonging to this group** in your **CP 525** user **program and** transfer it to the corresponding interface.

× 61.

Jobs for which data are fetched from the CPU

This group includes:

- process status list (XL)
- chained list (CL)
- update group inhibit bits (update GIB)
- update sequential message list (update SML)

To execute these jobs the ${\tt CP}$ 525 requires data from the CPU. Where the data is to be fetched from is decided when you program the job with COM 525.

The HDB SEND DIRECT n must have parameters assigned as follows:

GENERATE RLO = 1 !!

:JU FB120 NAME :SEND	
SSNR : KYO,O	CP 525 INTERFACE 1
A-NR : KY0,10	JOB NUMBER 10
ANZW : FW6	
QTYP : KSNN	INFORMATION COMES FROM CP 525
DENR : KYO,O	I RRELEVANT
QANF : KF+0	I RRELEVANT
QLAE : KF+0	IRRELEVANT
PAFE : FY199	PARAM. Ass. ERROR IN FY199

Under the job number above you must program the required job belonging to this group in your CP 525 user program and transfer it to the corresponding interface.

37

.

•

.

Jobs for which data are sent by the CPU

These PC jobs include the following:

sequential message list (sML)current message list (CML)

The CP 525 requires data from the CPU to execute these jobs. Which data the CPU is to send is decided when you assign <code>parameters</code> to the SEND DIRECT n.

ي المعربة ومراجعهم

The HDB SEND **DIRECT** n must have parameters **assign**(d as follows:

	•		
			GENERATE RLO = 1. ! !
	:		
	:JU	FB120	
NAME	:SEN	D	
SSNR	:	kyo ,0	CP 525 INTERFACE 1
A-NR	:	kyo ,10	JOB NUMBER 10
ANZW	: FW	6	
QTYP	:	KSDB	SOURCE = DATA BLOCK
DBNR	:	KYO ,11	Data block number: 11
QANF	:	KF+0	START ADDRESS ALWAYS O ! !
QLAE	:	KF+200	TRANSFER 200 DATA WORDS
PAFE	:	FY199	PARAM. ASS. ERROR IN FY199

Under **the** job number above **you** must program the required job belonging to this group in your **CP** 525 user program and **trans**fer it to the corresponding interface.

38

....

3.8 Evaluation

The following sections contain information about **how to evaluate** the parameter assignment error byte (**PAFE** byte) and the condition codeword **ANZW**.

3.8.1 The Parameter Assignment Error Byte (PAFE Byte)

When working with the handling blocks **SYNCHRON**, SEND DIRECT n, SEND ALL, RECEIVE DIRECT n and RESET DIRECT 200, you **must** specify a **parameter assignment error byte (PAFE-BYTE)**.

If the HDB recognizes an error, it writes the corresponding **error** number into the **PAFE** byte. If the block runs through without errors the HDB writes OOH into the **PAFE** byte.

In addition to the errors that can be attributed directly to the parameter assignment, the **PAFE** byte also indicates errors which occur during the **CPU/CP** 525 communicat ion.

If such errors occur you should immediately remedy the situation to avoid other messages appearing during operation.

In certain exceptional cases, errors are indicated in the **PAFE** byte during operation. These errors are caused by the software of the **CP** 525 in special situations as follows:

if more than 10 **DIRECI** jobs are processed in the cycle, i.e. ten jobs are already **signalled** as "running" and **you** want to start an eleventh job

if the whole or part of the $C\!P$ 525 user program is transferred to the $C\!P$ 525 with COM 525 during normal operation

 $i\;f\;a$ SEND job is started before the <code>SYNCHRON</code> job runs through <code>once</code> without errors.

During multiprocessor operation you can also receive a PAFE message if several CPUS access the same CP 525 interface at the same time. The access is prevented (with a PAFE message) until the current CPU/CP communication is completed. This PAFE message does not indicate an error but shows that access is momentarily blocked.

The following list provides a brief explanation of the **PAFE** error messages. Note that the significance of each error number depends on the PC type. You should check the instructions for the handling blocks for your PC (see /1/ in the list of further relevant documentation).

PAFE no. Significance

- 11H Wrong ORG format: source/destination parameter type wrong, area (start address, length) not permissible
- 21H DB/DX or memory area not present
- 31H Source/destination too small
- **41H** Source/destination area not present (not plugged in) Timeout (QVZ) from this area
- 51H Error in condition codeword

61H No source parameters with SEND ALL

- 71H Interface does not exist (QVZ for the CP 525) Check the jumper settings for the interface number on the CP 525
- 81H Interface nc: ready or not synchronized. Check that the SYNCHRON HDB is programmed in the start-up OBs andnms without errors.
- **91H** Interface overload; part of a program is being transferred from the PC to the **CP** 525.

1.0

- AlH CP interface being used by a different CPU in multiprocessor operation (S5-135U and S5-155U) when several CPUs access one CP interface.
- BlH Job number too high
- ClH Error in handshake with the CP 525:
 - CP does not reply within the monitoring time or
 - **CP** rejects handshake, e.g. because **more** than 10 jobs were triggered simultaneously (when 10 jobs are running bit 1 is set in the **ANZW**)
- **DlH** Field Length (SYNCHRON) wrong and group error for other handshake errors
- **ElH** DB1 call missing with indirect parameter assignment and group error for **HDB** software error
- **FlH** Double block call (when a block can be **interrupted** at **command** boundaries)

3.8.2 The Condition Codeword (ANZW)

The condition codeword with direct jobs:

The first part of the condition codeword contains information about the status of the job processing - the secondpart contains **information about** the length ofthedata transferred.

If you are working with the **direct** jobs SEND **DIRECT n**, FETCH **DIRECT n**, RECEIVE **DIRECT** n the condition **codeword** requires a **doubleword**.

Significance of the bits in the condition codeword of a DIRECT job:

. Чер самотности страницати and a second state of the second and the second second second second second second second second second second

- Bit 1 : job running O = SEND/FETCH enabled 1 = SEND/FETCH disabled
- Bit 2 : job *complete* without error
- Bit 3 : job complete with error
- Bit 4 : data transfer/data reception running

The manufacture of the second states and the second states of the second

- Bit 5 : data transfer complete
- Bit 6 : data reception complete
- Bit 7 : 0 = data transfer/reception enabled 1 = data transfer/reception disabled
- Bit 8-11 : error messages see the error tables in the user's guide "Event output and listing with the PT88/PT89 printer" in this manual
- Bit 12-15 : not used

In the second half of the condition code **doubleword** the SEND ALL enters the length of the transferred data. This is the number of pieces of data currently being transmitted.

Every direct job should have its own condition code **doubleword reserved**. The following areas can be used:

-theflagareaFWOto 252

or

- the data words 0 to 254 in the currently open data block (DB).

To ensure reliable data exchange it is sufficient when the STEP 5 program evaluates bits 1, 2 and 3 of the ANZW as follows:

- bit 1 indicates that the job was accepted for processing by the CP. The job can be triggered again only after bit 1 is reset
- bits 2 and 3 indicate that the CP completed processing the job.

s •

فميردي فالدمية الافتهار بالم

.

On completion of the **job** without errors (bit 2 = 1), new information data can be prepared for the next transfer.

If the job is terminated with an error (bit 3 = 1), the error number is entered in bits 8 to 11 of the ANZW. The list of error **numbers** can be found in the user's guide "Event output and listing with the **PT88/PT89** printer" in this manual.

The condition codeword with the ALL job:

With the SEND **ALL**, the condition codeword requires only one data word or flag word.

While the **ALL** function **is** running, the number of the DIRECT job for which the data is being transferred is indicated in the **ANZW**. If the function is run through without any data transfer, the value O is entered.

In addition, the **ALL** function updates **the condition** codeword of the corresponding direct job.

SIEMENS

We have checked the contents of this manual for agreement with the hardware and software described. Since deviations cannot be precluded entirely, we cannot guarantee full agreement. However, the data in this manual are reviewed regularly and any necessary corradons included in subsequent editions. Suggestions for improvement are welcomed Technical data subjectio change. The reproduction, transmission or use of this document or its contents is not permitted without express written authority.

Offenders will be liable for damages. All rights, ineluding rights created by patent grant es registration of a utility model or design. are resewed.

Copyright Siemens AG 1989 All Rights Reserved

Siemens Aktiengesellschaft

.

C79000-B8576-C547.06 Gerätewerk Karlsruhe Printed in the Federal Republic of Germany